THE CITY RECO THE CITY RECORD U.S.P.S. 0114-660 Official Journal of The City of New York

VOLUME CXLVI NUMBER 76

FRIDAY, APRIL, 19, 2019

	Supply Management	1908
	Human Resources Administration	
TABLE OF CONTENTS	Office of Contracts	
	Parks and Recreation	
PUBLIC HEARINGS AND MEETINGS	Revenue	
City Planning Commission	Probation	
Housing Authority	Contract Procurement	
Landmarks Preservation Commission	Sanitation	
Transportation	Agency Chief Contracting Office	
•	School Construction Authority	
COURT NOTICES	Contract Services	
Supreme Court	Transportation	
<i>Queens County</i>	Administration	
Court Notice Maps1964	Youth and Community Development	1910
PROPERTY DISPOSITION	CONTRACT AWARD HEARINGS	
Citywide Administrative Services	Youth and Community Development	1910
Office of Citywide Procurement		
Environmental Protection	AGENCY RULES	
Housing Preservation and Development1905	Campaign Finance Board	
Police	Finance	1955
PROCUREMENT	SPECIAL MATERIALS	
Aging	Citywide Administrative Services	1956
Contract Procurement and Support Services 1906	Housing Preservation and Development	1958
Contract Services	Changes in Personnel	1959
Education		
Contracts and Purchasing	LATE NOTICE	
Environmental Protection	Board of Correction	
Agency Chief Contracting Office	District Attorney - New York County	
Wastewater Treatment	Human Resources Administration	
Health and Mental Hygiene	Office of Contracts	
Agency Chief Contracting Office	Law Department	
Housing Authority	Office of the Mayor	1963

Price: \$4.00

THE CITY RECORD

BILL DE BLASIO Mayor

LISETTE CAMILO Commissioner, Department of Citywide Administrative Services

ELI BLACHMAN Editor, The City Record

JANAE C. FERREIRA Assistant Editor, The City Record

Published Monday through Friday except legal holidays by the New York City Department of Citywide Administrative Services under Authority of Section 1066 of the New York City Charter.

Subscription \$500 a year, \$4.00 daily (\$5.00 by mail). Periodicals Postage Paid at New York, NY OSTMASTER: Send address changes to THE CITY RECORD, 1 Centre Street, 17th Floor, New York, NY 10007-1602

Editorial Office/Subscription Changes: The City Record, 1 Centre Street, 17th Floor, New York, NY 10007-1602 (212) 386-0055

Visit The New City Record Online (CROL) at www.nyc.gov/cityrecord for a searchable database of all notices published in the City Record.

PUBLIC HEARINGS AND MEETINGS

See Also: Procurement; Agency Rules

CITY PLANNING COMMISSION

PUBLIC HEARINGS

NOTICE IS HEREBY GIVEN that resolutions have been adopted by the City Planning Commission, scheduling a public hearing on the following matters, to be held, at NYC City Planning Commission, Hearing Room, Lower Concourse, 120 Broadway, New York, NY, on Wednesday, April 24, 2019, at 10:00 A.M.



BOROUGH OF MANHATTAN No. 1 515 WEST 18TH STREET GARAGE

CD 4 C 190213 ZSM IN THE MATTER OF an application submitted by 18th Highline IN THE MATTER OF an application submitted by 18th Highline Associates, L.L.C., pursuant to Sections 197-c and 201 of the New York City Charter, for the grant of special permits, pursuant to Section 13-45 (Special Permits for additional parking spaces), and Section 13-451 (Additional parking spaces for residential growth), of the Zoning Resolution to allow an attended accessory parking garage with a maximum capacity of 180 spaces on portions of the ground floor and collar of a property located floor and cellar of a proposed mixed-use building on property, located at 515 West 18th Street (Block 690, Lots 12, 20, 29, 40, 54 and 1001-1026), in C6-2 and C6-3 Districts, within the Special West Chelsea District.

Plans for this proposal are on file with the City Planning Commission, and may be seen, at 120 Broadway, 31st Floor, New York, NY 10271-0001.

BOROUGH OF QUEENS No. 2 76TH DRIVE AND AUSTIN STREET REZONING CD 6

C 180399 ZMQ

IN THE MATTER OF an application submitted by Able Orthopedic & Sports Medicine, PC, pursuant to Sections 197-c and 201 of the New York City Charter, for an amendment of the Zoning Map, Section No. 14b, changing from an R2 District to an R3-2 District, property bounded by 76th Drive, a line perpendicular to the southeasterly street line of 76th Drive distant 55 feet northeasterly (as measured along the street line), from the point of intersection of the southeasterly street line of 76th Drive and northeasterly street line of Austin Street, 77th Avenue, and Austin Street;

as shown on a diagram (for illustrative purposes only), dated February 11, 2019, and subject to the conditions of CEQR Declaration E-522.

YVETTE V. GRUEL, Calendar Officer City Planning Commission 120 Broadway, 31st Floor, New York, NY 10271 Telephone (212) 720-3370

HOUSING AUTHORITY

■ MEETING

The next Board Meeting of the New York City Housing Authority, is scheduled for Wednesday, April 24, 2019, at 10:00 A.M., in the Board Room, on the 12th Floor of 250 Broadway, New York, NY (unless otherwise noted). Copies of the Calendar will be available on NYCHA's website, or may be picked up, at the Office of the Corporate Secretary, at 250 Broadway, 12th Floor, New York, NY, no earlier than 24 hours before the upcoming Board Meeting. Copies of the Minutes will also be available on NYCHA's website, or may be picked up, at the Office of the Corporate Secretary, no earlier than 3:00 P.M., on the Thursday following the Board Meeting.

Any changes to the schedule, will be posted here, and on NYCHA's website, at http://www1.nyc.gov/site/nycha/about/board-calendar.page, to the extent practicable, at a reasonable time before the meeting.

The meeting, is open to the public. Pre-Registration, at least 45 minutes before the scheduled Board Meeting, is required by all speakers. Comments are limited to the items on the Calendar. Speaking time will be limited to three minutes. The public comment period will conclude upon all speakers being heard or at the expiration of 30 minutes allotted by law, for public comment, whichever occurs first.

The meeting will be streamed live on NYCHA's website, at http://nyc.gov/nycha, and on http://on.nyc.gov/boardmeetings.

For additional information, please visit NYCHA's website or contact $\left(212\right)$ 306-6088.

Accessibility questions: Office of the Corporate Secretary (212) 306-6088, corporate.secretary@nycha.nyc.gov, by: Wednesday, April 10, 2019, 5:00 P.M.

Ci Large Print

a3-24

LANDMARKS PRESERVATION COMMISSION

■ PUBLIC HEARINGS

NOTICE IS HEREBY GIVEN that, pursuant to the provisions of Title 25, Chapter 3 of the Administrative Code of the City of New York (Sections 25-303, 25-307, 25-308, 25-309, 25-313, 25-318, 25-320) on Tuesday, April 30, 2019, a public hearing will be held, at 1 Centre Street, 9th Floor, Borough of Manhattan with respect to the following properties and then followed by a public meeting. The final order and estimated times for each application will be posted on the Landmarks Preservation Commission website the Friday before the hearing. Any person requiring reasonable accommodation in order to participate in the hearing or attend the meeting should contact the Landmarks Commission no later than five (5) business days before the hearing or meeting.

902 Shore Road - Douglaston Historic District LPC-19-29208 - Block 8014 - Lot 26 - Zoning: R1-1 CERTIFICATE OF APPROPRIATENESS

An Arts and Crafts/Colonial Revival style house, built in 1906 and designed by Wilbur S. Knowles. Application is to construct a rear addition, alter and relocate the existing garage, excavate the site for a below grade garage and terrace, and alter fenestration.

132 Calyer Street - Greenpoint Historic District LPC-19-35316 - Block 2594 - Lot 21 - Zoning: 13A CERTIFICATE OF APPROPRIATENESS

A frame house, built in 1868-69. Application is to legalize the installation of siding in non-compliance with Landmarks Preservation Commission permit(s).

14 Old Fulton Street - Fulton Ferry Historic District LPC-19-37589 - Block 200 - Lot 6 - Zoning: M2-1 CERTIFICATE OF APPROPRIATENESS

A one-story gas station. Application is to install a new commercial structure on the site, with signage and lighting.

163 Dean Street - Boerum Hill Historic District LPC-19-37861 - Block 189 - Lot 57 - Zoning: R6B CERTIFICATE OF APPROPRIATENESS

A Greek Revival style rowhouse, built c. 1850. Application is to install rooftop HVAC units.

372-374 Fulton Street - Individual and Interior Landmark LPC-19-36232 - Block 154 - Lot 17 - Zoning: C6-4.5 CERTIFICATE OF APPROPRIATENESS

A late Italianate style townhouse, built in the mid-1870s, with a Neo-Grec style storefront and a ground floor restaurant interior, installed in 1892. Application is to install fixtures and finishes within the designated interior space.

352-360 Clermont Avenue - Fort Greene Historic District LPC-19-20079 - Block 2121 - Lot 28 - **Zoning:** R6B **CERTIFICATE OF APPROPRIATENESS** A parking lot with garage. Application is to demolish the garage and construct five rowhouses.

154 Court Street - Cobble Hill Historic District LPC-19-35682 - Block 292 - Lot 31 - **Zoning:** R6/C2-3 **CERTIFICATE OF APPROPRIATENESS** A Gothic Revival style rowhouse, built in 1854. Application is to construct a rear yard addition and replace storefront infill.

700 Gerard Avenue - Grand Concourse Historic District LPC-19-32409 - Block 2473 - Lot 8 - Zoning: R8 CERTIFICATE OF APPROPRIATENESS An empty lot. Application is to construct a new building.

195 Broadway - Individual and Interior Landmark LPC-19-36246 - Block 80 - Lot 1 - Zoning: C5-5 CERTIFICATE OF APPROPRIATENESS

A Neo-Classical style office building, designed by William Welles Bosworth, and built in phases from 1912-1922 with a monumental Classical style interior. Application is to install glass railings and turnstiles, and to modify a directory at the designated interior.

200 Mercer Street - NoHo Historic District LPC-19-37269 - Block 523 - Lot 32 - Zoning: C6-2 CERTIFICATE OF APPROPRIATENESS A vernacular stable designed by John G Prague and

A vernacular stable, designed by John G. Prague, and built in 1870-71. Application is to paint the building base, replace side yard fence, gate, and paving, and install a canopy and site furnishings.

422 West Broadway - SoHo-Cast Iron Historic District Extension LPC-19-35420 - Block 502 - Lot 33 - Zoning: M1-5A CERTIFICATE OF APPROPRIATENESS

An Italianate style store and loft building, designed by John H. Whitenack and built in 1873-74. Application is to modify storefront infill.

422 West Broadway - SoHo-Cast Iron Historic District Extension LPC-19-30153 - Block 502 - Lot 33 - Zoning: M1-5A MODIFICATION OF USE AND BULK

An Italianate style store and loft building, designed by John H. Whitenack, and built in 1873-74. Application is to request that the Landmarks Preservation Commission, issue a report to the City Planning Commission, relating to an application for a Modification of Use and Bulk.

7th Avenue South and West 10th Street - Greenwich Village Historic District

LPC-19-35605 - Block - Lot - Zoning: C2-6 BINDING REPORT

Northwest corner of 7th Avenue South and West 10th Street. Application is to install a newsstand at the sidewalk.

5 West 16th Street -LPC-19-29595 - Block 818 - Lot 37 - Zoning: C6-2M CERTIFICATE OF APPROPRIATENESS A Greek Revival style house, built c. 1846. Application is to replace storefront infill.

200 West 83rd Street - Upper West Side/Central Park West Historic District LPC-19-36228 - Block 1230 - Lot 133 - Zoning: C2-7A

CERTIFICATE OF APPROPRIATENESS A Neo-Gree flats building, built 1880-81. Application is to modify an existing one-story addition, install new entrance infill, signage, and doors.

429 West 146th Street - Hamilton Heights/Sugar Hill Historic District

LPC-19-35891 - Block 2061 - Lot 22 - Zoning: R6A CERTIFICATE OF APPROPRIATENESS

A Renaissance Revival style rowhouse, designed by John P. Leo, and built in 1899-1900. Application is to modify masonry openings and install a rear deck.

150 West 79th Street - Upper West Side/Central Park West Historic District

LPC-19-28629 - Block 1150 - Lot 55 - **Zoning:** R10A **CERTIFICATE OF APPROPRIATENESS** A Neo-Renaissance style apartment building, designed by Schwartz & Gross and built in 1919-20. Application is to establish a master plan governing the future installation of windows.

267 West 138th Street - St. Nicholas Historic District LPC-19-33446 - Block 2024 - Lot 4 - Zoning: R7-2 CERTIFICATE OF APPROPRIATENESS

A Neo-Georgian style rowhouse, designed by Bruce Price and Clarence S. Luce, and built in 1891-92. Application is to modify masonry openings, install a planting bed, and construct a garage.

a17-30

NOTICE IS HEREBY GIVEN that, pursuant to the provisions of Title 25, Chapter 3 of the Administrative Code of the City of New York

ð

(Sections 25-303, 25-307, 25-308, 25-309, 25-313, 25-318, 25-320) on Tuesday, April 30, 2019, a public hearing will be held, at 1 Centre Street, 9th Floor, Borough of Manhattan with respect to the following properties and then followed by a public meeting. The final order and estimated times for each application will be posted on the Landmarks Preservation Commission website, the Friday before the hearing. Any person requiring reasonable accommodation in order to participate in the hearing or attend the meeting should contact the Landmarks Commission no later than five (5) business days before the hearing or meeting.

est 28th Street - 47 West 28th Street Building, Tin Pan Alley LP-2626 - Block 830 - Lot 11 - Zoning: ITEM PROPOSED FOR PUBLIC HEARING

An Italianate row house, built c. 1852 with later commercial alterations to house music publishers at the turn of the 20th century, part of a block known as "Tin Pan Alley.

49 West 28th Street - 49 West 28th Street Building, Tin Pan Alley LP-2627 - Block 830 - Lot 10 - Zoning:

ITEM PROPOSED FOR PUBLIC HEARING

An Italianate row house, built c. 1852 with later commercial alterations to house music publishers at the turn of the 20th century, part of a block known as "Tin Pan Alley."

51 West 28th Street - 51 West 28th Street Building, Tin Pan Alley LP-2628 - Block 830 - Lot 9 - Zoning: ITEM PROPOSED FOR PUBLIC HEARING An Italianate row house, built c. 1852 with later commercial alterations

to house music publishers, at the turn of the 20th century, part of a block known as "Tin Pan Alley."

53 West 28th Street - 53 West 28th Street Building, Tin Pan Alley LP-2629 - Block 830 - Lot 8 - Zoning: ITEM PROPOSED FOR PUBLIC HEARING

An Italianate row house, built c. 1859 with later commercial alterations to house music publishers at the turn of the 20th century, part of a block known as "Tin Pan Alley."

55 West 28th Street - 55 West 28th Street Building, Tin Pan Alley IP-2630 - Block 830 - Lot 7 - Zoning: ITEM PROPOSED FOR PUBLIC HEARING

An Italianate row house, built c. 1859 with later commercial alterations to house music publishers at the turn of the 20th century, part of a block known as "Tin Pan Alley."

a17-30

Ť

TRANSPORTATION

PUBLIC HEARINGS

PUBLIC NOTICE

NOTICE OF A JOINT PUBLIC HEARING of the Franchise and Concession Review Committee, and the New York City Department of Transportation ("DOT"), to be held on May 6, 2019, at 2 Lafayette Street, 14th Floor Auditorium, Borough of Manhattan, commencing at 2:30 P.M. relative to:

INTENT TO AWARD as a concession, a Sole Source License Agreement ("License"), to the 34th Street Partnership, Inc. ("Partnership"), whose address is 1065 Avenue of the Americas, Suite 2400, New York, NY 10018, to provide for the operation, management, and maintenance of a pedestrian plaza, located on Broadway and 6th Avenue between West 32nd and West 36th Streets, in the borough of Manhattan ("Licensed Plaza"), including through DOT-approved events, sponsorships, and subconcessions, including but not limited to providing for the sale of any of the following: prepared food, flowers, locally grown produce or locally manufactured products, merchandise (such as souvenirs or T-shirts) that promotes the neighborhood or Partnership, and other similar merchandise within the Licensed Plaza. Subconcessions would be awarded based on solicitations issued, by the Partnership in the basic form of a Request for Proposals or Request for Bids, subject to DOT's prior written approval of both solicitation and award. The Partnership will be required to invest any revenue generated by this concession into the maintenance and/or repair, including reasonable administrative costs, of the Licensed Plaza.

The License will provide for one (1) one-year term, commencing upon written Notice to Proceed, which may be renewed for up to three (3)additional five-year terms, exercisable at the sole discretion of DOT.

A draft copy of the License may be reviewed or obtained at no cost, commencing April 18 through May 3, 2019, between the hours of 10:00 A.M. and 4:00 P.M., excluding weekends and holidays, at the NYC Department of Transportation, located at the NYC Department of Transportation, Office of Cityscape & Franchises, 55 Water Street, 9th Floor New York NY 10041 Floor, New York, NY 10041.

This location is accessible to individuals using wheelchairs or other mobility devices. For further information on accessibility or to make a request for accommodations, such as sign language interpretation services, please contact the Mayor's Office of Contract Services (MOCS) via email, at DisabilityAffairs@mocs.nyc.gov, or via phone, at (212) 788-0010. Any person requiring reasonable accommodation for the public hearing should contact MOCS at least three (3) business days in advance of the hearing to ensure availability.

TELECOMMUNICATION DEVICE FOR THE DEAF (TDD) (212) 504-4115.

🕶 a19

PUBLIC NOTICE

NOTICE OF A JOINT PUBLIC HEARING of the Franchise and Concession Review Committee and the New York City Department of Transportation ("DOT"), to be held on May 6, 2019, at 2 Lafayette Street, 14th Floor Auditorium, Borough of Manhattan, commencing at 2:30 P.M. relative to:

INTENT TO AWARD as a concession, a Sole Source License Agreement ("License"), to the Meatpacking District Management Association, Inc. ("Meatpacking BID"), whose address is 32 Gansevoort Street, 5th Floor, New York, NY 10014, to provide for the operation, management, and maintenance of pedestrian plazas, located at 9th Avenue between Gansevoort Street and West 15th Street, in the borough of Manhattan ("Licensed Plaza"), including through DOTapproved events, sponsorships, and subconcessions, including but not limited to providing for the sale of any of the following: prepared food, flowers, locally grown produce or locally manufactured products, merchandise (such as souvenirs or T-shirts) that promotes the neighborhood or Meatpacking BID, and other similar merchandise within the Licensed Plaza. Subconcessions would be awarded based on solicitations issued by Meatpacking BID in the basic form of a Request for Proposals or Request for Bids, subject to DOT's prior written approval of both solicitation and award. Meatpacking BID will be required to invest any revenue generated by this concession into the maintenance and/or proper in playing and a subject to dot the solicitation and a subject to the solicitation and a solicitation maintenance and/or repair, including reasonable administrative costs, of the Licensed Plaza.

The License will provide for one (1) six-month term, commencing upon written Notice to Proceed, which may be renewed for up to three (additional five-year terms, exercisable at the sole discretion of DOT.

A draft copy of the License may be reviewed or obtained at no cost A draft copy of the License may be reviewed of obtained at no cost, commencing April 26 through May 3, 2019, between the hours of 10:00 A.M. and 4:00 P.M., excluding weekends and holidays at the NYC Department of Transportation, located at the NYC Department of Transportation, Office of Cityscape & Franchises, 55 Water Street, 9th Floor, New York, NY 10041.

This location is accessible to individuals using wheelchairs or other mobility devices. For further information on accessibility or to make a request for accommodations, such as sign language interpretation services, please contact the Mayor's Office of Contract Services (MOCS) via email, at DisabilityAffairs@mocs.nyc.gov or via phone, at (212) 788-0010. Any person requiring reasonable accommodation for the public hearing should contact MOCS at least three (3) business days in advance of the hearing to ensure availability.

TELECOMMUNICATION DEVICE FOR THE DEAF (TDD) (212) 504-4115.

🕶 a19

PUBLIC NOTICE

NOTICE OF A JOINT PUBLIC HEARING of the Franchise and Concession Review Committee and the New York City Department of Transportation ("DOT"), to be held on May 6, 2019, at 2 Lafayette Street, 14th Floor Auditorium, Borough of Manhattan, commencing at 2:30 P.M. relative to:

INTENT TO AWARD as a concession, a Sole Source License Agreement ("License"), to the 34th Street Partnership, Inc. ("Partnership"), whose address is 1065 Avenue of the Americas, Suite 2400, New York, NY 10018, to provide for the operation, management, and maintenance of a pedestrian plaza, located at West 33rd Street between 7th Avenue and 8th Avenue, in the borough of Manhattan ("Licensed Plaza"), including through DOT-approved events, sponsorships, and subconcessions, including but not limited to providing for the sale of any of the following: prepared food, flowers, locally grown produce or locally manufactured products, merchandise (such as souvenirs or T-shirts) that promotes the neighborhood or Partnership, and other similar merchandise within the Licensed Plaza. Subconcessions would be awarded based on solicitations issued by the Partnership in the basic form of a Request for Proposals or Request for Bids, subject to DOT's prior written approval of both solicitation and award. The Partnership will be required to invest any revenue generated by this concession into the maintenance and/or repair, including reasonable administrative costs, of the Licensed Plaza.

The License will provide for one (1) one-year term, commencing upon written Notice to Proceed, which may be renewed for up to three (3) additional five-year terms, exercisable at the sole discretion of DOT.

A draft copy of the License may be reviewed or obtained at no cost, commencing April 26 through May 3, 2019, between the hours of 10:00 A.M. and 4:00 P.M., excluding weekends and holidays at the NYC Department of Transportation, located at the NYC Department of Transportation, Office of Cityscape & Franchises, 55 Water Street, 9th Floor, New York, NY 10041.

This location is accessible to individuals using wheelchairs or other mobility devices. For further information on accessibility or to make a request for accommodations, such as sign language interpretation services, please contact the Mayor's Office of Contract Services (MOCS) via email, at DisabilityAffairs@mocs.nyc.gov or via phone, at (212) 788-0010. Any person requiring reasonable accommodation for the public hearing should contact MOCS at least three (3) business days in advance of the hearing to ensure availability.

TELECOMMUNICATION DEVICE FOR THE DEAF (TDD) (212) 504-4115.

ð

🖝 a19

COURT NOTICES

SUPREME COURT

QUEENS COUNTY

■ NOTICE

QUEENS COUNTY IAS PART 38 NOTICE OF PETITION INDEX NUMBER 705567/2019 CONDEMNATION PROCEEDING

IN THE MATTER OF the Application of the CITY OF NEW YORK Relative to Acquiring Title in Fee Simple to Property, located in Queens, including All or Parts of 142ND STREET AT ITS INTERSECTION WITH THE

SOUTHWEST CORNER OF 135TH AVENUE

in the Borough Queens, City and State of New York.

PLEASE TAKE NOTICE that the City of New York ("City"), intends to make an application to the Supreme Court of the State of New York, Queens County, IAS Part 38, for certain relief. The application will be made, at the following time and place: At the Queens County Courthouse, located at 88-11 Sutphin Avenue, Part 38, Courtroom 116 in the Borough of Queens, City and State of New York, on May 2, 2019, at 9:30 A.M., or as soon thereafter as counsel can be heard.

The application is for an order:

- a. authorizing the City to file an acquisition map in the Office of the City Register;
- b. directing that, upon the filing of the order granting the relief sought in this petition, together with the filing of the acquisition map in the Office of the City Register, title to the property shown on said map and sought to be acquired and more particularly described in this petition shall vest in the City in fee simple absolute;
- c. providing that the compensation which should be made to the owners of the real property sought to be acquired and described in this petition be ascertained and determined by the Court without a jury;
- d. directing that within thirty days of the entry of the order granting the relief sought in this petition, the City shall cause a Notice of Acquisition, to be published in at least ten successive issues of The City Record, an official newspaper published in the City of New York, and shall serve a copy of such notice by first class mail on each condemnee or his, her, or its attorney of record; and
- e. directing that each condemnee shall have a period of one calendar year from the vesting date for this proceeding, in which to file a written claim, demand or notice of appearance with the Clerk of this Court and to serve a copy of the same upon the Corporation Counsel of the City of New York, 100 Church Street, New York, NY 10007.

The City of New York, in this proceeding, intends to acquire title

in fee simple absolute to certain real property where not heretofore acquired for the same purpose, for roadway improvements, widening, and related work in the Borough of Queens, City and State of New York. The real property that is to be acquired in fee simple absolute in this proceeding is described as follows:

142ND STREET AT ITS INTERSECTION WITH THE SOUTHWEST CORNER OF 135TH AVENUE

All that certain plot, piece or parcel of land, situate, lying and being in the Borough and County of Queens, City and State of New York, being more particularly bounded and described as follows:

BEGINNING at a corner formed by the intersection of the southeasterly line of the 135th Avenue, with southwesterly line of the said 142nd Street as shown on Alteration Map No. 5014;

- 1. RUNNING THENCE, northeasterly, along the former southeasterly line of 135th Avenue, a distance of 24.52 feet to the former westerly line of 142nd Street, as eliminated by the said Alteration Map;
- 2. THENCE, southerly, deflecting to the left 104 degrees 04 minutes 34.5 seconds from the last mentioned course and along the said former westerly line of 142nd Street, a distance of 94.26 feet to an angle point in the westerly line of 142nd Street;
- 3. THENCE, northwesterly, along the said southwesterly line of 142nd Street, deflecting to the right 164 degrees 55 minutes 36.5 seconds from the last mentioned course, a distance of 91.44 feet back to the point of Beginning.

This parcel consists of part of tax lot 6 in Queens tax block 12095, as laid out on the "City Map" of the City of New York, Borough of Queens, and comprises an area of 1,121 square feet or 0.02573 acres.

Surveys, maps or plans of the property to be acquired are on file in the office of the Corporation Counsel of the City of New York, 100 Church Street, New York, NY 10007.

PLEASE TAKE FURTHER NOTICE THAT, pursuant to EDPL § 402(B)(4), any party seeking to oppose the acquisition must interpose a verified answer, which must contain specific denial of each material allegation of the petition controverted by the opponent, or any statement of new matter deemed by the opponent to be a defense to the proceeding. Pursuant to CPLR § 403, said answer must be served upon the office of the Corporation Counsel at least seven (7) days before the date that the petition is noticed to be heard.

Dated: New York, NY April 8, 2019

ZACHARY W. CARTER Corporation Counsel of the City of New York Attorney for the Condemnor 100 Church Street New York, NY 10007 Telephone (212) 356-4064

See map(s) in back of paper

a15-26

PROPERTY DISPOSITION

CITYWIDE ADMINISTRATIVE SERVICES

■ SALE

The City of New York in partnership with PropertyRoom.com posts vehicle and heavy machinery auctions online every week at: https://www.propertyroom.com/s/nyc+fleet

All auctions are open to the public and registration is free.

Vehicles can be viewed in person by appointment at:

FRIDAY, APRIL, 19, 2019

Kenben Industries Ltd., 1908 Shore Parkway, Brooklyn, NY 11214. Phone: (718) 802-0022

m30-s11

OFFICE OF CITYWIDE PROCUREMENT

■ NOTICE

The Department of Citywide Administrative Services, Office of Citywide Procurement is currently selling surplus assets on the Internet. Visit http://www.publicsurplus.com/sms/nycdcas.ny/browse/home

To begin bidding, simply click on 'Register' on the home page.

There are no fees to register. Offerings may include but are not limited to: office supplies/equipment, furniture, building supplies, machine tools, HVAC/plumbing/electrical equipment, lab equipment, marine equipment, and more.

Public access to computer workstations and assistance with placing bids is available at the following locations:

- DCAS Central Storehouse, 66-26 Metropolitan Avenue, Middle Village, NY 11379
- DCAS, Office of Citywide Procurement, 1 Centre Street, 18th Floor, New York, NY 10007

j2-d31

ENVIRONMENTAL PROTECTION

SALE

Environmental Protection, Natural Resources Division Forest Management Project # 5090 "Damsite" NOTICE OF PROJECT AVAILABILITY

Description: The City of New York will sell approximately 381 thousand board feet of mixed hardwood and softwood sawtimber and more than 198 cords of hardwood firewood through Forest Management Project **#5090**. The project is located within the approximately 120 acre *Damsite Forest Management Project* area, on New York City-Owned watershed land in the Ashokan Reservoir basin, Town of Olive, Ulster County, NY. Bid Solicitation information is available by contacting NYCDEP Regional Forester, Todd Baldwin, at (845) 340-7854, or requesting via email, at tbaldwin@dep.nyc.gov.

Show Dates: Prospective bidders are <u>required</u> to attend one of the two public showings in order to receive a bid package necessary to submit a valid bid. The showings will be held on Thursday, April 25, 2019, at 2:00 P.M., and Friday, April 26, 2019, at 9:00 A.M., and will begin at the Ashokan Fountain parking area, across from Ben Nesin Lab (#2389 Route 28A), Olivebridge, NY 12461.

Bidding: All bid proposals must be received by Todd Baldwin, 71 Smith Avenue, Kingston, NY 12401 (845-340-7854), NO LATER THAN Thursday, May 16, 2019, at 3:00 P.M., local time. Sealed bids will be publicly opened, at the DEP Office, 71 Smith Avenue, Kingston, NY, on Friday May 17, 2019, at 9:00 A.M., local time. The projected date for awarding the bid is on or about May 31, 2019.

a16-22

HOUSING PRESERVATION AND DEVELOPMENT

■ PUBLIC HEARINGS

All Notices Regarding Housing Preservation and Development Dispositions of City-Owned Property appear in the Public Hearing Section.

j9-30

POLICE

■ NOTICE

OWNERS ARE WANTED BY THE PROPERTY CLERK DIVISION OF THE NEW YORK CITY POLICE DEPARTMENT

The following list of properties is in the custody of the Property Clerk Division without claimants:

Motor vehicles, boats, bicycles, business machines, cameras, calculating

machines, electrical and optical property, furniture, furs, handbags, hardware, jewelry, photographic equipment, radios, robes, sound systems, surgical and musical instruments, tools, wearing apparel, communications equipment, computers, and other miscellaneous articles.

Items are recovered, lost, abandoned property obtained from prisoners, emotionally disturbed, intoxicated and deceased persons; and property obtained from persons incapable of caring for themselves.

INQUIRIES

Inquiries relating to such property should be made in the Borough concerned, at the following office of the Property Clerk.

FOR MOTOR VEHICLES (All Boroughs):

- Springfield Gardens Auto Pound, 174-20 North Boundary Road, Queens, NY 11430, (718) 553-9555
- Erie Basin Auto Pound, 700 Columbia Street, Brooklyn, NY 11231, (718) 246-2030

FOR ALL OTHER PROPERTY

- Manhattan 1 Police Plaza, New York, NY 10038, (646) 610-5906
- Brooklyn 84th Precinct, 301 Gold Street, Brooklyn, NY 11201, (718) 875-6675
- Bronx Property Clerk 215 East 161 Street, Bronx, NY 10451, (718) 590-2806
- Queens Property Clerk 47-07 Pearson Place, Long Island City, NY 11101, (718) 433-2678
- Staten Island Property Clerk 1 Edgewater Plaza, Staten Island, NY 10301, (718) 876-8484

j2-d31

PROCUREMENT

"Compete To Win" More Contracts!

Thanks to a new City initiative - "Compete To Win" - the NYC Department of Small Business Services offers a new set of FREE services to help create more opportunities for minority and Women-Owned Businesses to compete, connect and grow their business with the City. With NYC Construction Loan, Technical Assistance, NYC Construction Mentorship, Bond Readiness, and NYC Teaming services, the City will be able to help even more small businesses than before.

• Win More Contracts at nyc.gov/competetowin

"The City of New York is committed to achieving excellence in the design and construction of its capital program, and building on the tradition of innovation in architecture and engineering that has contributed to the City's prestige as a global destination. The contracting opportunities for construction/construction services and construction-related services that appear in the individual agency listings below reflect that commitment to excellence."

HHS ACCELERATOR

To respond to human services Requests for Proposals (RFPs), in accordance with Section 3-16 of the Procurement Policy Board Rules of the City of New York ("PPB Rules"), vendors must first complete and submit an electronic prequalification application using the City's Health and Human Services (HHS) Accelerator System. The HHS Accelerator System is a web-based system maintained by the City of New York for use by its human services Agencies to manage procurement. The process removes redundancy by capturing information about boards, filings, policies, and general service experience centrally. As a result, specific proposals for funding are more focused on program design, scope, and budget.

Important information about the new method

- Prequalification applications are required every three years.
- Documents related to annual corporate filings must be submitted on an annual basis to remain eligible to compete.
- Prequalification applications will be reviewed to validate compliance with corporate filings, organizational capacity, and relevant service experience.

• Approved organizations will be eligible to compete and would submit electronic proposals through the system

The Client and Community Service Catalog, which lists all Prequalification service categories and the NYC Procurement Roadmap, which lists all RFPs to be managed by HHS Accelerator may be viewed at http://www.nyc.gov/html/hhsaccelerator/html/ roadmap/roadmap.shtml. All current and prospective vendors should frequently review information listed on roadmap to take full advantage of upcoming opportunities for funding.

Participating NYC Agencies

HHS Accelerator, led by the Office of the Mayor, is governed by an Executive Steering Committee of Agency Heads who represent the following NYC Agencies:

Administration for Children's Services (ACS) Department for the Aging (DFTA) Department of Consumer Affairs (DCA) Department of Corrections (DOC) Department of Health and Mental Hygiene (DOHMH) Department of Homeless Services (DHS) Department of Probation (DOP)

Department of Small Business Services (SBS)

Department of Youth and Community Development (DYCD)

Housing and Preservation Department (HPD) Human Resources Administration (HRA) Office of the Criminal Justice Coordinator (CJC)

To sign up for training on the new system, and for additional information about HHS Accelerator, including background materials, user guides and video tutorials, please visit www.nyc.gov/hhsaccelerator

AGING

CONTRACT PROCUREMENT AND SUPPORT SERVICES

AWARD

Human Services / Client Services

SENIOR SERVICES - BP/City Council Discretionary Community Council Inc., 128 Brighton Beach Avenue, 4th Floor, Brooklyn, NY 11235.

City Council/ Borough President discretionary - funds for this contract have been provided through a discretionary award, to enhance services to New York City's older adults.

🖝 a19

CONTRACT SERVICES

■ INTENT TO AWARD

Human Services / Client Services

TIER 1 EVIDENCE-BASED PROGRAMMING - Negotiated Acquisition - Judgment required in evaluating proposals -PIN#12519N0001 - Due 4-23-19 at 5:00 P.M.

The NYC Department for the Aging (DFTA), which oversees a wide range of programs for older New Yorkers (www.nyc.gov/aging), aims to deliver Tier 1 Evidence-Based programs throughout its senior service programs portfolio within the five boroughs, addressing the health concerns of its constituents. Tier 1 Evidence-Based programs are targeted to individual adults who are age 60 or above, with the goal of improving their independence, dignity and quality of life. DFTA is seeking one provider to provide National Council on Aging (NCOA) approved Tier 1 Evidence-Based Program interventions for Senior Center program participants, addressing a host of health concerns through lectures and/or physical activity.

DFTA has identified the following four organizations, which DFTA, intends to negotiate a possible contract with and award to one of the following organizations:

1. IPRO

2. LiveOnNY

- 3. The Institute of Family Health
- 4. Temple University

Expressions of Interest must include a narrative addressing each of the bulleted qualification requirements listed below. Expressions of interest will be evaluated, but please be aware that submission of an Interest will be evaluated, but please be aware that submission of an expression of interest does not guarantee consideration, and DFTA may determine to continue negotiations with the four vendors originally identified. DFTA will send the solicitation document to the four identified entities, and any additional entities who express interest and are deemed qualified.

All interested organizations that meet the requirements, will be required to submit a written application that will be due no later than 3:00 P.M., on May 7, 2019.

- 1. Experience delivering NCOA Evidence-Based programs to older adults 60 and over.
- 2. Experience with development, implementation, management, training and evaluation of evidence-based health promotion programs promoted and supported by the US Administration on Community Living.
- 3. Experience developing a systems change project.
- 4. Experience with assessing qualitative and quantitative outcomes.
- 5. Experience in convening stakeholder meetings.
- 6. Experience with research as it pertains Evidence-Based
- programming. 7. Experience with Medicaid/Medicare reimbursement for the delivery of Evidence-Based programming.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor prequalification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above.

Åging, 2 Lafayette Street, 4th Floor, New York, NY 10007. Michelle Biondi (212) 602-7747; mbiondi@aging.nyc.gov

a16-22

EDUCATION

CONTRACTS AND PURCHASING

■ SOLICITATION

Goods and Services

REQUIREMENTS CONTRACT FOR ENVIRONMENTAL TESTING OF AIR AND WATER - Competitive Sealed Bids -PIN#B3322040 - Due 5-20-19 at 4:00 P.M.

To download, please go to https://infohub.nyced.org/resources/vendors/ open-doe-solicitations/request-for-bids. If you cannot download, send an email to vendorhotline@schools.nyc.

gov, with the RFB number and title in the subject line.

For all questions related to this RFB, please email tellis@schools.nyc. gov, with the RFB number and title in the subject line of your email. Provide testing and analysis of drinking water and indoor air quality in DOE schools and administrative offices.

There will be a Pre-Bid Conference, on Tuesday, April 30, 2019, at 3:00 P.M., at 65 Court Street, 12th Floor, Conference Room 1201, Brooklyn, NY 11201

The New York City Department of Education (DOE), strives to give all businesses, including Minority and Women-Owned Business Enterprises (MWBEs), an equal opportunity to compete for DOE procurements. The DOE's mission is to provide equal access to procurement opportunities for all qualified vendors, including MWBEs, from all segments of the community. The DOE works to enhance the ability of MWBEs to compete for contracts. DOE is committed to ensuring that MWBEs fully participate in the procurement process.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor pre-qualification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above Éducation, 65 Court Street, Room 1201, Brooklyn, NY 11201. Vendor Hotline (718) 935-2300; vendorhotline@schools.nyc.gov

ð

🕶 a19

ENVIRONMENTAL PROTECTION

AGENCY CHIEF CONTRACTING OFFICE

AWARD

Construction Related Services

GI-S-DESIGN-2: TASK ORDER CONTRACT FOR DESIGN SERVICES FOR GREEN INFRASTRUCTURE - Request for Proposals - PIN#82617GINYDOE - AMT: \$3,850,000.00 - TO: NV5 New York-Engineers, Architects, Landscape Architects and Surveyors, 32 Old Slip, 4th Floor, New York, NY 10005.

🖝 a19

WASTEWATER TREATMENT

■ SOLICITATION

Services (other than human services)

REPAIR OF BOILERS AT VARIOUS WASTEWATER TREATMENT PLANTS, PUMP STATIONS AND ASSOCIATED

DEP FACILITIES (NORTH REGION) - Competitive Sealed Bids - PIN#82619B0052 - Due 5-14-19 at 11:30 A.M.

Project Number: 1477-BLR(R), Document Fee: \$80.00, Project Manager: Mohd Alam, Engineers Estimate: \$2,265,000.00 - \$3,000,000.00.

There will be a Pre-Bid on 4/26/19, at 11:00 A.M., located at 96-05 Horace Harding Expressway, 2nd Floor Conference Room, Collections, Flushing, NY 11373. Last day for questions 5/1/19.

Please email Agency contact Fheras@dep.nyc.gov, all questions.

Bidders are hereby advised that this contract is subject to the Project Labor Agreement ("PLA"), entered into between the City and the Building and Construction Trades Council of Greater New York ("BCTG") affiliated local unions. Please refer to the bid documents for further information.

The procurement is subject to participation goals for MBEs and/or WBEs as required by Local Law 1.

4 percent M/WBE Subcontracting goals.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor prequalification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above.

Énvironmental Protection, 59-17 Junction Boulevard, 17th Floor, Bid Room, Flushing, NY 11373. Fabian Heras (718) 595-3265; fheras@dep.nyc.gov

Ä

• a19

REPAIR AND MAINTENANCE OF BOILERS AT VARIOUS WWTP'S, PUMP STATIONS AND ASSOCIATED DEP FACILITIES (SOUTH REGION) - Competitive Sealed Bids -PIN# 82619B0053 - Due 5-14-19 at 11:30 Å.M.

Project Number: 1478-BLR(R), Document Fee: \$80.00, Project Manager: Mohd Alam, Engineers Estimate: \$2,400,000.00 - \$3,300,000.00.

There will be a Pre-Bid on 4/26/19, at 11:00 A.M., located at 96-05 Horace Harding Expressway, 2nd Floor Conference Room, Collections Flushing, NY 11373. Last day for questions 5/1/19.

Please email Agency contact Fheras@dep.nyc.gov, all questions.

The procurement is subject to participation goals for MBEs and/or WBEs as required by Local Law 1.

Bidders are hereby advised that this contract is subject to the Project Labor Agreement ("PLA"), entered into between the City and the Building and Construction Trades Council of Greater New York ("BCTG") affiliated local unions. Please refer to the bid documents for further information.

3 percent M/WBE Subcontracting goals.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor prequalification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above.

Environmental Protection, 59-17 Junction Boulevard, 17th Floor, Bid Room, Flushing, NY 11373. Fabian Heras (718) 595-3265; fheras@dep.nyc.gov

L	2
ŀ	OT.

🕶 a19

HEALTH AND MENTAL HYGIENE

■ INTENT TO AWARD

Goods

CLARKE MOSQUITO CONTROL PRODUCTS INC. - Sole Source - Available only from a single source - PIN#20EN005001R0X00 - Due 4-29-19 at 11:30 A.M.

DOHMH, intends to enter into a Sole Source agreement with Clarke Mosquito Control Products Inc., for the purchase of The ANVIL and DUET products. In accordance with purchasing by DOHMH's Bureau of Environmental Health Administration, to ensure the prevention of illnesses related to environmental health risks. DOHMH has determined that Clarke Mosquito Control Products Inc., is the sole authorized distributor of MGk's The ANVIL, and The DUET products in the United States. There are no other authorized distributor of these products in the United States. Clarke Mosquito Control Products Inc., requires these uniquely formulated products, to ensure New York City has adequate means to protect the public from the introduction and spread of Zika virus and other mosquito-borne diseases.

Any questions regarding this sole source contract, should be addressed in writing to the contracting officer identified. Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor pre-qualification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above. Health and Mental Hygiene, 42-09 28th Street, 17th Floor, CN 110, Long Island City, NY 11101. Marcella Napolitano (347) 396-6680; Fax: (347) 396-6759; mnapolitano0210@gmail.com

a15-19

AGENCY CHIEF CONTRACTING OFFICE

■ INTENT TO AWARD

Goods **PURCHASE OF QUIDEL TESTING PRODUCTS** - Sole Source - Available only from a single source - PIN#20LB008301R0X00 -Due 4-26-19 at 11:00 A.M.

DOHMH, intends to enter into a Sole Source contract with QuidelCorporation, to procure D3, Ultra, Duet and Fastpoint L-DFA product lines (including service and maintenance). These testing equipments and reagents, will be used in the NYC Public Health Laboratory, for the detection of several different viruses including Influenza, Parainfluenzas, Herpes, VZV, and CMV. DOHMH has determined that Quidel Corporation is a sole source vendor for D3, Ultra, Duet, and Fastpoint L-DFA product lines as they are the sole manufacturer, and sole provider of these products. Any vendor who believes they can provide these products are welcome to submit an expression of interest via email. All questions and concerns regarding this sole source should also be submitted via email.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor pre-qualification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above. Health and Mental Hygiene, 42-09 28th Street, 17th Floor, Long Island City, NY 11101. Chassidi Miner (347) 396-6754; Fax: (347) 396-6758; cminer@health.nyc.gov

a15-19

HOUSING AUTHORITY

■ SOLICITATION

Construction / Construction Services

LOCAL LAW 11 CONTRACT FOR: EXTERIOR RESTORATION AT VARIOUS DEVELOPMENTS - Competitive Sealed Bids -PIN#BW1835764 - Due 5-10-19 at 11:00 A.M. RFQ#68310

There will be a Pre-Bid Meeting on 4/26/2019, at 11:00 A.M., at 90 Church Street, 11th Floor, Room 11-002, New York, NY 10007. Although attendance is not mandatory, it is strongly recommended that you attend. NYCHA staff will be available to address all inquiries relevant to this contract.

• ROOFING, RAILING, REPLACEMENT AND ROOFTOP AT FOREST HOUSES - Competitive Sealed Bids - PIN#RF1803947 -Due 5-10-19 at 11:30 A.M.

There will be a Pre-Bid Meeting on 4/26/2019, at 10:00 A.M., at 1010 Trinity Avenue, Bronx, NY 10456. Although attendance is not mandatory, it is strongly recommended that you attend. NYCHA staff will be available to address all inquiries relevant to this contract.

Bid documents can be picked up and dropped off Monday through Friday, 9:00 A.M. to 4:00 P.M., on the 6th Floor, CPD Bid Reception Window for a \$25.00 fee. Documents can also be obtained by registering with I-Supplier and downloading documents. Please note that original bid bonds are due at time of bid opening.

Please note that in the event only one bidder has submitted a bid in connection with the contract on or before the original bid submission deadline, the bid submission deadline shall automatically be extended for fourteen (14) calendar days. The foregoing extension does not in any way limit NYCHA's right to extend the bid submission deadline for any other reason.

These contracts shall be subject to the New York City Housing Authority's Project Labor Agreement, if the Bidder's price exceeds \$250,000.00.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor prequalification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above. Housing Authority, 90 Church Street, New York, NY 10007. Latrena Johnson (212) 306-3223; latrena.johnson@nycha.nyc.gov

🖝 a19

SUPPLY MANAGEMENT

■ SOLICITATION

Construction / Construction Services

SMD REPLACEMENT OF SUBMERSIBLE WASTEWATER SEWAGE PUMPS - MARCY HOUSES - BROOKLYN - Request for Quote - PIN#68245 - Due 5-9-19 at 10:00 A.M.

Removal of six (6) existing submersible wastewater sewage pumps: (EBARA Model 150 DLFU 6152-20 HP), valves and associated ancillary equipment.

Please note: This contract will be subject to the New York City Housing Authority's Project Labor Agreement (PLA), if a Bidder's quote total is \$250,000.00 or more. As part of its bid and no later than three (3) business days after the bid opening, the Bidder must submit Letters of Assent to the Project Labor Agreement, signed by the Bidder and each of the Bidder's proposed Subcontractors. Failure to submit all required signed Letters of Assent within three (3) business days after the bid opening, shall result in a determination that the Bidder's bid is non-responsive.

Interested firms may obtain a copy and submit it on NYCHA's website: Doing Business With NYCHA. http://www.nyc.gov/html/nycha/html/ business/business.shtml: Vendors are instructed to access the "Register Here" link for "New Vendors"; if you have supplied goods or services to NYCHA in the past and you have your log-in credentials, click the "Log into iSupplier" link under "Existing Vendor". If you do not have your log-in credentials, click the "Request a Log-in ID" using the link under "Existing Vendor". Upon access, reference applicable RFQ number per solicitation.

Vendors electing to submit a non-electronic bid (paper document), will be subject to a \$25 non-refundable fee; payable to NYCHA by USPS-Money Order/Certified Check only for each set of RFQ documents requested. Remit payment to NYCHA Finance Department, 90 Church Street, 6th Floor; obtain receipt and present it to 6th Floor, Supply Management Department Procurement Group. A bid package will be generated at the time of request. Note (*): Vendor/Supplier submitting sealed non-electronic ("paper") bids must include a copy of your receipt as proof of purchase.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor prequalification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above.

Housing Authority, 90 Church Street, 6th Floor, New York, NY 10007. Mimose Julien (212) 306-8141; Fax: (212) 306-5109; mimose.julien@nycha.nyc.gov

🕶 a19

HUMAN RESOURCES ADMINISTRATION

OFFICE OF CONTRACTS

■ INTENT TO AWARD

Human Services / Client Services

IMMIGRANT OPPORTUNITY INTIATIVE PROGRAM (IOI) - Renewal - Due 4-22-19 at 5:00 P.M.

PIN	#09618]	[0008002	R001 -	LEGAL	SERVICE	S NYC
PIN	#09615]	[0016001	R001 -	THE LE	GAL AID	SOCIETY
DIN	# 0001 F1	001000	DOO1	TIDDAN	TIOMICE	CENTED

PIN#09615I0016002R001 ·	URBAN JUSTICE CENTER
DINTUGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGGG	APPRILIC DIT I DO LI LOOION

PIN#09618I0008001R001 - NEW YORK LEGAL ASSISTANCE GROUP INC.

The New York City Human Resources Administration through its Office of Civil Justice, intends to renew four (4) contracts with contractors that currently provide legal assistance to low-income immigrant New Yorkers, as part of the Immigrant Opportunity Initiative Program (IOI). The renewal term of the contracts will be from 7/1/19 to 6/30/22. Anyone having comments on the contractors' performance or the proposed renewal of the contracts, may contact Annabelle Villegas at (929) 221-5398. This Notice is for informational purposes only.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor prequalification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above.

Human Resources Administration, 150 Greenwich Street, 31st Floor, New York, NY 10007. Annabelle Villegas (929) 221-5398; villegasan@hra.nyc.gov

🕶 a19

AWARD

Human Services / Client Services

PROVISION OF SOCIAL SERVICES AND COMMUNITY LEADERSHIP DEVELOPMENT TO LGBT AND HIV-AFFECTED SURVIVORS OF ALL FORMS OF VIOLENCE. - BP/City Council Discretionary - PIN#09619L0107001 - AMT: \$128,500,00 - TO: NYC Gay and Lesbian Anti-Violence Project, 116 Nassa Street, 3rd Floor, New York, NY 10038. Term: 7/1/2018 - 6/30/2019

🕶 a19

PARKS AND RECREATION

VENDOR LIST

Construction Related Services

PREQUALIFIED VENDOR LIST: GENERAL CONSTRUCTION, NON-COMPLEX GENERAL CONSTRUCTION SITE WORK ASSOCIATED WITH NEW YORK CITY DEPARTMENT OF PARKS AND RECREATION ("DPR" AND/OR "PARKS") PARKS AND PLAYGROUNDS CONSTRUCTION AND **RECONSTRUCTION PROJECTS**

NYC DPR is seeking to evaluate and pre-qualify a list of general contractors (a "PQL") exclusively to conduct non-complex general construction site work involving the construction and reconstruction of NYC DPR parks and playgrounds projects not exceeding \$3 million per contract ("General Construction").

By establishing contractor's qualification and experience in advance, NYC DPR will have a pool of competent contractors from which it can draw to promptly and effectively reconstruct and construct its parks, playgrounds, beaches, gardens and green-streets. NYC DPR will select contractors from the General Construction PQL for non-complex general construction site work of up to \$3,000,000.00 per contract through the use of a Competitive Sealed Bid solicited from the PQL generated from this RFQ.

The vendors selected for inclusion in the General Construction PQL, will be invited to participate in the NYC Construction Mentorship. NYC Construction Mentorship focuses on increasing the use of small NYC contracts, and winning larger contracts with larger values. Firms participating in NYC Construction Mentorship will have the opportunity to take management classes and receive on-the-job training provided by a construction management firm.

NYC DPR will only consider applications for this General Construction PQL from contractors who meet any one of the following criteria:

- 1) The submitting entity must be a Certified Minority/Woman Business enterprise (M/WBE)*:
- 2) The submitting entity must be a registered joint venture or have a valid legal agreement as a joint venture, with at least one of the entities in the joint venture being a certified M/WBE*;
- 3) The submitting entity must indicate a commitment to sub-contract no less than 50 percent of any awarded job to a certified M/WBE for every work order awarded.

Firms that are in the process of becoming a New York City-Certified MWBE, may submit a PQL application and submit a M/WBE Acknowledgement Letter, which states the Department of Small Business Services has begun the Certification process.

Application documents may also be obtained online at: http://a856-internet.nyc.gov/nycvendoronline/home.asap.; or http:www.nycgovparks.org/opportunities/business.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor pre-qualification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above. Parks and Recreation, Olmsted Center Annex, Flushing Meadows – Corona Park, Flushing, NY 11368. Alicia H. Williams (718) 760-6925; Fax: (718) 760-6885; dmwbe.capital@parks.nyc.gov

i2-d31

AWARD

Goods VELOCITY FLOW METER AND RICE LAKE RESEARCH **EQUIPMENT** - Innovative Procurement - Other - PIN#219963846 -AMT: \$27,650.07 - TO: Gus Paul Swimming Pools Inc., 16 Sintsink Drive East, Port Washington, NY 11050. Greyline area velocity flow meter 50 feet and 25 feet. It includes local readout standard sensor, cables, relays and 4-20 MADC analog outputs with a 9-32 VDC Power input. SDX Pressure sensor and desiccant cartridge.

Rice Lake 4-20 plus indicator with analog output option card with Platform, remote junction box and cables. All includes installation, calibration and to be fabricated. ZL 6 Cellular data logger to allow access to data via internet. Usage with all meter sensors and solar charging.

Contracts awarded, pursuant to the Innovative Procurement Method, under PPB Rule 3-12 (M/WBE Purchase Method).

🕶 a19

REVENUE

■ SOLICITATION

Services (other than human services)

REQUEST FOR PROPOSALS FOR AN OUTDOOR CAFE AND SNACK BAR, CONSERVATORY WATER, CENTRAL PARK - Competitive Sealed Proposals - Judgment required in evaluating proposals - PIN# M10-15SB-2019 - Due 6-3-19 at 3:00 P.M.

In accordance with Section 1-13 of the Concession Rules of the City of New York, the New York City Department of Parks and Recreation ("Parks"), is issuing, as of the date of this notice, a significant Request for Proposals ("RFP"), for the renovation, operation and maintenance of an outdoor café and snack bar, at Conservatory Water in Central Park, Manhattan.

There will be a recommended proposer site tour on Tuesday, April 23, 2019, at 11:30 A.M. We will be meeting at the proposed concession site (Block # 1111 and Lot #1), Conservatory Water, which is located in Central Park at about East 74th Street, in the borough of Manhattan. If you are considering responding to this RFP, please make every effort to attend this recommended meeting. All proposals submitted in response to this RFP, must be submitted no later than Monday, June 3, 2019, at 3:00 P.M.

Hard copies of the RFP can be obtained, at no cost, commencing on Friday, Åpril 12, 2019 through Monday, June 3, 2019, between the hours of 9:00 A.M. and 5:00 P.M., excluding weekends and holidays, at the Revenue Division of the New York City Department of Parks and Recreation, which is located at 830 Fifth Avenue, Room 407, New York, NY 10065.

The RFP is also available for download, on Friday, April 12, 2019 through Monday, June 3, 2019, on Parks' website. To download the RFP, visit http://www.nyc.gov/parks/businessopportunities, and click on the "Concessions Opportunities at Parks" link. Once you have logged in, click on the "download" link that appears adjacent to the RFP's description.

For more information or to request to receive a copy of the RFP by mail, prospective proposers may contact Eric Weiss, Project Manager, at (212) 360-3483, or at eric.weiss@parks.nyc.gov.

TELECOMMUNICATION DEVICE FOR THE DEAF (TDD) (212) 504-4115.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor pre-qualification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above. Parks and Recreation, The Arsenal, Central Park, 830 Fifth Avenue, Room 407, New York, NY 10065. Eric Weiss (212) 360-3483; Fax: (917) 849-6639; eric.weiss@parks.nyc.gov

a12-25

PROBATION

CONTRACT PROCUREMENT

AWARD

Human Services / Client Services

ICM PLUS STATEN ISLAND - Competitive Sealed Proposals/ Pre-Qualified List - Judgment required in evaluating proposals -PIN#78119I0002001 - AMT: \$362,916.67 - TO: New York Center for Interpersonal Development, Inc., 130 Stuyvesant Place, 5th Floor, Staten Island, NY 10301.

The contractor has been awarded by means of the HHS Accelerator Method, pursuant to Section 3-16 of the Procurement Policy Board Rules.

SANITATION

AWARD

AGENCY CHIEF CONTRACTING OFFICE

Goods and Services

ENERGY EFFICIENT LIGHTING PARTS - Innovative Procurement - Other - PIN# 20194090383 - AMT: \$100,000.00 - TO: Edge Electronics, 75 Orville Drive, Bohemia, NY 11716. MWBE Award.
 SOLARWINDS SOFTWARE - Innovative Procurement - Other - PIN# 20191600131 - AMT: \$49,400.00 - TO: Quality and Assurance Disheductory Core 12 Marine Pidem Divide Piden NY 11961

Technology Corp., 18 Maringewood Drive, Ridge, NY 11961. MWBE Award. 🕶 a19

DULEVO SWEEPER REPLACEMENT PARTS - Innovative Procurement - Other - PIN#20195090242 - AMT: \$100,000.00 - TO: Mudu Multimedia Enterprises, 1369 Broadway, Suite 210849, Brooklyn, NY 11211. MWBE Award.

🕶 a19

SCHOOL CONSTRUCTION AUTHORITY

CONTRACT SERVICES

■ INTENT TO AWARD

Human Services / Client Services

REQUEST FOR EXPRESSION OF INTEREST (RFQEI) FOR PAYROLL PROCESSING SERVICES - Request for Qualifications - PIN# 19-XXXXX - Due 5-1-19 at 4:30 P.M.

This is a Request for Expression of Interest by the New York City School Construction Authority ("SCA"), to conduct a preliminary survey seeking responses from potential proposers, who are interested in and qualified to provide Payroll Processing Services.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor prequalification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above.

School Construction Authority, 30-30 Thomson Avenue, First Floor, Long Island City, NY 11101. June Thompson (718) 752-5229; jthompson@ nvcsca.org

a17-23

TRANSPORTATION

ADMINISTRATION

■ SOLICITATION

Services (other than human services)

EVENT PRODUCTION, SPONSORSHIP RECRUITMENT AND MARKETING SERVICES - Request for Proposals -PIN#84119MBAD319 - Due 5-22-19 at 2:00 P.M.

This procurement is subject to participation goals for Minority-Owned Business Enterprises (MBEs), as required by Section 6-129 of the New York Administrative Code. The M/WBE goal for this project, is 24 percent. A printed copy of the proposal can also be purchased. A deposit of \$50.00 is required for the proposal documents in the form of a Certified Check or Money Order payable to: New York City Department of Transportation. NO CASH ACCEPTED. Company address, telephone and fax numbers are required when picking up contract documents. (Entrance is located on the South Side of the Building facing the Vietnam Veterans Memorial). Proper government issued identification is required for entry to the building (driver's license, passport, etc.). A Pre-Proposal Conference (Optional), will be held on April 29, 2019, at 10:00 A.M., at 55 Water Street, Ground Floor Conference Room, New York, NY 10041. The deadline for submission of written questions is May 1, 2019, by 5:00 P.M. For additional information, please contact David Maco, at (212) 839-9400.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor prequalification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time

specified above. Transportation, Contract Management Unit, 55 Water Street, Ground Floor, New York, NY 10041. Bid Window (212) 839-9435.

Ť

YOUTH AND COMMUNITY DEVELOPMENT

■ INTENT TO AWARD

Goods and Services

CACFP FOOD RENEWAL - Renewal - PIN#26016B0001001 - Due 4-22-19 at 9:00 A.M.

In accordance with Section 4-04 of the of the Procurement Policy Board rules, the Department of Youth and Community Development, intends to renew the Adult Care Food Program, at Community Centers, located in various New York City Housing Authority developments. The contractor will provide and deliver to the Centers complete, safe and nutritious meals that conform to the United States Department of Agriculture Required Meal Patterns and Food Specifications, and also conform to the New York City Food Standards. The term of this contract shall be July 1, 2019 through June 30, 2020. The contractor's, contract number, address and PIN number is indicated below;

EPIN: 26016B0001001 AMOUNT: \$3,554,400 The Maramont Corporation 5600 First Avenue Brooklyn, NY 11220

Please be advised that this publication is for information purposes only. If you wish to contract DYCD for further information, please send an email to ACCO@dycd.nyc.gov.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor prequalification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above.

Youth and Community Development, 2 Lafayette Street, New York, NY 10007. Dana Cantelmi (646) 343-6330; acco@dycd.nyc.gov

🖝 a19

CONTRACT AWARD HEARINGS

NOTE: LOCATION(S) ARE ACCESSIBLE TO INDIVIDUALS USING WHEELCHAIRS OR OTHER MOBILITY DEVICES. FOR FURTHER INFORMATION ON ACCESSIBILITY OR TO MAKE A REQUEST FOR ACCOMMODATIONS, SUCH AS SIGN LANGUAGE INTERPRETATION SERVICES, PLEASE CONTACT THE MAYOR'S OFFICE OF CONTRACT SERVICES (MOCS) VIA EMAIL AT DISABILITYAFFAIRS@MOCS.NYC.GOV OR VIA PHONE AT (212) 788-0010. ANY PERSON REQUIRING REASONABLE ACCOMMODATION FOR THE PUBLIC HEARING SHOULD CONTACT MOCS AT LEAST THREE (3) BUSINESS DAYS IN ADVANCE OF THE HEARING TO ENSURE AVAILABILITY.

YOUTH AND COMMUNITY DEVELOPMENT

■ PUBLIC HEARINGS

NOTICE IS HEREBY GIVEN that a Contract Public Hearing, will be held, on Friday, May 3, 2019, at 2 Lafayette Street, 14th Floor Hearing Room, in the Borough of Manhattan, commencing at 10:00 A.M., on the following:

IN THE MATTER OF (1) proposed FY19 Tax Levy Discretionary contract between the Department of Youth and Community Development, and the Contractor listed below, for services related to youth and family development services, along with special events and social activities to community residents. The term of the contract shall be from July 1, 2018 to June 30, 2019, with no option to renew.

The Contract numbers, Contractor, Contract amount and Contractor address is indicated below.

		Contract	
<u>Contract #</u>	Contractor Name	Amount	Address
26019022751Q	Bergen Basin Community Development Corporation	\$110,950.00	2331 Bergen Avenue Brooklyn, New York 11234

The proposed award is being funded through Line Item Appropriations or Discretionary Funds, pursuant to Section 1-02 (e) of the Procurement Policy Board (PPB) Rules.

A draft copy of the proposed contract will be available for public inspection, at the Department of Youth and Community Development, Office of the Agency Chief Contracting Office, at 2 Lafayette Street, 14th Floor, from April 19, 2019 to May 3, 2019, during the hours of 9:00 A.M. to 5:00 P.M. excluding Saturdays, Sundays and legal holidays.

	🖝 a19

NOTICE IS HEREBY GIVEN that a Contract Public Hearing, will be held on Friday, May 3, 2019, in Conference Room 1421, at the office of the Department of Youth and Community Development, 2 Lafayette Street, New York, NY 10007, commencing at 10:00 A.M., on the following:

IN THE MATTER OF the proposed contract between the Department of Youth and Community Development, and the contractor listed below, to provide Payroll related services for DYCD's Summer Youth Employment Program (SYEP) participants. The term will be from April 1, 2019 to March 31, 2020.

The provider's name, address, and amount is as follows:

PIN: 26011P0003002N002 Amount: \$1,993,833.00 Contractor: Community Software Solutions, Inc. Address: 30 Jefferson Plaza, Princeton, New Jersey 08540

The proposed contractor is being selected by Negotiated Acquisition Extension, pursuant to Section 3-04 (b)(2)(iii) of the Procurement Policy Board Rules.

A Draft copy of the proposed contract will be made available for public inspection, at the Department of Youth and Community Development, Office of the Agency Chief Contracting Officer, 2 Lafayette Street, 14th Floor, New York, NY 10007, on business days between the hours of 9:00 A.M. and 5:00 P.M., from April 19, 2019 to May 3, 2019, excluding weekends and holidays.

🕶 a19

NOTICE IS HEREBY GIVEN that a contract public hearing, will be held on Friday, April 26, 2019, in Conference Room 1421, at the office of the Department of Youth and Community Development, 2 Lafayette Street, New York, NY 10007, commencing at 10:00 A.M., on the following:

IN THE MATTER OF the proposed contract between the Department of Youth and Community Development and the contractor listed below, to operate a Drop-In Center for Runaway Homeless Youth (RHY), and youth at risk for homelessness in New York City (City) and their families. Drop-In Centers have a broad role. They are resource hubs for the whole borough in which they are located, and they conduct outreach, provide information, and educate the public about youth homelessness and its causes. They also provide a range of direct services for RHY and youth at risk for homelessness and their families. These services include provision of basic amenities, crisis intervention, needs assessments, counseling and case management services. The term will be from January 1, 2019 to June 30, 2022, with options to renew up to two additional years.

PIN: 260190009565

Amount: \$1,310,854.00

Contractor: Sheltering Arms Children and Family Services, Inc. Address: 305 $7^{\rm th}$ Avenue, New York, NY 10001

The proposed contractor is being selected by Competitive Sealed Proposals, pursuant to Section 3-16 HHS Accelerator Method of the Procurement Policy Board (PPB) Rules.

A draft copy of the proposed contract will be made available for public inspection at the Department of Youth and Community Development, Office of the Agency Chief Contracting Officer, 2 Lafayette Street, 14th Floor, New York, NY 10007, on business days between the hours of 9:00 A.M. and 5:00 P.M., from April 19, 2019 to April 26, 2019, excluding weekends and holidays.

AGENCY RULES

CAMPAIGN FINANCE BOARD

■ NOTICE

Notice of Public Hearing and Opportunity to Comment on Proposed Rules

What are we proposing? We are proposing amendments to the Campaign Finance Board's ("Board") rules regarding disclosure, contributions, expenditures, public funds payments, disclosure of independent expenditures, and various other aspects of the New York City Campaign Finance Program (the "Program"). These amendments are being made to clarify and update content to help support public understanding of, and compliance with, the Board rules. The proposed rules also include minor plain language revisions and reorganization of text.

When and where is the hearing? The Board will hold a public hearing on the proposed rules. The public hearing will take place at 1:00 P.M., on June 13, 2019. The hearing will be in the Board's board room at 100 Church Street, 12th Floor, New York, NY 10007.

This location has the following accessibility option(s) available:

- Wheelchair Accessible
- Sign Language Interpretation

How do I comment on the proposed rules? Anyone can comment on the proposed rules by:

- Website. You can submit comments to the Board through the NYC rules website at http://rules.cityofnewyork.us.
- Email. You can email written comments to Rules@nyccfb.info.
- Mail. You can mail written comments to Hillary Weisman, General Counsel, Campaign Finance Board, 100 Church Street, 12th Floor, New York, NY 10007.
- Fax. You can fax written comments to the Board at (212) 409-1705.
- Speaking at the hearing. Anyone who wants to comment on the proposed rules at the public hearing must sign up to speak. You can sign up before the hearing by calling Hillary Weisman, General Counsel, at (212) 409-1858. You can also sign up in the hearing room before the hearing begins. You may speak for up to three minutes.

Is there a deadline to submit written comments? Yes, written comments must be submitted by June 12, 2019.

Do you need assistance to participate in the hearing? You must tell Hillary Weisman if you need a reasonable accommodation of a disability at the hearing. You must tell us if you need a sign language interpreter. You can tell us by mail at the address given above. You may also tell us by telephone at (212) 409-1800. You must tell us by June 10, 2019.

Can I review the comments made on the proposed rules? You can review the comments made online on the proposed rules by going to the website at http://rules.cityofnewyork.us/. A few days after the hearing, copies of all comments submitted online, copies of all written comments, and a summary of oral comments concerning the proposed rules will be available to the public on the Board's website at <u>www.</u> nyccfb.info.

What authorizes the Board to make these rules? Sections 1043, 1052(a)(8), and 1052(a)(15) of the New York City Charter ("City Charter") and Sections 3-701 *et seq.* of the Administrative Code of the City of New York authorize the Board to make these proposed rules. These rules were included in the Board's regulatory agenda for the 2019 fiscal year.

Where can I find the Board's rules? The Board's rules are in title 52 of the Rules of the City of New York and on the Board's website at http://www.nyccfb.info/act-program/rules/.

What laws govern the rulemaking process? The Board must meet the requirements of Section 1043 of the City Charter when creating or changing rules. This notice is made according to the requirements of Section 1043 of the City Charter.

Statement of Basis and Purpose of Proposed Rules

The Campaign Finance Board ("CFB" or "the Board") is a nonpartisan, independent City agency that empowers New Yorkers to make a greater impact in elections. The CFB administers the City's campaign finance system, overseeing and enforcing the regulations related to campaign finance and holding candidates accountable for using public funds responsibly. The CFB publishes detailed public information about money raised and spent in City elections by candidates and independent spenders, and engages and educates voters through community outreach, the Voter Guide, and the Debate Program.

The CFB is proposing amendments to its rules in order to clarify and update content to help support public understanding and compliance. While some substantive changes are proposed, the majority of the amendments would make minor revisions and reorganize text for clarity.

The following is a summary of the changes.

Summary of Proposed Rules

Throughout the rules

The chapters are reorganized to correspond roughly with the chronology of a typical campaign. Sections concerning special and runoff elections are moved into separate chapters so that candidates in such elections can use those chapters as a quick reference and brief overview of their responsibilities. Certain definitions previously contained in other chapters are moved into Chapter 1. The only definitions remaining in other chapters are for terms that may have a different meaning in the context of that particular chapter than in the rest of the Board rules.

Certain sections whose substance was duplicative are combined, and sections containing confusing or unclear phrasing are edited for clarity.

Short titles are added where possible to clarify the purpose and scope of individual rules.

Changes are made to standardize punctuation, capitalization, numbering, and other grammatical and stylistic issues, in order to conform to the CFB's internal style guide.

Certain amendments are made to allow for evolving technology and banking practices. For example, filing requirements are described more broadly to encompass both printed and electronic submissions, and all references to "charge cards" are deleted.

Use of the words "candidate" and "participant" is standardized, with preference given to the word "candidate." For example, it is not necessary to use the word "participant" in a section concerning public funds payments, as such sections are by definition applicable only to participating candidates. Additionally, the word "campaign," which is not legally defined, is replaced wherever possible with the term "candidate," whose definition includes the treasurer, authorized committee, and any agent of the candidate, thus effectively encompassing the campaign in most instances. The word "committee" is used in instances where the use of "candidate" could cause ambiguity as to whether the reference was to the candidate in the candidate's personal capacity and, in specific instances described below, the phrase ^{*}participant or nonparticipant or an agent of the participant or nonparticipant" is used to refer to the individual candidate, to the exclusion of the treasurer and committee. Moreover, in most instances, where a section referenced "the candidate or an agent of the candidate," the phrase "or an agent of the candidate" is removed, as the definition of "candidate" includes all agents thereof. In certain instances, where it was deemed necessary for clarity, the reference to a candidate's agent remains in the section.

All references to limited participants, and sections that apply only to such candidates, are removed. Pursuant to Section 3-718 of the Campaign Finance Act (the "Act"), a candidate may register as a limited participant. Limited participants do not receive public funds, but are subject to all applicable requirements of the Act and Board rules, including (unlike non-participants) the expenditure limits, except that limited participants may contribute an unlimited amount to their campaigns from their own personal funds. Since 2004, when the Act was amended to add section 3-718, not a single candidate has opted to register as a limited participant. Candidates who register as limited participants remain subject to all applicable provisions in the Act and the Board rules.

Similarly, all references to disclosure statement resubmissions are removed, as the amendment process subsumes any resubmission that would be made.

The proposed rules also include plain language revisions throughout.

Chapter 1: Definitions and General Provisions

Previous section 11-01, detailing the scope of Chapter 11, regarding Transition and Inauguration Entities ("TIEs"), is incorporated into Section 1-01, which details the scope of each chapter.

Section 1-02 is amended to add a definition for the terms "loan" and "segregated account," and to separate out the definition of

"other receipts" from the definition of "receipts." Definitions are added for the following terms, which apply in the context of TIEs: "donation," "inauguration expenses," "transition and inauguration entity," and "transition expenses." The latter definition incorporates the presumption (previously contained in Section 11-04(e)(ii)) that incumbent elected candidates do not have transition expenses. Additionally, the definitions of "Certification" and "Filer Registration" are amended to reflect evolving technological processes, which may include electronic submission of information in lieu of paper forms. The definition of "fair market value," previously located in the chapter governing independent expenditures, is moved to section 1-02, as the term appears elsewhere in the rules. The definitions of "federal form," "multicandidate committee," and "rule" are deleted, as the terms are used rarely in the rules and the definitions are not ambiguous. The definition of "iunspent campaign funds" is deleted, because the Board no longer seeks to collect unspent campaign funds as previously defined.

Section 1-03(b) (previously Section 7-06) is amended to delete extraneous language regarding the ethical standards of the Board.

Sections 1-04(a) and (b) are added to clarify the Board's standards for computation of days and procedures for submitting documentation in accordance with Board deadlines. Additionally, section 1-04(b) is amended to delete previous section 1-09(a)(3)(iv), which stated that a candidate who fails to deliver a complete disclosure statement in a timely manner is in violation of the Act and subject to penalty. This provision was redundant with other provisions of the Act and rules requiring complete and timely disclosure, and stating that any violation of the Act and rules may be the basis for a penalty determination.

Chapter 2: Registration and Certification

Section 2-02(d) (previously Section 2-01(c)) is amended to allow candidates who have already submitted a Filer Registration to file a Certification in electronic format, unless a new treasurer has been appointed.

Section 2-03 (previously Sections 1-11(b) and 2-01(d)) is amended to require that, if any information required to be listed on the Filer Registration or Certification changes within five years after the election, the candidate must submit an amendment within 30 days of the change. Previously the prescribed period was within six years of the election, and there were certain exceptions to the 30-day deadline. The five-year requirement is consistent with the new record retention requirement (*see* section 4-02).

Section 2-04 (previously Section 2-11) is amended to remove a reference to non-participants being permitted to accept contributions from unregistered political committees, consistent with the amendment to section 3-707 of the Act enacted by Local Law Number 171 for the year 2016.

Section 2-06 (previously Section 2-12) is amended to provide that candidates must attend the required training session on or before the deadline for submission of the 32-day pre-election primary disclosure statement. Additionally, the section is amended to require all candidates to attend the pre-election training; previously, the requirement applied only to participants.

Chapter 3: Eligibility to Receive Public Funds

Section 3-01(c) (previously Section 5-01(g)) is amended to delete the statement that a payment of public funds does not constitute the final determination of the amount of public funds for which the candidate is eligible. This statement is contained in Section 7-01(a).

Section 3-01(d) (previously Section 5-01(f)) is amended to clarify that public funds may be denied if a candidate has committed fraud or material misrepresentation, been found in breach of certification, or engaged in conduct detrimental to the Program that is in violation of any other applicable law at any time in the course of that candidate's Program participation, which may include previous election cycles. Additionally, the requirement that, in order to be eligible to receive public funds, a candidate who loses the primary election but remains on the ballot for the general must certify to the Board that such candidate will actively campaign for office in the general election, is amended to require such certification on or before the 32-day pregeneral election disclosure statement deadline.

Finally, Section 3-01(d) is bifurcated to distinguish between bases for denial of public funds that apply only to a pre-election public funds payment, and those that apply to both pre-election and post-election public funds payments. A candidate may be deemed ineligible to receive a pre-election public funds payment, but not a post-election payment, based on: failing to file a disclosure statement, failing to provide documentation or records in response to a Board request, and if there is reason to believe that the candidate has committed a violation of the Act and/or Board rules not otherwise listed in this section. Failing to file a disclosure statement, failing to provide documentation or records, and committing other violations of the Act and Board rules are in this category because they can be remedied, and prior to the election, denying a public funds payment to a candidate who fails to demonstrate compliance creates an incentive for the candidate to resolve the issue as quickly as possible in order to become eligible to receive public funds. For example, if the review of a candidate's pre-election disclosure statement reveals a possible over-the-limit or prohibited contribution, a pre-election payment could be denied based on there being reason to believe that the candidate has committed a violation of the Act and Board rules. This encourages the candidate to refund the contribution immediately, which promotes fairness and reduces the corrupting influence of large contributions by minimizing the amount of time for which the candidate's campaign had the benefit of those funds. However, after the election, the timing of the refund is less important because the candidate's opponent is no longer at a greater disadvantage based on how long the candidate has the benefit of the contribution. This also applies to the filing of disclosure statements; the value of public disclosure is greatest prior to the election, so that the voters have as much information as possible when making their choice.

Moreover, after the election, the candidate would be subject to a penalty for accepting and failing to refund the contribution. The same is true of failing to file a disclosure statement and failing to provide documents and records. The only conduct that may result in a penalty as well as a denial of public funds both pre-election and post-election is the type of conduct underlying the most serious violations of the Act and Board rules, which effectively cannot be remedied: exceeding the expenditure limit, fraud or material misrepresentation, and any other action that may cause a finding of breach of certification pursuant to Section 3-01(e). In addition to being severe enough to warrant ineligibility to receive public funds on top of the penalty, such conduct is frequently not known to the Board until the post-election audit, so it would not be reasonable to limit the resulting public funds ineligibility to the pre-election 3-01(e), which includes the remaining bases for eligibility listed in previous Section 5-01(f).

Section 3-02 (previously Section 2-07) is amended to remove the requirement that candidates who are disqualified from the ballot provide immediate notification of such disqualification to the Board, as the CFB itself generally collects the information.

Section 3-04 (previously Section 2-09) is amended to require that candidates notify the Board of a request for reversal of termination of their candidacy within five days of receiving the notice of termination. Previously the request was required to be submitted within 10 days. Section 3-04 is also amended to provide that the Board will notify the candidate of the termination within five business days of receiving a Verification of Terminated Candidacy form. Finally, the section is amended to remove separate references to limited participants and non-participants, as its provisions are applicable to all candidates.

Section 3-05(a) (previously Section 3-11(a)) is amended to provide that candidates must demonstrate compliance with Conflicts of Interest Board ("COIB") reporting requirements on or before the deadline established by the COIB or by the reporting deadline immediately preceding the date on which the candidate seeks to receive a public funds payment, whichever is earlier. This change is made to provide for a situation in which the COIB deadline is set on a date following the date on which the CFB makes determinations for a particular public funds payment.

Chapter 4: Records and Reporting

Section 4-01(b)(ii)(E) (previously Section 4-01(c)) is amended to clarify that the fair market value of an in-kind contribution must be reported.

Section 4-01(c)(iii) (previously Section 4-01(e)(3)) is amended to remove language concerning "charge cards" and to clarify that receipts provided for credit card purchases must include information about the underlying purchase, including but not limited to the vendor.

Section 4-01(c)(iv) (previously Section 4-01(d)) is amended to clarify that candidates must maintain written documentation showing that a bill has been forgiven or settled for a lesser amount. Additionally, the requirement that an additional record created be "contemporaneous" is removed and incorporated by the reference to Section 4-01(a), which provides for the creation of contemporaneous records.

Section 4-01(c)(vi) (previously Sections 4-01(e)(4), 4-01(m)) is amended to require that a record of an advance must be signed by the individual making the advance purchase. Additionally, the requirement that candidates obtain vouchers for advance reimbursements is deleted.

Section 4-01(c)(ix) (previously Section 4-01(o)) is amended to provide that the documentation exception for public transportation within New York City does not apply in its entirety to unlimited-use MetroCards. For such cards, candidates must maintain a contemporaneous record containing the date on which each card was purchased and the name of any individual who was assigned a particular card for long-term use.

Section 4-01(j), which required candidates to maintain a list identifying the address of each campaign office, is deleted.

Section 4-02 (previously Section 4-03(a)) is amended to change the required record retention period from six years after the date of the election to five years after the last disclosure statement deadline for

that election, or until the Board has issued the candidate's final audit report and the candidate has extinguished all outstanding liabilities resulting from the applicable election including payment of any penalties or repayment of public funds owed to the Board, whichever is later.

Section 4-03(b), which required candidates to notify the Board in writing of any change in the location of their records on or before the deadline for the next disclosure statement, is deleted.

Section 4-05(a) (previously Section 9-01) is amended to require that all backup documentation accompanying disclosure statements, including bank statements, be submitted electronically via the CFB's C-SMART (Candidate Software for Managing and Reporting Transactions) software. Section 4-05(a)(i) (previously Sections 1-09(c), 3-07, and 9-03(b)) is amended to clarify that a disclosure statement may (rather than will) be deemed incomplete and rejected if it does not meet certain submission requirements. Section 4-05(a)(iii) (previously Sections 3-08 and 9-03(a)) is amended to remove the provision for non-electronic submission of disclosure statements, as disclosure statements are now required to be submitted electronically. Section 4-05(a)(iv) (previously Section 9-02(a)(1)) is amended to remove the exemption for small campaigns from the electronic filing requirements, as small campaigns are not required to file disclosure statements at all. Section 3-06, which stated that disclosure statements must be submitted in such form and manner as shall be provided by the Board, is deleted, as its substance is incorporated into section 4-05(a)(i). Sections 9-02(a)(2), which provided that state and federal forms may not be submitted to the Board unless specifically permitted, and 9-03(c), which required that electronic submission be accompanied by printed versions, are deleted.

Section 3-02(f)(3), which provided for deferred filing for candidates who raised less than \$2,000 and spent not more than 45% of the expenditure limit since the last disclosure statement, is deleted.

Section 4-05(c)(ii) (previously Sections 1-09(c), 3-03(c)(1), 3-03(c)(4), 3-04(a), 3-04(b), and 3-04(c)) is amended to delete the requirement that disclosure statements indicate whether the candidate has met the threshold for eligibility to receive public funds; to provide that candidates must itemize contributions received from a contributor who is an employee of the candidate, even if such contributions total \$99 or less; to clarify that for each contribution returned to a contributor, candidates must report the account from which the funds originated and the number of the refund check used; to require that matchable contributions be reported in the disclosure statement covering the period in which they are received; and to clarify that for each contribution received, candidates must indicate in their reporting if the contribution was accepted for a runoff or rerun election or for a segregated account.

Section 4-05(c)(iii) is added to clarify the reporting requirements for other receipts, which are payments received by a candidate that are not contributions or loans, such as interest, dividends, proceeds from sales or leases of assets, and any other sources of income.

Sections 4-05(c)(iv)(A) and (B) (previously Section 3-03(e)(1)) are amended to require that, for each expenditure, in addition to all other information previously required to be reported or otherwise provided to the Board, candidates must report an explanation, the vendor address, and an exempt code (if any), and must provide to the Board the payment method and the amount of any remaining outstanding liability to the vendor or payee. This reflects existing CFB policy.

Section 4-05(c)(iv)(F) (previously Section 3-03(e)(5)) is amended to provide that the amount above which contributions to political committees from all candidates' personal funds are considered to be expenditures in furtherance of their campaigns is the doing business limit applicable to mayor, rather than to all three citywide candidates (which are currently the same). This results in no change to the current threshold, but decouples the citywide offices in case the Act is changed.

Section 4-05(c)(iv)(G) is added to require that candidates properly report and document expenditure refunds.

Section 4-05(c)(v) (previously Section 3-03(c)(7)) is amended to provide that candidates need not report an intermediary who is a member of the candidate's immediate family, and to delete certain reporting requirements for fundraising agents.

Section 4-05(c)(vi) (previously Section 3-03(c)(2)) is amended to clarify the information and documentation required to be provided for transfers, as well as the method for attribution of contributions underlying the transfer, and to state that transfers shall not contain public funds.

Section 4-05(c)(vii) (previously Section 3-03(c)(3)) is amended to require that if an advance is reimbursed by check, the candidate must report the bank account from which the check is issued.

Section 3-03(c)(10), which required a list of joint fundraising events to be provided with disclosure statements, is deleted.

Section 4-05(c)(viii) (previously Section 3-03(d)) is amended to clarify the information required to be reported for loan repayments.

Chapter 5: Contributions, Loans, and Other Receipts

Section 5-01(b) (previously Section 7-05(a)) is amended to reflect the fact that the contribution limit adjustment was performed in 2018, and that the next such adjustment shall occur in 2022.

Section 5-01(d)(1), which provided that cash or money order contributions aggregating in excess of \$100 from a single contributor are not matchable, is deleted, as it is redundant with the amended version of section 5-05(g) (previously Section 5-01(d)(4)), which provides that prohibited or excess contributions are not matchable, and sections 5-03(e) and (f), which prohibit contributions aggregating in excess of \$100 from a single contributor for both cash and money orders.

Section 5-02(a) (previously Section 1-08(n)) is amended to require that the statement required to be included on fundraising solicitations be in the same language as the solicitation itself.

Section 5-03 is added to enumerate the categories of prohibited contributions.

Section 5-03(e) is added to provide that candidates may not accept a cash contribution or other cash receipt aggregating in excess of \$100 from a single source. This requirement is consistent with 14-118(2) of the New York State Election Law.

Section 5-03(f) is added to provide that candidates may not accept money order contributions or other money order receipts from a single contributor aggregating in excess of \$100. This limit is imposed to mitigate the risks associated with such contributions, which, because money orders are not traceable, are substantially similar to those associated with cash contributions.

Section 5-03(g) (previously Section 1-04(s)) is amended to clarify that candidates may not accept any contribution made, received, solicited, or otherwise obtained in violation of any local, state, or federal law.

Section 5-04(c) (previously Section 1-04(k)) is amended to provide that the contemporaneous documentation to substantiate the amount of a joint contribution may include a contribution card.

Section 5-05 is added to enumerate the categories of non-matchable contributions.

Section 5-05(l) (previously Section 5-10(d)(20)) is amended to clarify that contributions made, received, solicited, or otherwise obtained in violation of any local, state, or federal law are not matchable.

Section 5-05(u) (previously Sections 5-01(d)(10), (21), and (25)) is reworded to ensure that it accurately reflects existing CFB policy. Filing a disclosure statement after the applicable deadline may result in a violation and penalty, but not the invalidation of all matching claims contained in the statement. In order for the validity of all matching claims contained in a disclosure statement to be preserved, the statement must be filed timely. Contributions first reported in an amendment to a disclosure statement – rather than in the statement itself – are not matchable. Due to the administrative processes associated with reviewing disclosure statements and issuing public funds payments, it is crucial that matchable contributions be claimed and reported in a timely manner, and that all contributions that will be claimed for match are reported with the initial statement filing. If a contribution first reported in an amendment were permitted to be claimed for match, the auditor would be required to restart the review of the disclosure statement being amended, which might not allow sufficient time for all matching claims to be reviewed prior to the next public funds payment date. Additionally, the phrase "contributions that were not received within the reporting period" is deleted for redundancy, because it is covered by the requirement that the contributions be reported timely and as part of the original statement rather than an amendment, and by section 4-05(c)(ii)(B)(1), which states that matchable contributions must be reported in the statement covering the reporting period in which they were received.

Section 5-05(v) is added to clarify that a contribution is not matchable if the corresponding matching claim was submitted after December 31 in the year of the election, even if the contribution itself was timely reported. This reflects existing CFB policy.

Section 5-05(w) (previously Sections 5-01(d)(10), (17), and (18)) is amended to clarify that a contribution may not be matchable if the information on the documentation submitted to substantiate the contribution does not match the information reported by the campaign. This includes the provision in the previous section that contributions made by post-dated check (thus reflecting a receipt date other than that reported by the campaign) are not matchable.

Section 5-05(z) is added to provide that contributions for which a matching claim was previously made and withdrawn are not matchable. This reflects existing CFB policy.

Section 5-06(a) (previously Section 1-04(g)(1)) is amended to clarify that the amount of an in-kind contribution, including those arising from a forgiven debt, counts toward both the contribution and expenditure limits, and that the candidate may rebut the presumption that the date of an in-kind contribution is the date of the associated expenditure.

Section 5-06(b) (previously Section 1-04(g)(2)) is amended to state that a candidate may use a reasonable estimate to document the fair market value of an in-kind contribution if no invoice is available and obtaining one would not be practicable.

Section 5-06(c) (previously Section 1-04(g)(3)) is amended to clarify that the amount of any applicable tax is included in the calculation of an in-kind contribution.

Section 5-06(d) (previously Section 1-04(g)(4)) is amended to provide that an extension of credit by a corporation, limited liability company, or partnership that is not commercially reasonable and is not made in the regular course of business is presumed to be a prohibited in-kind contribution. Previously, if a candidate demonstrated that a creditor that was a corporation, limited liability company, or partnership did not intend to make a contribution, the extension of credit would not in itself result in the candidate being deemed to have accepted a contribution from that entity.

Section 5-06(f) (previously Section 1-04(g)(7)) is amended to provide that an outstanding liability not contemporaneously reported may be presumed to be an in-kind contribution if other indicia of a contribution exist.

Section 5-07(a) (previously Section 1-04(c)(1)) is amended to acknowledge that excess or prohibited contributions may be returned to the New York City Election Campaign Finance Fund (the "Fund") as an alternative to being returned to the contributor. The section further clarifies that nominee or anonymous contributions must be returned to the Fund, because a refund to the reported contributor would be impossible (in the case of anonymous contributions) or would compound the illegality (in the case of nominee contributions).

Section 5-07(b) (previously Section 1-04(c)(1)) is amended to delete the provision stating that no violation shall exist if the excess portion of a doing business contribution is refunded within 20 days of notification, as the provision was redundant with the Act and other sections of the rules.

Section 5-07(d) (previously Section 1-04(c)(2)) is amended to specify that a contribution returned due to the source or intermediary involved must involve a reputational interest associated with such source or intermediary.

Section 5-07(e) (previously Section 1-04(c)(1)) is amended to provide that a disgorgement to the Fund made because a contribution refund is impracticable must be disclosed as a refund to the contributor.

Section 5-07(f) (previously Section 1-04(c)(1)) is amended to reflect the lack of matchability of contributions intermediated by individuals doing business with the city, pursuant to Local Law No. 167 of 2016 (codified in Admin. Code § 3-702(3)(i)). The section is further amended to provide that, for purposes of calculating compliance with the doing business limits, while contributions received from a contributor who appears on the Doing Business Database ("DBDB") are aggregated with all contributions received from that contributor during the same cycle, contributions accepted when the contributor did not appear on the DBDB are not required to be refunded. For example: a candidate for City Council accepts a \$1,000 contribution during the first year of an election cycle from a contributor who is not on the DBDB. One year later, the contributor begins doing business with the City. This contributor has already reached the doing business limit of \$250, but the candidate is not required to return the excess, as the contributor did not appear on the DBDB when the contribution was made. However, the candidate may not accept any additional contributions from that contributor while the contributor is on the DBDB.

Section 5-08(a) (previously Section 1-03(a)) is amended to clarify that funds remaining in a segregated account after the election may not be transferred into another account or used for any future election. Such funds must be returned to contributors or to the Fund or used to pay outstanding liabilities from the election for which the funds were raised. Additionally, previous Section 1-03(a)(4), which provided that after a participant first receives public funds for an election, the principal committee for that election may not make expenditures to pay expenses or debt from a previous election (other than a primary election held in the same calendar year), is deleted. Participants are permitted to make such expenditures, but the amount of the expenditure will be deducted from the amount of public funds payable to the participant, in accordance with Section 7-07(a)(ii)(B).

Section 5-08(c)(i) (previously Section 1-07(c)) is amended to add that candidates have the burden of demonstrating that surplus funds and transfers of funds from committees not otherwise involved in the covered election do not derive from contributions in violation of State or Federal law.

Section 5-09(i) (previously Section 1-05(g)) is amended to provide that a loan issued after the Board has issued a final determination for the candidate, for the purposes of paying penalties or repaying public funds pursuant to such determination, is not subject to the contribution limits. Previously, this provision applied to all loans made after the election, including those made prior to the issuance of the final Board determination. Section 5-10(c)(i) (previously Section 1-04(p)(1)) is amended to delete the statement that the section does not prohibit a candidate from making a contribution to any other candidate or political committee, provided that such contributions may result in reduced public funds payments pursuant to previous section 5-01(n). The statement is deleted because it is clear from the plain meaning of the section that candidates may contribute to other candidates and political committees in a manner consistent with all applicable contribution limits and restrictions.

Section 5-11(a)(iii) (previously Section 2-06) is amended to clarify that candidates must use a merchant account for collecting credit card contributions, and that funds accepted for an election may not be commingled with personal or business funds or funds accepted for a transition and inauguration entity ("TIE").

Section 5-12 (previously Section 1-04(o)) is amended to provide that candidates in court-ordered rerun elections that are canceled have the burden of demonstrating that any portion of the contributions raised may be reasonably attributed to expenses incurred for the rerun election before its cancellation. Previously, this burden existed only for contributions in excess of the limit.

Chapter 6: Expenditures

Section 6-01(h)(iii)(B) is added to state that expenditures incurred after the date of a contested or write-in primary election will be attributed to the general election.

Section 6-01(h)(iv)(B)(1) (previously Section 1-08(c)(2)(iii)) is amended to provide that an affidavit submitted with a petition to demonstrate that a candidate reasonably anticipated a primary election must be sworn to or affirmed by the candidate.

Section 1-08(d)(2), which provided that, to be permitted to spend the total aggregate amount of the out-year, primary election, and general election expenditure limits, a candidate must spend at least the amount of the out-year limit during the three years preceding the year of the election, is deleted, because the Act clearly provides that the out-year expenditure limit applies to the three years preceding the election year.

Section 6-01(h)(iv)(C) (previously Section 1-08(c)(2)(iv)) is amended to clarify that expenditures made after a primary election is no longer reasonably anticipated are subject to the general election expenditure limit.

Section 6-01(i)(ii) (previously Section 1-04(d)(4)(ii)) is amended to provide that expenditures for childcare services made pursuant to Admin. Code § 3-702(21)(a)(13) are exempt from the expenditure limits up to \$20,000, as provided by Local Law No. 196 of 2018, codified in Admin. Code § 3-706(5). This local law allows for pre-election childcare services provided during the year of the election or the year preceding the election to be paid for with private campaign funds, provided that the candidate is a primary caregiver for one or more children under 13 years of age; that either the need for the services would not exist but for the campaign, or the candidate has experienced a significant loss of salary or wages that would not exist but for the campaign; and that the candidate submits a statement of childcare eligibility that is approved by the Board. Additionally, the law provides that such expenditures are not qualified. *See also* section 6-02(a)(ii)(V).

Section 6-01(i)(v) (previously Section 1-08(1)(3)) is amended to remove the provision excluding expenditures attributed to a period ending on or before the closing date of the 10 day post-primary election disclosure statement from the general election expenditure limit, provided that the candidate was subject to the primary election expenditure limit. The CFB's current audit standards and methodology provide for close scrutiny of transactions reported on the 10 day post-primary statement, in order to determine on a case-by-case basis which expenditures should be subject to the general election expenditure limit.

Sections 6-01(k)(i) and (ii) (previously Sections 1-08(e)(i) and (ii)) are amended to clarify that the amount raised and the amount spent are not added together for the purpose of determining whether expenditure limit relief should be granted.

Section 6-02(a)(i) (previously Section 1-08(g)) is amended to delete the statement that public funds may only be used for expenditures in furtherance of the candidate's campaign and during the calendar year of the election, as those requirements are covered in new section 6-02(a)(ii).

Section 6-02(a)(ii)(C) (previously Section 1-08(g)(2)(iii)) is amended to provide that expenditures made in violation of the Board rules are not qualified expenditures.

Section 6-02(a)(ii)(U) is added to state that expenditures to facilitate, support, or otherwise assist in the execution or performance of the duties of public office are not qualified, as provided by Local Law No. 190 of 2016 (codified in Admin. Code § 3-704(2)(1)).

Previous Section 1-08(g)(3), which provided that bills that are first reported in a disclosure statement submitted later than the 10 day or

27 day post-election disclosure statement applicable to that election, and bills first reported in a disclosure statement amendment that is made after the last election in an election year, or after a special election, are presumed not to be qualified expenditures, is deleted. Candidates are encouraged to add and amend any remaining bills or bill payments in filing the final disclosure statement for each cycle, which is due after the disclosure statements listed in the previous section. Thus, previous section 1-08(g)(3) is deleted to conform to CFB practices and procedures.

Section 6-02(a)(ii)(V) is added to state that expenditures related to childcare services are not qualified, *i.e.*, they may not be made with public funds, as provided by Local Law No. 196 of 2018, codified in Admin. Code § 3-704(2)(m). This local law allows for pre-election childcare services provided during the year of the election or the year preceding the election to be paid for with private campaign funds, provided that the candidate is a primary caregiver for one or more children under 13 years of age; that either the need for the services would not exist but for the campaign, or the candidate has experienced a significant loss of salary or wages that would not exist but for the campaign; and that the candidate submits a statement of childcare eligibility that is approved by the Board. Additionally, the law provides that such expenditures are exempt from the spending limits up to \$20,000. See also section 6-01(i)(ii).

Section 6-03(d) (previously Section 1-08(h)(1)) is amended to delete the sentence stating that the amount spent by a candidate for a joint expenditure is subject to the expenditure limit unless otherwise exempt. This sentence is unnecessary because all expenditures are subject to the limit unless exempt, and in particular, section 6-03 makes clear that non-proportional payment for a joint expenditure may result in the presumption of an in-kind contribution, which is also considered an expenditure by the campaign receiving such contribution.

Section 6-04(e)(ii) (previously Section 1-08(f)(5)(ii)) is amended to describe the types of expenditures that will be presumed to be in-kind contributions from a party committee or constituted committee to a candidate as those made "for the purpose of promoting or facilitating the nomination or election of" the candidate. This description consolidates the list of types of expenditures contained in the previous section.

Section 6-04(e)(iii) (previously Section 1-08(f)(5)(ii)) is amended to clarify that if a candidate makes or promises reimbursement to a party committee or constituted committee for an expenditure, but has not reported an outstanding liability for the reimbursement, then the expenditure will be considered an in-kind contribution from the party or constituted committee to the candidate until the reimbursement by the candidate is made. The previous section provided that, under such circumstances, the Board could require the candidate to demonstrate that the expenditure was not an in-kind contribution; the new section still allows for such documentation to be provided, but clarifies that the presumption of an in-kind exists whether or not the documentation is requested by the Board.

Section 6-04(g) is added to provide that certain routine interactions between campaigns and outside entities will not, absent other indicia of non-independence, lead to a determination of non-independence. This section is a codification of CFB Advisory Opinion No. 2013-1 (January 10, 2013).

Section 6-06(c) (previously Section 2-13(c)) is amended to clarify the requirements for identification of communications broadcast over the internet, as well as the distinctions between text, audio, and video communications.

Chapter 7: Pre-Election Public Funds Payments

Section 7-01(e)(ii) (previously Section 5-02(b)) is amended to remove the requirement that a petition for a candidate to receive public funds after having been disqualified from the ballot include a demonstration that hardship will exist if the funds are not paid.

Section 7-02(a) (previously Section 5-01(i)) is amended to require that a candidate seeking to receive the optional early public funds payment declare the specific office to which such candidate is seeking nomination or election. In order to receive the early payment, candidates must certify their intention to meet all the requirements of law to appear on the ballot; for such certification to occur in good faith, the candidate must know what office such candidate seeks. Additionally, both the threshold for eligibility to receive public funds and the maximum amount of the early payment depend on the office sought.

Section 7-02(b) (previously Section 5-01(b)(i)) is amended to clarify the process of preliminary review of disclosure statements, including to specify that, in order for the Board to issue a public funds payment to a candidate within four business days of receiving that candidate's disclosure statement, the statement must be received by the Board by 5:00 P.M. on the due date.

Section 7-02(c) (previously Section 5-01(p)) is amended to describe clearly and in additional detail the process of responding to invalid

matching claims reports, and to clarify that failure to respond may lead to a delay in the issuance of public funds payments. The section is further amended to delete the provision requiring that candidates respond to an invalid matching claims report by the March 15 after the election or 60 days after receiving the report, whichever is sooner. CFB practice is for each invalid matching claims report to include a deadline; the amount of the response time may differ depending on the circumstances.

Section 7-04(a) (previously Section 5-01(a)(2)) is amended to change the deadline for filing a Certified Statement of Need. Such statements must be filed no later than the 32-day pre-election disclosure statement deadline applicable to the election in which the participant is a candidate, except that if the candidate's basis for filing the statement of need arises for the first time after that deadline, then the statement immediately preceding the next public funds payment date. For example, if, after the 32-day pre-election disclosure statement deadline, a candidate receives an endorsement as described in § 3-705(7)(b)(2)of the Act, and no other basis for filing a statement of need previously existed in that race, the candidate's opponent may file a statement of need up until the disclosure statement immediately preceding the subsequent payment date.

Section 7-06 (previously Section 5-01(e)) is amended to define more broadly the circumstances under which the Board may withhold a portion of a public funds payment, and to clarify that such withholdings are subject to post-election audit.

Section 7-08 (previously Section 5-01(h)) is amended to delete the sentence providing that, after the Board notifies a candidate of the difference between the amount of matching claims and the amount of the corresponding payment, subsequent payments may be adjusted upward or downward to reflect further review and auditing of previous matching claims. The sentence was self-evident and redundant with the remainder of the chapter, which makes clear that a preliminary public funds determination may be followed by subsequent determinations based on further audit review and submission of additional reporting and documentation.

Section 7-09(a) (previously Section 5-02(a)(1)) is amended to require that a petition for review of a Board determination include either a request to appear before the Board, or an affirmative waiver of the right to such an appearance.

Section 7-10(a) (previously Section 7-03(c)) is amended to delete the presumption that cash on hand after an election consists of contributions that are available for expenditures in the next election, to maintain consistency with the requirement that participating candidates return their final bank balance to the Board after the election.

Section 7-10(b)(i) (previously Sections 7-03(d) and (e)) is amended to add deadlines for submission of petitions for reconsideration; to require that such petitions include sufficient specificity and a request for a hearing, if desired; and to delete the sentence providing that failure to file required disclosure forms may constitute a sufficient basis for a determination, and must be addressed in the response to the petition.

Section 7-10(b)(vi) (previously Section 7-03(j)) is amended to provide that a new petition for review of a preliminary public funds determination must be based on new evidence.

Section 7-10(b)(vii) (previously Section 7-03(k)) is amended to require that the Board respond to such petitions within five business days of their filing.

Chapter 8: Post-Election Public Funds Payments

Section 8-02 (previously Sections 1-08(g)(4), 3-02(b)(3), and 5-01(l)) is amended to detail the bases for post-election public funds payments, and to clarify that such a payment may only be made if all three bases exist, and that the amount to be paid is the lesser of the applicable amounts. For example, if a candidate has unpaid public funds claims of \$5,000, a qualified expenditure surplus of \$7,500, and documented outstanding liabilities of \$10,000, the post-election payment amount is \$5,000. All three factors must be present because each one is a necessary component of eligibility to receive a post-election payment. To receive a post-election payment, a candidate must have submitted valid matching claims corresponding to an amount of public funds greater than the amount previously received, must have documented qualified expenditures in excess of the amount of public funds previously received, and must have timely and properly reported liabilities that remain outstanding after the election, and which will be reduced by the amount remaining in the campaign's bank account. In order to receive a post-election payment, a candidate may therefore be required to submit any bank statements not previously provided. This enables CFB staff to verify the amount of the final bank balance in the committee account, which is required to calculate the amount the candidate is eligible to receive based on documented outstanding liabilities. These changes to the section reflect existing Board policy.

Section 8-04 (previously Section 5-01(m)) is amended to move the deadline by which a disclosure statement may be amended in order

for the content of such statement to serve as a basis for a public funds payment from December 31 in the year of the election to January 15 in the year following the election, and to provide that a payment may be made based on amendments or resubmissions in response to expenditure sample reports to which the CFB has requested a response after January 15 in the year after the year of the election.

Chapter 9: Public Funds Repayments

Section 9-01 (previously Sections 5-03(a) and (g)) is amended to clarify that public funds repayments must be paid to the Board separately from, and in addition to, any penalties owed. Accordingly, candidates may not use their final bank balance to pay penalties, because the bank balance exists as a wholly independent amount that must also be repaid. This reflects existing Board policy.

Section 9-02(c)(iii) is added to clarify the requirements regarding post-election bonus payments to staff. A campaign that received public funds may make a bonus payment to an employee after the election, provided that the bonus is related to services performed prior to the election; is based on quantifiable, merit-based criteria; is not conditioned upon the result of the election or whether the campaign received public funds or has funds remaining after the election; is clearly detailed in a contract executed before the services in question were provided; and does not constitute more than 10% of total payments to the employee.

Section 9-02(d) is added to clarify that where more than one ground exists for a candidate or committee to repay public funds to the Board, the applicable repayment amount is the larger of the amounts. For example, if a committee has a qualified expenditure deficit of \$1,000 and an overpayment of \$500, the committee's repayment obligation is \$1,000. This reflects existing Board policy.

Chapter 10: Audit and Enforcement

Section 10-02(a) (previously Section 2-12(b)) is amended to clarify that, as provided by the Act, the 14- or 16-month deadline for the issuance of a candidate's final audit report may be satisfied by issuing the candidate's notice of alleged violations and recommended penalties and/or notice of recommended public funds repayment by such deadline.

Sections 10-02(b) and (c) (previously Section 4-05(b)) are amended to state the extended final audit report deadlines for candidates who do not complete a required training by the applicable deadline, pursuant to Admin. Code § 3-710(1)(b).

Section 10-03(a)(ii) (previously Section 7-02(c)(1)) is amended to clarify who has the authority to waive the right to an adjudication pursuant to section 1046 of the City Charter on behalf of a campaign, a TIE, and an independent spender. Such authority is held by the individual candidate, the elected candidate, and the independent spender or authorized representative, respectively, or an agent of one of those individuals. The word "candidate" is therefore replaced with the phrase "participant or nonparticipant" in order to clarify that only the individual candidate (rather than the treasurer or the committee, who are included in the definition of the term "candidate"), or someone acting directly on the candidate's behalf, such as an attorney or compliance consultant, has the authority to waive the right to such an adjudication unilaterally. A waiver received from any of the individuals listed in the section is deemed an effective waiver on behalf of the treasurer and committee, the TIE, and the independent spender, as applicable. Conversely, for example, a treasurer does not have the authority to waive adjudication on the candidate's behalf.

Section 10-03(b) (previously Section 7-02(c)(2)) is amended to clarify that extensions to respond to enforcement notices may be granted only upon good cause shown, and must be requested prior to the response deadline. The section specifies that a "complete response" must be submitted by the deadline, clarifying that piecemeal responses, with a portion submitted after the deadline, are not permitted. Information or documentation submitted after the deadline shall be disregarded, except that the Board may choose, under extraordinary circumstances, to accept such information or documentation at its sole discretion. Such circumstances are events beyond the control of the campaign, TIE, or independent spender that make it impossible for the candidate, treasurer, or independent spender/authorized representative to respond in a timely manner. Examples include serious health issues, disasters, and accidents. Campaigns, TIEs, and independent spenders will be required to submit documentation explaining and demonstrating these circumstances.

Section 10-03(e) (previously Section 7-02(g)) is amended to provide that aggravating and mitigating circumstances "may" be considered in the Board's assessment of whether to deviate from the Penalty Guidelines. Previously the section provided that such factors "shall" be considered. The amendment reflects existing Board policy as stated in the existing Penalty Guidelines.

Chapter 11: Procedural Rules for Formal Adjudications

No substantive changes are made to Chapter 11.

Chapter 12: Complaints and Investigations

Section 12-02(b) (previously Section 7-01(c)) is amended to require that a formal complaint filed with the Board include the complainant's full name and contact information.

Section 12-02(c) (previously Section 7-01(d)) is amended to provide that, within ten days of receiving a complaint, the Board will determine whether to reject such complaint as moot, facially meritless, or not in substantial compliance with the Board's requirements. If a complaint is not rejected, the Board will send each party complained about a copy of the complaint. The section is further amended to provide that, even if the complaint is not in substantial compliance, the Board may investigate its subject matter, but need not follow the procedural requirements of the rules.

Section 12-02(d) (previously Section 7-01(e)) is amended to extend the period in which the Board can specify a shorter response time from 40 days before the election to 90 days before the election, and to require that answers to complaints be accompanied by copies of all documentary evidence available to the respondent. The section is further amended to provide that, based on the answer received, a complaint may be dismissed as moot.

Chapter 13: Transition and Inauguration Entities

Section 13-01 (previously Section 11-02) is amended to provide that TIEs may accept donations prior to registration for the purpose of paying fees and depositing any minimum balance required to open a bank account for the TIE; to clarify the dates on which a TIE registration may be submitted; to delete the provision for submission of registration in a non-electronic format; and to clarify that each elected candidate may maintain only one TIE, with one bank account, per election.

Sections 13-02(a) and (b) (previously Section 11-03(a)) are amended to provide that only donations exceeding \$99 must be itemized, and to specify the information that must be reported for each in-kind donation. Additionally, section 13-02(a) is amended to align the disclosure requirements for TIEs more closely with the analogous requirements for candidates.

Section 13-02(d) (previously Section 11-03(c)) is amended to require monthly reports (previously bimonthly) and to modify the due dates for each such report. Additionally, the provision for filing of such reports in non-electronic format is deleted.

Section 13-03(a)(ii) (previously Section 11-04(b)) is amended to prohibit money order donations aggregating in excess of \$100 from a single donor. This aligns with the limit imposed on money order contributions to campaigns by section 5-03(f).

Section 13-03(a)(v) (previously Section 11-04(e)(i)) is amended to remove the provisions regarding expenditures, which are now contained in section 13-03(b), and special elections, which are now contained in section 15-07.

Section 13-03(b)(i) (previously Section 11-04(b)(4)) is amended to add a list of presumptively TIE-related expenditures.

Section 13-03(b)(ii) is added to provide that purchases of durable goods are presumed not to be TIE-related.

Section 13-03(b)(iii) (previously Section 11-04(b)(6)) is amended to clarify the distinction between transition liabilities and inauguration liabilities, and to add an exception allowing for expenditures after January 31 in the year after the election related to fundraising to satisfy liabilities incurred prior to that date.

Section 13-03(b)(iv) is added to provide that expenditures incurred after January 31 in the year following a general election are presumed to be for the next election in which the elected candidate is a candidate, and not for a TIE.

Section 13-03(b)(vi) is added to provide that only one single inaugural event is presumed to be TIE-related.

Section 13-03(c)(i) is added to require that TIEs be terminated after extinguishing all liabilities or on April 30 in the year following the election, whichever is earlier.

Section 13-04(b) (previously Section 11-05(b)) is amended to describe in more detail the records required to be kept by TIEs, including requiring that the written records for donations include the date and amount of the donation and, for cash and money order donations, the donor's signature.

Section 13-04(d) (previously Section 11-05(e)) is amended to modify the record retention period for TIEs from six years to five years.

Chapter 14: Disclosure of Independent Expenditures

Section 14-01 (previously Section 13-01) is amended to add a definition for the term "authorized representative."

Section 14-02(b)(ii) is added to provide that a mass mailing is considered to have been distributed on the date it was postmarked, or accepted by a common carrier.

Section 14-02(c)(i) (previously Section 13-02(c)(1)) is amended to require that invoices for independent expenditures be itemized.

Section 14-03(b) (previously Section 13-03(b)) is amended to provide that the reporting period for ballot proposals begins on January 1 in the year in which the proposal will appear on the ballot. Previously, the reporting period began on the date the first communication was distributed or first expenditure was made.

Section 14-04(a) (previously Section 13-04(a)) is amended to ensure that the Board's requirements for identification of communications are consistent with the requirements provided in New York State law.

Previous Section 3-10, which provided for a political committee making expenditures supporting or opposing a ballot proposal to file voluntary disclosure statements with the Board, is deleted. Expenditures regarding ballot proposals are covered by the provisions of Chapter 14.

Chapter 15: Special Elections

Section 15-01 (previously Section 1-06) is amended to remove the reference to determining the amount of surplus funds for special elections. For special election candidates who already have a committee registered with the CFB from a previous election, that committee becomes a special election committee, but surplus funds remaining in the committee account may not be used for the special election.

Section 15-02(a) is added to provide that a Certification filed for a special election may not be rescinded.

Section 15-02(c) is added to provide that candidates who have already been raising funds for other elections within the same election cycle may use their existing committees and bank accounts in the special election, provided that they amend their registration with the Board of Elections to reflect the change in election, and further provided that once that committee and account have been used for the special election, they shall not be used for any future election, including any other elections during the same election cycle.

Section 15-02(d)(i) is added to provide that disclosure statements reporting financial activity that occurred after a candidate's most recent election prior to a special election will be considered to be for the special election.

Section 15-02(d)(ii)(A) (previously Section 3-03(a)(3)) is amended to provide that if a candidate is already registered for a future election and an intervening special election is called, the candidate must reregister with the Board for the special election and re-submit all of the candidate's financial activity with the first special election disclosure statement.

Section 15-02(d)(vii) is added to require that each disclosure statement submitted for a special election include a copy of the candidate's most recent bank statements, as is required for primary and general election disclosure statements.

Sections 15-03(b) and 15-04(b) are added to provide that expenditures made and contributions received after a candidate's most recent election and prior to a special election are presumed to be for the special election.

Section 15-03(c) (previously Section 1-08(c)(4)) is amended to provide that expenditures incurred prior to the date a special election was reasonably anticipated may be presumed to be subject to the special election expenditure limit, beginning when a candidate has registered for a covered office or has begun raising and/or spending funds.

Section 15-03(f) is added to clarify certain types of expenditures that may be made with public funds in a special election.

Section 15-04(c) is added to provide that, for contributions received on or before the date the special election was first reasonably anticipated that are in excess of the contribution limit, the candidate must refund the excess portion to the contributor.

Section 15-04(e) (previously Section 5-01(p)) is amended to remove the set deadline for responding to invalid matching claims reports. See also Section 7-02(c).

Section 15-04(g) (previously Section 2-06(d)) is amended to clarify that receipts accepted for a special election may not be commingled with receipts accepted for another election, with receipts accepted for a TIE, or with personal or business funds, and to provide that such receipts may not be used for any other election until the special election is actually held.

Section 15-04(h) is added to provide that receipts accepted for a special election that is not held, or in which the candidate is not on the ballot, must be returned to the contributors or to the Fund. Candidates who received public funds must first return receipts to the Fund, up to the amount of public funds received, before returning any receipts to contributors.

Section 15-05 is added to require that candidates in a special election must attend a compliance training session designed specifically for that election, and that the training must be completed on or before the financial disclosure cut-off date of the 11-day pre-election disclosure statement. Section 15-06(d) (previously Section 13-03(a)(3)(ii)) is amended to provide that the reporting period for the first disclosure statement for independent expenditures related to special election ballot proposals begins on the date on which the first reportable expenditure is incurred. This was not explicitly stated in the previous section because the reporting period began on the same date for primary and general elections, as it did for special elections, which is no longer the case. See section 14-03(b).

Section 15-07 (previously Section 11-04(e)(i)) is amended to provide that a special election TIE may not incur expenditures more than 30 days after the candidate's inauguration, except to satisfy liabilities incurred within that time period or for fundraising to pay such liabilities. The section further provides that a special election TIE must terminate after it has extinguished all liabilities or 60 days after inauguration, whichever is earlier.

Chapter 16: Runoff Elections

Section 16-05(c) (previously Section 1-04(q)(3)) is amended to clarify that no single contribution may be divided between runoff and non-runoff accounts.

Section 16-05(d) is added to detail the documentation requirements for runoff contributions, including runoff-specific contribution cards.

Section 16-05(e)(i) (previously Section 5-01(j)), regarding flat grants for runoff elections, is amended to remove the provision stating that the amount of public funds paid "may be adjusted to reflect further review and auditing of matchable contribution claims for the preceding election." The amendment reflects the Board's practice, which is that the flat grant is calculated as a percentage of the amount actually paid to the candidate for the preceding election, not the amount the candidate may theoretically have been eligible to receive.

Section 5-01(i)(1)(ii) is amended to remove the provision stating that no public funds would be paid in a runoff election prior to the day after the day of the primary election held to nominate candidates for the runoff. Section 5-01(i)(1)(iii), which provided that no public funds would be paid in a runoff special election prior to the day after the day of the special election, is deleted. These provisions were redundant, as no public funds can be paid for an election until such election is reasonably anticipated and the identities of the candidates have been determined.

Section 16-05(e)(ii)(B) is added to provide that no public funds shall be issued to candidates in a runoff election until the Board has determined that the election is reasonably anticipated.

Section 16-05(b) is added to provide that receipts accepted for a runoff election that is not held, or in which the candidate is not on the ballot, must be returned to the contributors or to the Fund. Candidates who received public funds must first return receipts to the Fund, up to the amount of public funds received, before returning any receipts to contributors.

Section 16-05(f) (previously Section 2-06(c)(1)(ii)) is amended to clarify that receipts accepted for a runoff election may not be commingled in any account with receipts accepted for another election, with receipts accepted for a TIE, or with personal or business funds, and to provide that such receipts may not be used for any other election held in the year that the runoff primary election is held or anticipated.

Section 16-05(g) (previously Section 2-06(c)(iii)) is amended to provide that receipts accepted for an anticipated runoff election that is not held, or for a runoff election in which the candidate is not on the ballot, must be returned to the contributors or to the Fund, after the earlier of (i) the first January 12 after the date of the election for which the runoff election was anticipated, or (ii) the date upon which all the candidate's liabilities from the election for which the runoff election was anticipated have been extinguished. Candidates who received public funds must first return receipts to the Fund, up to the amount of public funds received, before returning any receipts to contributors.

Section 16-06(a)(i) (previously Section 2-06(c)(1)(i)) is amended to provide that transfers may occur from a primary election account to a runoff primary election account after the primary election is held, and from a runoff primary election account to a general election account after the runoff primary is held.

Section 16-06(a)(ii) is added to require that runoff bank accounts be disclosed to the Board as an amendment to the candidate's Filer Registration or Certification.

Section 16-06(b)(i) is added to clarify that runoff special elections are held only for the office of mayor. New York State Election Law requires a runoff election if no candidate receives 40% or more of the vote in a primary election for the office of mayor, public advocate, or comptroller, but does not provide for runoff special elections for any office. See N.Y. Elec. Law § 6-162(1). However, the City Charter requires runoff special elections for the office of mayor. See New York City, N.Y., Charter § 10(c)(10). This section is added because both the Act and Board rules refer to runoff special elections, but do not specify to which office they apply.

FRIDAY, APRIL, 19, 2019

Sections 16-06(b)(ii), (iii), and (iv) are added to provide that funds raised for a runoff special election are subject to the same requirements as those for a runoff primary election.

Section 16-07(a) is added to provide that the reporting period for the first disclosure statement for independent expenditures related to a runoff election begins on the day the Board issues a determination stating that the runoff election is reasonably anticipated.

Chapter 17: Voter Education and Engagement

Section 17-02(b)(i)(C) (previously Section 10-02(b)(1)(iii) is amended to provide that photographs submitted with candidate statements for the print edition of the Voter Guide must comply with size and resolution requirements as determined by the Board. The previous section provided that only photographs must "not exceed" such requirements; this clarification is to allow for the possibility of minimum size and resolution requirements in addition to maximum requirements.

Sections 17-02(b)(i)(D) (previously Section 10-02(b)(1)(iv)) and 17-02(b) (ii)(A) (previously Section 10-02(b)(2)(i)) are amended to provide that candidate print and video statements for the Voter Guide may not refer to any opposing candidate by name, and must not include statements, gestures, or materials that are patently offensive. A prohibition on the unauthorized use of copyrighted material or invasion of privacy in such statements is replaced by a broader prohibition on the violation of any city, state, or federal law, which mirrors the analogous provision related to the video Voter Guide (see section 17-02(b)(ii)(A)(6)).

Section 17-02(b)(ii)(C) (previously Section 10-02(b)(2)(iii)) is amended to state the Board's authority to approve and set deadlines for the submission of video Voter Guide statements.

Section 17-02(b)(ii)(H) (previously Section 10-02(b)(2)(viii)) is amended to delete the statement that the filming schedule for the video Voter Guide will permit candidates filing designating and nominating petitions to participate.

Sections 17-02(b)(iii)(A) and (B) (previously Sections 10-02(b)(3)(i) and (ii)) are amended to remove the requirement that candidates be named in designating or nominating petitions in order to appear in the Voter Guide for the primary and general elections.

Previous Section 2-07(d), which provided that write-in candidates not appearing on the ballot would not appear in the Voter Guide for that election, is deleted. The substance of this section is covered in new sections 17-02(b)(iii)(A) and (B).

Chapter 18: Public Access to Information

Section 18-02 (previously Section 6-01(a)) is amended to delete the list of the Records Access Officer's duties, as the procedure is adequately detailed in the rest of the chapter.

Section 18-03(a) is added to provide that a candidate may request access to records previously provided to the Board by contacting the Candidate Guidance and Policy Unit, which may, in its discretion, provide the records without a Freedom of Information Law ("FOIL") request.

Sections 18-03(b) and (c) (previously Section 6-01(c)) are amended to provide additional detail regarding the procedure and requirements of submitting a FOIL request.

Section 18-03(d) is added to state that if the Board is unable to locate responsive records, the Board will, upon request, certify either that it is not the custodian of the records, or that the records cannot be found after a diligent search.

Section 18-03(e) (previously Section 6-01(b)) is amended to clarify the procedure for inspection of records and to require the requester to promise that the records will not be removed, damaged, or modified in any way.

Sections 18-04(a) and (b) (previously Section 6-01(d)) are amended to provide that appeals of FOIL denials should be directed to the General Counsel, to state that the appeal must be filed within 30 days of receiving the denial, and to specify that the appeal must include a copy of the original request and a reasonable description of the records to which access was denied.

New York City Campaign Finance Board Proposed Rules

New material is underlined.

[Deleted material is in brackets.]

"Shall" and "must" denote mandatory requirements and may be used interchangeably in the rules of the Board, unless otherwise specified or unless the context clearly indicates otherwise.

Section 1. Title 52 of the rules of the City of New York is REPEALED, and a new Title 52 is added to read as follows:

<u>Title 52: Campaign Finance Board</u>

Chapter 1: Definitions and General Provisions

§ 1-01 Scope of Rules

Chapters 1 through 8 contain requirements applicable

to candidates seeking nomination or election to the office of mayor, comptroller, public advocate, borough president, or member of the City Council ("Council member") during the pre-election period.

<u>Chapters 9 through 11 pertain to post-election audit and</u> <u>enforcement of candidates seeking nomination or election to the office</u> <u>of mayor, comptroller, public advocate, borough president, or Council</u> <u>member, as well as independent spenders.</u>

<u>Chapter 12 pertains to investigations conducted by, and</u> complaints filed with, the Board.

<u>Chapter 13 contains requirements for transition and</u> <u>inauguration entities ("TIEs"), which apply to all candidates elected</u> to the office of mayor, comptroller, public advocate, borough president, or Council member, regardless of whether the elected candidate is a participant in the voluntary Campaign Finance Program.

<u>Chapter 14 contains requirements for disclosure of</u> <u>independent expenditures related to candidates seeking nomination</u> <u>or election to the office of mayor, comptroller, public advocate, borough</u> <u>president, or Council member, as well as ballot proposals.</u>

Chapter 15 contains requirements for special elections for the office of mayor, comptroller, public advocate, borough president, or Council member.

Chapter 16 contains requirements for runoff elections for the office of mayor, comptroller, public advocate, borough president, or Council member.

<u>Chapter 17 pertains to the Voter Guide and voter</u> engagement and applies to candidates seeking nomination or election to the office of mayor, comptroller, public advocate, borough president, or Council member, as well as city ballot proposals or referenda.

<u>Chapter 18 contains requirements for public access to</u> information as provided by the Freedom of Information Law.

Except as otherwise specified, the requirements of these rules related to authorized committees do not apply to an authorized committee that does not, at any time, aid or otherwise take part in an election in which the candidate is a participant or non-participant. Aiding or otherwise taking part in an election includes accepting contributions, loans, or other receipts, and making expenditures, including expenditures of surplus funds, for such election.

§ 1-02 Definitions

"Act" means the New York City Campaign Finance Act, codified in Chapter 7 of Title 3 of the Code (§ 3-701, *et seq.*).

<u>"Administrative law judge" means the hearing officer</u> assigned to preside over a case that is referred to the Office of Administrative Trials and Hearings.

<u>"Advance" means a payment for goods or services on behalf</u> of a candidate made with the expectation that the payment will be reimbursed by the candidate.

<u>"Authorized committee</u>" means an authorized committee as defined in the Act.

"Board" means the Campaign Finance Board.

"Board of Elections" means the New York City Board of Elections, unless otherwise specified as the New York State ("State") Board of Elections.

<u>"Business dealings with the city" means business dealings</u> with the City of New York as defined in the Act.

"Candidate" means a candidate as defined in Article 14 of the New York State Election Law. Except as otherwise provided in these rules, a "candidate" includes every authorized committee of the candidate, the treasurer of each such committee, and any other agent of the candidate.

"CAPA" means the City Administrative Procedure Act, §§ 1041 to 1047 of the Charter.

"Certification" means the submission of the information required pursuant to section 2-02 in order to join the Program.

"Charter" means the New York City Charter.

"Code" means the Administrative Code of the City of New York.

"Contribution" means a contribution as defined in the Act.

"Covered election" means any election for the office of mayor, public advocate, comptroller, borough president, or Council member.

"Disclosure statement" means the campaign finance disclosure statement filed with the Board under Chapter 4 of these rules.

"Doing business database" means the computerized database containing the names of individuals and entities engaged in business dealings with the city as defined in the Act. **"Domestic partner"** means a domestic partner as defined in § 1-112(21) of the Code.

"Donation" means a gift, subscription, advance, payment, or deposit of money or any thing of value, made by an individual or entity, in connection with the transition or inauguration expenses of an elected candidate, including but not limited to compensation for the personal services rendered in connection with such transition or inauguration expenses without charge. A loan is deemed to be a donation, subject to the limits and restrictions of the Act, to the extent the loan is not repaid by the date that the elected candidate is sworn into office. The term "donation" shall not include:

(a) The value of personal services provided without compensation by individuals volunteering a portion or all of their time on behalf of a TIE, provided that such an individual may not provide any paid services to a TIE at the same time as such individual serves as a volunteer for that TIE;

(b) The use of real or personal property and the cost of invitations, food, and beverages voluntarily provided by an individual to a TIE on the individual's residential premises for TIE-related activities to the extent such services do not exceed \$500 in value; and

(c) The travel expenses of any individual who, on the individual's own behalf, volunteers personal services to any TIE to the extent such expenses are unreimbursed and do not exceed \$500 in value.

"Election" means any primary, runoff primary, special, runoff special, or general election for nomination or election.

"Election cycle" means the period beginning on the first. January 12 following the most recent general election for the specific office to which a candidate is seeking nomination or election and ending on the first January 11 following the next general election for that office.

<u>"Electronic means" means facsimile transmission, email, or</u> any other electronic manner of communication that shall be prescribed by the Board.

<u>"Entity" means any organization of one or more individuals,</u> and includes any parent, subsidiary, branch, division, department, or local unit thereof.

"Expenditure" means an expenditure as defined in the Act.

"Fair market value" means: (1) for goods, the price of those goods when received in the market in which they ordinarily would have been purchased; and (2) for services, other than those provided by an unpaid volunteer, the hourly or piecework charge for the services at a commercially reasonable rate prevailing when the services were rendered.

"Filer Registration" means the submission of the information required pursuant to section 2-01 prior to the filing of disclosure statements.

"Fund" means the New York City Election Campaign Finance Fund established by the Act.

"Fundraising agent" means any of the following individuals or entities that have accepted or may accept contributions on behalf of the candidate: (1) paid or volunteer full-time campaign workers; or (2) commercial fundraising firms retained by the candidate and the agents thereof.

"Hearing officer" means the person assigned to preside over a case before OATH.

"Inauguration expenses" means expenses for an inaugural event held within seven days before or 30 days after the elected candidate is officially sworn into office. Factors used by the Board in determining whether an event is an inaugural event include but are not limited to: (1) the celebratory or commemorative nature of the event; (2) the location of the event in relation to the geography of the elected official's district; and (3) the inclusion of non-celebratory and/ or commemorative functions, including but not limited to constituent outreach or services. The burden of proving that an event is an inaugural event rests with the TIE.

"In-kind contribution" means: (1) a gift, subscription, loan, advance of, or payment for, any thing of value (other than money) made to or for any candidate; or (2) the payment by any individual or entity other than an authorized committee of compensation for the personal services of another individual or entity which are rendered to the candidate without charge. "In-kind contribution" does not include personal services provided without compensation by individuals volunteering a portion or all of their time on behalf of a candidate.

"Intermediary" means an intermediary as defined in the Act.

"Labor organization" means a labor organization as defined in the Act.

"Loan" means a monetary payment made to an authorized

committee with the expectation that the funds will be repaid by such committee.

<u>"Matchable contribution</u>" means a matchable contribution as defined in the Act.

"Mobile fundraising vendor" means any persons or entities that provided services to a candidate related to the processing or receipt of any text message contribution.

"Non-participant" means a candidate for nomination or election to a covered office who has not filed a Certification as a participant. Except as otherwise provided in these rules, a "nonparticipant" includes the candidate, every political committee authorized by the candidate for the covered election, the treasurer of each such committee, and any other agent of the candidate.

"OATH" means the Office of Administrative Trials and Hearings.

"On the ballot" means on the ballot as provided in Article 6 of the New York State Election Law and as recorded on the Board of Elections nomination or designation ledgers or contest lists.

<u>"Optional early public funds payment" means the</u> <u>disbursement of optional public financing occurring prior to two weeks</u> <u>after the last day to file designating petitions for a primary election.</u>

"Other receipts" means payments received by a candidate that are not contributions or loans, such as interest, dividends, proceeds from sales or leases of assets, and any other sources of income.

"Participant" means a candidate for nomination or election to a covered office who has chosen to join the Program for an election by submitting a Certification pursuant to § 3-703(1)(c) of the Code. Except as otherwise provided in these rules, a "participant" includes the candidate, the principal committee authorized by the candidate pursuant to § 3-703(1)(e) of the Code, the treasurer of such committee, and any other agent of the candidate.

"Political committee" means a political committee as defined in the Act.

<u>"Principal committee</u>" means the principal committee as defined in the Act.

"Program" means the New York City Campaign Finance Program established by the Act.

"Public funds" means monies disbursed from the Fund.

"Receipts" means monetary and in-kind contributions, loans, and any other payment received by a candidate.

"Registered user" means the individual registered with the wireless carrier to use the specific mobile device from which a contribution made via text message was initiated.

"Reporting period" means a time period covered by a disclosure statement, as described in section 4-05.

"Segregated account" means a bank account that may be established by a participating candidate in accordance with section 7-07(b).

"Single source" means any individual, individuals in combination, entity, or entities in combination that establish, maintain, or control another entity and every entity so established, maintained, or controlled, including every political committee established, maintained, or controlled by the same individual, individuals in combination, entity, or entities in combination.

<u>"State form" means a statement of campaign receipts and</u> expenditures required to be filed by a candidate or political committee with the State Board of Elections.

<u>"Text message contribution" means a text message</u> contribution as defined in the Act.

"Transfer" means any exchange of funds or any other thing of value between political committees, other than multicandidate committees, authorized by the same candidate pursuant to § 14-112 of the New York State Election Law. In section 16-06 the term "transfer" means funds exchanged between different bank or other depository accounts.

"Transition and inauguration entity" or "TIE" means an entity established by an elected candidate to raise and spend private funds for transition or inauguration expenses.

"Transition expenses" means expenses relating to an elected candidate's transition into office for goods and services received, used, or rendered before the elected candidate's date of inauguration. Transition expenses shall be limited to those incurred solely for the purpose of preparing to take office, such as those listed in section 13-03(b)(i), and may not be incurred after the elected candidate has taken office. Incumbent elected candidates shall not incur transition expenses. "Treasurer" means the treasurer of any authorized committee involved in a covered election, except as otherwise provided in these rules.

§ 1-03 The Board

(a) The Board is nonpartisan

(i) Pursuant to § 3-708 of the Code, the Board consists of five <u>members.</u>

(A) Two Board members are appointed by the Speaker of the City Council ("Speaker"). No more than one of them shall be enrolled in any one political party.

(B) Two Board members are appointed by the Mayor. No more than one of them shall be enrolled in any one political party.

(C) The chair of the Board is appointed by the Mayor in consultation with the Speaker.

(ii) The Board shall conduct all its activities in a strictly nonpartisan manner.

(b) Board members and staff are governed by ethical standards

(i) The Act and the Charter. Board members and staff are subject to the standards set forth in the Act and in Chapter 46 of the Charter.

(ii) Ethical guidelines. The Board shall establish and publish ethical guidelines governing the conduct of its members and staff.

(iii) Additional city ethical guidelines. Board members and staff are subject to the standards set forth in Chapter 68 of the Charter.

(c) Board administration of the Fund. To safeguard the administration of the Fund and assure candidates that sufficient public funds will be available to make all payments required by the Act in

funds will be available to make all payments required by the Act in upcoming elections, the Board shall:

(i) make budget requests for the Fund sufficient to cover all anticipated Fund obligations in the upcoming fiscal year and to maintain a reserve for contingencies;

(ii) when it has determined that monies in the Fund are insufficient or likely to be insufficient for payments to candidates, report this determination to the Commissioner of Finance and provide its estimate of the additional amount which will be necessary to make such payments pursuant to the Act (together with a detailed statement of the assumptions and methodologies on which the estimate is based), as required by § 1052(a)(10) of the Charter, not more than four days after which the Commissioner of Finance is required by § 1052(a)(10) of the Charter to transfer an amount equal to the Board's estimate from the city's general fund to the Fund;

(iii) take steps to ensure that the Fund is maintained in a separate account, credited with all sums appropriated therefor and all earnings accruing thereon, in the custody of the comptroller on behalf of the Board, as required by \$ 1052(a)(10) of the Charter;

(iv) take steps to ensure that the Fund and its administration are insulated from the risk of improper action by any city official or agency or any agent or contractor thereof;

(v) subject the Fund to periodic audits by independent outside auditors; and

(vi) take such other actions as are necessary and proper to ensure the integrity of the Fund.

(d) Advisory opinions. Upon the written request of a candidate or any other individual or entity, the Board shall issue an advisory opinion interpreting the Act and these rules, or otherwise respond in writing to the request, within 30 days of receipt of such request, or within 10 business days of receipt if such request is received less than 30 days before a covered election, to the extent practicable. At its discretion, the Board may issue advisory opinions in the absence of a request. The Board shall make public its advisory opinions and the questions of interpretation for which advisory opinions will be considered by the Board, including by publication on its website.

(e) Public petitions for rulemaking.

(i) Procedures for submitting petitions.

(A) Any individual or entity may petition the Board to consider the adoption of a rule. The request must be sent to the Executive Director and contain the following information:

(1) the rule to be considered, with proposed language for adoption; (2) a statement of the Board's authority to promulgate the rule and its purpose;

(3) arguments in support of adoption of the rule;

(4) the period of time the rule should be in effect; and

(5) the name, address, telephone number, and signature of the petitioner or the petitioner's authorized representative.

(B) Any change in the information provided pursuant to clause (5) of subparagraph (A) must be communicated promptly in writing to the Executive Director.

(C) All requests should be typewritten or submitted electronically, if possible, but handwritten petitions will be accepted, provided they are legible.

(ii) **Responses to petitions.** Within 60 days from the date the petition was received, the Board shall either deny such petition in a written statement containing the reasons for denial, or shall state in writing the Board's intention to grant the petition and to initiate rulemaking by a specified date. In proceeding with such rulemaking, the Board shall not be bound by the language proposed by petitioner, but may amend or modify such proposed language at the Board's discretion. The Board's decision to grant or deny a petition is final.

§ 1-04 Deadlines

(a) Computation of days.

(i) Counting calendar days. Where a number of days is specified as a period from a certain day, the days will be counted as the number of calendar days except for the first day of the counting period.

(ii) Where holiday falls within two-day counting period. If the counting period is two days in duration, then Saturday, Sunday, or a legal holiday must be excluded if it falls on the first or last day of the counting period.

(iii) Weekends and holidays. If the deadline for submitting a Certification or for filing a disclosure statement, other than a daily pre-election disclosure statement, falls on a Saturday, Sunday or legal holiday, the next business day becomes the deadline.

(b) Meeting Board deadlines.

(i) Submission in person. Submissions filed in person on weekdays between the hours of 9:00 A.M. and 5:00 P.M. at the offices of the Board, unless otherwise provided, are deemed submitted upon receipt, subject to review and acceptance.

(ii) Submission by electronic means, non-electronic mail or common carrier.

- (A) <u>Electronic submissions. A submission sent</u> electronically shall be deemed filed when received by the Board, subject to review and acceptance.
- (B) Non-electronic submissions.
 - (1) With postmark. A submission sent by nonelectronic mail or common carrier shall be deemed to have been received, subject to review and acceptance, on the date it was postmarked or date stamped by the carrier.
 - (2) Without postmark. Submissions sent by non-electronic mail in an envelope without a postmark will be presumed to have been sent three days prior to receipt, subject to review and acceptance, unless evidence presented to the Board, such as a post office receipt with a date stamp indicating when the submission was sent, demonstrates otherwise.

(iii) Board evaluation of submissions that arrive after the deadline. Submissions of disclosure statement documentation that arrive after 5:00 P.M. on the date of the deadline, even if submitted on or before 11:59 P.M. on such date, may prevent the Board from making a timely determination regarding payment of public funds. The Board shall make such a determination at such time as is practicable.

§ **1-05 Legibility of submissions.** The Board will not accept any electronic disclosure statement or other document, or any part thereof, that is infected with a virus, damaged, blank, improperly formatted, or otherwise unreadable or illegible.

<u>§ 1-06 Severability</u>

(a) If any provision of these rules or portion thereof is adjudged invalid by a court of competent jurisdiction, such determination shall not affect or impair the validity of the remainder of these rules. (b) If the application of any provision of these rules or portion thereof to any individual, entity, or circumstances is adjudged invalid by a court of competent jurisdiction, such determination shall not affect or impair the application thereof to other individuals, entities, and circumstances.

Chapter 2: Registration and Certification

<u>§ 2-01 Filer Registration.</u> A candidate must submit a Filer Registration, prior to filing disclosure statements, in the form and manner required by the Board, unless such candidate has previously submitted a Certification for the same election.

(a) Not a statement of intent. The submission of a Filer Registration shall not be construed as a statement of intent to run for any particular office or to join the Program.

(b) Applicable requirements. Because the requirements of the Act and these rules apply to financial transactions that take place before a candidate joins the Program, the Board advises candidates to begin compliance with all applicable requirements set forth in the Act and these rules prior to joining the Program.

(c) Deadline. A candidate must submit a complete Filer Registration no later than the day that the candidate files the first disclosure statement for an election.

(d) Form. The Filer Registration must contain any signatures and notarizations as may be required by the Board.

(e) Contents. The Filer Registration must include:

(i) the candidate's name, residential address information and telephone numbers, email address, and employment information;

(ii) the name and mailing address, and treasurer name, treasurer residential address information and telephone numbers, treasurer email address, and treasurer employment information, of every political committee authorized by the candidate that has not been terminated, and, in the case of a participant or limited participant, an indication of which such committee is the principal committee;

(iii) the name, mailing address, email address, and telephone number of any person designated by the candidate to act as liaison with the Board for each committee filing disclosure statements;

(iv) identification of all bank accounts and other depository accounts, including merchant and payment processor accounts, into which receipts have been, or will be, deposited, and all bank accounts used for the purpose of repaying debt from a previous election; and

(v) other information as required by the Board.

(f) Small campaign registration

(i) If neither the expected total cumulative receipts nor the expected total cumulative expenditures of a campaign, including expenditures made with the candidate's personal funds, exceeds an amount equal to the amount applicable to qualify for the exception provided in § 14-124(4) of the New York State Election Law, the candidate may, instead of submitting a Filer Registration, submit a small campaign registration form, which must contain such information as may be required by the Board. The small campaign registration form must also include an affirmation stating that neither the total cumulative receipts nor the total cumulative expenditures of the campaign, including expenditures made with the candidate's personal funds, will exceed the amount applicable to qualify for the exception provided in § 14-124(4) of the State Election Law, and that if such amount is exceeded, beginning on or before the deadline to file the next disclosure statement, the candidate will submit a Filer Registration and all subsequent required disclosure statements, which must include all prior financial activity beginning at the inception of the campaign.

(ii) A candidate who has filed a small campaign registration form pursuant to this section need not submit disclosure statements. If a candidate who has filed a small campaign registration form raises or spends an amount exceeding the amount necessary to qualify for the exception provided in § 14-124(4) of the New York State Election Law, the candidate must submit a Filer Registration and all subsequent required disclosure statements, beginning on or before the deadline to file the next disclosure statement. The first such statement filed must include all prior financial activity beginning at the inception of the campaign.

§ 2-02 Certification. To join the Program, a candidate must submit a Certification by the election year's deadline date as provided in § 3-703(1)(c) of the Code. A candidate may submit a Certification, in lieu of the Filer Registration, prior to filing disclosure statements. (a) Applicability. The Certification applies to all covered elections that are held in the same calendar year or to a special election to fill a vacancy in an office covered by the Act. A candidate need only file one Certification for the primary and general election, and any related runoff election. Special elections and all other elections require a separate Certification.

(b) Deadlines. The deadline for filing a Certification shall be: (i) in the case of a primary or general election, the later of the tenth of June in the year of the covered election or the thirtieth day after a special election is held to fill a vacancy for the office sought by the candidate; or (ii) in the case of the declaration of an extraordinary circumstance, on or before the seventh day following the declaration by the Board of the extraordinary circumstance.

(c) Rescission; Failure to timely certify. A candidate may rescind such candidate's Certification by submitting a Certification rescission form on or before the ninth Monday preceding the primary election or prior to the receipt of public funds, whichever occurs first. A candidate who rescinds a Certification in a timely manner or who does not file a timely Certification shall be deemed to be a non-participant.

(d) Form. The Certification must contain any signatures and notarizations as may be required by the Board. A candidate who has already submitted a Filer Registration may submit a Certification in electronic format, except that if a new principal committee treasurer has been appointed since the Filer Registration was submitted, the Certification must be submitted in non-electronic format. A Certification submitted in a non-electronic format must contain an original notarized signature from both the candidate and the principal committee treasurer.

(e) Contents. The Certification must include all filer registration information required by section 2-01 and such other information as required by the Board, including all information necessary to receive payment by electronic funds transfer. In the Certification, the candidate shall designate a principal committee.

(f) Legal effect. The candidate must comply fully with Program requirements in all elections for which the Certification is submitted, regardless of the office sought and regardless of whether the candidate: (1) meets all the requirements of law to have such candidate's name on the ballot in the election; (2) meets the Act's threshold for eligibility for public funds; (3) accepts public funds; or (4) is otherwise not eligible to receive public funds in the election.

<u>§</u> 2-03 Amendments to Filer Registration or Certification. The candidate must notify the Board of any material change in the information required to be listed on the candidate's Filer Registration or Certification occurring prior to the covered election or within a period of five years following the covered election, including any new information, or any change to any required information, concerning any political committee, bank account, merchant or payment processor account, candidate or treasurer employment, address, telephone number, or email address, in such manner as may be provided by the Board. Such notification must be submitted no later than the next deadline for filing a disclosure statement, or, in the case of changes that occur after the deadline for the last disclosure statement required to be filed, no later than 30 days after the date of the change; provided, however, that if the candidate has extinguished all outstanding liabilities resulting from the applicable election, including payment of any penalties or repayment of public funds owed to the Board, the candidate need not notify the Board of any material change after the issuance of the candidate's final audit report.

§ 2-04 Non-participants. A non-participant is not eligible to receive public funds pursuant to § 3-705 of the Code and shall not be subject to the expenditure limitations provided in § 3-706 of the Code. A nonparticipant is not subject to the contribution limitations set forth in §§ 3-703(1)(f) and (1-a) of the Code when such contributions are made from the non-participant's personal funds or personal property, including funds or property jointly held with the non-participant's spouse, domestic partner, or unemancipated children. A non-participant is subject to the contribution and disclosure requirements provided in § 3-719 of the Code, and may be subject to penalties pursuant to §§ 3-710.5 and 3-711 of the Code for violations of the Act and of these rules.

§ 2-05 Petition for extraordinary circumstances

(a) Pursuant to § 3-703(1)(c) of the Code, a Certification may be filed on or before the seventh day after the occurrence of an extraordinary circumstance in a covered election. A candidate in such election may file a petition setting forth facts alleged to constitute an extraordinary circumstance within seven days of the date on which the candidate reasonably believes that the extraordinary circumstance occurred. The Board, following review of the petition, or on its own initiative, may declare an extraordinary circumstance.

(b) An "extraordinary circumstance" includes: (i) the death of a candidate in an election, (ii) the resignation or removal of the person holding the office sought, and (iii) the submission to the Board of a written declaration, sworn to or affirmed by the holder of the office sought, terminating such office holder's campaign for reelection (which may be submitted together with a petition under subdivision (a)).

§ 2-06 Training. A candidate or such candidate's representative must attend a training provided by the Board concerning compliance with the requirements of the Program and use of the disclosure software. Such training must be completed on or before the financial disclosure cut-off date of the 32-day pre-election primary disclosure statement. The individual attending the training may be the candidate, the candidate's campaign manager or treasurer, or another individual with significant managerial control over a campaign (not including campaign consultants). The training attendee must be listed on the candidate's Filer Registration or Certification.

Chapter 3: Eligibility to Receive Public Funds

§ 3-01 Candidates must demonstrate eligibility

(a) No payments from the Fund shall be made to a candidate unless the Board has determined that such candidate has demonstrated that such candidate has met all eligibility requirements set forth in the Act and these rules, including the threshold for eligibility pursuant to <u>§</u> 3-703(2) of the Code.

(b) Ballot status. In order to be eligible to receive public funds, a candidate in a covered election must meet all of the requirements to appear on the ballot as provided in Article 6 of the New York State Election Law, and be opposed by at least one other candidate on the ballot, or, for the optional early public funds payment, certify that such candidate intends to meet all the requirements of law to have such candidate's name on the ballot for the primary or general election.

(c) Preliminary determinations. Throughout the audit process or as a result of an investigation, the Board may make a preliminary determination that a candidate is ineligible to receive public funds. In the event of a preliminary determination of ineligibility, the Board will send written notification to the candidate and the candidate may request reconsideration of such determination pursuant to Section 7-09.

(d) Basis for ineligibility determination

(i) **Pre-election.** The Board may determine that a preelection public funds payment shall not be paid to a candidate if:

(A) the candidate fails to submit a disclosure statement required by these rules;

(B) the candidate fails to provide to the Board, upon its request, documents or records required by Chapter 4 of these rules, or other information that verifies campaign activity; or

(C) there is reason to believe that the candidate has committed a violation of the Act or these rules not otherwise enumerated in paragraph (ii) of this subdivision.

(ii) **Pre-election or post-election.** The Board may determine that neither a pre-election nor a post-election public funds payment shall be paid to a candidate if:

(A) the candidate has failed to meet one of the eligibility criteria of the Act or these rules;

(B) the candidate is required to repay public funds previously received, as described in sections 9-01 and 9-02, or the candidate has failed to pay any outstanding claim of the Board for the payment of civil penalties or the repayment of public funds against such candidate or such candidate's authorized committee or an authorized committee of such candidate from a prior covered election, provided that the candidate has received written notice of the potential payment obligation and potential ineligibility determination in advance of the certification deadline for the current covered election and an opportunity to present reasons for such candidate's eligibility for public funds to the Board;

(C) previous public fund payments to the candidate for the election equal the maximum permitted by the Act;

(D) the candidate fails to demonstrate compliance with §12-110 of the Code, as required pursuant to §3-703(1)(m) of the Code and section 3-05;

(E) the candidate endorses or publicly supports such candidate's opponent for election pursuant to §3-705(9) of the Code:

(F) the candidate loses in the primary election but remains on the ballot for the general election and fails to certify to the Board, as required by \$3-705(10) of the Code, that such candidate will actively campaign for office in the general election, provided that such certification must be complete on or before the 32-day pre-general election disclosure statement deadline; or the candidate certifies to the Board that such candidate will actively campaign for office in the general election but thereafter fails to engage in campaign activity that shall include but not be limited to, raising and spending funds, and broadly soliciting votes; (G) the candidate has exceeded the applicable expenditure limits provided in § 3-706 of the Code;

(H) the candidate has been found by the Board, in the course of Program participation, to have committed fraud or material misrepresentation or to be in breach of certification pursuant to section 3-01(e); or

(I) there is reason to believe that, in the course of Program participation, the candidate has engaged in conduct detrimental to the Program that is in violation of any other applicable law.

(e) Breach of certification

(i) The Board considers any of the following to be a fundamental breach of a candidate's certification:

(A) the submission to the Board of documentation or information that the candidate knew or should have known was false or fictitious in whole or in part, including a disclosure statement which the candidate knew or should have known includes substantial fraudulent matchable contribution claims;

(B) the misrepresentation of a material fact in any submission of such documentation or information to the Board;

(C) the falsifying or concealment of any such documentation or information;

(D) the use of public funds to make or reimburse substantial campaign expenditures that the candidate knew or should have known were fraudulent;

(E) coordination in alleged independent expenditures, whereby material or activity that directly or indirectly assists or benefits a candidate's nomination or election, which is purported to be paid by independent expenditures, was in fact authorized, requested, suggested, fostered, or cooperated in by the candidate; and

(F) the use of a political committee or other entity over which a candidate exercises authority to conceal from the Board expenditures that directly or indirectly assist or benefit the candidate's nomination or election.

(ii) In the event of a fundamental breach of a candidate's certification, the candidate will be deemed by the Board to be ineligible for public funds for the covered election and to have forfeited all public funds previously received for the elections covered by the Certification. Additionally, the candidate will be subject to such civil and criminal sanctions as are applicable under § 3-711 of the Code and other applicable law.

(iii) This section is not intended to be an enumeration of all circumstances that may constitute a fundamental breach of a candidate's certification, as may be determined by the Board.

§ **3-02 Disqualification from ballot.** A candidate who has been disqualified from the ballot, or who is no longer opposed by a candidate who appears on the ballot, is not eligible to receive public funds.

(a) Notice of appeal. The candidate must notify the Board immediately, in writing, if the disqualified candidate is seeking to appeal or otherwise remedy a disqualification. This notice must indicate whether a judicial appeal is being taken as of right or by permission and the specific nature of any other judicial remedy sought.

(b) Disqualification reversed. The candidate must immediately inform the Board, in writing, if the disqualification of the candidate or the opponent is reversed by a court of competent jurisdiction.

§ **3-03 Write-in candidates.** A candidate who is seeking nomination or election to a covered office as a write-in candidate must promptly notify the Board in writing.

§ 3-04 Termination of candidacies

(a) The Board may send a notice to a candidate that such candidate's candidacy has been deemed terminated if such candidate: (1) has not received public funds; (2) has not submitted a petition for payment after final disqualification from the ballot, pursuant to section 7-01(e) (ii); and (3) is not on the ballot for that election.

(b) If the terminated candidate is seeking nomination or election as a write-in candidate, or, in the case of a participant, intends to submit a petition for public funds pursuant to section 7-01(e)(ii), the candidate must notify the Board within five business days after receiving the notice of termination, in which case the Board may reverse the termination. A candidate whose termination has been reversed must continue to submit disclosure statements as required by these rules.

(c) A candidate may also request that the Board deem such candidate's candidacy terminated because such candidate has satisfied all three requirements of subdivision (a), or has satisfied requirements (1) and (2) of subdivision (a) and has ceased campaigning and has verified that

fact in the form and manner required by the Board.

(d) Terminated candidates are required to abide by Program obligations, such as maintaining requisite records, submitting documentation or information in response to requests by the Board, and paying penalties for violations of the Act and these rules.

<u>§ 3-05 Proof of filing with the Conflicts of Interest Board</u> (COIB); payment of penalties

(a) **Requirements**. In order to be eligible to receive public funds, a candidate must comply with the requirements in § 12-110 of the Code, including payment of any penalties assessed by the conflicts of interest board. The Board may obtain confirmation of the candidate's compliance from the conflicts of interest board. The failure of a candidate to demonstrate such compliance by the deadline established by the conflicts of interest board or by the reporting deadline immediately preceding the date on which the candidate seeks to receive a public funds payment, whichever is earlier, may result in a delay of any payment by the Board of public funds the candidate may otherwise be eligible to receive until the next scheduled payment date.

(b) Due dates. A candidate may submit proof of compliance with the Board and such proof shall be considered timely submitted if it is submitted to the Board on or prior to the last business day of July in the year of the covered election, except as provided by Subdivision (a).

Chapter 4: Records and Reporting

§ 4-01 Records to be kept

(a) Generally

(i) Candidates must keep records that enable the Board to verify the accuracy of disclosure statements, substantiate that expenditures were made in furtherance of the campaign, were qualified expenditures, or were permissible post-election expenditures, and confirm any matchable contributions claimed. Candidates must maintain and may be required to produce originals and copies of checks, bills, or other documentation to verify contributions, expenditures, or other transactions reported in their disclosure statements. Candidates must maintain clear and accurate records sufficient to show an audit <u>trail that demonstrates compliance with the Act and these rules.</u> The records must be made and maintained contemporaneously with the transactions recorded, and maintained and organized in a manner that facilitates expeditious review by the Board. Nothing in this chapter shall be construed to modify the requirements of § 14-118 of the New York State Election Law. The records maintained for each campaign finance transaction, whether maintained on paper or electronically, must be accurate and, if necessary, modified promptly to ensure continuing accuracy.

(ii) If at any time a candidate becomes aware that a record of an expenditure is missing or incomplete, the candidate may create a new record or modify an existing record, provided that the record so created or modified is clearly identified by the candidate as such, and provided that the candidate also creates a record, in the form of a signed, dated, and notarized statement by the candidate, treasurer, or other campaign representative having first-hand knowledge of the matter, explaining the reasons for and the circumstances of the creation or modification of the missing or incomplete record. If the missing or incomplete record is an invoice or contract from a vendor, the candidate must in the first instance attempt to get a duplicate or more complete record directly from such vendor. The Board reserves the right not to accept such non-contemporaneous record created or modified pursuant to this paragraph if it deems that the record is not sufficient to document the actual transaction.

(b) Receipts

(i) Deposit slips. Candidates must maintain copies of all deposit slips. The deposit slips must be grouped together with the monetary instruments representing the receipts deposited into the bank or other depository accounts held by the candidate for an election. Where the bank or depository does not provide itemized deposit slips, candidates must make a contemporaneous written record of each deposit. Such written record must indicate the date of the deposit, the amount of each item deposited, whether each item deposited was a check, a money order, or cash, and the total amount deposited.

(ii) Contribution and loan records

(A) Generally. For each contribution received, all candidates must maintain records demonstrating the source and details of the contribution as described herein. All records required to be maintained must be provided to the Board upon request.

(1) Cash contributions. For each contribution received from an individual contributor via cash,

the record must be in the form of a contribution card.

(2) Money order contributions.

(I) For each contribution received via money order, the record must include a copy of the money order made out to the authorized committee.

(II) The candidate must also maintain a contribution card, if the contributor's name and residential address are not printed on the money order by the issuer.

(3) Check contributions.

(I) For each contribution received via check, the record must include a copy of the check made out to the authorized committee and signed by the contributor.

(II) For each contribution received from an individual contributor via check, the candidate must also maintain a contribution card, if the check used to make the contribution is not signed by the contributor.

(4) Credit card contributions.

(I) For each contribution received via credit card, including contributions received over the internet, the record must have been provided by the merchant or processor and must contain: the contributor's name, residential address, credit card account type, credit card account number, credit card expiration date, the amount of the contribution, and an indicator showing that the contribution was charged to the contributor's account and processed. In the case of credit card contributions made over the internet, the contributor must actively agree online to an affirmation statement, as required by subparagraph (C)(1) of this paragraph, and the candidate must maintain a copy of all website content concerning the solicitation and processing of credit card contributions.

(II) The candidate must also maintain copies of the merchant account or payment processor agreement, all merchant account statements, credit card processing company statements and correspondence, transaction reports, or other records demonstrating that the credit card used to process the transaction is that of the individual contributor (including proof of approval by the credit card processor for each contribution and proof of real time address verification), the account's fee schedule, and the opening and closing dates of the account. Merchant account statements must be provided in such form as may be required by the Board.

(5) Text message contributions. For each contribution received via text message, the record must have been provided by the mobile fundraising vendor and must contain: the contributor's name, residential address, and phone number; the amount of the contribution; and the name, residential address, and phone number of the registered user of the specific mobile device used to initiate the contribution, to the extent that such information may be reasonably obtained under law. The candidate must also maintain the following records for each text message contribution received:

> (I) copies of all relevant third-party vendor agreements between the candidate and mobile fundraising vendor, copies of records maintained by a mobile fundraising vendor listing contributors and amounts pledged and paid, receipts indicating fees paid by the candidate to a mobile fundraising vendor and fees deducted by such vendor, and similar records relating to the solicitation or receipt of text message contributions;

(II) copies of any content used by the candidate to solicit text message contributions; and (III) copies of any templates or scripts used by a mobile fundraising vendor to communicate with a contributor in facilitating and processing a text message contribution.

(6) Segregated account documentation.

(I) Segregated account contribution cards. For each contribution from an individual contributor that the candidate deposits into a segregated bank account pursuant to section 7-07(b), the record must be in the form of a contribution card.

(II) Segregated account bank statements, contribution cards, and checks. Candidates seeking to comply with the exception contained in section 7-07(b) must submit segregated account contribution cards and copies of segregated account bank statements and checks to the Board in. the manner and to the extent provided by section 7-07 with each disclosure statement filing.

(7) Intermediaries. For each contribution accepted from an intermediary, including any contributions delivered to a fundraising agent, or solicited by an intermediary where such solicitation is known to the candidate, the candidate must maintain a separate record in the form of an intermediary statement. The intermediary statement must contain: the intermediary's name, residential address, employer and business address; the names of the contributors; and the amounts contributed. This record must be signed by the intermediary, or if the intermediary is unable to sign such intermediary's name, marked with an "X" by the intermediary and signed by a witness. Adjacent to the signature or mark, the intermediary signed or marked the form.

(B) Contribution cards

(1) Contribution cards must contain the contributor's name and residential address, the amount of the contribution, the authorized committee's name, and the contributor's selection of an instrument code corresponding to the instrument used to make the contribution. Credit card contribution cards must also contain the credit card account type, account number, and expiration date.

(2) Contribution cards must be signed by the contributor or, if the contributor is unable to sign such contributor's name, marked with an "X" by the contributor and signed by a witness to the contribution. Adjacent to the signature or mark, the contributor must write the date on which such contributor signed or marked the contribution card.

(3) After a contribution card has been signed, it may not be corrected, modified, or altered by anyone other than the contributor.

(4) The Board shall provide a template of all contribution cards required to be maintained pursuant to this subparagraph.

(5) A contribution card that contains any additional information and signatures required by Section 7-07(b) must also satisfy the requirements of that section.

(C) Affirmation statements

(1) Unless otherwise specified herein, above the line for the contributor's signature, contribution cards must state: "I understand that State law requires that a contribution be in my name and be from my own funds. I hereby affirm that I was not, nor, to my knowledge, was anyone else, reimbursed in any manner for this contribution; that this contribution is not being made as a loan; and that this contribution is being made from my personal funds or my personal account, which has no corporate or business affiliation."

(2) For text message contributions, the candidate must maintain records demonstrating that the contributor has certified via text message the following statement: "I certify I am the registered user of this phone and will pay the amount specified from my personal funds."

(3) Segregated account contribution cards must state, above the line for the contributor's signature: "I understand that this entire contribution will be used only (i) to pay expenses or debt from a previous election; (ii) by the candidate for an election other than the election for which this contribution is made; or (iii) to support candidates other than the candidate to whose campaign this contribution is made, political party committees, or political clubs. I further understand that this contribution be in my name and be from my own funds. I understand that State law requires that a contribution be in my name and be from my own funds. I hereby affirm that I was not, nor, to my knowledge, was anyone else, reimbursed in any manner for this contribution; that this contribution is not being made as a loan; and that this contribution is being made from my personal funds or my personal account, which has no corporate affiliation."

(4) Intermediary statements must state, above the line for the intermediary's signature: "I hereby affirm that I did not, nor, to my knowledge, did anyone else, reimburse any contributor in any manner for a contribution, and that none of the submitted contributions were made by the contributor as a loan. The making of false statements in this document, which will be submitted to the Campaign Finance Board, is punishable as a class E felony pursuant to § 175.35 of the Penal Law or a Class A misdemeanor pursuant to § 210.45 of the Penal Law."

(D) Transfers. Candidates must maintain all records specified by the Board regarding transfers. In the case of a transfer to an authorized committee from a committee not otherwise involved in the covered election, unless the transferring committee is another authorized committee disclosure statements with the Board in a timely manner, the candidate must maintain records demonstrating that the funds underlying the transfer derive from contributions intended for the committee receiving the transfer. Such records must include, for each contribution to be transferred, a record indicating the contributor's intent to designate the contribution for the covered election. The record must be obtained prior to receipt of the transfer and must be signed by the contributor, or, if the contributor is unable to sign such contributor's own name, marked with an "X" and signed by a witness. Adjacent to the signature or mark, the contributor must write the date of the contribution. Above the line for the contributor's signature, the record must state: "I understand that this contribution will be used by the candidate for an election other than that for which the contribution was originally made. I further understand that State law requires that a contribution be in my name and be from my own funds. I hereby affirm that I was not, nor, to my knowledge, was anyone else, reimbursed in any manner for this contribution; that this contribution is not being made as a loan; and that this contribution is being made from my personal funds or my personal account, which has no corporate or business affiliation."

(E) In-kind contributions. For each in-kind contribution, candidates must maintain a written record that provides the date the contribution was made, the name and residential address of the contributor, a detailed description of the goods or services provided, the fair market value of the contribution, and such further information and documentation necessary to show how the fair market value of the contribution was determined. The Board shall provide a specimen of such a record.

(F) Loans. For each loan received, including loans made by the candidate, candidates must maintain a loan agreement, documentation of each loan repayment, and, if applicable, documentation that shows that the loan has been forgiven. The loan agreement must be contemporaneous and in writing, must be signed and dated by both parties, and must provide all terms and conditions of the loan, including the amount and term of the loan and whether interest is being charged. The candidate must retain copies of loan and loan repayment checks and records of electronic transactions showing the source of the funds.

(G) Business dealings with the City. For each individual or entity making a contribution, loan, guarantee or other security for such loan in excess of the amounts set forth in <u>§3-703(1-a) of the Code, candidates must maintain all records</u> specified by the Board concerning whether such individual or entity has business dealings with the City.

(iii) Photocopies of checks and other monetary instruments. Candidates must maintain a photocopy of each check or other monetary instrument, other than cash, representing a contribution or other monetary receipt. In order for a contribution in the form of a check signed by an authorized agent of the contributor to be matchable, candidates must maintain:

(A) a copy of the check upon which is printed the name of the actual contributor; and

(B) a document, signed by the contributor, which indicates:

(1) that the person signing the check is authorized to do so;

(2) the date and amount of the contribution; and

(3) the principal committee's name.

(c) Disbursements

(i) By check or debit. Candidates must make all_ disbursements by check or debit from the committee bank_ account, except for petty cash.

(ii) Petty cash. Candidates may maintain a petty cash fund of no more than \$500 out of which they may make disbursements not in excess of \$100 to any individual or entity per purchase or transaction. If a petty cash fund is maintained, the candidate must maintain a petty cash journal in C-SMART including the name of every individual or entity to whom any disbursement is made, as well as the date, amount, and purpose of the disbursement.

(iii) Credit card purchases. Candidates must maintain a monthly billing statement and customer receipt for each disbursement from a committee credit card showing the underlying purchases, including the vendor for each charge.

(iv) Bills

(A) Candidates must maintain bills for disbursements for goods or services provided.

(B) Documentation for goods or services must be contemporaneous and must provide the date the vendor was retained or the date the goods or services were provided, the vendor's name and address, the amount of the expenditures, and a detailed description of the goods and services provided. If the invoice supplied by the vendor does not meet these requirements, the vendor may provide an additional, more detailed document or replacement document with sufficient detail. If the vendor does not provide such a document, the candidate must create an additional record containing the necessary information, and such record must be signed by the candidate. The newly created record must satisfy the requirements of Section 4-01(a).

(C) For wages, salaries, and consulting fees, candidates must maintain a contemporaneous record, signed by the employee or consultant and the candidate, and dated, providing the name and address of the employee or consultant, a detailed description of the services, the amount of the wages, salary or consulting fees, the date(s) on which the work was performed, the period for which the individual was retained, and, if applicable, a detailed breakdown of the number of hours worked. The Board shall provide specimens of records for employees and consultants, including timesheets for election day workers and consultant agreements.

(D) Candidates must maintain written documentation showing that a bill has been forgiven or settled for a lesser amount.

(v) Vendors. In addition to obtaining and keeping contemporaneous documentation (such as bills) for all goods and services provided by vendors, including campaign consultants and attorneys, and employees, when a candidate retains or otherwise authorizes an individual or entity (including an employee) to provide goods or services to the campaign, and the candidate knows or has reason to believe that the goods or services to be provided directly or indirectly by this vendor will exceed \$1,000 in value during the campaign, the candidate must:

> (A) keep a copy of the contemporaneously written contract with the vendor, which must, at a minimum, provide the name and address of the vendor, be signed and dated by both parties, state the terms of the

contract including the terms of payment and a detailed description of the goods or services to be provided, and must include, if the contract was at any time amended, a contemporaneously written amendment, signed and dated by both parties and describing in detail the changes to the terms and conditions of the contract, or

(B) if no contemporaneously written contract has been entered into, keep a contemporaneously written record that includes the date the vendor is retained or otherwise authorized by the candidate, the name and address of the vendor, and the terms of the agreement or understanding between the candidate and the vendor including the terms of payment and a detailed description of the goods or services the vendor is expected to provide. If the agreement or understanding was at any time amended, the candidate must create and maintain a contemporaneously written record describing in detail the changes to the terms and conditions of the agreement or understanding.

(C) In addition to the records to be kept pursuant to subparagraphs (A) or (B) above, the candidate must keep evidence sufficient to demonstrate that the work described in the contract was in fact performed and completed. Such evidence may include samples or copies of work product, emails, time records, phone records, and photographs or other documentary evidence. Where such evidence is nonexistent or unavailable, the candidate must maintain affidavits signed by the vendor and either the candidate, treasurer, or other campaign representative having first-hand knowledge, describing the goods or services provided and the reason(s) why documentary evidence is nonexistent or unavailable.

(vi) Advance purchases and reimbursement of advances. Candidates must maintain records of advances, which must include the name and address of each individual or entity that made an advance on behalf of the campaign, the amount so advanced, the date of the advance, the name and address of the payee to whom advanced funds were paid, the amount paid, the purpose of the payment, and the manner of payment, including check number, credit card name, or cash. The record of the advance must be signed by the individual making the advance purchase. A receipt, bill, or invoice from the vendor must be attached to the record.

(vii) Subcontracted goods and services. Candidates required to itemize the cost of subcontracted goods and services pursuant to Section 4-05(c)(iv)(D) must obtain and maintain documentation from the vendor, consultant, or other individual or entity who or which subcontracts, containing all information required to be disclosed pursuant to that section.

(viii) Political advertisements and literature. Pursuant to § 14-106 of the New York State Election Law, candidates must maintain copies of all advertisements, pamphlets, circulars, flyers, brochures, letterheads, and other printed matter purchased or produced and a broadcast record of all radio or television time purchased and scripts used therein.

(ix) Travel. Candidates must obtain and maintain copies of all checks, bills, or other documentation to verify campaignrelated travel transactions reported in disclosure statements. In addition to the above, for all travel candidates must the campaign-related purpose of the travel, the complete travel itinerary, the dates of the travel, and the names of all individuals who participated in the travel; provided, however, that such records shall not be required for travel by public transportation within New York City, with the exception of unlimited-use MetroCards, for which candidates must create and maintain a contemporaneous record containing the dates on which each such card was purchased and, if a card is assigned to a single individual, the name of each such individual. For travel by private car, candidates must create and maintain a contemporaneous travel log providing, for each trip and each vehicle, the names of the driver and passengers, the date(s) and purpose of each trip, the itinerary, including all the locations of any campaign events and other stops, the beginning and ending mileage, and the total mileage. Travel between two stops is considered an individual trip for logging purposes even if the stops are part of a multi-stop itinerary. For the purposes of reporting and reimbursing campaign expenditures, candidates must calculate expenditures for travel by private car based on mileage according to the purposition of Directing Sin of the Nur mileage according to the provisions of Directive Six of the New York Čity Comptroller.

(d) Bank records. Candidates must maintain the following records received from or in connection with banks and other depositories relating to accounts, and must submit with each disclosure statement a copy of all such records not previously provided: (i) all periodic bank or other depository statements in chronological order, maintained with any other related correspondence received with those statements, such as credit and debit memos, deposit slips, and contribution checks returned because of insufficient funds; and

(ii) the front and back of all returned and cancelled disbursement checks, including substitute checks which may be returned by the bank in lieu of cancelled checks.

(e) Fundraisers. Candidates must maintain records for all fundraising events, which must contain: the date and location of the event; the individual(s) or organization(s), other than the candidate's authorized committee, hosting the event; an itemized list of all expenses incurred in connection with the event, including all expenses whether or not paid or incurred by the authorized committee; and the contributor name and amount of each contribution received at or in connection with the event. This subdivision does not apply to activities at an individual's residential premises, including house parties, to the extent that the total costs of each such activity do not exceed \$500 and are not contributions pursuant to \$ 3-702(8)(ii) of the Code.

§ 4-02 Record retention. The candidate must retain all records and documents required to be kept under section 4-01 for five years after the last disclosure statement deadline set by the Board for the covered election, or until the Board has issued the candidate's final audit report and the candidate has extinguished all outstanding liabilities resulting from the applicable election including payment of any penalties or repayment of public funds owed to the Board, whichever is later.

§ 4-03 Assistance to candidates. In order to promote compliance with the requirements of the Act and these rules, the Board's staff shall offer assistance to candidates in developing campaign procedures for gathering campaign finance information and keeping records and shall, to the extent feasible, provide model recordkeeping forms and templates.

§ 4-04 Failure to maintain or provide records. A candidate's failure to keep or provide records or other information to the Board, upon its request or as required by these rules, may result in a determination that matchable contribution claims are invalid; a determination that the candidate must repay public funds to the Board; the withholding of all or a portion of a public funds payment; and the assessment of penalties.

<u>§4-05 Disclosure statements</u>

(a) Form. Disclosure statements must be generated by C-SMART, and such statements, as well as all supporting documentation, including bank statements, must be submitted using C-SMART. Daily preelection disclosure statements must be submitted using C-SMART within 24 hours after a contribution, loan, or expenditure that meets the pre-election daily disclosure requirement has been accepted or made.

(i) **Deficient submissions.** A submission is deficient and may be rejected or deemed incomplete if: (1) it is not submitted in a format or manner authorized by the Board; (2) it is not submitted with the backup documentation substantiating each matchable contribution claimed within the particular statement, or said records are not accessible or legible; or (3) it is not submitted with all of the committee's bank statements from the applicable disclosure period. Any document that is illegible, improperly annotated, damaged, blank, improperly formatted, or otherwise unreadable by the Board, shall be deemed not to have been submitted.

(ii) <u>C-SMART upgrades.</u> The Board may issue upgrades or system improvements of C-SMART or its user instructions. To the extent reasonably practicable, the Board shall provide candidates with reasonable advance notice of such upgrades.

(iii) Verification. The candidate or treasurer must verify that the submission is true and complete to the best of such candidate's or treasurer's knowledge, information, or belief. The disclosure statement must contain such signatures as may be required by the Board.

(iv) Exceptions. Any candidate who seeks to submit disclosure statements, or a portion thereof, in any format or manner other than that permitted by this section, including but not limited to non-electronic formats and electronic formats not generated by C-SMART, must submit a written request for authorization to the Board no later than four weeks before the filing date for the first disclosure statement for which the candidate desires an exception from such requirements, or in the case of a special election, as soon as possible but no later than seven days before such disclosure statement filing date. Such written request must be in a form and manner as prescribed by the Board. Candidates who demonstrate that submission of disclosure statements in an electronic format would pose a substantial hardship shall be permitted, upon request, to submit disclosure statements to the Board in non-electronic formats. Board authorization shall be in writing and shall apply only to the candidate, paper forms, and electronic submission form and manner specified therein. The authorization shall indicate whether it applies to one or more disclosure statements.

(b) Timing of submissions

(i) Disclosure statements must be received by the Board no later than 11:59 P.M. on the due date.

(ii) Filing dates. The Board shall publish a schedule of disclosure statement filing dates on its website on or before. March 1 in the first year of each election cycle, or as soon as is practicable after the State Board of Elections has published its schedule.

(A) Semi-annual disclosure statements are due on January 15 and July 15 in each year of the election cycle and on January 15 in the year after the election.

(B) Pre-election disclosure statements are due 32 and 11 days before the election and, at the Board's discretion, on March 15 and May 15 in the year of the election.

(C) Post-election disclosure statements are due 10 days after a primary election and 27 days after a general or special election.

(D) Weekends and holidays. The Board's published schedule of disclosure statement filing dates shall reflect that if a disclosure statement is due to be submitted on a Saturday, Sunday, or legal holiday, submission must be made on the next business day.

(iii) **Reporting period.** The Board shall publish a schedule of the reporting periods for each disclosure statement.

(A) First disclosure statement. The reporting period for a candidate's first disclosure statement shall begin on the day the candidate first raises or spends funds in furtherance of the candidate's election for a covered office.

(B) All subsequent disclosure statements. The reporting period for each disclosure statement shall: (i) begin on the third day before the deadline for the submission of the candidate's preceding disclosure statement; and (ii) conclude on and include the fourth day before the deadline for the submission of that disclosure statement.

(c) Content

(i) Summary information. Each disclosure statement must include the following information about the committee involved in the election: (A) the cash balance at the beginning and end of the reporting period; (B) total itemized and unitemized contributions, loans, and other receipts accepted during the reporting period; and (C) total itemized and unitemized expenditures made during the reporting period. A separate disclosure statement must be submitted for each committee involved in the election. All data reported in disclosure statements and amendments must be accurate as of the last day of the reporting period.

(ii) Contributions

(A) Reporting requirements. To fully report a contribution accepted during the reporting period, the candidate must report, for each contribution:

(1) the contributor's and intermediary's (if any) full name, residential address, occupation, employer, and business address;

(2) the date the contribution was received by the candidate;

(3) the amount of the contribution;

(4) the form of the contribution (cash, check, money order, credit card, text, or other);

(5) the number of the check or money order, if applicable;

(6) the date and amount of each contribution returned to a contributor, the account from which the funds used to make the return originated, and the number of the bank or certified check used to issue the return of funds;

(7) each previously reported contribution for which the check was returned unpaid;

(8) in the case of contributions claimed as matchable or in excess of the amounts set forth in §3-703(1-a) of the Code, whether the contributor has business dealings with the City as defined in the Act; (9) whether the contribution was accepted for a runoff or rerun election in accordance with Chapter 16 and Section 5-12;

(10) whether the contribution was accepted to be deposited into a segregated bank account in accordance with Section 7-07(b); and

(11) such other information as the Board may require.

(B) Matchable contribution claims

(1) Contemporaneous reporting. Matchable contributions must be reported in the disclosure statement covering the reporting period in which they were received. The candidate's disclosure statement must state the amount of matchable contributions claimed in a reporting period, and must indicate which contributions are claimed for match.

(2) Backup documentation. For each matchable contribution claimed in the disclosure statement, candidates must submit a copy of the records required to be maintained pursuant to Section 4-01(b).

(C) Contributions totaling \$99 or less from a single source. Contributions totaling \$99 or less from a single source need not be separately itemized in a disclosure statement, unless such contributor is an employee of the candidate or of the spouse, or domestic partner of such candidate or of a business entity in which such candidate, spouse, or domestic partner has an ownership interest of 10% or more or in which such candidate, spouse or domestic partner holds a management position, such as the position of officer, director, or trustee.

(D) Must itemize all contributions from a single source that exceeds \$99. If a candidate has accepted contributions totaling more than \$99 from a single source, all contributions comprising the total (including previously unitemized contributions) must be fully reported (i.e., itemized) in the same disclosure statement. All subsequent contributions from that single source must be fully reported as well.

(E) Affiliated contributors. Affiliated contributors considered to be a "single source" under Sections 1-02 and 5-10(b) must be reported.

(iii) Other receipts. To fully report other receipts accepted during the reporting period, the candidate must report, for <u>each receipt:</u>

(A) the date of receipt;

(B) the amount of the receipt;

(C) the type of receipt; and

(D) such other information as the Board may require.

(iv) Expenditures

(A) **Reporting requirements.** To fully report expenditures, including outstanding liabilities, made by the candidate during the reporting period, the disclosure statement must be itemized to include the following information:

(1) the name and address of each vendor or payee;

(2) the bill or invoice date and amount;

(3) the purpose code and explanation of each expenditure;

(4) the exempt code, if any; and

(5) such other information as the Board may require.

(B) In addition to the information required in subparagraph (A), candidates must provide the following information concerning each payment:

> (1) the date and amount of each payment, including exempt amount, if any;

(2) the payment method, including check number and committee bank account; and

(3) the amount of remaining outstanding liability to the vendor or payee; and

(4) such other information as the Board may require.

(C) Expenditures of less than \$50. Expenditures of less than \$50 do not need to be separately itemized in a disclosure statement; however, public funds may not be used to pay for unitemized expenditures.

(D) Subcontractors

(1) In addition to reporting any expenditures to the vendor, if aggregate payments by the vendor to a subcontractor exceed \$5,000, the candidate must report:

(I) the name and address of that

subcontractor;

(II) the amount(s) expended to the subcontractor; and

(III) the purpose code(s) of the subcontracted goods or services.

(2) Disclosure must occur either beginning in the reporting period in which such cost first exceeds
\$5,000 or in the first post-election disclosure statement for the election to which the expenditure relates.

(E) Credit card purchases. For expenditures paid with a credit card, the candidate must report the vendor and purchase price of any goods or services purchased. Disbursements to credit card accounts must not be itemized as such.

(F) Contributions to political committees. Political contributions to political committees that support or oppose candidates in New York City (except political committees of other candidates), including state party committees that are made by a candidate with such committees, that are made by a candidate with such candidate's personal funds and that, in the aggregate for any single political committee, exceed the contribution <u>limit applicable to the office of mayor for contributors</u> having business dealings with the City pursuant to Section 3-703(1-a) of the Code, are presumed to be expenditures in furtherance of the campaign's candidate and contributions from the candidate to the candidate's campaign, and, as such, must be reported to the Board. The candidate may rebut this presumption by providing evidence indicating that the contributions were not in furtherance of the campaign's candidate. Such contributions are subject to all applicable expenditure and contribution limits, except that contributions made to committees registered with the State Board of Elections or the Federal Election Commission as independent expenditure committees are not subject to such limits. Candidates must create and maintain records of such contributions. Contributions made with a candidate's personal funds as provided in this subparagraph shall not be the basis for a deduction from such candidate's public funds payment pursuant to Section 7-07(a).

(G) Expenditure refunds. For expenditures of which all or a portion was refunded to the candidate by the vendor, the candidate must report the refund and provide an amended invoice or other documentation from the vendor specifying the amount, date, and reason for the refund, as well as proof of receipt of the refund in the form of a check, bank statement, or other financial documentation.

(v) Intermediaries. In addition to fully reporting contributions, candidates must fully report any intermediary that solicited or delivered a contribution as provided in Paragraph (ii) of this subdivision.

(A) <u>Exceptions</u>

(1) The candidate need not report an intermediary for aggregate contributions of \$500 or less collected from a contributor in connection with an event held at an individual's residence, unless the expenses related to such event exceed \$500.

(2) The candidate need not report an intermediary for contributions collected at an event organized by a candidate to raise funds for such candidate and paid for in whole or in part by such candidate's authorized committee.

(3) The candidate need not report an intermediary who is a spouse, domestic partner, parent, child, or sibling of the candidate.

(B) <u>Contributions collected at a non-campaign</u> <u>sponsored fundraising event with multiple</u> <u>hosts.</u> In the case of contributions collected at a fundraising event neither organized by the candidate nor paid for in whole or in part by such candidate's authorized committee, where there are multiple hosts, the hosts must designate one host who must be reported by the candidate as the intermediary for all such contributions.

(C) <u>Contributions delivered by an intermediary's</u> <u>agent.</u> The candidate must report as the intermediary an individual who solicits contribution(s) and directs such individual's agent to deliver them to the candidate or fundraising agent. The candidate must not report the agent as an intermediary.

(vi) Transfers

(A) Transfers of funds shall consist entirely of contributions previously raised by the transferor committee and shall not include any public funds.

(B) Candidates must report contemporaneously: (1) the aggregate amount of each transfer and (2) each contribution the transfer consists of (using a last-in/ first out attribution), including the name and residential address of the contributor and the amount and date of the contribution.

(C) In the case of a transfer to an authorized committee from a committee not otherwise involved in the covered election, unless the transferring committee is another authorized committee of the same candidate that has filed contemporaneous disclosure statements with the Board in a timely manner, participants must (1) report to the Board, in the same disclosure statement in which the transfer is reported, any expenditures incurred by the transfer committee in connection with raising or administering the transferred contributions, regardless of when the expenditures were incurred, and (2) upon request by the Board, disclose all expenditures made by the transferor committee during the covered election cycle for purposes other than raising or administering the transferred contributions. Further, candidates must submit contemporaneously the records required to be maintained pursuant to section 4-01(b)(ii)(D).

(vii) Advances and reimbursements

(A) For advance payments, the candidate must report in each disclosure statement:

(1) the name and address of each individual or entity, including the candidate, that has made purchases on behalf of the committee during the reporting period with the expectation of being reimbursed by the committee;

(2) the date and amount of each purchase;

(3) the name and address of the individual or entity from whom the purchase has been made;

(4) the payment method;

(5) the purpose code and explanation of the purchase; and

(6) such other information as the Board may require.

(B) For advance reimbursements, the candidate must report in each disclosure statement:

(1) the name of each individual or entity, including the candidate, whom the candidate reimbursed for purchases made on behalf of the committee during the reporting period;

(2) the date and amount of each reimbursement;

(3) the payment method of each reimbursement, provided that, if the reimbursement is done by check, the candidate must also report the bank account and check number from which the check was issued; and

(4) such other information as the Board may require.

(viii) Loans. Each disclosure statement shall include the following information about loans accepted, forgiven, or repaid by the candidate during the reporting period:

(A) for each loan accepted, the lender's, guarantor's or other obligor's full name, residential address, occupation, employer, and business address;

(B) the date and amount of each loan, guarantee, or other security for a loan accepted;

(C) for each loan repayment made, the date, amount, check number, name of bank account or credit card, and name of any third party payor; and (D) the date and amount of any portion of a loan which has been forgiven.

(ix) Documentation. Together with each disclosure statement, the candidate must submit documentation to verify the accuracy of the data reported, including all bank records and deposit slips required to be maintained pursuant to sections 4-01(b)(i) and 4-01(d)(i) not previously submitted.

(d) Amendments to disclosure statements are prohibited unless expressly authorized or requested by the Board.

§ 4-06 Daily disclosures during the two weeks preceding the election. If a candidate, during the 14 days preceding an election, accepts aggregate contributions and/or loans from a single source in excess of \$1,000 or makes aggregate expenditures to a single vendor in excess of \$20,000, the candidate must report, in a disclosure to the Board, all contributions and loans accepted from such source or expenditures made to such vendor during that 14-day period. The first such disclosure must be received by the Board within 24 hours after the contribution or loan that causes the total to exceed \$1,000 is accepted or the expenditure that causes the total to exceed \$20,000 is made. Each subsequent disclosure must be received by the Board within 24 hours after any additional contribution or loan is accepted or expenditure is made. Information reported in these disclosures must also be included in the candidate's next post-election disclosure statement.

§ **4-07 Terminated candidacy**. A candidate need not submit any disclosure statements for periods subsequent to the termination of such candidate's candidacy, as described in Section 3-04, unless the termination is reversed pursuant to Section 3-04(b).

§ **4-08 Write-in candidates.** Any candidate who is seeking nomination or election as a write-in candidate must file all disclosure statements for the election as required by Section 4-05(b).

§ 4-09 Candidates not in primary or general elections

(a) Not in primary election. A candidate need not submit the two pre-primary and the 10 day post-primary election disclosure statements if the candidate is not a candidate in a primary election unless the candidate is a participant claiming that expenditures are subject to a primary election spending limit, pursuant to Section 6-01(h)(ii)(A) or 6-01(h)(iv). If the candidate is not a candidate in the primary, daily disclosures during the two weeks preceding the primary need not be submitted.

(b) Not in general election. A candidate need not submit the two pre-general election and 27 day post-general election disclosure statements or daily disclosures during the two weeks preceding the general election, if the candidate is not a candidate in the general election.

§ 4-10 Other political committees. The financial records of any committees of a candidate are subject to Board review for purposes of monitoring the candidate's compliance with the requirements of the Act and these rules and must be made available to the Board upon its request.

<u>Chapter 5: Contributions, Loans, and Other Receipts</u>

§ 5-01 Contribution limits

(a) Candidates may not accept contributions from a single source in excess of the limits set forth in § 3-703(1)(f) of the Code.

(b) Adjustment. Pursuant to § 3-703(7) of the Code, not later than the first day of March in the year 2018, and every fourth year thereafter, the Board shall adjust the contribution limits. Such adjustment shall be made in accordance with changes in the consumer price index for the metropolitan New York-New Jersey region published by the United States Bureau of Labor Statistics. The adjustment shall be based on the difference between the average consumer price index over the 12 months preceding the calendar year of such adjustment, and either (a) the calendar year as may be appropriate pursuant to any amendment to the Act.

§ 5-02 Solicitation of contributions

(a) Fundraising solicitations. Each written solicitation of contributions (including, but not limited to, letters, contribution cards, flyers, and other invitations to contribute or attend events) by or on behalf of a candidate, whether in paper or electronic format, must include the following statement, written in a conspicuous and clearly recognizable manner: "State law prohibits making a contribution in someone else's name, reimbursing someone for a contribution made in your name, being reimbursed for a contribution made in your name, or claiming to have made a contribution when a loan is made." If the solicitation is written in a language other than English, the statement must be written in the same language as the solicitation.

(b) Solicitation of contributions for elections not subject to the Act. If a candidate makes a solicitation for a contribution for an election not subject to the requirements of the Act, the solicitation must specify that the contribution is being solicited for an election that is not subject to the requirements of the Act.

§ 5-03 Prohibited contributions

(a) Contributions from corporations, limited liability companies, and partnerships, including professional corporations and limited liability partnerships

(i) A candidate may not accept a contribution from a prohibited organization, *i.e.*, a corporation, limited liability company, or partnership, including professional corporations and limited liability partnerships.

(ii) A candidate may not accept a loan from a prohibited organization unless the loan is made in the regular course of the lender's business.

(iii) A candidate may not accept a guarantee or other security for a loan from a prohibited organization.

(iv) This prohibition does not apply to contributions by political committees.

(b) Nominee contributions. A candidate may not accept a contribution made in the name of an individual or entity for which the source of the funds is a different individual or entity.

(c) Anonymous contributions. A candidate may not accept a contribution from an unidentified source.

(d) Contributions from foreign nationals. A candidate may not accept a contribution from a person who is not a naturalized U.S. citizen or a holder of a green card.

(e) Cash contributions in excess of \$100. A candidate may not accept cash receipts aggregating in excess of \$100 from a single source.

(f) Money order contributions in excess of \$100. A candidate may not accept money order receipts aggregating in excess of \$100 from a single source.

(g) Contributions in violation of state or federal law. A candidate may not accept a contribution that was made, received, solicited, or otherwise obtained in violation of any Local, State, or Federal law.

§ 5-04 Contributions subject to special requirements

(a) Contributions from political committees

(i) A candidate may accept a contribution from a political committee, provided that the political committee:

> (A) was registered with the Board for the period that includes the candidate's next covered election at the time the contribution was received; or

(B) registers with the Board within 10 days of receipt of the contribution.

(ii) A political committee must register in such form and manner as shall be determined by the Board. Such registration shall include but is not limited to:

(A) the name and address of the committee, and the name, address, and employer of the chairperson, treasurer, and liaison of the committee;

(B) an indication whether the committee is a political action committee, a candidate committee (and if so, identification of the candidates supported by the committee), or another kind of political committee;

(C) identification of the governmental agency or agencies with which the committee files disclosure statements;

(D) an indication whether the committee makes monetary contributions, in-kind contributions, or independent expenditures, and the name, address, and employer of each individual or entity with the authority to determine the candidates for whom the committee makes contributions or independent expenditures; and

(E) an indication whether the committee accepts contributions from corporations, limited liability companies, or partnerships, and an agreement not to use funds from such entities for contributions to candidates.

(iii) A list of registered political committees for each election cycle will be maintained on the Board's website. Candidates have the burden to check the list of registered political committees published by the Board to ensure that each political committee contribution accepted is from a registered committee.

(iv) The registration shall remain in effect through the end of the election cycle, unless there has been a material change in the information included in the registration.

(v) If there has been a material change in the information filed in the registration, the original registration is null and void until the committee files an amendment correcting the registration.

(vi) Political committees that do not submit the information

required by the Board, or any required signatures or notarizations, will not be considered to be registered.

(vii) Earmarked contributions from political committees

(A) A candidate may accept a contribution given to a political committee by a contributor who limited the committee's choice or directed the selection of the recipient, but the contribution shall be considered to be from both the original contributor and from the political committee.

(B) A contribution that a political committee has received solely because of its actions as an intermediary as defined in these rules is not an earmarked contribution.

(C) A nominee contribution is not an earmarked contribution.

(b) Contributions from minors. A candidate may accept a contribution from an individual under 18 years of age, provided that:

(i) the decision to contribute was made knowingly and voluntarily by the minor;

(ii) the funds, goods, or services contributed were owned and controlled exclusively by the minor, such as income earned by the minor, or a bank account opened and maintained exclusively in the minor's name; and

(iii) the contribution was not made from the proceeds of a gift, the purpose of which was to provide funds to be contributed.

(c) Contributions from joint contributors

(i) A candidate may accept a single contribution made jointly by two or more individuals or entities, provided that the contribution check or monetary instrument includes the signature and pre-printed information of each individual making the contribution or, in the case of an entity contribution, the signature of each authorized individual.

(ii) If a check or other monetary instrument representing a joint contribution, or a contribution card or other contemporaneous document related to the contribution, does not indicate the amount to be attributed to each contributor, the contribution shall be attributed equally to each contributor.

(d) Post-election contributions. Contributions accepted after an election may be used to pay liabilities incurred in that election, subject to the applicable contribution limit and prohibitions, only if deposited in and disbursed from an account established and maintained for that election, as provided in Sections 5-11(a)(iii) and (iv).

(e) Text message contributions. A text message contribution is received on the date it is delivered to the candidate, after payment of the contributor's wireless bill, by a wireless carrier or other mobile fundraising vendor.

§ 5-05 Non-matchable contributions. The following are not matchable:

- (a) <u>In-kind contributions and loans.</u> In-kind contributions and loans, regardless of whether such loans are considered contributions pursuant to § 3-702(8) of the Code.
- (b) Purchase price. Contributions in the form of the purchase price paid for an item with significant intrinsic and enduring value, or paid for or otherwise induced by a chance to participate in a raffle, lottery, or a similar drawing for valuable prizes.
- (c) Certain contributions from vendors. Contributions from individuals, other than employees of the candidate's principal committee, who are vendors to the candidate or individuals who have an interest in a vendor to the candidate, unless the expenditure to the vendor is reimbursement for an advance. For the purposes of this subdivision, "individuals who have an interest in a vendor" means individuals having an ownership interest of 10% or more in a vendor or control over the vendor. An individual shall be deemed to have control over the vendor firm if the individual holds a management position, such as the position of officer, director or trustee.
- (d) Doing business contributions. Contributions from individuals having business dealings with the city as defined in § 3-702(18) of the Code, and contributions from lobbyists as defined in § 3-211 of the Code, or persons required to be included in a statement of registration filed pursuant to §§ 3-213(c)(1) or 3-213(d) of the Code.
- (e) <u>Contributions intermediated by a person doing business</u> with the City. Contributions for which any individual or entity listed in the doing business database at the time of the contribution acted as an intermediary.
- (f) <u>Contributions from business accounts</u>. Contribution checks drawn on business accounts, or accounts that bear indicia of being business accounts, such as the contributor's professional title.

- (g) Prohibited or excess contributions. Contributions from contributors subject to the prohibitions of §§ 3-703(1)(k) or (l) of the Code or that are otherwise prohibited by these rules, or that, in the aggregate for a given contributor, exceed the contribution limits applicable under the Act;
- (h) <u>**Returned contributions.**</u> Contributions that are returned or refunded to, or not paid by, the contributor.
- (i) **Unitemized contributions**. Contributions for which required information is missing from or illegible in a disclosure statement.
- (j) Contributions with unreported occupation, employer, and business address. Contributions for which the candidate has not reported the contributor's occupation, employer, and business address, where such contributions (i) total more than \$99; or (ii) total \$99 or less, and the contributor is an employee of the candidate, of the spouse or domestic partner of such candidate, or of a business entity in which such candidate, spouse, or domestic partner has an ownership interest of 10% or more or in which such candidate, spouse, or domestic partner holds a management position, such as the position of officer, director, or trustee.
- (k) <u>Contributions from minors.</u> Contributions from individuals under 18 years of age.
- (l) <u>Contributions in violation of law.</u> Contributions that were made, received, solicited, or otherwise obtained in violation of any federal, state, or local law, including the Act and these rules.
- (m) <u>Contributions from entities.</u> Contributions received from entities other than individuals, including political committees.
- (n) <u>Contributions not from New York City residents.</u> Contributions from contributors whose residential addresses are not within New York City, or for whom other indicia exist that the contributor is not an individual New York City resident.
- (o) <u>Contributions made after the election year</u>. Contributions made later than December 31 of the year of the covered election in which the participant is a candidate.
- (p) <u>Contributions for other elections.</u> Contributions originally received for elections other than the election in which the participant is currently a candidate, as described in Section 5-08.
- (q) <u>Claims exceeding the gross amount of the contribution</u>. Matchable contribution claims that exceed the gross amount of the contribution.
- (r) Excess matching claims. Matchable contribution claims that would yield more than the maximum public funds payment per contributor as provided in §3-705(2)(a) of the Code.
- (s) Contributions made with checks drawn by someone other than the contributor. Checks drawn by an individual or entity other than the contributor, except for checks signed by a contributor's authorized agent, where the documentation required under Sections 4-01(b)(ii)(A)(3) and 4-01(b)(iii) has been maintained and provided.
- (t) <u>Contributions to other committees.</u> Contributions made payable to, or originally received by, entities other than the principal committee, including committees not otherwise involved in the covered election;
- (u) <u>Contributions not timely and properly reported.</u> Contributions that were not timely reported and itemized in disclosure statements, or that were reported for the first time in an amendment to a disclosure statement.
- (v) <u>Contributions claimed after the year of the election.</u> <u>Contributions not claimed as matchable on or before December 31</u> in the year of the election.
- (w) Contributions not properly documented. Contributions for which a record required under Chapter 4 was not kept or provided upon request, for which complete supporting documentation required by section 4-05(c)(ii)(B)(2) has not been submitted, or for which the information on such documentation is not consistent with the information reported in the disclosure statement.
- (x) Contributions made with consecutively numbered money orders. Contributions purportedly from different contributors that were made by money orders bearing consecutive serial numbers, or other indicia that they were purchased simultaneously.
- (y) <u>Contributions for a runoff election.</u> Contributions solicited for, or required to be deposited into an account established for, a runoff election, as provided in section 16-06.
- (z) Withdrawn matching claims. Contributions for which a matching claim was previously withdrawn by the candidate.
- (aa) <u>Non-matchable contributions</u>. Contributions that are otherwise not matchable contributions within the meaning of the <u>Act.</u>

- (bb) Additional factors. In addition, the Board shall consider the following factors in determining whether matchable contribution claims are invalid and in projecting a rate of invalid matchable contribution claims:
 - (i) <u>any information that suggests that a contribution has not</u> <u>been processed or reported in accordance with Program</u> <u>requirements;</u>
 - (ii) <u>any other information that suggests that matchable</u> <u>contribution claims may be invalid; and</u>
 - (iii) <u>arithmetical errors in totals reported.</u>

§ 5-06 In-kind contributions

(a) An in-kind contribution is an expenditure. An in-kind contribution to a candidate's authorized committee is also considered an expenditure made by such committee, and the amount of the inkind contribution thus counts toward the candidate's contribution and expenditure limits. If a debt, other than a loan, incurred by a candidate's authorized committee is forgiven, the act of forgiving the debt is an in-kind contribution to, and an expenditure by, the candidate. An in-kind contribution is received on the date the goods or services are received or rendered, which is presumed to be the date of the associated expenditure, unless the candidate provides evidence to demonstrate that the contribution should be deemed received on a different date.

(b) The value of an in-kind contribution

(i) Candidates must maintain invoices or other written records supporting the valuation of all in-kind contributions.

(ii) If no invoice is available and obtaining an invoice is not practicable, a candidate must use a reasonable estimate of value in documenting the fair market value of an in-kind contribution.

> (A) The fair market value of goods is the price of those goods in the market from which they ordinarily would have been purchased when the goods were received.

(B) The fair market value of services is the hourly or piecework charge for the services at a commercially reasonable rate prevailing when the services were rendered.

(c) Good or service purchased below fair market value. Where goods or services are purchased by a candidate for less than fair market value, the difference between the fair market value when they were received and the amount charged to the candidate, including any applicable tax, is an in-kind contribution.

(d) Extension of credit that is not commercially reasonable is an in-kind contribution

(i) Generally. A creditor that extends credit to a candidate for a period beyond 90 days has made a contribution equal in value to the credit extended, unless the creditor is making a commercially reasonable attempt to collect the debt.

(ii) Prohibited in-kind contribution. Extension of credit by a corporation, limited liability company, or partnership that is not commercially reasonable and is not made in the regular course of business is presumed to be a prohibited in-kind contribution.

(e) Debt forgiven is an in-kind contribution

(i) Generally. A debt owed by a candidate, which is forgiven or settled for less than the amount owed, is an in-kind contribution, unless the debt was forgiven or settled by the creditor in a commercially reasonable manner.

(ii) Commercially reasonable treatment of debts. The Board will consider as evidence of commercially reasonable treatment that: (A) all commercially reasonable efforts have been taken to satisfy the outstanding debt; and (B) the creditor has pursued its remedies in the same manner as that customarily employed by creditors of other debtors.

(iii) Advances. An advance is considered to be an in-kind contribution from the individual or entity making the advance until it has been reimbursed by the candidate, and a candidate may not accept an advance from a prohibited source.

(f) Failing to report a liability leads to a presumption of an inkind contribution. An outstanding liability not contemporaneously reported as outstanding by a candidate may be presumed to be an in-kind contribution to the candidate, if other indicia of a contribution exist.

§ 5-07 Refunding prohibited and over-the-limit contributions

(a) Generally. When a candidate knows or has reason to know that such candidate has accepted a contribution or aggregate contributions from a single source in excess of the applicable contribution limit, including a contribution or contributions from a contributor having

business dealings with the city, or from a source prohibited by the Act or the Charter or by state or federal law, the candidate must promptly refund the excess portion or prohibited contribution by bank or certified check made out to the contributor or to the Fund; provided, however, that when a candidate knows or has reason to know that such candidate has accepted a nominee or anonymous contribution, the candidate must promptly disgorge the amount of such contribution by bank or certified check made out to the Fund.

(b) Contribution refunds must be timely.

- (i) When a candidate knows or has reason to know that such candidate has accepted a prohibited or over-the-limit contribution, the candidate must return or refund the contribution, or the over-the-limit portion thereof, on or before the next disclosure statement filing deadline or the deadline set by the Board.
- (ii) When a candidate is notified by the Board that such candidate has accepted a prohibited or over-the-limit contribution, the candidate must return the contribution by the date specified in the notice sent by the Board.

(c) Contribution refunds must be documented and reported. If a candidate issues a refund for a contribution after it has been deposited in the committee's account, the contribution and corresponding refund must be documented and reported to the Board.

(d) Restrictions on return. After receiving public funds for an election, a candidate may not return a contribution, unless directed by the Board to do so, until any required repayments to the Fund have been made, except if the contribution: (i) exceeds the contribution limit, including the limit applicable to contributors having business dealings with the city, (ii) is otherwise illegal, (iii) is returned because of the candidate's reputational interest in light of the particular source or intermediary involved, or (iv) was deposited in a separate account pursuant to Section 16-06 for a runoff election that is not held.

(e) Where refund is impracticable. If a timely refund of a contribution to the contributor is impracticable, the candidate may pay the Fund an amount equal to the amount of the refund. Such payment must be disclosed as a refund to the contributor.

(f) Over-the-limit contributions from contributors doing business with the city

(i) When a candidate reports an over-the-limit contribution from a contributor having business dealings with the City:

> (A) The Board will provide such notification to the candidate within 20 days of the reporting of the contribution; provided, however, that if such twentieth day is a Saturday, Sunday, or legal holiday, notification by the Board on the next business day shall be considered timely. In the case of a contribution reported during the six weeks preceding the candidate's next covered election, the Board shall provide such notification within three business days.

(B) The candidate must:

(1) refund the excess portion of the contribution within 20 days of the date of the notice; or

(2) demonstrate to the Board, within 20 days of notification, that the contributor identified by the Board as having business dealings with the city has applied to the Mayor's Office of Contract Services for removal from the doing business database and that such application is pending.

(C) For purposes of calculating compliance with the limits applicable to contributors having business dealings with the city, contributions accepted from a contributor who appears on the doing business database shall be aggregated with all other contributions accepted from such contributor during the same election cycle; provided, however, that contributions accepted while the contributor did not appear on such database shall not be required to be refunded pursuant to this section.

(ii) Where a contributor has made an application for removal from the doing business database, the candidate may retain any contribution received from the contributor until the Board notifies the candidate that the Mayor's Office of Contract Services has denied the application for removal, in which case the candidate shall have 20 days from the date of such second notice to refund the excess portion of the contribution.

(iii) Contributions from or intermediated by individuals who have applied for removal from the doing business database shall not be considered matchable unless and until the contributor or intermediary is removed from the doing business database.

<u>§ 5-08 Contributions originally received for other elections</u> (a) Transfers (i) Generally. Notwithstanding Section 5-11(a), participants may transfer funds previously received for another election to the principal committee. Additionally, participants may transfer funds from the principal committee to another authorized committee, after all excess funds have been returned to the Board as provided by Sections 5-11(c)(ii) and 9-02(c).

(ii) Segregated accounts. Notwithstanding paragraph (i) of this subdivision, if a candidate has funds remaining in a segregated account after the election, such funds must be refunded to the contributors or paid to the Fund, or used to pay outstanding liabilities related to the preceding election. Such funds may not be transferred into any other account or used for any future election.

(b) Surplus funds. The cash balance reported in a non-participant's first semi-annual Board disclosure statement at the beginning of the first reporting period for an election is the total amount of surplus funds the committee had from a previous election; except that the amount deemed to be surplus funds may be reduced by the following:

(i) the total amount of debts and obligations outstanding at the beginning of the reporting period;

(ii) the total amount subsequently transferred to a political committee that is not involved in a covered election; and

(iii) if the candidate was a participant in the previous election, the total amount of public funds subsequently repaid.

(c) Additional requirements for transfers and surplus funds

(i) Candidates have the burden of demonstrating that surplus funds and transfers of funds from committees not otherwise involved in the covered election do not derive from:

> (A) contributions in excess of the Act's contribution limits, including contributions that would exceed the Act's contribution limits when aggregated with other contributions accepted from the same source; or

(B) contributions from sources prohibited by the Act, the Charter, or state or federal law.

(ii) Participants have the burden of demonstrating that funds transferred from a committee, other than another authorized committee of the same candidate that has filed contemporaneous disclosure statements with the Board in a timely manner, derive solely from contributions for which records demonstrating the contributors' intent to designate the contributions for the covered election have been submitted and maintained as required pursuant to sections 4-05(c)(vi) and 4-01(b)(ii)(D), respectively.

(iii) For purposes of enforcing the contribution limit and contribution prohibitions, the Board shall attribute surplus funds and such transfers to the last monetary contributions, loans, and other receipts received by:

> (A) the candidate on or before the date of the cash balance described in Subdivision (b), in the case of surplus funds; or

(B) the transferor committee before making the transfer.

(iv) For any prohibited or over-the-limit contribution, the candidate must either:

(A) promptly refund the excess portion or amount of the prohibited contribution, or

(B) keep the excess portion or amount of the prohibited contribution in the transferor committee bank account, not to be used in a covered election.

(d) Related expenditures. Expenditures incurred in connection with raising or administering funds transferred from a committee not otherwise involved in a covered election, other than from another principal committee of the same candidate, are presumed to be subject to the expenditure limits of the Act. Participants have the burden of demonstrating that any such expenditures are not subject to the expenditure limits of the Act, as provided in section 6-01(j).

<u>§ 5-09 Loans</u>

(a) Deposit. All loans must be accepted and deposited, or rejected and returned, within 10 business days after receipt.

(b) Loans over \$100 may not be made by cash or credit card.

(c) Loans forgiven. Any portion of a loan that is forgiven is a monetary contribution from the source of the loan, subject to the contribution limits and prohibitions.

(d) Third party repayment of loan. If any portion of a loan is repaid by an individual or entity other than the committee that received the loan, the portion thus repaid is a contribution by that individual or entity. (e) **Repayment by next election.** If a loan is not repaid by the date of the next election in which the candidate appears on the ballot, the loan, guarantee, or other security for the loan will be considered a contribution subject to the Act's contribution limits and prohibitions.

(f) Loans made in regular course of business. A loan made in the regular course of the lender's business shall be deemed, to the extent not repaid by the date of the next election in which the candidate appears on the ballot, a contribution by the lender and by any other individual or entity endorsing, cosigning, guaranteeing, collateralizing, or otherwise providing security for the loan.

(g) Loans not made in regular course of business. A loan not made in the regular course of the lender's business shall be deemed, to the extent not repaid by the date of the next election in which the candidate appears on the ballot, a contribution by the lender.

(h) Attributing a loan to an election. A loan is presumed to be accepted for the next election in which the candidate seeks nomination or election following the day that the loan is received, except as otherwise provided in section 5-09(i), and except that:

(i) in the case of a state or local election, loans received before the first January 12 after an election will be presumed to be accepted for that election; and

(ii) in the case of a federal election, loans received before the first January 1 after the election will be presumed to be accepted for that election, except as may otherwise be provided under federal law and regulations.

(i) Post-election loans

(i) Loans received after an election are considered contributions attributed to the previous election, and may be used for such election as provided in this section.

(ii) Loans received after an election must be deposited in the committee bank account for that election.

(iii) A loan made by a candidate, after a final determination of the Board has been issued with respect to such candidate, for the purpose of paying penalties or making required repayments to the Fund pursuant to such determination, is not subject to the contribution limit.

§ 5-10 Attributing contributions

(a) Attributing a contribution to an election. A contribution is presumed to be accepted for the first covered election in which the candidate seeks nomination or election following the day that the contribution is received, unless the contribution was properly transferred to the committee from another committee of the candidate for a different election, and except that:

(i) in the case of a State or Local election, contributions received before the first January 12 after the election will be presumed to be accepted for that election; and

(ii) in the case of a Federal election, contributions received before the first January 1 after the election will be presumed to be accepted for that election, except as may otherwise be provided under federal law and regulations.

(b) Attributing multiple contributions to a single source and contribution limit

(i) Multiple contributions from a single source will be totaled to determine the candidate's compliance with the applicable contribution limit.

(ii) General factors for determining a "single source." Factors for determining whether an individual, individuals in combination, entity, or entities in combination establish, maintain, or control another entity include, but are not limited to:

> (A) whether the persons or entities make decisions or establish policy for the other entity, including determinations of the recipients of its contributions and the purposes of its expenditures;

(B) whether the persons or entities have the authority to hire, appoint, discipline, discharge, demote, remove, or otherwise influence other persons who make decisions or establish policies for the other entity;

(C) whether contributions made by the persons or entities and the other entity reflect a similar pattern; and

(D) whether the persons or entities know of and have acquiesced in public representations by the other entity that it is acting on their behalf or under their direction.

(iii) Attributing single source contributions from labor organizations. Notwithstanding Paragraph (i), different labor organizations shall not be considered to be a single source for the purpose of compliance with the applicable contribution limit if the candidate demonstrates that the contributors satisfy the four criteria below:

(A) the labor organizations do not share a majority of members of their governing boards;

(B) the labor organizations do not share a majority of the officers of their governing boards;

(C) the labor organizations maintain separate accounts with different signatories; and

(D) the labor organizations make contributions from separate accounts.

(iv) Attributing single source "doing business" contributions. If a candidate accepts multiple contributions from a single source consisting of at least one contribution from an individual having business dealings with the city and one or more contributions from an entity established, maintained, or controlled by that individual, the limit applicable to persons having business dealings with the city shall apply.

(v) Burden is on the candidate. If multiple contributions appear to be from a single source in excess of the contribution limit, the candidate has the burden of demonstrating that they are from different sources. Candidates must review the relationship between contributors who appear to constitute a single source before accepting and depositing contributions.

(c) Attributing jointly raised contributions; endorsements

(i) If a candidate cooperates in joint fundraising with another candidate or political committee, including in the form of endorsements, and any of the jointly raised contributions exceed the amount of the contribution limit applicable to the candidate, or comes from a source that would be prohibited to the candidate by the Act or the Charter, the candidate shall have the burden of demonstrating that the contribution was not used in a manner that directly or indirectly assisted or benefited the candidate in violation of the applicable limit or prohibition.

(ii) To ensure compliance with the contribution limits, candidates who run together as a "ticket" and make joint expenditures to raise contributions, must additionally abide by the requirements of this subdivision.

> (A) When paying a share of joint fundraising expenditures (by direct payment or reimbursement), the payor shall have the burden of demonstrating that the contributions raised pursuant to said fundraising expenditures would not exceed the other candidate's contribution limit, if those contributions were aggregated with contributions previously received by the other candidate.

> (B) No disbursement for joint fundraising expenditures shall be made before the candidate is able to account fully for the disbursement with contributions that would not exceed the other candidate's contribution limit, if so aggregated. Failure to make reimbursement for the expenditure, however, will result in a deduction in public funds payments otherwise due to the candidate to be reimbursed, pursuant to section 7-07(a), and failure to make reimbursement within 90 days will result in treatment of the expenditure as an in-kind contribution to the candidate failing to make reimbursement, pursuant to section 5-06(d).

(d) Attributing contribution amount for fundraiser ticket or fundraising item

(i) **Fundraiser ticket.** When attendees purchase entrance to a fundraiser, the entire amount paid to attend a fundraising event is a contribution.

(ii) Fundraising item. When an individual buys an item sold by a candidate to raise funds, the entire amount paid as the purchase price for such item is a contribution.

<u>§ 5-11 Use of receipts</u>

(a) Bank accounts

(i) Candidates must maintain a committee account with check writing privileges. Checks issued from such account must be signed by the candidate or an individual so authorized by the candidate.

(ii) Candidates must maintain a bank account in the name of the principal committee. Such bank account:

(A) must be used for the deposit of receipts, including public funds, and the making of expenditures, for the covered election;

(B) must be used exclusively for the purpose of furthering the candidate's nomination or election in the covered election, and for post-election expenditures as provided in subdivision (c) of this section; and

(C) must not contain personal, business, or other non-campaign funds.

(iii) Receipts accepted for one election may not be commingled in an account with receipts accepted for another election, with receipts accepted for a TIE, or with personal or business funds. Notwithstanding the foregoing, candidates:

- (A) may use one account for both the primary and general election in the same calendar year;
- (B) <u>may open a segregated account pursuant to section</u> <u>7-07(b);</u>
- (C) <u>must use a court-ordered rerun account for public funds</u> received for a court-ordered rerun election; and

(iv) Separate accounts for different elections. A candidate seeking election both to an office subject to the Act and to a state or federal office may maintain a separate allocation account for shared expenses in accordance with Advisory Opinion No. 1996-2 (July 18, 1996).

(v) Deposit of receipts. All monetary contributions must be accepted and deposited into the account(s) listed on the candidate's Filer Registration or Certification and in disclosure statements filed with the Board, or rejected and returned to a contributor, within 20 business days after receipt, except contributions made in the form of cash must be accepted and deposited, or rejected and returned to a contributor, within 10 business days after receipt. All monetary receipts accepted for an election must be deposited into the candidate's authorized committee bank account. A monetary receipt is received on the date it is delivered to the candidate.

(b) Receipts may be used only to further a candidate's nomination or election. A candidate may use receipts only for the covered election for which that account was established or an associated runoff election.

(i) Expenditures not in furtherance of the campaign. In determining whether or not an expenditure is in furtherance of a candidate's nomination or election, the Board may consider the factors from the following non-exhaustive list:

- (A) the timing of the expenditure;
- (B) whether the candidate has already purchased duplicative services or equipment;
- (C) the nature of the goods or services purchased;
- (D) whether an unusually high proportion of funds was spent on a specific category of expenditure;
- (E) whether a high total dollar amount or proportion of payments was made to individuals rather than to entities;
- (F) whether the candidate has demonstrated a pattern of making other expenditures not in furtherance of the campaign or impermissible post-election expenditures; and

(G) whether an expenditure made less than one month prior to the election, or after the election, is accompanied by the reporting of a corresponding outstanding liability.

(c) Post-election use of receipts

(i) After the last day of the final reporting period of a covered election, a candidate in that election may not expend, transfer, or use receipts accepted for another election, unless the receipts have been deposited in and are disbursed from a separate account, as provided in Sections 5-11(a)(iii) and (iv). Funds accepted and separately deposited for the previous election may be transferred to this account only after any required repayments to the Fund have been made and any fines or civil penalties assessed pursuant to the Act have been paid. Contributions and loans accepted for the previous election after such election are subject to Sections 5-04(d) and 5-09(i).

(ii) A participant may not use receipts for any purpose other than disbursements in the preceding election until all unspent campaign funds have been repaid, except as otherwise provided in Subparagraph (i) of this paragraph and Section 5-07(d). Notwithstanding the presumption of Section 6-01(h)(i), a participant has the burden of demonstrating that a post-election expenditure is for the preceding election.

§ 5-12 Court ordered rerun elections

(a) No contributions may be accepted until court contest has begun. A candidate may not accept contributions for a court-ordered rerun election until a complaint has been filed in a court of competent jurisdiction concerning the canvass of returns or the conduct of the election.

(b) Where rerun election is canceled, contributions must be reasonably related to expenditures. If a rerun election is ordered by a court but subsequently canceled, a candidate who would have been on the ballot has the burden of demonstrating that any portion of contributions raised may be reasonably attributed to expenses incurred for the rerun election before its cancellation.

Chapter 6: Expenditures

§ 6-01 Expenditure limits

(a) Candidates participating in the Program may not exceed the applicable expenditure limits provided in § 3-706 of the Code.

(b) Adjustment. Pursuant to § 3-706(1) of the Code, not later than the first day of March in the year 2010, and every fourth year thereafter, the Board shall adjust the expenditure limits. Such adjustment shall be made in accordance with changes in the consumer price index for the metropolitan New York-New Jersey region published by the United States Bureau of Labor Statistics. The adjustment shall be based on the difference between the average consumer price index over the 12 months preceding the calendar year of such adjustment, and either (a) the calendar year preceding the year of the last such adjustment to any amendment to the Act.

(c) Candidates have the burden of monitoring their expenditures to be sure that they do not exceed the limit.

(d) Applicability. All expenditures made by a participant to further the participant's nomination or election, including expenditures made for the purpose of furthering or facilitating the defeat of the nomination or election of an opponent or prospective opponent, are subject to the expenditure limit applicable under the Act. All expenditures made by the participant shall be totaled to determine the participant's compliance with the applicable expenditure limit. Expenditures incurred after the last election in an election year in which the participant is a candidate, or a special election, are not subject to the expenditure limits for that election.

(e) Expenditures made during the three calendar years preceding the election. Expenditures made by a participant during the three calendar years preceding the year of a covered election are subject to the applicable expenditure limit set forth in § 3-706(2) of the Code. A participant is permitted to make expenditure in excess of this limit. However, pursuant to § 3-706(2a)(a), the amount by which the limits of § 3-706(2) are exceeded will be counted against the participant's first election year expenditure limit under § 3-706(1).

(f) Expenditures made during the year of the election. Expenditures made on or after the first day of January in the year of a covered election, and expenditures attributed to the year of the election pursuant to paragraph (v) of subdivision (h) of this section, are subject to the applicable expenditure limit set forth in § 3-706(1) of the Code.

(g) Expenditures for or against a ballot proposal. Expenditures made by a participant for the purpose of advocating a vote for or against a proposal on the ballot in an election that is also a covered election, regardless of whether the expenditures were also made to further or facilitate the participant's nomination or election, shall be subject to the applicable expenditure limit for the covered election.

(h) Attributing an expenditure

(i) An expenditure is presumed to be made for the first covered election in which the participant or non-participant is a candidate following the day it is made, except:

(A) expenditures made before the first January 12 after a state or local election will be presumed to be made for that election; and

(B) expenditures made before the first January 1 after a federal election will be presumed to be made for that election, except as may otherwise be provided under federal law and regulations.

(ii) No contested primary. If there is no contested primary election in any party for an office, expenditures made by a candidate seeking that office will be entirely attributed to the general election.

(iii) Contested primary or write-in primary

(A) If there is a contested or write-in primary election in any party for an office, every participant or limited participant seeking that office, regardless of whether the participant or limited participant is in the primary election, may make expenditures subject to the primary election expenditure limit of §3-706(1) of the Code, provided the participant or limited participant files the two pre-primary and 10 day post-primary election disclosure statements and daily disclosures pursuant to Sections 4-05(b)(ii)(B), (C), and 4-06 in a timely manner. In this case, the general election expenditure limit will first apply after the date of the primary election.

(B) Expenditures incurred after the date of the contested or write-in primary election will be attributed to the general election.

(C) Expenditures incurred before the primary election by a candidate in a contested primary election are attributed to the primary election, regardless of whether the candidate also received the nomination of another party without a primary election.

(iv) Reasonably anticipated primary. Expenditures may be attributed to a primary election that the Board has determined is reasonably anticipated.

(A) If a participant demonstrates to the Board that for a period preceding the primary election the participant had reasonably anticipated a primary election in any party for the office the participant seeks, the participant may attribute expenditures made before and during that period to the primary election expenditure limit of §3-706(1) of the Code, provided the participant files the two pre-primary and 10 day post-primary election disclosure statements and daily disclosures pursuant to sections 4-05(b)(ii)(B), (C), and 4-06 in a timely manner. In this case, the general election expenditure limit will first apply after that period.

(B) In order to demonstrate to the Board that for a period preceding the primary election the participant had reasonably anticipated a primary election, the participant must file a petition, consisting of an affidavit with supporting documentation, with the Board no later than 10 business days following the date the last remaining candidate other than the participant was finally disqualified from the ballot as set forth in Section 7-01(e)(i).

(1) <u>The affidavit must:</u>

(I) be sworn to or affirmed by the candidate;

(II) specify the period of time during which it was reasonable to anticipate that a primary election would be held;

(III) identify the prospective candidate(s); and

(IV) provide a factual basis for the participant's belief that a primary election was reasonably anticipated during the specified period of time.

(2) The supporting documentation must demonstrate that the prospective candidate(s) engaged in activities that would lead a reasonable person to believe that such candidate(s) would participate in the primary election. Such activities may include:

> (I) authorizing a political committee with the Board of Elections for the primary election;

(II) filing a Filer Registration or Certification with the Board;

(III) engaging in petitioning activity, including the filing of petitions with the Board of Elections;

(IV) producing and/or distributing campaign communications related to the primary election; and

(V) campaigning for office or otherwise publicly declaring an intent to participate in the primary election.

(C) Once it is determined that no primary election will be held for nomination to an office, or that such primary election is no longer reasonably anticipated, subsequent expenditures will be subject to the general election expenditure limit.

(v) Timing of expenditures. As provided and described in §3-706 (1) and (2) of the Code, an expenditure for goods or services is made when the goods or services are received, used, or rendered, regardless when payment is made. Expenditures for goods or services received, used, or rendered in more than one year, including campaign websites, shall be attributed in a reasonable manner to the expenditure limits of \$3-706(1) or (2) of the Code, as appropriate.

> (A) Expenditures for campaign advertising or other campaign communications shall be attributed to the expenditure limit in effect when the advertisement or communication is distributed, broadcast, or published. For the purposes of this paragraph, "campaign advertising or other campaign communications" shall not include a campaign website. A communication that is mailed shall be considered to have been "distributed" on the date on which it was postmarked.

(B) Expenditures for services performed or deliverables provided over a period that includes both the primary and the general elections shall be attributed in a reasonable manner to the expenditure limits of § 3-706(1) and (2) of the Code, as appropriate.

(C) Notwithstanding the requirements of this subdivision, the Board may require a candidate to demonstrate that an expenditure should be attributed to the expenditure limit provided in §3-706(1) or (2) of the Code, as appropriate, based on the timing, nature, and purpose of the expenditure.

(i) Exempt expenditures

(i) The following shall not be subject to the expenditure limits:

(A) expenses made for the purpose of bringing or responding to any action, proceeding, claim, or suit before any court or arbitrator or administrative agency to determine a candidate's or political committee's compliance with the requirements of this chapter, including eligibility for public funds payments, or pursuant to or with respect to election law or other law or regulation governing candidate or political committee activity or ballot status;

(B) expenses to challenge or defend the validity of petitions of designation or nomination or certificates of nomination, acceptance, authorization, declination or substitution, and expenses related to the canvassing or re-canvassing of election results;

(C) expenses related to the post-election audit, except as provided in paragraph (ii) of this subdivision; and

(D) expenditures for childcare services made pursuant to § 3-702(21)(a)(13) of the Code for an aggregate amount of \$20,000 or less.

(ii) Exempt expenses related to the post-election audit shall include pre-election expenses for organizing and copying existing records in preparation for submission during the post-election audit, but shall not include pre-election expenses for:

> (A) Ordinary compliance activities, such as the review of records to identify missing documents, evaluating whether documents meet Board standards, and identifying, preventing, and correcting any potential violation;

(B) Post-election work for which an invoice is issued or paid prior to the election;

(C) Salaries or other payments to campaign managers, finance chairpersons, treasurers, accountants, advisors, or other consultants;

(D) Legal or accounting fees;

(E) Costs associated with record creation and retention;

(F) Costs associated with running an office or business, such as standard bookkeeping, maintaining checkbook registers, petty cash journals, bank records, and loan records;

(G) Bookkeeping for payroll or vendor payments; and

(H) Other standard practices that political committees routinely perform as entities that raise and spend funds.

(iii) Candidates have the burden of demonstrating that expenditures are exempt pursuant to § 3-706(4) of the Code. To meet this burden, a candidate must maintain contemporaneous, detailed records that demonstrate that each individual expenditure is exempt in accordance with the Act and these rules, and submit such documentation as required. Expenditures not demonstrated to be exempt will be included in the expenditure limit calculation.

(iv) Notwithstanding anything otherwise provided for in this subdivision, the reimbursement of an advance shall not be considered an exempt expenditure.

(v) For purposes of this subdivision, in determining whether a participant's total expenditures exceed the expenditure limits of § 3-706(1) or § 3-706(3)(a)(i) of the Code, as appropriate, expenditures made in the first three years of the election cycle, to the extent such expenditures do not exceed the limit applicable under § 3-706(2) of the Code, shall be excluded.

(j) Expenditure limit compliance for a transfer between political committees

(i) A committee of a participant that receives a transfer of funds from another political committee, other than another principal committee of the same candidate, must:

(A) allocate to the transferred contributions any expenditures incurred by the transferor committee in connection with raising transferred contributions, and any expenditures incurred by the transferor committee during the covered election cycle in connection with administering transferred contributions; and

(B) upon request, provide documentation related to any such expenditures, including copies of federal forms or disclosure statements filed with the New York State Board of Elections on behalf of the transferor committee.

(ii) Expenditures will be applied towards the expenditure limit in effect at the time of the transfer; provided, however, that in the case of transfers made prior to the covered election cycle, expenditures will be applied towards the expenditure limits of § 3-706(2) of the Code.

(iii) The participant has the burden of demonstrating, for the purpose of compliance with the expenditure limits, that an expenditure made by the transferor committee was not made in connection with raising or administering the transferred contributions.

(k) Expenditure limit relief

(i) Pursuant to §3-706(3)(a) of the Code, where the Board has determined that a non-participating candidate has spent or contracted or become obligated to spend, or received in loans or contributions, an amount which, in the aggregate, exceeds half the applicable expenditure limit pursuant to §3-706(1)(a) of the Code, the expenditure limit applicable to participating candidates in the election for that office will be increased to 150% of the expenditure limit.

(ii) Pursuant to §3-706(3)(b) of the Code, where the Board has determined that a non-participating candidate has spent or contracted or become obligated to spend, or received in loans or contributions, an amount which, in the aggregate, exceeds three times the applicable expenditure limit pursuant to §3-706(1)(a) of the Code, the expenditure limit will no longer apply to participating candidates in the election for that office.

§ 6-02 Restrictions on expenditures

(a) Spending public funds; qualified expenditures

(i) Public funds may be used only for qualified expenditures.

(ii) The following are not considered qualified expenditures:

(A) expenditures for any purpose other than to further the candidate's nomination or election;

(B) expenditures not incurred during the calendar year of the election;

(C) expenditures in violation of any law or these rules;

(D) payments to the candidate or a spouse, domestic partner, child, grandchild, parent, grandparent, brother, or sister of the candidate or spouse or domestic partner of such child, grandchild, parent, grandparent, brother, or sister, or to a business entity in which the candidate or any such person has a 10% or greater ownership interest;

(E) payments in excess of the fair market value of services, materials, facilities, or other things of value received;

(F) expenditures made after the candidate has been finally disqualified or such candidate's petitions have been finally declared invalid by the Board of Elections or a court of competent jurisdiction, except that such expenditures may be made (1) as otherwise permitted pursuant to § 3-709(7) of the Code, or (2) for a different election (other than a special election to fill a vacancy) held later in the same calendar year in which the candidate seeks election for the same office;

(G) expenditures made after the only remaining opponent of the candidate has been finally disqualified or such opponent's petitions have been finally declared invalid by the Board of Elections or a court of competent jurisdiction, except that such expenditures may be made for a different election (other than a special election to fill a vacancy) held later in the same calendar year in which the candidate seeks election for the same office;

(H) expenditures made for any other election, if the public funds were originally received for a special election to fill a vacancy;

(I) payments in cash;

(J) contributions, transfers, or loans made to another candidate or political committee;

(K) gifts, except brochures, buttons, signs, and other printed campaign material;

(L) expenditures to challenge or defend the validity of petitions of designation, or nomination, or certificates of nomination, acceptance, authorization, declination, or substitution, and expenses related to the canvassing of election results;

(M) expenditures for which records required by section 4-01 are not maintained or obtained by the candidate and submitted to the CFB;

(N) expenditures that are not itemized in a disclosure statement;

(O) payments that are not made or reimbursed from an account disclosed by the candidate pursuant to Section 2-01 or 2-02(e);

(P) reimbursement for advances, except in the case of individual purchases in excess of \$250, provided that the individual purchase is not otherwise not a qualified expenditure on the basis of any of the other subparagraphs of paragraph (ii);

(Q) expenditures made in connection with any action, claim, or suit before any court or arbitrator;

(R) expenditures made primarily for the purpose of expressly advocating a vote for or against a ballot proposal, unless such expenditures were also made to further the participating candidate's nomination or election;

(S) payments of any penalty or fine imposed pursuant to Federal, State, or Local law;

(<u>T</u>) payments for services that were never received, including payments for legal services pursuant to a retainer agreement to the extent payments for such services exceed the value of the services rendered;

(U) expenditures to facilitate, support, or otherwise assist in the execution or performance of the duties of public office;

(V) expenditures related to childcare services; or

(W) payments for liabilities that were not reported in the disclosure statement covering the reporting period in which the liability was incurred.

(b) Expenditures in cash. A candidate may not use cash for an expenditure of more than \$100.

(c) Volunteer services. Candidates may not pay volunteers for services already performed on a voluntary basis for that election, but may hire them as paid employees or retain them as consultants for future services. Candidates may not accept professional services on a volunteer basis from individuals who previously provided, on a paid basis, services of a similar nature to the same candidate during the same election cycle. Candidates may not accept volunteer services from any entity, or from an individual having an ownership interest of 10% or more in, or control over, any entity that provided paid services to the same candidate during the same election cycle. Notwithstanding the foregoing, after the election, candidates may accept volunteer services from individuals who previously provided paid services.

(d) Candidates may not enter into contracts or agreements that provide for payments to vendors or employees that are conditional upon the receipt of public funds.

§ 6-03 Joint expenditures; endorsements

(a) Generally. Candidates may engage in joint campaign activities, including joint fundraising, joint petitioning, and the production of joint campaign literature, subject to applicable expenditure limits.

(b) Candidates must pay proportionally. A candidate must pay for the portion of the joint expenditure that is proportionally equivalent to such candidate's campaign's exposure in or benefit from the joint activity, unless the benefit is *de minimis*.

> (i) Candidates and other individuals or entities may present information to the Board establishing such a *de minimis* benefit pursuant to Section 12-02.

> (ii) Factors for proportional payment. Among the factors the Board will consider in determining whether the benefit to a candidate is proportionally equivalent to its expenditure or is <u>de minimis are</u>:

(A) the geographic distribution or location of the material or activity;

(B) the subject matter of the communication;

(C) the references to the candidate or the candidate's appearances therein;

(D) the relative prominence of a candidate's references or appearances in the communication, including the size and location of references to the candidate and any photographs of the candidate;

(E) the timing of the communication; and

(F) other circumstances of the communication.

(c) Non-proportional payment may lead to in-kind contribution. To the extent a candidate does not pay for the proportional benefit it receives, the candidate is considered to have received an in-kind contribution from the other candidate. An in-kind contribution is also an expenditure by the candidate receiving the contribution.

(d) Not joint expenditures. The following activities, by themselves, are not joint expenditures:

(i) the act of appearing with another candidate;

(ii) endorsing a candidate, or communicating about an endorsement or appearance in an insubstantial way, such as where the endorser's name is one of several names appearing on a communication and is of equivalent prominence as the other names; and

(iii) giving fundraising assistance to another candidate in the form of *de minimis* written communications, such as allowing the use of one's name or signature on a letter soliciting funds for another candidate or on fundraising material where the endorser's name appears alone or with other names and is of equivalent prominence as the other names.

§ 6-04 Independent expenditures

(a) Factors for determining independence. In determining whether an expenditure is independent, the Board may consider any of the factors from the following non-exhaustive list:

(i) whether the person or entity making the expenditure is also an agent of a candidate;

(ii) whether any person authorized to accept receipts or make expenditures for the person or entity making the expenditure is also an agent of a candidate;

(iii) whether a candidate has authorized, requested, suggested, fostered, or otherwise cooperated in any way in the formation or operation of the person or entity making the expenditure;

(iv) whether the person or entity making the expenditure has been established, financed, maintained, or controlled by any of the same persons or entities as those that have established, financed, maintained, or controlled a political committee authorized by the candidate;

(v) whether the candidate shares or rents space for a campaign-related purpose with or from the person or entity making the expenditure;

(vi) whether the candidate has solicited or collected funds on behalf of the person or entity making the expenditure, during the same election cycle in which the expenditure is made;

(vii) whether the candidate, or any public or private office held or entity controlled by the candidate, including any governmental agency, division, or office, has retained the professional services of the person making the expenditure or a principal member or professional or managerial employee of the entity making the expenditure, during the same election cycle in which the expenditure is made; and (viii) whether the candidate and the person or entity making the expenditure have each consulted or otherwise been in communication with the same third party or parties, if the candidate knew or should have known that the candidate's communication or relationship to the third party or parties would inform or result in expenditures to benefit the candidate.

(b) Board determination. Upon consideration of the factors described in subdivision (a), the Board may determine by a preponderance of evidence that an expenditure was not independent. Prior to such determination, the candidate and/or the person or entity making the expenditure shall have an opportunity to provide evidence indicating that such expenditure was independent.

(c) Presumed non-independent expenditures. Financing the dissemination, distribution, or republication of any broadcast or any written, graphic, or other form of campaign materials prepared by a candidate is presumed to be a non-independent expenditure, unless the candidate can show that the activity was not in any way undertaken, authorized, requested, suggested, fostered, or otherwise cooperated in by the candidate.

(d) Non-independent expenditures are contributions and expenditures. An expenditure for the purpose of furthering or facilitating the nomination or election of a candidate which is determined not to be an independent expenditure, is a contribution to, and an expenditure by, the candidate.

(e) Expenditures made by party committees or constituted committees. Communication between, or common agents shared by, parties and their nominees will not require a conclusion that all spending by the party's constituted committees and party committees in an election is an in-kind contribution to the nominee.

> (i) The following expenditures made by party committees or constituted committees are not considered in-kind contributions to a candidate unless it is demonstrated that the candidate in some way cooperated in the expenditure and that the expenditure was intended to benefit that candidate:

> > (A) materials or activities that promote the party, or oppose another party, by name, platform, principles, history, theme, slogans, issues, or philosophy, without reference to particular candidates in an upcoming election subject to the requirements of the Act.

(B) materials or activities in connection with candidates and elections not subject to the requirements of the Act.

(C) training, compensating, or providing materials for poll watchers appointed by the party pursuant to § 8-500 of the New York State Election Law.

(D) promoting party enrollment or voter turnout without reference to particular candidates in an upcoming election subject to Program requirements, including research, polling, recruitment of party employees and volunteers, and development and maintenance of voter and contributor lists.

(E) raising funds for the party without reference to particular candidates in an upcoming election subject to the requirements of the Act.

(F) mailing of absentee ballot applications in a special or general election in which an office not subject to the requirements of the Act is on the ballot.

(ii) The Board may require a candidate to demonstrate that expenditures made by a party committee or constituted committee for the purpose of furthering or facilitating the nomination or election of a candidate, including expenditures for the purpose of furthering or facilitating the defeat of the nomination or election of such candidate's opponent or prospective opponent, are not in-kind contributions to the candidate.

(iii) Where a candidate has, without making public disclosure of an outstanding liability in a timely manner, promised or made reimbursement or other payment to a party committee or constituted committee for an expenditure, such expenditure will be considered an in-kind contribution to the candidate during the time preceding the reimbursement or other payment by the candidate.

(f) Running as a "ticket." If candidates announce they are running together as a "ticket" for which they have chosen to join together in a broad spectrum of activities to promote each other's election, the Board will presume that expenditures made by one candidate's campaign for materials or activities that clearly identify the other candidate are in-kind contributions to the second candidate. If the expenditures are

THE CITY RECORD

in-kind contributions, the expenditures are subject to apportionment requirements as joint expenditures. The following factors would increase the burden a candidate would have in overcoming this presumption:

(i) the campaigns have staff, consultants, office space, or telephone lines in common; and

(ii) other in-kind contributions, expenditure refunds, advances, or joint expenditures have been made between these candidates, or one of the candidates has reported a liability owed to the other candidate(s).

(g) Certain routine interactions with entities. Certain routine interactions with entities, absent other indicia of non-independence, will not in themselves lead to a determination of non-independence pursuant to Section 6-04(b), including:

(i) Discussions of logistics, including scheduling, regarding a non-fundraising event hosted by the entity;

(ii) Communications directly related to an entity's endorsement process, including questionnaires and interviews; and

(iii) Requesting, obtaining, or distributing publicly available materials such as a candidate's photograph, biography, position paper, or press release, but not including leaflets, posters, or other similar materials, nor video or audio materials.

§ 6-05 Expenditures made by other committees established for the candidate

(a) A candidate has the burden of demonstrating that expenditures made by a committee authorized by such candidate for a different election should not be attributed to the covered election.

(b) Failure to meet this burden will result in the application of all Program requirements to these committees for the covered election, including attribution of expenditures to the covered election.

(c) Expenditures for fundraising for more than one election. When a candidate makes expenditures for a single event or other activity to raise funds for more than one office, and the next election that will be held is:

(i) a covered election, the full amount of such expenditures is subject to the expenditure limit.

(ii) not a covered election, a portion of such expenditures will be subject to the expenditure limit in the same proportion as the funds raised for the covered election bears to the total funds raised, unless the candidate can demonstrate to the Board that a different apportionment is applicable.

§ 6-06 Identification of communications

(a) "Paid for by." When a candidate makes expenditures for any literature, advertisement, or other communication, the communication must include the words "paid for by" followed by the first and last name of the candidate or the name of the candidate's authorized committee, or, if the candidate has more than one authorized committee, the candidate's principal committee; provided that, if the name of the committee does not include the first or last name of the candidate, the words "paid for by" must be followed by the first and last name of the candidate, either instead of or in addition to the name of the committee.

(b) "Authorized by." When a candidate authorizes any individual or entity, other than the candidate, to pay for any literature, advertisement, or other communication in support of or in opposition to any candidate in any covered election, the communication must include the words "authorized by" followed by the first and last name of the candidate or the name of the candidate's authorized committee, or, if the candidate has more than one authorized committee, the candidate's principal committee; provided that, if the name of the committee does not include the first or last name of the candidate, the words "authorized by" must be followed by the first and last name of the candidate, either instead of or in addition to the name of the committee.

(c) Form. The identification required by subdivision (a) or (b) of this section must be in the following form:

(i) For printed material, an internet text advertisement, or a website, the identification must be written in a font of conspicuous size and style and contained in a box within the borders of the communication.

(ii) For an audio communication broadcast on radio or over the internet, the identification must be clearly spoken at the beginning or end of the communication.

(iii) For a video communication broadcast by television, satellite, cable, internet, or similar medium, the identification must be clearly spoken at the beginning or end of the communication and, simultaneous with the spoken disclosure, written in a font of conspicuous size and style contained in a box within the borders of the communication. (iv) For a telephone communication, the identification must be clearly spoken at the beginning or end of the communication. If the identification is spoken at the end of the communication, the name of the candidate must also be clearly spoken at the beginning of the call.

(d) Languages other than English. For communications primarily in a language other than English, all required written or spoken identification required by this section must be in such language.

(e) Where identification would be impractical. This requirement may be modified by the Board concerning items upon which identification would be impractical.

§ 6-07 Routine communication sent by a political club to its members. A routine communication sent by a political club to its members that includes the name of a candidate is not a contribution or an expenditure, provided that:

(a) the candidate is already a member of the club;

(b) the club has fewer than 500 members; and

(c) the communication does not solicit funds on behalf of or otherwise promote the candidate's campaign for a covered election.

Chapter 7: Pre-Election Public Funds Payments

§ 7-01 Eligibility

(a) A determination that a candidate has met all eligibility requirements of the Act and these rules, as of the date on which payment is to be made, shall not constitute the Board's final determination of the amount, if any, for which the candidate qualifies. The Board shall provide specific notice of any such final determination.

(b) Failure to respond to a request for audit documentation or information by the Board may be a basis for a non-payment determination.

(c) Payment amount. A candidate in any covered primary, general, or special election, having demonstrated eligibility to receive public funds, including by meeting the threshold for eligibility for public funding pursuant to § 3-703(2) of the Code, may receive public matching funds based on valid matchable contribution claims and the matching rate set forth in § 3-705(2)(a) of the Code, up to the maximum amount set forth in § 3-705(2)(b) or § 3-705(7) of the Code, as applicable. Payments are subject to withholdings and deductions as set forth in the Act and these rules.

(d) Approval by Board subject to correction of limited, isolated, and easily corrected compliance issues. The Board may, at its discretion, approve a public funds payment to a candidate, notwithstanding limited, isolated, and easily corrected compliance issues, conditioned upon a demonstration by the candidate that such candidate has taken the action specified by the Board, by the deadline specified by the Board, to correct the compliance issues. The candidate shall have the burden of demonstrating to the Board that such candidate has taken the action specified by the Board by the applicable deadline.

(e) Ballot disqualification; unopposed candidates. Pursuant to § 3-703(1)(a) and (5) of the Code, the Board will not make payment to any candidate disqualified from the ballot by the Board of Elections or by a court, or to any candidate for an election in which all other candidates have been disqualified from the ballot by the Board of Elections or by a court, until such candidate or other candidate is restored to the ballot by a court of competent jurisdiction. A candidate who appears as the only candidate on the ballot in an election shall not be eligible to receive public funds, notwithstanding any write-in candidates in that election, except as otherwise provided in Paragraph (ii) below.

(i) The Board will consider a candidate to be finally disqualified from the ballot on the earlier of the date of an administrative or judicial determination disqualifying the candidate for which there is no appeal as of right (unless the disqualification is reversed on appeal or otherwise) or of the election.

(ii) Payment petition. Payment may be made to a candidate who was temporarily on the ballot or opposed in an election, pursuant to a determination of the Board of Elections or a court of competent jurisdiction, but then ultimately disgualified from the ballot or unopposed in that election, only when the Board's audit of the candidate's campaign has been completed and only if the candidate has, within 30 days after the date of the final disgualification, as provided herein, filed a written petition, sworn to or affirmed, and supporting documentation that demonstrates that: (1) liabilities in qualified campaign expenditures incurred before the date of final disgualification remain unpaid; (2) the total amount of cash on hand is insufficient to pay these liabilities; (3) all expenditures made and liabilities incurred after the final disgualification were reasonable and necessary; and (4) the candidate was otherwise eligible and in compliance with all other Program requirements as of the date of final disqualification and has remained in compliance at all times since that date. The petition must also include an undertaking to use all public funds received to extinguish the liabilities enumerated pursuant to subparagraph. (1) of this paragraph.

(f) Write-in candidates. A candidate who is seeking election exclusively as a write-in candidate, or who is only opposed by a candidate who is seeking election exclusively as a write-in candidate, is not eligible to receive public funds.

§ 7-02 Timing

(a) Three payment dates in the 30 days prior to an election. The Board shall schedule at least three payment dates in the 30 days prior to a covered primary, general, or special election.

(i) No public funds shall be paid to candidates in a primary or general election any earlier than four business days after the final day to file a written Certification for such election pursuant to paragraph (c) of subdivision 1 of § 3-703 of the Code.

(ii) Pursuant to § 3-703(1)(a) and (5) of the Code, public funds are not payable to a participant who has not met the legal requirements to have such participant's name on the ballot, who is unopposed, or, for the optional early public funds payment, who has not certified that such participant intends to meet all the requirements of law to have such participant's name on the ballot and stated the specific office to which such participant is seeking nomination or election.

(b) Preliminary review of disclosure statements; reasons for delay. In order to enable payment within four business days after receipt of disclosure statements, or as soon thereafter as is practicable, pursuant to § 3-705(4) of the Code, the Board shall conduct a preliminary review of all disclosure statements filed.

(i) If a candidate's complete disclosure statement is not actually received by 5:00 P.M. on the due date, or if the candidate submits a disclosure statement that fails to comply substantially with the requirements of the Act or these rules, the Board may be unable to make payment within four business days, and may review the disclosure statement for possible payment determinations at the next payment date.

(ii) A preliminary review may also be delayed for reasons including, but not limited to, consideration of whether a basis exists for an ineligibility determination. A delayed preliminary review may result in a payment determination being delayed until such time as it is practicable.

(c) Preliminary review of matching claims. Prior to making a preelection payment determination, the Board may issue a report to the candidate indicating any matching claims determined to be invalid based on preliminary review of each disclosure statement and of the matchable contribution claims reported therein. The candidate may respond to such report by providing information or documentation demonstrating that any such matching claims should be considered valid. Failure by the candidate to respond to such report by the deadline set by the Board may result in a delay in payment of public matching funds.

(d) Characterization of payments as for the primary or general election.

(i) If a participant is on the ballot and has an opponent on the ballot in both a primary and the general elections, payments made after the primary election will be characterized initially as follows:

(A) As a primary election payment, if the payment is made on the basis of contribution and expenditure information reported in or before the disclosure statement due 10 days after the primary election, except as otherwise provided in subparagraph (B).

<u>(B) As a general election payment, to the extent that any further</u> primary election payments would exceed a maximum applicable in the primary election pursuant to the Act.

(C) As a general election payment, if the payment is made on the basis of contribution and expenditure information reported in disclosure statements due later than 10 days after the primary election.

(ii) If the Board determines that payments characterized initially as either primary or general election payments were, in fact, used for qualified campaign expenditures incurred in the other election, the payments will be recharacterized accordingly, and additional payments may be made or repayments required, if appropriate.

§ 7-03 Payment by electronic funds transfer. All payments of public funds shall be by electronic funds transfer unless the Board determines, at its sole discretion, to use an alternative payment method. In order to receive prompt payment, the participating candidate must provide the Board with a voided committee check and such additional information as shall be required by the Board.

§ 7-04 Public funds cap

(a) Pursuant to § 3-705(7)(b) of the Code, a candidate shall not be eligible to receive more than one quarter of the applicable maximum pursuant to § 3-705(2)(b) of the Code unless the candidate submits a certified signed statement attesting to, and stating the reason for, such candidate's need for additional public funds. The statement of need must be filed with the Board no later than the due date for the applicable disclosure statements as follows, except that, if the basis for filing the statement of need arises after such due date, and no such basis existed prior to such due date, then the statement of need shall be due by the deadline for the disclosure statement immediately preceding the next date on which a public funds payment is to be made:

- (i) <u>Candidates in the primary election must file the statement of</u> <u>need no later than the due date for the 32-day pre-primary</u> <u>election disclosure statement.</u>
- (ii) <u>Candidates in the general election must file the statement of</u> <u>need no later than the due date for the 32-day pre-general</u> <u>election disclosure statement.</u>

(b) The Board shall verify the truthfulness of any certified signed statement submitted pursuant to § 3-705(7)(b) of the Code and determine whether supporting documentation demonstrates the existence of the condition or conditions described in such statement. For the purposes of making a determination pursuant to § 3-705(7) (b)(1) of the Code, a non-participating candidate shall be presumed to have the ability to self-finance when it is demonstrated through supporting documentation that such candidate has readily available funds in excess of one fifth of the applicable expenditure limit and that such candidate can reasonably be expected to spend such funds for such candidate's election.

§ 7-05 Small primaries

(a) A candidate on the ballot in one or more primary election(s) in which the number of persons eligible to vote for party nominees in each such election totals fewer than 1,000 shall not receive public funds in excess of \$5,000 for qualified campaign expenditures in such election or elections; provided, however, that the foregoing limitation shall not apply to such candidate if the candidate is opposed in a primary election by (i) a participant who is not subject to such limitation or (ii) a non-participant who has spent or contracted or become obligated to spend in excess of \$10,000 for such primary election. The Board shall determine whether a non-participant has exceeded such \$10,000 level pursuant to \$ 3-705(6) of the Code.

(b) For the purposes of Subdivision (a), the number of persons eligible to vote for party nominees in a primary election shall be as determined by the Board of Elections for the calendar year of the primary election pursuant to § 5-604 of the New York State Election Law. If such determination for any primary election is not available from the Board of Elections as of the day before the due date for filing a Certification pursuant to § 3-703(1)(c) of the Code, the most recent determination by the Board of Elections of the persons eligible to vote for party nominees for the office for which such primary election is held shall be relied upon.

§ 7-06 Withholdings. The Board may withhold up to 5% of the amount of public funds payable to a candidate until the final preelection payment for any election in which the candidate is eligible to receive public funds. In addition, the Board may withhold from preelection public funds payments: (a) a reasonable amount based on a projection of the rate of invalid matching claims; and (b) a reasonable amount if the Board determines that there is reason to believe that the candidate has failed to comply with the Act and these rules, including by failing to adequately respond to a Board request for information or documentation. Such withholdings shall be subject to post-election audit.

§ 7-07 Deductions from payments

(a) The total amount of public funds payable to a candidate for a covered primary, general, or special election shall be reduced by the sum of the following:

(i) the amount of outstanding civil penalties assessed by the Board as a result of the candidate's failure to comply with the Act and these rules during the current covered election; and

(ii) the amount of the candidate's:

(A) transfers and other disbursements from a political committee that is involved in an election in which the candidate is currently a participant, to a political committee that is not involved in that election;

(B) expenditures made to pay expenses for or debt from a previous election, including repayments of public funds and payment of penalties owed to the Board for a previous election;

(C) contributions to other political committees that do not meet the requirements provided in §3-705(8) of the Code for contributions that shall not be a basis for reducing public funds payments;

(D) loans to other candidates that are not repaid within 30 days or by the date of the election, whichever is

THE CITY RECORD

earlier, or spending for other candidates, including joint expenditures, to the extent such expenditures benefit another candidate, and independent expenditures; provided that independent expenditures made by the principal committee of a candidate shall not be a basis for reducing public funds payments to that candidate, where such expenditures do not, in the aggregate, exceed the amount provided in §§ 3-705(8)(i), (ii), or (iii) of the Code, as applicable, for contributions to political committees;

(E) loans to or spending for political party committees and political clubs that are not reimbursed within 30 days or by the date of the election, whichever is earlier, provided that if the candidate demonstrates that the expenditure was for a tangible item that directly promotes the candidate's election, such as an advertisement in a fundraising journal, this subparagraph shall not apply to the fair market value of that item; and

(F) expenditures made for the purpose of furthering the candidate's election to the position of Speaker of the City Council.

(b) Disbursements that would otherwise result in a deduction pursuant to Paragraph (ii) of Subdivision (a) of this section shall not result in any such deduction if:

(i) such disbursements are made out of a segregated bank account;

(ii) at no time does the segregated bank account contain any funds other than contributions received by the candidate and deposited directly into the account pursuant to this section, and bank interest paid thereon;

(iii) funds deposited into the segregated bank account are not used for any purpose other than disbursements governed by Paragraph (ii) of Subdivision (a) above or payment of bank fees associated with the segregated bank account;

(iv) contributors whose contributions are deposited into the segregated bank account have confirmed in writing, pursuant to Section 4-01(b)(ii)(B), that they understand that these contributions will only be used for such disbursements and will not be matched with public funds;

(v) copies of such written confirmations are submitted to the Board by the due date for the disclosure statement in which such contributions are required to be reported pursuant to these rules;

(vi) copies of checks for each disbursement out of the segregated bank account are submitted to the Board by the due date for the disclosure statement in which such disbursements are required to be reported pursuant to these rules;

(vii) a copy of each bank statement for the segregated bank account is submitted to the Board by the due date for the next disclosure statement; and

(viii) for each individual contribution deposited into the segregated bank account, and each disbursement out of the segregated bank account, the candidate has complied with all other applicable provisions of the Act and these rules, including but not limited to the record keeping and reporting provisions.

(c) Candidates must deposit the entire amount of a contribution into the segregated bank account provided for in subdivision (b), and may not divide the contribution between different accounts.

(d) Contributions deposited into a segregated bank account pursuant to this section will not be matched with public funds.

(e) Any funds remaining in a segregated bank account after the election must be returned to the contributors whose contributions were deposited into the account, or, if that is impracticable, to the Fund, on or before December 31 in the year following the year of the election.

(f) A candidate who establishes a segregated bank account pursuant to this section, but fails to comply with any provision of Subdivisions (b), (c), (d), or (e) of this section, shall no longer be entitled to the exception from Subdivision (a) contained in Subdivision (b) of this section.

(g) Funds deposited into, and disbursements made from, a segregated bank account established and maintained in compliance with this section for the purpose of making expenditures to pay expenses for or debt from a previous election, including repayments of public funds owed to the Board, will not be considered to be raised or spent for the current covered election for purposes of the participant's expenditure limit calculation.

§ 7-08 Notice to candidates. The Board shall notify the candidate in writing of any non-payment determination or of the difference between the amount of public funds claimed and the amount paid.

§ 7-09 Petitions for review

(a) After the Board provides a candidate a written determination specifying the basis for payment or non-payment of public funds prior

to the election, the candidate may petition the Board in writing for reconsideration of such determination. Such petition must state the grounds for reconsideration and must also include either a request to appear before the Board concerning the petition or a statement that the candidate waives such candidate's right to appear. The Board shall review the determination that is the subject of the petition within five business days of the filing of such petition. If the Board is unable to convene within five business days, the Board may delegate to the Chair of the Board or the Chair's designee authority to make a determination regarding the petition.

(b) To be considered by the Board, a petition for review of a preelection payment or non-payment determination must not include any documentation or factual information not submitted to the Board prior to the determination under review, unless the participating candidate can demonstrate good cause for the previous failure to submit such documentation or information and for any failure to communicate on a timely basis with the Board.

(c) The Board shall timely issue a written determination on the subject of the petition. If the petition is denied, the determination shall inform the candidate of the right to appeal such determination pursuant to Article 78 of the Civil Practice Law and Rules.

§ 7-10 Review of contributions and expenditures

(a) Facial Determinations. The Board may find that a candidate's registration and financial disclosure submitted to the Board or to the State Board of Elections provide on their face a sufficient basis for a determination of the amount such candidate has spent or contracted or become obligated to spend, or received in loans or contributions. The Board will presume that contributions and loans are accepted, disbursements are made, and liabilities are incurred by a candidate for such candidate's next election. In the absence of contrary evidence, the Board will rely upon the date of contributions, loans, disbursements, and liabilities as they appear on disclosure statements filed with the Board or the State Board of Elections.

(b) Petitions for reconsideration

(i) Contents. A candidate may file a petition for reconsideration of a preliminary Board determination made pursuant to subdivision (a). Such petition must be written and must:

(A) be sworn to or affirmed;

(B) indicate on whose behalf the petition is being submitted;

(C) contain sufficient specificity and detail for the Board to make a decision, including:

(1) if alleging that information submitted to the Board or the State Board of Elections is inaccurate or incomplete, the petition must describe the inaccuracies or omissions in detail, or the Board may presume that any information on file is true and complete.

(2) if alleging fundraising or spending activities to have taken place after the close of a reporting period, the petition must describe the activities in detail;

(3) if alleging that a third party is providing unreported contributions, goods, services, or benefits to a candidate in coordination with that candidate, the petition must include all relevant names, dates, and amounts, including information concerning the fair market value of services, materials, facilities, advertising, or other things of value reported to have been received or expended; and

(4) if alleging that statements have been made, the petition must include content of statements, name, dates, how speaker is affiliated with a candidate or other actor, and any other relevant information.

(D) include a request for a hearing, if desired;

(E) be submitted no later than 10 days after the determination or decision; and

(F) be accompanied by evidence that supports the allegations made in the petition. The petitioner has the burden of demonstrating a sufficient basis for a determination.

(ii) Notice of petition

(A) The Board shall send a copy of the petition

to any opposing entities named in the petition or known by the Board to have an opposing interest in the petition.

(B) The notice shall specify:

(1) the deadline for the opposing entities to submit a response; and

(2) the date and time at which the Board will conduct a hearing on the petition, if any.

(iii) **Response to the petition**. The opposing entities may submit a response to the petition. A failure to submit a response may be deemed an admission of the allegations in the petition and sufficient reason to deny a request to reconsider any determination based on the petition. The response must:

(A) be submitted in the manner and by the deadline set forth by the Board;

(B) be sworn to or affirmed;

(C) either acknowledge that fundraising or spending on behalf of the respondent is sufficient to permit the Board to make a determination, regardless of the merit of the petitioning candidate's particular allegations, or contend that a determination would be inappropriate and specify the allegations in the petition that are denied and those that are admitted;

(D) explain any failure to disclose any information to the Board or the City or State Board of Elections;

(E) include evidence that supports the response, including, but not limited to, all relevant names, dates, and amounts, including information concerning the fair market value of services, materials, facilities, advertising, or other things of value reported to have been received or expended;

(F) include a request for a hearing, if desired; and

(G) be signed by the opposing entity.

(iv) Hearing

(A) The Board may decide to conduct a hearing upon request or on its own initiative.

(B) The hearing date will be no earlier than 10 days after the petition is received, except:

(1) for good cause shown; or

(2) for petitions received less than 30 days before the election, in which case the hearing date will be no earlier than three days after the petition is received.

(C) The respondent and each opposing candidate must notify the Board no less than one day in advance whether they will be represented at the hearing. Representation of a party at a hearing must be by a person with knowledge of the facts at issue.

(D) At the hearing, testimony given may be under oath and no new documentary evidence will be considered unless good cause is shown for the party's failure to submit it with its petition or response.

(v) Notice of determination. Following the review of disclosure statement filings; the petition and response, if any; and testimony and evidence presented at a hearing, if any, the Board will:

(A) decide whether there is a sufficient basis for a determination; and

(B) provide written notice to the petitioner and opposing entities, if any, of its determination.

(vi) New petition for a preliminary public funds determination may be submitted based on new evidence. The Board's decision not to award public funds shall not be construed to preclude the subsequent submission of any petition based on new evidence for a determination respecting that candidate.

(vii) Request for reconsideration. The petitioner or an opposing party may request reconsideration of a preliminary public funds determination.

(A) A request for reconsideration must be made in writing no later than 10 days after the determination, and must include: (1) an affidavit of the party affirming that neither fundraising nor spending on behalf of the respondent that has taken place after the close of the most recent reporting period is sufficient to permit the Board to make a determination; and

(2) an undertaking by the respondent to notify the Board immediately in writing if and when either fundraising or spending on behalf of the respondent becomes sufficient to permit the Board to make a determination pursuant to subdivision (a).

(B) The Board shall consider the evidence, may conduct a hearing, and shall notify the petitioner of its determination.

(C) The Board may consider a respondent's failure to respond to the petition, or the failure of the party requesting reconsideration to be represented at a hearing on the petition, to be a sufficient basis for denying a request for reconsideration. The Board shall notify the respondent, all opposing candidates, and the petitioner whether it will reconsider the determination pursuant to subdivision (a) or decision not to make such a determination, as the case may be.

(D) The Board will respond within five business days of the filing of such petition.

(viii) Appeal. A party may bring a special proceeding as set forth in Article 78 of the Civil Practice Law and Rules within four months of the Board's final determination denying the petition for reconsideration.

Chapter 8: Post-Election Public Funds Payments

§ 8-01 Payment determinations

(a) Candidates who fail to demonstrate compliance with the Act and these rules, including candidates who satisfy one or more criteria for ineligibility as provided in section 3-01(d), are not eligible to receive a post-election public funds payment.

(b) Candidates have the burden of demonstrating eligibility to retain public funds received prior to the election and to receive additional public funds after the election. Candidates who fail to demonstrate eligibility to retain all or a portion of the public funds previously received may be required to repay such amount to the Fund.

(c) A candidate's post-election payment, if any, will be reduced by the amount of any applicable deductions pursuant to section 7-07(a), if such amounts were not deducted from pre-election public funds payments.

(d) The post-election payment, repayment, or nonpayment determination shall be the final determination regarding that candidate's public funds payment status, except as provided in section <u>8-05.</u>

§ 8-02 Amount of post-election payment

(a) Reasons for post-election payment. A post-election payment shall only be made if, at the conclusion of the post-election audit, a candidate has demonstrated unpaid matching claims, a qualified expenditure surplus, and documented outstanding liabilities.

(i) Unpaid matching claims. Candidates may be entitled to receive a post-election payment equal to the amount by which the candidate's total valid matchable claims multiplied by the applicable matching rate pursuant to § 3-705(2)(a) of the Code exceed the total pre-election payments received by the candidate.

(ii) Qualified expenditure surplus. Candidates may be entitled to receive a post-election payment equal to the amount by which the candidate's total qualified expenditures exceed the total pre-election payments received by the candidate.

(iii) Documented outstanding liabilities. Candidates may be entitled to receive a post-election payment equal to the amount of the candidate's properly reported and documented liabilities that remain outstanding, less the amount remaining in the candidate's principal committee bank account.

(A) Prior to issuing a post-election payment, the Board may require the candidate to submit any bank statements not previously provided.

(B) In order to be the basis for a post-election payment, an outstanding liability must be reported on or before the due date for the final disclosure statement required to be submitted for the covered election to which the liability relates. (C) To document an outstanding liability for the purpose of receiving a post-election payment, the candidate must provide documentation demonstrating that the reported payee has made a commercially reasonable attempt to collect the debt.

(iv) Post-election payment is smallest amount. The amount of the post-election payment shall be the lowest of the three amounts detailed in paragraphs (i), (ii), and (iii) of this subdivision. If any of these amounts is zero, the candidate is not entitled to receive a post-election public funds payment.

(b) Statutory maximum. Combined with the total pre-election payments received by the candidate, the post-election payment shall not exceed the maximum payment allowed for each election in which the candidate was a participant pursuant to § 3-705(2)(b) of the Code.

§ 8-03 Use of final post-election payment. Before the Board makes the final post-election payment determination, if eligible, the candidate must submit to the Board bills or other documentation of outstanding debt for which such payment will be used. Within 60 days after the final public funds payment, the candidate must demonstrate that the public funds were used to pay such outstanding debt. If such demonstration is not made, the candidate must repay the public funds to the Board.

§ 8-04 Disclosure statement amendments. The Board shall not make payments based on disclosure statement amendments filed after January 15 in the year following the year of the election; provided, however, that the Board may make payments based upon such amendments solely if they are made in response to invalid matching claims reports or expenditure sample reports to which the Board has requested a response after January 15 in the year following the year of the election.

§ 8-05 Post-election petitions for review

(a) After the Board provides a candidate a written determination specifying the basis for payment or non-payment of public funds after the election, the candidate may petition the Board in writing for reconsideration of such determination.

(b) A petition for review of a post-election payment determination must be submitted within 30 days of the candidate's final audit report, and must include:

(i) a statement of the grounds for reconsideration;

(ii) information or documentation that was unavailable to the Board previously and is material to such determination;

(iii) a showing that the candidate had good cause for the previous failure to provide such information or documentation; and

(iv) either a request to appear before the Board concerning the petition or a statement that the candidate waives such candidate's right to appear.

(c) The Board shall timely issue a written determination on the subject of the petition. If the petition is denied, the determination shall inform the candidate of the right to appeal such determination pursuant to Article 78 of the Civil Practice Law and Rules.

Chapter 9: Public Funds Repayments

§ 9-01 Candidates must promptly repay public funds upon a determination made by the Board that a repayment is required. Candidates returning public funds must make a check payable to the "New York City Election Campaign Finance Fund." Candidates may not reclaim public funds they have returned. Public funds must be repaid to the Board separately from, and in addition to, any penalties owed by the candidate.

§ 9-02 Reasons for repayment. The Board shall determine whether a candidate or such candidate's principal committee is required to repay public funds previously received as follows:

(a) **Overpayment of public funds.** Pursuant to § 3-710(2)(a) of the Code, the candidate's principal committee is required to pay to the Board an amount equivalent to the amount of public funds received in excess of the amount which the candidate is eligible to receive.

(b) Qualified expenditure deficit. Pursuant to § 3-710(2)(b) of the Code, the candidate's principal committee is required to pay to the Board an amount equivalent to the amount of public funds paid to the candidate that were not spent on qualified expenditures.

(i) Determination of personal liability. A candidate may be personally liable for paying to the Board the amount of public funds not spent on qualified expenditures, or a portion of that amount. In considering whether and to what extent a participating candidate shall be personally liable for repayment of public funds, the Board shall act in accordance with the following:

(A) where credible documentation supporting each qualified expenditure exists but is incomplete, the Board

shall not impose such liability for such expenditure;

(B) where there is an absence of credible documentation for each qualified expenditure, the Board may impose such liability upon a showing that such absence of credible documentation for such expenditure arose from a lack of adequate controls including, but not limited to, trained staff, internal procedures for following Board guidelines and internal procedures for employing standard financial controls.

(c) Final bank balance. Pursuant to §3-710(2)(c) of the Code, the candidate and the principal committee are required to pay to the Board an amount equivalent to the campaign funds remaining in such candidate's principal committee account. The candidate must promptly pay to the Board such remaining funds; provided, however, that all remaining funds for a candidate shall be immediately due and payable to the Board upon a determination by the Board that the candidate has delayed the post-election audit process. The candidate must promptly a determination made by the Board at a subsequent date. The amount of remaining campaign funds shall be presumed to be equal to the candidate's authorized committee bank account balance on January 11 in the year following the election, or for special elections, on the last day of the reporting period for the final disclosure statement the candidate is required to file with the Board for such election, less any permissible documented post-election expenditures pursuant to paragraph (i) of this subdivision. The Board may also consider information revealed in the course of an audit or investigation in making a remaining funds determination, including, but not limited to, the fact that campaign expenditures were made in violation of law, that expenditures were made for any purpose other than the furtherance of the candidate's nomination or election, or that the candidate has not maintained or provided requested documentation. If a candidate repays to the Fund all funds remaining in such candidate's authorized committee bank account on or before December 31 in the year of the election, or, for special elections, on or before the last day of the month following the month in which the election took place, such candidate shall be presumed not to owe additional remaining funds, provided that any contributions received and expenditures made after such funds are repaid must be raised and spent in compliance with the Act and these rules.

> (i) Before repaying campaign funds remaining in the committee bank account, a candidate may make post-election expenditures only for routine activities involving nominal cost associated with winding up a campaign and responding to the post-election audit. Such expenditures may include: payment of utility bills and rent; reasonable staff salaries and consultance foce for reconciling to a post election and consultancy fees for responding to a post-election audit; reasonable staff salaries and legal fees incurred prior to the date of the issuance of the candidate's final audit report and associated with defending against a claim that public funds must be repaid; a post-election event for staff, volunteers, or supporters held within 30 days of the election; reasonable moving expenses related to closing the campaign office; a holiday card mailing to contributors, campaign volunteers, and staff; thank you notes to contributors, campaign volunteers, and staff; payment of taxes and other reasonable expenses for compliance with applicable tax laws; and interest expense. Routine post-election expenditures that may be paid for with remaining campaign funds do not include such items as post-election mailings other than as specifically provided for in this paragraph; making contributions; or making bonus payments or gifts to staff or volunteers. Campaign funds remaining in the committee account may not be used for transition and inauguration activities.

(ii) Notwithstanding the restriction on the use of receipts provided in Subdivision (c)(i), a candidate who has outstanding liabilities from the election, including a candidate who owes public funds or penalties to the Board, may make post-election expenditures for the purpose of raising funds to repay such debt, provided, however, that such expenditures and any contributions received shall be included in the calculation of the candidate's remaining funds, unless such expenditures and contributions are incurred or received after the date of the issuance of the candidate's final audit report.

(iii) Notwithstanding the prohibition on post-election bonus payments provided in Subdivision (c)(i), bonus payments to employees may be made after the election, provided that:

> (A) the bonus is paid pursuant to a contract executed before the services provided by the employee that serve as the basis for the bonus are performed;

(B) such contract states in detail the amount of the bonus and the criteria that must be satisfied in

order for the bonus to be paid;

(C) such criteria are quantifiable, merit-based, and related to services provided prior to the election;

(D) the bonus is not conditioned, either in existence or amount, on the result of the election, the amount of funds remaining in the committee account, or whether the candidate received public funds; and

 (\underline{E}) the bonus does not constitute more than 10% of the total amount paid to the employee.

(d) Repayment is largest amount. Where more than one grounds for repayment exists as provided in subdivisions (a), (b), and (c) of this section, the amount to be paid to the Board shall be the largest of the amounts.

(e) Breach of certification. A candidate found to have breached such candidate's certification pursuant to section 3-01(e) may be deemed ineligible to receive public funds and may be required to return all public funds previously received.

(f) Ballot disqualification. Pursuant to §3-709(7) of the Code, a candidate who has been finally disqualified or whose designating or nominating petitions have been finally declared invalid by the Board of Elections or a court of competent jurisdiction, may not thereafter spend public funds for any purpose other than the payment of previous liabilities incurred in qualified campaign expenditures. All public funds in excess of such liabilities previously incurred shall be promptly repaid to the Board; the amount to be repaid shall be determined in accordance with §3-710(2)(b) of the Code and subdivision (b) of this section. A repayment made pursuant to §3-709(7) shall not preclude a determination that an additional repayment is required pursuant to that or any other provision of the Act.

(g) Ballot fraud. Pursuant to § 3-710(3) of the Code, a candidate who has been disqualified by a court of competent jurisdiction on the grounds that such candidate committed fraudulent acts in order to obtain a place on the ballot, if such decision is not reversed, must pay to the Board an amount equal to the total public funds paid to the candidate. Repayments pursuant to this subdivision must be made promptly upon the court's determination of disqualification.

(h) Other reasons for repayment. The Board may require a candidate to repay public funds because:

(i) the candidate failed to maintain copies of checks or contribution cards that document matchable contributions;

(ii) the public funds paid were based on contributions that were returned or contribution checks that have not been paid;

(iii) the candidate has failed to demonstrate eligibility for the public funds paid or compliance with Program requirements; or

(iv) a determination pursuant to §§ 3-705(6) or (7) of the Code is reversed following reconsideration pursuant to Section 7-10(b) (vii).

Chapter 10: Audit and Enforcement

<u>§ 10-01 Audits</u>

(a) The Board shall conduct desk and field audits of participants and non-participants, regardless of whether the candidates receive public funds. Field audits may be conducted before or after an election, as the Board deems appropriate. In conducting audits, the Board may use random sampling of data and other analytic techniques, as appropriate. The Board shall conduct campaign audits in accordance with generally accepted government auditing standards.

(b) Candidates are required to respond to all Board requests for information and documentation during the audit process. Failure to respond in a timely fashion may cause delay in the issuance of a final audit report. In addition, the Board may assess penalties for failing to respond or responding late to requests for documentation or information pursuant to § 3-711 of the Code.

§ 10-02 Audit deadlines; optional post-election training. The Board shall issue all draft and final audit reports in accordance with the deadlines provided in § 3-710(1)(a) and (b) of the Code subject to any applicable exceptions to those deadlines provided in § 3-710(1)(d), (e), and (f) of the Code.

(a) Candidates who attend a post-election training. Pursuant to § 3-710(1) of the Code, where the candidate, the campaign manager, or the treasurer has attended a post-election audit training provided by the Board, the Board will issue the candidate's notice of alleged violations and recommended penalties, notice of recommended public funds payment or repayment, or final audit report within 14 months after the deadline for submission of the final disclosure report for the covered election, in the case of city council and borough-wide races, and within 16 months after the deadline for submission of the final disclosure report for the covered election in the case of Citywide races.

(b) Candidates who do not attend a post-election training. For candidates who do not attend a post-election training, the Board will

issue a notice of alleged violations and recommended penalties, notice of recommended public funds payment or repayment, or final audit report:

> (i) For City Council and borough-wide races, within 16 months after the deadline for submission of the final disclosure report for the covered election; and

(ii) For citywide races, within 18 months after the deadline for submission of the final disclosure report for the covered election.

(c) Deadline for attending post-election training. To be subject to the deadlines provided in subdivision (a) of this section, candidates must attend a post-election training on or before the earlier of 20 days following issuance of the candidate's draft audit report or one year after the deadline for submission of the final disclosure report for the covered election.

<u>§ 10-03 Board determinations concerning violations,</u> infractions, penalties, and repayment of public funds

(a) Notice and opportunity to contest. The Board will notify the candidate, the TIE, or the independent spender and its authorized representative, as applicable, in writing of any infractions or violations alleged to have been committed by such party, as well as any corresponding recommended penalties, and any recommended public funds repayment. Such notice shall:

(i) set forth in detail the factual and legal basis for staff's recommendation that a violation, infraction, or public funds repayment obligation be found;

(ii) inform the party that hearings are conducted in accordance with the requirements for adjudications contained in § 1046 of the Charter unless such procedures are waived by, as applicable: the participant or nonparticipant, the elected candidate, the independent spender or authorized representative, or an agent thereof;

(iii) notify the party of the opportunity to submit information and documentation for the Board's consideration within a time period to be specified in such notice; and

(iv) if sent prior to the election, notify the party of the opportunity to appear at a hearing before the Board to contest the alleged violations or infractions, penalties, and public funds repayment obligation; or

(v) if sent after the election, notify the party of the opportunity to appear at a hearing before the Board or an administrative law judge to contest the alleged violations or infractions, penalties, and public funds repayment obligation.

(b) Response to the notice

(i) The noticed party must submit a complete response to the notice by the due date. Extensions may be granted for good cause shown. Extension requests must be made in writing and must be submitted prior to the deadline.

(ii) Unless specifically notified to the contrary by the Board, the opportunity to submit information and documentation in response to the notice shall be the only such opportunity, and any information and documentation that is not timely received by the Board shall be disregarded. Under extraordinary circumstances, the Board may, at its sole discretion, choose to accept information or documentation not timely received.

(c) Final Board determination. Based on the evidence, the Board will make a final Board determination as to the party's violations or infractions, penalties, and public funds repayment obligation, and shall provide the party with a written copy of its determination. Such determination shall notify the party that it may be appealed within four months as set forth in Article 78 of the Civil Practice Law and Rules.

(d) **Payment.** If the Board has determined that the party must pay penalties or repay public funds, the party must make payment by the deadline set forth in the final Board determination. If these amounts are not paid by the payment deadline, they may be sought through appropriate legal action or, in the case of civil penalties against participants, by deduction from any public funds otherwise due for any election.

(e) Penalty Guidelines. The Board shall publicize staff guidelines for presumptive penalties to be recommended for certain violations of the Charter, Act, and these rules, subject to consideration of aggravating and mitigating factors that may be considered in determining the appropriate penalty assessment for the violation.

§ **10-04 Submission of false information.** If the Board has reason to believe that any individual or entity has knowingly submitted false information or fabricated evidence or has knowingly withheld or concealed information in a written or oral submission or representation to the Board, the Board may refer the matter to the appropriate agency for criminal prosecution and may commence a civil action to recover public funds and/or civil penalties, if appropriate.

Chapter 11: Procedural Rules for Formal Adjudications § 11-01 Definitions

Except as otherwise provided, the definitions set forth in section 1-02 apply in this chapter. In addition, the following terms mean:

"Case" means an adjudication pursuant to CAPA, § 1046 of the Charter.

"Filing" means submitting papers to the hearing officer, whether in person, by mail, or by electronic means, for inclusion in the record of proceedings in a case.

"Petition" means a document filed by the Board, analogous to a complaint in a civil action, which states the claims to be adjudicated.

"Petitioner" means the Board.

"Respondent" means a party against whom claims are asserted by the Board.

§ 11-02 Applicability

(a) This chapter applies solely to cases that are subject to CAPA, including hearings, pre-hearing and post-hearing matters, brought by the Board pursuant to the New York City Campaign finance Act, Title 3 of the New York City Administrative Code (the "Administrative Code").

(b) In the event of any inconsistency between this chapter, other chapters of this title, and the rules of OATH, this chapter shall govern.

§ 11-03 Proceedings before designation of hearing officer

Proceedings before the case is referred to a hearing officer shall be governed by § 1046 of the Charter and Chapters 9 and 10 of the Board rules.

§ 11-04 Designation of hearing officer

The Board shall designate, at its sole discretion, one or more hearing officers to preside over cases pursuant to this chapter. The designated hearing officer(s) shall be: (i) one or more members of the Board; (ii) another person or persons assigned by the Board; or (iii) the chief administrative law judge of OATH or such administrative law judge as the chief administrative law judge may assign.

§ 11-05 Commencement of proceedings and pleadings

(a) The petition

(i) The Board shall institute proceedings pursuant to this chapter by serving a petition, sworn to or affirmed as to the truth thereof, on the respondent(s). The petition shall comply with the requirements in OATH Rules of Practice § 1-22. The petition shall set forth the facts which, if proved, constitute violations of the Act or these rules or require repayment of public funds.

(ii) The petition shall be accompanied by the following: notice of the respondent's right to file an answer, the deadline to file an answer, and the place(s) to serve and file an answer; notice that failure to serve and file a timely answer shall be deemed an admission of all allegations contained in the petition; notice of the respondent's right to representation by an attorney or other representative; and notice that a person representing the respondent(s) must file a notice of appearance with the hearing officer.

(b) Service of the petition

(i) The Board shall serve the petition upon the respondent(s).

(ii) Service of the petition may be made by regular first-class mail or electronic means to the respondent's last known residential or business address or last known email address or fax number.

(iii) Service of the petition shall be complete upon mailing. if service was by first-class mail, or upon the date of transmission, if service was by electronic means.

(c) Answer

(i) The respondent(s) must file an answer to the petition with the hearing officer and serve the same answer on the Board.

(ii) If the petition was served on the respondent(s) by regular first-class mail, the answer must be filed and served within 26 calendar days of the date the petition was postmarked. If the petition was served on the respondent(s) by electronic means, the answer must be filed and served within 21 calendar days of the date of the transmission of the petition.

(iii) The answer may include affidavits or affirmations, documentary exhibits, or other evidentiary material in rebuttal of the petition. The answer may be accompanied by a memorandum of law.

(iv) If the respondent fails to serve and file a timely answer, all allegations of the petition shall be deemed admitted, and the case shall proceed as scheduled. If the answer fails to specifically address any allegation in the petition, such allegation shall be deemed admitted.

(v) The time to serve and file an answer may be extended only upon the consent of all parties or upon application to the hearing officer for good cause shown.

(d) Amendment of Pleadings. Pleadings shall be amended in accordance with OATH Rules of Practice § 1-25.

§ 11-06 Filing of Papers

(a) Generally. The notice accompanying the petition shall notify the parties of the designated hearing officer(s) and of the place to file papers. Papers may be filed with the hearing officer in person, by mail or by electronic means.

(b) Headings. If an OATH index number has been assigned pursuant to Section 11-07(c), the subject matter heading for each document must indicate the OATH index number.

(c) Means of service on adversary. Submission of papers to the hearing officer by electronic means, mail, or personal delivery without providing equivalent method of service to all other parties shall be deemed to be an ex parte communication.

(d) Proof of service. Proof of service must be maintained by the parties for all papers filed with the hearing officer. Proof of service must be in the form of an affidavit by the person effecting service, or in the form of a signed acknowledgement of receipt by the person receiving the papers. A written acknowledgment of service by the person served is adequate proof of service. Proof of service for papers served by electronic means, in addition to the foregoing, may be in the form of a record confirming delivery or acknowledging receipt of the electronic transmission.

§ 11-07 Docketing the Case at OATH

(a) Only cases referred to OATH must be docketed, and only the Board may docket a case at OATH.

(b) Following service of the petition upon the respondent, the Board shall docket a case by delivering to OATH a completed intake sheet, with a petition and appropriate proof of service of the petition.

(c) When a case is docketed, OATH shall place it on the trial calendar, the conference calendar, or on open status. Cases involving the same respondent(s) shall be scheduled for joint hearings or conferences. The case shall be given an index number and assigned to a hearing officer pursuant to OATH Rules of Practice § 1-26(b).

(d) After docketing the case at OATH and selecting a hearing or conference date, the Board shall serve a notice of hearing or notice of conference on the respondent within five business days. The notice shall be served by first class mail or electronic means, and appropriate proof of service shall be maintained by the Board.

(e) A conference or hearing shall be scheduled for a date that is at least two weeks after the date the answer must be served and filed.

(f) The administrative law judge may determine that the case is not ready for a conference or hearing and may adjourn the conference or hearing, or may remove the case from the conference or hearing calendar and place it on open status.

<u>§ 11-08 Disqualification of hearing officers. Motions for</u> disqualification of administrative law judges serving as hearing officers must be filed in accordance with OATH Rules of Practice § 1-27.

§ 11-09 Conferences

(a) Only cases referred to OATH are eligible for conferences.

(b) Conferences may be held for the formulation and simplification of issues, the possibility of obtaining admissions or stipulations of fact and of admissibility and authenticity of documents, the order of proof and of witnesses, discovery issues, legal issues, pre-hearing applications, scheduling, and settlement of the case.

(c) Conferences may be scheduled at the discretion of the administrative law judge, whether or not a case has been placed on the OATH conference calendar, on the application of either party or sua sponte. At the discretion of the administrative law judge, conferences may be conducted by telephone.

(d) All parties are required to attend conferences as scheduled unless timely application for additional time is made to the administrative law judge.

(e) Settlement conferences and agreements must be conducted in accordance with OATH Rules of Practice § 1-31.

(f) If the case is not settled at the conference, outstanding prehearing matters, including discovery issues, must be raised during the conference. If the case is not settled at the conference, a hearing date may be set. The parties must know their availability and the availability of their witnesses for a hearing.

§ 11-10 Notice of conference or hearing

(a) When a case is placed on either the conference calendar or the trial calendar, the Board shall serve the respondent(s) with notice in accordance with OATH Rules of Practice § 1-28.

(b) The notice of conference or hearing shall be served by first class mail or electronic means, and appropriate proof of service shall be maintained.

§ 11-11 Adjournments

Applications for adjournments of conferences or hearings before OATH shall be governed by OATH Rules of Practice § 1-32.

<u>§ 11-12 Discovery.</u> Discovery in OATH proceedings shall be conducted in accordance with OATH Rules of Practice § 1-33.

<u>§ 11-13 Pre-hearing motions.</u> Pre-hearing motions pursuant to OATH proceedings shall be filed in accordance with OATH Rules of Practice § 1-34.

§ 11-14 Appearances at OATH. Appearances at OATH shall comply with the requirements set forth in OATH Rules of Practice § 1-11.

<u>§ 11-15 Ex-parte communications.</u> No *ex parte* communications, other than those related to ministerial matters regarding a conference or hearing, shall be made to a hearing officer.

<u>§ 11-16 Role of the hearing officer. In the conduct of an adjudication, a hearing officer may:</u>

(a) administer oaths and affirmations, examine witnesses, rule upon offers of proof, receive written and oral testimony, and oversee and regulate discovery procedures;

(b) upon the request of any party, or subject to the hearing officer's discretion, subpoend the attendance of witnesses and the production of books, records, or other information;

(c) regulate the course of the hearing in accordance with section 11-20;

(d) dispose of procedural requests or similar matters;

(e) make findings of fact or decisions, determinations or orders, as authorized by law; and

(f) take any other action authorized by law or agency rule consistent therewith.

<u>\$ 11-17 Consolidation; separate hearings.</u> All or portions of separate cases may be consolidated for the hearing, or portions of a single case may be severed for separate hearings, in accordance with OATH Rules of Practice § 1-41.

§ 11-18 Witnesses and documents. The parties must have all of their witnesses available on the hearing date. A party intending to introduce documents into evidence must bring to the hearing copies of those documents for the hearing officer, the witness, and the other parties. Repeated failure to comply with this section may be cause for sanctions, as set forth in OATH Rules of Practice § 1-13(e).

§ 11-19 Subpoenas. Subpoenas may be issued in accordance with OATH Rules of Practice § 1-43.

§ 11-20 Order of proceedings. Testimony and argument on the law and facts shall be presented in the following order: petitioner; witnesses called by the petitioner, if any; cross-examination; the respondent(s); witnesses called by the respondent(s); crossexamination; and closing statements. Each party shall be afforded an opportunity to present rebuttal testimony, if deemed appropriate by the hearing officer. Closing statements, if any, shall be made first by petitioner. The order of proceedings may be modified at the discretion of the hearing officer.

<u>§ 11-21 Interpreters.</u> The hearing officer will make reasonable efforts to provide language assistance services to a party or its witnesses who are in need of an interpreter to communicate at a hearing or conference.

§ 11-22 Failure to appear. All parties, counsel and other representatives are required to be present at the hearing and prepared to proceed at the time scheduled for commencement of the hearing, subject to the requirements set forth in OATH Rules of Practice § 1-45.

§ **11-23 Evidence at the hearing.** Evidence shall be governed by OATH Rules of Practice §§ 1-46 and 1-47.

§ 11-24 Official notice

(a) In reaching a decision, the hearing officer may take official notice, before or after submission of the case for decision, on request of a party or sua sponte on notice to the parties, of any fact which may be judicially noticed by the courts of this state, pursuant to OATH Rules of Practice § 1-48(a).

(b) Official notice may be taken, without notice to the parties, of rules

published in the Rules of the City of New York or in The City Record, pursuant to OATH Rules of Practice § 1-48(b).

§ **11-25 Public access to proceedings.** Other than conferences, all proceedings shall be open to the public, in accordance with OATH Rules of Practice § 1-49.

§ 11-26 Hearing motions. Motions may be made during the hearing orally or in writing, in accordance with OATH Rules of Practice § 1-50.

<u>§ 11-27 Transcript.</u> Hearings shall be stenographically or electronically recorded, and the recordings shall be transcribed, in accordance with OATH Rules of Practice § 1-51.

<u>§ 11-28 Decision made on the record.</u> The hearing officer may conclude a case by making a report and recommendation on the record.

§ 11-29 Written comments. Once the hearing officer has issued a recommended final determination, each party shall have 20 days from issuance of the recommendation to submit written comments to the Board. The comments should raise any objections to the recommended determination, and objections not raised in the comments will be deemed waived in any further proceedings. Comments shall be limited to the record of the adjudicatory proceeding. Comments shall be served upon all other parties. Upon application filed with the Board's General Counsel, the General Counsel may shorten or extend the time for comments for good cause shown. No personal appearances shall be made before the Board unless the Board specifically requests that the parties appear.

§ 11-30 Final determination. The Board shall provide a final determination affirming, rejecting, or modifying the hearing officer's recommended final determination within 30 days of the conclusion of the written comments period, or as soon thereafter as is practicable. The final determination shall notify the candidate of the commencement of the four-month period during which a special proceeding may be brought to challenge the Board's determination pursuant to Article 78 of the Civil Practice Law and Rules. If the Board rejects or modifies the hearing officer's recommended final determination, the Board shall provide a written determination stating the basis for any assessed penalty or public funds determinations made by the Board pursuant to this chapter may not be appealed to the Board unless the Board specifically provides otherwise in its determination.

Chapter 12: Investigations and Complaints

§ **12-01 Investigations.** The Board may conduct an investigation into possible violations of the Charter, Act, or these rules. In its investigation, the Board may use its investigative powers pursuant to § 1052(a)(5) of the Charter and §§ 3-708(5) and 3-710(1) of the Code. An investigation may include, but is not limited to, field investigations, desk and field audits, the issuance of subpoenas, the taking of sworn testimony, the issuance of document requests and interrogatories, and other methods of information gathering.

§ 12-02 Complaints

(a) Filing a complaint. A complaint must be filed by hand delivery, mail, fax, or scanned and sent as an attachment via email to the Board's offices.

(b) Complaint requirements. Complaints must:

(i) be made in writing;

(ii) be sworn to or affirmed;

(iii) be based on personal knowledge, if possible, and if a complaint is based on information and belief, the complainant must state the source of that information and belief;

(iv) specify the conduct alleged to be in violation of the Charter, Act, or these rules, and to the extent known:

(A) the date(s) and time(s) of the conduct,

(B) the place(s) the conduct occurred, and

(C) the names of witnesses, if any.

(v) be accompanied by copies of all documentary evidence available to the complainant; and

(vi) contain the complainant's full name, current address, telephone number, and email address.

(c) Initial review of complaint

(i) If in compliance. Upon receipt, the complaint will be reviewed for substantial compliance with the above requirements. If the complaint is in compliance, the Board shall, within 10 days of receipt, send each party complained about a copy of the complaint.

(ii) Deficient complaints.

(A) If the complaint is moot, facially meritless, or not in substantial compliance, it may be rejected and the complainant so notified. (B) If the complaint is not in substantial compliance, the Board may investigate the subject matter of the complaint, but need not follow the procedural requirements of this chapter.

(d) Response to complaint; possible dismissal

(i) The party complained about may submit an answer to the complaint:

- (A) within 20 days of the date the complaint was sent by the Board, or
 - (B) within such lesser time as may be specified by the Board for complaints received less than 90 days before an election.

(ii) The answer must:

(A) be sworn to and affirmed,

- (B) set forth a response to the allegations contained in the complaint, including any reasons why the Board should dismiss the complaint in full or in part, if applicable, and
- (C) be accompanied by copies of all documentary evidence available to the respondent.

(iii) Based on the answer, a complaint may be dismissed as moot.

(e) Board action. Based on its review of the complaint and any answer filed, and of any investigation it chooses to conduct, the Board may:

(i) determine the complaint to be moot or lacking in merit and dismiss the complaint; or

(ii) determine that violations or infractions of the Charter, Act, or these rules have been substantiated, and initiate an enforcement proceeding pursuant to the procedures outlined in Chapter 10.

Chapter 13: Transition and Inauguration Entities

§ 13-01 Registration

(a) Before any private funds are raised or spent for transition or inauguration into office, the elected candidate must register a TIE with the Board; provided, however, that donations may be accepted prior to the registration of the TIE for the purpose of paying fees and depositing any minimum balance required to open a bank account for the TIE.

(b) The registration may be submitted at any time between the day after the general election and the due date of the first disclosure statement following the date of the candidate's election, at such time as the form is made available by the Board.

(c) The elected candidate must register no more than one TIE per election, and such TIE must have no more than one bank account. The elected candidate may not use any other bank accounts, including those of the candidate or any political committee.

(d) The registration must be submitted in such form and manner as shall be provided by the Board and must contain all information required by the Board, including: (i) the name and address of the elected candidate and of the TIE; (ii) the name, address, and employment information of the treasurer and of the liaison, if any; (iii) information about the TIE's bank account and merchant account; and (iv) any signatures and notarizations as may be required by the Board.

<u>§ 13-02 Periodic disclosure reports</u>

(a) Required information. Disclosure reports must contain all information for the reporting period required by the Board, including:

(i) the cash balance at the beginning and end of the reporting period;

(ii) total itemized and unitemized donations, loans, and other receipts accepted during the reporting period;

(iii) total itemized and unitemized expenditures made during the reporting period;

(iv) for each donation accepted in excess of \$99, the contributor's and intermediary's (if any) full name, residential address, occupation, employer, and business address;

(v) the date and amount of each donation accepted or other receipt;

(vi) the form of the donation (cash, check, money order, credit card, text, or other);

(vii) the number of the check or money order, if applicable;

(viii) the date and amount of each donation returned to a donor, the account from which the funds used to make the return originated, and the number of the bank or certified check used to issue the return of funds;

(ix) each previously reported donation for which the check was returned unpaid;

(x) for each loan accepted, the lender's, guarantor's or other obligor's full name, residential address, occupation, employer, and business address;

(xi) the date and amount of each loan, guarantee, or other security for a loan accepted;

(xii) for each loan repayment made, the date, amount, check number, name of bank account or credit card, and name of any third party payor;

(xiii) the date and amount of any portion of a loan which has been forgiven;

(xiv) the bill or invoice date and amount, name and address of the vendor or payee, the purpose code and explanation of each expenditure;

(xv) the date and amount of each payment;

(xvi) the payment method, including check number and committee bank account;

(xvii) the amount of remaining outstanding liability to the vendor or payee;

(xviii) for each in-kind donation, the date the donation was made, the name and residential address of the donor, a detailed description of the goods or services provided, the amount of the fair market value of the donation, and such further information and documentation as necessary to show how the fair market value of the donation was determined; and

(xix) such other information as the Board may require.

(xx) Expenditures of less than \$50 do not need to be separately itemized in a disclosure statement.

(b) All information reported in disclosure reports and amendments must be accurate as of the last day of the reporting period.

(c) Electronic submissions. Disclosure report submissions must follow the submission standards applicable to candidates under Chapter 4 of these rules.

(d) Monthly reports

(i) The first report is due on the fifth business day of the month immediately following the month during which the TIE first registers.

(ii) The first report must cover the period from the day of the first financial activity of the TIE through the last day of the calendar month immediately preceding the month in which the report is due. Each subsequent report shall cover the next consecutive calendar month, and shall be due on the fifth business day of the subsequent month.

(iii) The final report must cover the period from the day after_ the conclusion of the preceding report through the day the TIE terminates its activities and must be filed within five business days of such termination.

(iv) Disclosure reports that fail to comply substantially with the disclosure requirements described in the Act or these rules will not be accepted by the Board. Amendments to disclosure reports are prohibited unless expressly authorized or requested by the Board.

(v) The disclosure report must contain a verification from the elected candidate, treasurer, or other officer designated in the registration that the disclosure report is true and complete to the best of such elected candidate, treasurer, or other designee's knowledge, information, and belief, and must contain such signatures as may be required by the Board.

§ **13-03 Limits and requirements.** The following limits and requirements apply pursuant to § 3-801 of the Code.

(a) Donations and receipts

(i) An elected candidate must not: (A) authorize or register a political committee to raise or spend funds for transition or inauguration into office; (B) use funds accepted by a political committee authorized by the candidate for transition or inauguration into office; (C) authorize or register a previously existing entity to raise or spend funds for transition or inauguration into office; (D) continue a TIE in existence after it has paid all liabilities it incurred for transition and inauguration into office and otherwise has disposed of all funds pursuant to Paragraph (ii) of Subdivision (c) below; or (E) continue in existence an entity registered under section 13-01 after (1) April 30 in the year following the year of the election, or

(2) in the case of a special election, 60 days after inauguration. (ii) A TIE must not:

(A) accept donations or other receipts from any political committee authorized by the elected candidate;

(B) accept donations from a political committee that has not registered with the Board under § 3-801(3) of the Code on a form that includes the information set forth in Section 5-04(a) (political committees registered for an election cycle pursuant to § 3-707 of the Code shall be deemed registered for purposes of § 3-801(3) of the Code with respect to transition and inauguration expenditures immediately following an election held in that cycle);

(C) accept donations in excess of the donation limits set forth in § 3-801(2)(b) of the Code;

(D) accept cash or money order donations totaling more than \$100 from a single donor;

(E) accept any donations after all liabilities are paid;

(F) accept any donations from any corporation, limited liability company, limited liability partnership, or partnership;

(G) accept any donations from any individual or entity whose name appears in the doing business database as of the date of such donation; provided, however, that this limitation on donations shall not apply to any donation made by a natural person who has business dealings with the city where such donation is from the elected candidate, or from the elected candidate's parent, spouse, domestic partner, sibling, child, grandchild, aunt, uncle, cousin, niece or nephew by blood or by marriage; or

(H) commingle receipts in any account with personal or business funds, or funds accepted for an election.

(iii) An elected candidate may donate to such elected candidate's TIE with such elected candidate's personal funds or property, make in-kind donations to such elected candidate's TIEs with such elected candidate's personal funds or property, and make advances to such TIE with such elected candidate's personal funds or property, without regard to the donation limits of §3-801(2)(b) of the Code. An elected candidate's personal funds or property shall include such elected candidate's funds or property shall include with such elected candidate's spouse, domestic partner, or unemancipated children.

(iv) In determining compliance with the donation limits of § 3-801(2)(b) of the Code, the Board shall total all donations from a single source to the TIE, using the definition of "single source," and the factors for so determining, pursuant to sections 1-02 and 5-10(b).

(v) Any donation, loan, or other receipt received by a TIE after January 31 in the year following the date of a regularly scheduled general election shall be presumed to be for the next election in which the elected candidate is a candidate following the day that it is received, and not received for the purposes of transition or inauguration.

(vi) Loans are deemed to be donations, subject to the limits and restrictions of the Act, to the extent the loan is not repaid by, or is made after, the date of the elected candidate's inauguration into office.

(b) Expenditures

(i) Funds raised for a TIE may not be used for any purpose other than transition or inauguration expenses. Expenses related to the holding of office, or related to any past or future election, are prohibited. The following are examples of types of expenditures that are presumed to be TIE-related:

(A) Transition

- (1) Conferences and seminars related to city government and elected service;
- (2) Costs related to seeking and selecting city office staff;
- (3) Payroll and consulting fees directly related to transition; and
- (4) Costs related to fundraising to pay for transition expenses.

(B) Inauguration

(1) Payroll and consulting fees directly related to holding an inauguration event;

- (2) Printing and postage for inauguration event invitations;
- (3) Inauguration event site and equipment rental;
- (4) Inauguration event catering, decoration, entertainment, and photography; and
- (5) Costs related to fundraising to pay for inauguration event.

(ii) There is a rebuttable presumption that the purchase of durable goods is not TIE-related.

(iii) Transition liabilities may not be incurred nor expenditures made after the elected candidate assumes office, and inauguration liabilities may not be incurred nor expenditures made after January 31 in the year following the election, except for:

(A) expenditures made to satisfy liabilities incurred prior to January 31,

(B) expenditures related to fundraising to satisfy liabilities incurred prior to January 31, and

(C) routine and nominal expenditures associated with and necessary for satisfying such liabilities and terminating the entity, such as bank fees, taxes, and other reasonable expenses for compliance with applicable tax laws;

(iv) Unless permitted by paragraph (iii) of subdivision (b) above, any expenditure incurred after January 31 in the year following the date of a regularly scheduled general election by a TIE shall be presumed to be for the next election in which the elected candidate is a candidate following the day that it is incurred, and not incurred for the purposes of transition or inauguration.

(v) A TIE may maintain a petty cash fund of no more than \$500 out of which it may make disbursements not in excess of \$100 per transaction. The TIE must maintain a petty cash journal including the name of every individual or entity to whom any disbursement is made, as well as the date, amount, and purpose of the disbursement.

(vi) Single inaugural event. No more than one inaugural event shall be presumed to be TIE-related.

(c) Termination

(i) The TIE must be terminated after it has paid all liabilities and otherwise disposed of all funds pursuant to paragraph (ii) below and, in any event no later than April 30 in the year following the election.

(ii) If a TIE has funds remaining after all liabilities have been paid, it shall refund such funds to one or more of its donors, or if that is impracticable, to the Fund.

<u>§ 13-04 Records and audit</u>

(a) Each TIE must keep records that enable the Board to verify the accuracy of disclosure reports and compliance with all requirements of § 3-801 of the Code and this chapter. The records kept must be clear, accurate, and sufficient to demonstrate compliance. The records must be made and maintained contemporaneously with the transactions recorded, and maintained and organized in a manner that facilitates expeditious review by the Board.

(b) The records that must be kept include:

(i) Copies of all deposit slips;

(ii) A copy of each check or other monetary instrument, other than cash, representing a donation or other monetary receipt;

(iii) For each cash, credit card, or money order donation received, a written record containing (i) the donor's residential address, employer, business address, and occupation, (ii) the date and amount of the donation, (iii) if any, the intermediary's name, address, employer, business address, and occupation, and (iv) for cash and money order donations, the donor's signature;

(iv) For each in-kind donation, a receipt or other written record showing how the value of the donation was determined;

(v) Each contract, invoice, bill, and receipt for goods or services provided;

(vi) Written documentation for each loan received, loan repayment, and loan forgiven;

(vii) A monthly billing statement or customer receipt for each disbursement to a TIE credit card account showing vendors underlying the disbursement; and (viii) The following from banks and other depositories relating to accounts:

> (A) all periodic bank or other depository statements in chronological order, maintained with any other related correspondence received with those statements, such as credit and debit memos and donation checks returned because of insufficient funds; and

(B) the front and back of all returned and cancelled disbursement checks.

(c) All of the TIE's records are subject to Board review and must be made available to the Board upon its request, within such time as shall be specified by the Board.

(d) Each TIE must retain all records and documents required to be kept pursuant to Section 13-04 for five years after the date of its registration.

§ 13-05 Other provisions concerning TIEs. See Sections 9-02(c)(i) (Final bank balance); 10-03 (Enforcement); 15-07 (Special elections); Chapter 12 (Complaints).

Chapter 14: Disclosure of Independent Expenditures

§ 14-01 Definitions. Except as otherwise provided, the definitions set forth in section 1-02 apply in this chapter. In addition, the following terms mean:

"Authorized representative" means an officer or owner of an independent spender who is authorized to submit, and is jointly responsible with the spender for, independent expenditure reports.

"Ballot proposal" means a municipal ballot proposal, initiative, measure, or referendum. Activities associated with ballot proposals initiated by petition include the circulation of petitions to place the proposal on the ballot. Activities associated with ballot proposals not initiated by petition include those conducted after the measure or language is filed with the City Clerk.

"Clearly identified" means: (1) the name of the candidate or ballot proposal appears; (2) a photograph or drawing of the candidate appears; or (3) the identity of the candidate or ballot proposal is otherwise apparent by unambiguous reference.

"Contribution" means any monetary gift made to an entity. Contributions do not include membership dues paid by an individual or revenue from goods and services.

"Covered communication" means an express advocacy communication or electioneering communication.

"Covered election" means a covered election as defined in § 1052(a)(15)(a)(iii) of the Charter.

"Covered expenditure" means an expenditure directly related to the design, production, or distribution of a covered communication. An expenditure made during the ordinary conduct of business in connection with covering or carrying a news story, commentary, or editorial by any broadcasting station (including a cable television operator, programmer, or producer), website, newspaper, magazine, or other periodical publication, including any internet or electronic publication, is not a covered expenditure.

"Electioneering communication" means a communication that: (1) is disseminated by radio, television, cable, or satellite broadcast, a paid advertisement such as in a periodical or on a billboard, or a mass mailing; (2) is disseminated within 30 days of a billboard, or a mass mailing; (2) is disseminated within 30 days of a covered primary or special election, or within 60 days of a covered general election; and (3) refers to one or more clearly identified ballot proposals or candidates for a covered election. Electioneering communication shall not include a campaign-related communication made by an organization operating and remaining in good standing made by an organization operating and remaining in good standing under \$ 501(c)(3) of the Internal Revenue Code of 1986.

"Entity" means an entity as defined in § 1052(a)(15)(a)(ii) of the Charter.

"Express advocacy communication" means a communication that contains a phrase including, but not limited to, "vote for," "re-elect," "support," "cast your ballot for," "[Candidate] for [elected office]," "vote against," "defeat," "reject," or "sign the petition for," or a campaign slogan or words that in context and with limited reference to external events, such as the proximity to the election, reference to external events, such as the proximity to the election, can have no reasonable meaning other than to advocate the election, passage, or defeat of one or more clearly identified ballot proposals or candidates in a covered election, and is disseminated by: (1) radio, television, cable, or satellite broadcast; (2) telephone communication; (3) mass mailing; (4) other printed material; or (5) any other form of paid electoral advertising. Paid electoral advertising shall not include communications over the internet, except for: (1) communications placed for a fee on another individual or entity's website; or (2) websites formed primarily for, or whose primary purpose is, the election, passage, or defeat of a candidate in a covered election or of a ballot proposal.

"Independent expenditure" means an independent expendit 1052(a)(15)(a)(i) of the Charter. ure as defined in §

"Independent spender" means an individual or entity that makes an independent expenditure, or an agent of such individual or entity.

"Mass mailing" means a mailing by United States mail, common carrier, or facsimile of more than 500 pieces of mail matter of an identical or substantially similar nature within any 30-day period.

"Principal owner" means an individual or entity that owns or controls 10% or more of an entity, including stockholders and partners.

"Telephone communication" means 500 or more telephone calls, whether live or recorded, of an identical or substantially similar nature within any 30-day period.

14-02 Disclosure statements. Each disclosure statement must include the following information:

(a) Filer information

(i) All independent spenders must provide their:

(A) name, mailing address, telephone number, email address, and employer information;

(B) contact person's name, mailing address, email address, telephone number, and employer information; and

(C) other similar information that may be required by the Board.

(ii) Independent spenders that are entities must also provide their:

(A) website URL;

(B) authorized representative's name, mailing address, email address, telephone number, and employer information;

(C) type of organization;

(D) name(s) of principal owners, and employer information for any such owners who are individuals; and

(E) name and employer information of board members and officers or their equivalents.

(iii) Independent spenders that are individuals must also provide their occupation and employer information.

(b) Communications

(i) When an independent spender makes covered expenditures aggregating \$1,000 or more during an election cycle for communications that refer to a specific candidate or ballot proposal, it must report these communications and each future communication associated with an expenditure of \$100 or more that refers to that candidate or ballot proposal. Expenditures of less than \$100 shall not be covered expenditures for the purposes of this subdivision. Each communication must be disclosed in the reporting period in which it in first published gived on otherwise distributed which it is first published, aired, or otherwise distributed, <u>except that no communication is required to be disclosed</u> before the \$1,000 threshold has been reached. For each communication, the independent spender must provide:

(A) The type of communication;

(B) Its distribution date;

(C) The names of the candidates and/or ballot proposals referred to in the communication;

(D) For a printed communication, an electronic or paper copy of the communication as it was distributed to the public;

(E) For a broadcast or internet communication, an audio, video, or source file of the communication as it was distributed to the public, except that if a source file is not available for an audio communication then a script will be accepted; and

(F) Such other similar information as the Board may require.

(ii) A mass mailing shall be considered to have been distributed on the date on which it was postmarked, or accepted by a common carrier.

(c) Expenditures

(i) When a covered communication has been reported, each covered expenditure of \$100 or more associated with that

communication must be reported. If the communication refers to more than one candidate or ballot proposal, only those expenditures for candidates or ballot proposals that have met the \$1,000 threshold need be reported. Each expenditure must be disclosed in the reporting period in which the expenditure is incurred, except that no expenditure is required to be disclosed prior to the reporting of its associated communication. For each expenditure, the independent spender must provide:

(A) The names of the candidates or ballot proposals referred to by the associated communication;

(B) The name of the vendor;

(C) The date, amount, and purpose;

(D) The names of the individuals or entities paying for the expenditure, if not the independent spender;

(E) The itemized invoice detailing the expenditure, when it becomes available; and

(F) Such other similar information as the Board may require.

(ii) Valuation. All expenditures must be reported at their fair market value. If it is not practicable to obtain an invoice, the independent spender must use a reasonable estimate of fair market value and must provide a written record supporting the estimate.

(iii) Apportionment. For reporting purposes, an expenditure associated with a communication that refers to more than one candidate or ballot proposal shall be apportioned among the candidates or ballot proposals based on the following factors:

(A) The focus of the communication;

(B) The geographic distribution or location of the communication;

(C) The subject matter of the communication;

(D) The relative prominence of references to or the appearance of a candidate or ballot proposal in the communication, including the size and location of references to the candidate or ballot proposal and any photographs of the candidate;

(E) The timing of the communication; and

(F) Other circumstances of the communication.

(d) Contributions

(i) When an independent spender that is an entity makes covered expenditures of \$100 or more aggregating \$5,000 or more in the 12 months preceding the election for communications that refer to any single candidate, it is required to report:

> (A) All contributions accepted from other entities since the first day of the calendar year preceding the year of the covered election; and

(B) All contributions aggregating \$1,000 or more accepted from an individual during the 12 months preceding the election.

(ii) Each contribution must be disclosed in the reporting period in which it was received. For each contribution, the independent spender must provide:

> (A) For each contribution accepted from another entity, the entity's name, address, and type of organization and the names of the entity's principal owners, partners, board members and officers, or their equivalents, or, if no natural persons exist in any such role, the name of at least one natural person who exercises control over the activities of such entity;

> (B) For each entity from which contributions aggregating \$50,000 or more have been accepted in the 12 months preceding the election (the "major contributor"), (A) the name, address, and type of each entity that contributed \$25,000 or more to the major contributor in the 12 months preceding the election; and (B) the name, residential address, occupation, and employer of each individual who contributed \$25,000 or more to the major contributor in the 12 months preceding the election;

(C) For each contribution accepted from an individual, the individual's name, residential address, occupation, and employer;

(D) The date of receipt and amount of each such contribution accepted by the independent spender or the major contributor; and

(E) Such other similar information as the Board may require.

(iii) Exemption for earmarked contributions.

Contributions to independent spenders or major contributors that are earmarked for an election that is not a covered election, or for an explicitly stated non-electoral purpose, are not required to be reported; provided, however that records of such contributions to independent spenders must be maintained and may be requested by the Board to verify their qualification for this exemption.

(e) Verification. Each independent spender filing a disclosure statement must verify that each reported expenditure referring to a candidate was not authorized, requested, suggested, fostered, or cooperated in by such candidate, an agent of such candidate, an opponent of such candidate, or an agent of such candidate, independent spender must verify that the disclosure statement is true and complete to the best of the filer's knowledge, information, and belief. The disclosure statement must contain such signatures or notarizations and other verifications as may be required by the Board.

(f) Format. All disclosure statements must be submitted electronically as specified by the Board. Supporting documents must be submitted electronically, by hand, or by common carrier.

§ 14-03 Disclosure dates

(a) Filing dates. The Board shall publish a schedule of disclosure statement filing dates and reporting periods for primary and general elections, and for ballot proposals, based on the following:

> (i) Disclosure statements are due on January 15, March 15, May 15, and July 15 of the election year.

(ii) Additional disclosure statements are due:

(A) For a primary election: 32 and 11 days before and 10 days after the election.

(B) For a general election: 32 and 11 days before and 27 days after the election.

(iii) During the 14 days before a primary or general election, or a related runoff election, an independent spender must submit a disclosure statement to the Board within 24 hours of distributing any reportable communication, making any reportable expenditure, or receiving any reportable contribution.

(iv) The Board's published schedule of disclosure statement filing dates shall reflect that if a disclosure statement is due to be submitted on a Saturday, Sunday, or legal holiday, submission shall be considered timely if made on the next business day.

(v) An independent spender that has not distributed any reportable communications, made any reportable expenditures, or received any reportable contributions within a reporting period is not required to file a disclosure statement for that period.

(b) Reporting periods

(i) First disclosure statement

(A) The reporting period for the first disclosure statement containing communications or expenditures related to candidates in a primary or general election begins on the first day of the election cycle.

(B) The reporting period for the first disclosure statement for communications or expenditures related to ballot proposals begins on the first day of January in the year in which the proposal appears on the ballot.

(ii) Each subsequent disclosure statement. The reporting period for each subsequent disclosure statement begins on the third day before the deadline for the submission of the previous disclosure statement.

(iii) Conclusion of reporting period. The reporting period for all disclosure statements concludes on and include the fourth day before the deadline for the submission of that disclosure statement.

(iv) Weekends and holidays. The Board's published schedule of reporting period dates shall reflect that if a disclosure statement is due to be submitted on a Saturday, Sunday, or legal holiday, the reporting period will conclude on and include the fourth day before the next business day.

<u>§ 14-04 Identification of communications</u>

(a) Independent spender identification. When an independent spender makes covered expenditures of \$100 or more aggregating \$1,000 or more during an election cycle, the communication associated with the expenditure that meets the \$1,000 threshold and all subsequent communications, regardless of dollar value, must include:

(i) Printed material. For printed material, the words "Paid for by" must appear, followed by (i) the name of the independent spender; (ii) if the spender is an entity: (A) the name of any individual or entity that owns or controls more than 50% of the independent spender, (B) the name of the independent spender's chief executive officer or equivalent, if any, and (C) the independent spender's top donors as described in subdivision (b) of this section; and (iii) the words "Not expressly or otherwise authorized or requested by any candidate or the candidate's committee or agent. More information at nyc.gov/FollowTheMoney". Such words must appear in a conspicuous size and style and must be enclosed in a box within the borders of the communication.

(ii) Television, internet video, other video. For television, internet videos, or other types of video communications, the words "Paid for by" followed by the name of the independent spender must be clearly spoken at the beginning or end of the communication in a pitch and tone substantially similar to the rest of the communication. Additionally, simultaneous with the spoken disclosure, in a conspicuous size and style and enclosed in a box, the words "Paid for by" must appear followed by: (i) the name of the independent spender; (ii) if the spender is an entity, the spender's top donors as described in subdivision (b) of this section; and (iii) the words "Not expressly or otherwise authorized or requested by any candidate or the candidate's committee or agent. More information at nyc.gov/FollowTheMoney".

(iii) Radio, internet audio, automated telephone calls. For radio, internet audio, or automated telephone calls, the words "Paid for by" followed by (i) the name of the independent spender; (ii) if the spender is an entity, the spender's top donors as described in subdivision (b) of this section; and (iii) the words "Not expressly or otherwise authorized or requested by any candidate or the candidate's committee or agent. More information at nyc. gov/FollowTheMoney", must be clearly spoken at the end of the communication in a pitch and tone substantially similar to the rest of the communication. For radio and internet audio communications of 30 seconds in duration or shorter, subparagraph (ii) of this paragraph may be omitted.

(iv) Non-automated telephone calls longer than 10 seconds. For non-automated telephone calls lasting longer than 10 seconds, the words "This call is paid for by" followed by the name of the independent spender and the words "Not expressly or otherwise authorized or requested by any candidate or the candidate's committee or agent. More information is available at nyc.gov/FollowTheMoney" must be clearly spoken during the call in a pitch and tone substantially similar to the rest of the call.

(b) "Top donor(s)" identification. Identification of an independent spender's top donors in a communication shall be as follows:

(i) A spender's top donors are its largest aggregate contributors during the 12 months preceding the election, in descending order by aggregate amount, who have contributed at least \$5,000.

(ii) If there are at least three such donors:

(A) Printed identification shall be the words "Top Three Donors" followed by the names of such donors;

(B) Video identification shall be the words "The top three donors to the organization responsible for this advertisement are" followed by the names of such donors;

(C) Audio identification shall be the words "with funding provided by" followed by the names of such donors;

(D) If the third largest donor has donated the same amount as the fourth largest donor, the independent spender may choose which three donors to include, so long as no donor is included that has donated less than any other donor that is not included.

(iii) If there are only two such donors, the words "Top Donors" must replace "Top Three Donors."

(iv) If there is only one such donor, the words "Top Donor"

<u>must replace "Top Three Donors."</u>

(v) If there are no such donors, all references to donors must be removed from the identification.

(c) Languages other than English. All written or spoken identification required by this section must be in the primary language of the communication, except that the web address nyc. gov/Follow/TheMoney, if required to be written or spoken in the identification, must be in English.

(d) Modification. The requirements of this section may be modified by the Board concerning items upon which disclosures cannot be reasonably printed, pursuant to \$1052(a)(15)(c)(i) of the Charter or any other items whose disclosures are not otherwise provided for in \$1052(a)(15)(c) of the Charter.

(e) This section shall not apply to communications required to include a disclosure pursuant to § 3-703(16) of the Code.

§ 14-05 Non-independent expenditures. An expenditure by an individual or entity referring to a candidate, that was authorized, requested, suggested, fostered by, or cooperated in by such candidate or the opponent of such candidate, or their agents, is not an independent expenditure and is not governed by this Chapter. Factors used by the Board to determine whether an expenditure is independent or non-independent are referenced in section 6-04(a).

§ 14-06 Guidance for independent spenders. Prior to dissemination of a communication, an individual or entity may seek guidance regarding whether the communication would be subject to these rules by submitting a written request in the form and manner prescribed by the Board, including a description or sample of the proposed communication and a description of its proposed dissemination. If the request is received more than 14 days before the election, guidance shall be provided within seven days, and if such request is received during the 14 days prior to the election, guidance shall be provided within 24 hours. Such guidance applies only to the communication exactly as submitted and disseminated exactly as described, and constitutes neither approval nor disapproval of the content, appropriateness, or accuracy of the communication.

<u>**§ 14-07 Document retention**</u>. Independent spenders must maintain for three years after the date of the election to which they apply:

(a) records that enable the Board to verify the accuracy of disclosure statements, including bills for goods or services used to produce and disseminate a communication, as well as records of all contributions received by independent spenders that are entities; and

(b) copies of all advertisements, pamphlets, circulars, flyers, brochures, and other printed communications disseminated, and a schedule of all radio or television time purchased, and scripts used therein or the audio or video source files. Documents previously provided to and received by the Board do not need to be maintained by the independent spender.

§ **14-08 Penalties**. Any independent spender who violates any provision of this chapter, and any agent of an independent spender who commits such a violation, shall be subject to a civil penalty in an amount not in excess of \$10,000.

<u>§ 14-09 Other provisions concerning independent expenditures.</u> See sections 15-06 (Special elections); 16-07 (Runoff elections); 10-03 (Enforcement); Chapter 12 (Complaints).

Chapter 15: Special Elections

<u>§ 15-01 Special requirements and procedures.</u> If a special election to fill a vacancy is declared, the Board, after considering the date of the election and any other relevant factors, may provide for special requirements and procedures for candidates, including:

(a) a standard by which contributions, loans, and expenditures are presumed to be accepted or made for the special election, notwithstanding sections 5-10(a), 5-09(h), and 6-01(h)(i); and

(b) such other requirements and procedures as the Board may deem necessary to implement the provisions of the Act in the special election fully and effectively.

§ 15-02 Filing requirements

(a) <u>Certification.</u> Candidates who wish to participate in the Program for a special election must file a Certification with the Board specifically for such election. The Certification must be filed on or before the 14th day after the proclamation of such election and may not be rescinded.

(b) Filer Registration. Candidates who do not wish to participate in the Program are not required to file a Certification, but must file a Filer Registration specifically for the special election with the Board.

(c) Candidates raising funds for other elections within the same election cycle. Candidates who have already been raising funds for other elections occurring after the special election, but within the same election cycle, may use their existing committees and bank accounts in the special election, provided that they amend their

registration with the Board of Elections to reflect the change in election, and further provided that once such committee and account have been used for the special election, they must not be used for any future election, including any other elections during the same election cycle.

(d) **Disclosure statements**

(i) Post-election financial activity. Financial activity that occurred after a candidate's most recent election prior to a special election will be considered to be for such special election.

(ii) First disclosure statement. The first disclosure statement for a special election is due 32 days before the election unless otherwise provided by New York State Election Law. As provided pursuant to New York State Election Law, if the first disclosure statement for a special election is otherwise due within a period of five days of a required semi-annual disclosure statement, the candidate may file a single combined statement on the date on which the latter of the two separate statements is required to be filed.

> (A) The reporting period for the first disclosure statement shall conclude on the 36th day before the election, unless otherwise provided pursuant to New York State Election Law. If a candidate is already registered for a future election and an intervening special election is called, the candidate is required to re-register with the Board for the special election and resubmit all of the candidate's financial activity with such candidate's first special election disclosure statement.

(B) Submissions required by paragraph (ii) above shall cover the reporting periods of the missing disclosure.

(iii) Semi-annual disclosure statements

(A) Semi-annual disclosure statements are due on January 15 and July 15 in each year of the election cycle and on January 15 in the year after the election.

(B) Notwithstanding subparagraph (A) above:

(1) for candidates in a special election who proceed to raise or spend funds for the following primary or general election, the 27 day post-election disclosure statement described in paragraph (iv) shall be the last statement required for the special election; provided, however, that if there is a runoff special election, the semi-annual disclosure statement described in paragraph (iv) shall be the last disclosure statement required for all candidates in the special election who continue to raise or spend funds for the following primary or general election, regardless of whether they were candidates in the runoff special election; and

(2) for candidates in a special election who do not proceed to raise or spend funds for the following primary or general election, the first semi-annual disclosure statement due following the date of the special election shall be the last statement required for the special election; provided, however, that if the first semi-annual disclosure statement following the date of the special election is due less than 30 days after the deadline for filing the 27 day post-election disclosure statement, the second semi-annual disclosure statement after the date of the special election shall be the last statement required for the special election.

(iv) Daily disclosures during the two weeks preceding the election. If a candidate, during the 14 days preceding an election, accepts aggregate contributions and/or loans from a single source in excess of \$1,000 or makes aggregate expenditures to a single vendor in excess of \$20,000, the candidate must report, in a disclosure to the Board, all contributions and loans accepted from such source or expenditures made to such vendor during that 14-day period. The first such disclosure must be received by the Board within 24 hours after the contribution or loan that causes the total to exceed \$1,000 is accepted or the expenditure that causes the total to exceed \$20,000 is made. Each subsequent disclosure must be received by the Board within 24 hours after any additional contribution or loan is accepted or expenditure is made. Information reported in these disclosures must also be included in the candidate's next post-election disclosure statement.

(v) Post-runoff special election disclosure statements. For runoff special election candidates, post-election disclosure statements must be filed 27 days after the special election and on the first January 15 or July 15 following the election. Candidates in the special election must file both post-runoff special election disclosure statements, regardless of whether they were on the ballot for the runoff special election.

(vi) Disclosure statement amendments. The Board shall not make payments based on disclosure statement amendments filed after the final disclosure statement; provided, however, that the Board may make payments based upon such amendments solely if they are made in response to invalid matching claims reports to which the Board has requested a response after the final disclosure statement.

(vii) Bank statements. Each disclosure statement submitted by the candidate must include a copy of the most recent bank statements for the candidate's special election account.

(viii) Special election due dates for compliance with § 12-110 of the Code. If the deadline for filing financial disclosure reports with the conflicts of interest board pursuant to § 12-110(b)(2) of the Code is before the due date for the first disclosure statement required to be filed with the Board pursuant to section 15-02(d)(ii), the candidate's compliance with the requirements in § 12-110 of the Code shall be considered timely demonstrated to the Board if the Board receives confirmation of the candidate's compliance on or prior to the disclosure statement due date. If the deadline for filing financial disclosure statement required to be filed with the Board pursuant to section 15-02(d)(ii), the candidate's compliance with the conflicts of interest board pursuant to § 12-110(b)(2) of the Code is on or after the due date for the first disclosure statement required to be filed with the Board pursuant to section 15-02(d)(ii), the candidate's compliance with the requirements in § 12-110 of the Code shall be considered timely demonstrated to the Board if the Board receives confirmation of the candidate's compliance no later than one business day after the last day for filing disclosure reports with the conflicts of interest board.

<u>§ 15-03 Expenditures</u>

(a) Limit. The expenditure limit for a special election is the same as the limit for a primary or general election.

(b) Expenditures made after a candidate's most recent election and prior to a special election are presumed to be made for such special election.

(c) Expenditures are presumed to be subject to the special election expenditure limit on and after the date a special election was first reasonably anticipated, as determined by the Board. Expenditures incurred prior to the date such election was reasonably anticipated may also be presumed to be subject to the special election expenditure limit beginning when a candidate committee has already registered for the covered office that is the subject of such special election, or has begun raising or spending funds.

(d) Candidates may present evidence to the Board demonstrating the date a special election was first reasonably anticipated.

(e) Candidates who have received public funds for a special election may not make expenditures after such election, except for narrowly defined purposes as provided in section 9-02(c), until any required public funds repayments are made to the Board. Expenditures made after a special election by the special election committee are presumed to be for the next election and are covered by the expenditure limit applicable to such election, pursuant to section 6-01(h)(i).

(f) Qualified expenditures

(i) Public funds received for a special election to fill a vacancy may be used only for expenditures made by a candidate to further the candidate's nomination or election to fill such vacancy, and may not be used for any expenditure that is not qualified as defined in § 3-704 of the Code and Section 6-02(a). An expenditure made prior to the date on which a vacancy was declared shall be presumed not to be in furtherance of the campaign's nomination or election, unless the candidate provides evidence demonstrating that the vacancy was reasonably anticipated when the expenditure was made.

(ii) <u>It is presumed that the following bills for goods and</u> services are not qualified campaign expenditures:

(A) bills for a special election that are first reported in a disclosure statement submitted later than the first postelection disclosure statement applicable to that special election; and

(B) bills first reported in an amendment to a disclosure statement that is made after the special election.

§ 15-04 Contributions

(a) Limit. The contribution limit for a special election is one half of the contribution limit applicable to the office for which the special election is occurring.

(b) Contributions received after a candidate's most recent election and prior to a special election are presumed to be for such special election, except that contributions received on or before the date the special election was first reasonably anticipated, which do not exceed the applicable contribution limit, will not be presumed to be for the special election.

(c) For contributions received on or before the date the special election was first reasonably anticipated which are in excess of the contribution limit for the special election, the candidate must refund the excess portion to the contributor.

(d) Candidates may present evidence to the Board demonstrating the date a special election was first reasonably anticipated.

(e) Public funds. To receive public matching funds, candidates in a special election must meet the same threshold and eligibility requirements as candidates in a primary or general election. A candidate in a special election must respond to an invalid matching claims report no later than the deadline set by the Board.

(f) Bank account. Candidates must use a special election account.

(g) Receipts accepted for a special election may not be commingled in any account with receipts accepted for another election, with receipts accepted for a TIE, or with personal or business funds, and may not be used for any other election until the special election is actually held.

(h) Receipts accepted for an anticipated special election that is not held, or for a special election in which the candidate is not on the ballot, must be returned either to the contributors or to the Fund. If the candidate received public funds for the special election, receipts must be returned to the Fund, up to the amount received in public funds, before any receipts are returned to contributors.

§ **15-05 Training.** A candidate in a special election, or such candidate's representative, must attend a compliance training session designed specifically for such election. Such training must be completed on or before the financial disclosure cut-off date of the 11-day pre-election disclosure statement.

§ 15-06 Independent expenditure disclosure

(a) The reporting period for the first independent expenditure disclosure statement in a special election concerning candidates begins on the day the vacancy is declared.

(b) Disclosure statements are due 32 and 11 days before and 27 days after the election. If the first disclosure statement for a special election is otherwise due within a period of five days of a required disclosure statement, a single combined statement may be filed on the date on which the latter of the two separate statements is required to be filed.

(c) During the 14 days before the election, an independent spender must submit a disclosure statement to the Board within 24 hours of distributing any reportable communication, making any reportable expenditure, or receiving any reportable contribution.

(d) For ballot proposals anticipated to be voted on at a special election, additional disclosure statements are due on January 15 and July 15 of the year prior to the year of the election. The reporting period for the first disclosure statement begins on the date the first reportable expenditure is incurred.

§ 15-07 Transition and inauguration entities

(a) A special election TIE must not incur liabilities or make expenditures after 30 days following the candidate's inauguration, except for expenditures made to satisfy liabilities incurred within 30 days of inauguration, or expenditures related to fundraising to satisfy such liabilities.

(b) Unless permitted by subdivision (a) above, any donation or other receipt received after 30 days following the date of the elected candidate's inauguration by a TIE shall be presumed to be for the next election in which the elected candidate is a candidate following the day that it is received, and not received for the purposes of transition or inauguration.

(c) Unless permitted by subdivision (a) above, any expenditure incurred after 30 days following the date of the elected candidate's inauguration by a TIE shall be presumed to be for the next election in which the elected candidate is a candidate following the day that it is incurred, and not incurred for the purposes of transition or inauguration.

(d) The TIE must be terminated after it has paid all liabilities and otherwise disposed of all funds pursuant to the requirements set forth in this chapter and, in any event, no later than 60 days after inauguration.

Chapter 16: Runoff Elections

§ 16-01 Demonstrating that a runoff is reasonably anticipated

(a) A candidate may not accept contributions or make expenditures for a runoff election unless the Board has determined that a runoff election is reasonably anticipated. (b) A candidate may not accept contributions or make expenditures for a runoff election once the Board determines that it is no longer reasonable to anticipate such an election.

(c) In determining whether a runoff election is reasonably anticipated, the Board shall consider factors including, but not limited to:

n_	(i) the history of runoff primaries in a particular party for the office at issue;
ш.	(ii) media reports discussing the likelihood of a runoff primary;
	(iii) polling information;
	(iv) the proximity to the date of the primary;
	(v) whether an incumbent office holder is a candidate in the race;
	(vi) the number of candidates running;
	(vii) the amount of funds raised by the candidates; and
	(viii) endorsements made to the candidates in the race.

§ **16-02 Certification.** A candidate's Certification for a primary, general, or special election applies to any related runoff or runoff special election.

§ 16-03 Reporting

(a) Pre-election disclosure statements must be filed four days before a runoff election.

(b) Post-election disclosure statements must be filed 10 days after a runoff primary election.

(c) Each disclosure statement submitted by the candidate must include a copy of the most recent bank statements for the candidate's runoff election account.

§ **16-04 Expenditure limit.** The expenditure limit for a runoff election is one half the amount of the applicable limitation for the associated primary, general, or special election.

§ 16-05 Contributions

(a) Limit. The contribution limit for a reasonably anticipated runoff election is one half of the contribution limit applicable to the office for which the runoff election is occurring.

(b) Solicitation of contributions. Until a primary or special election is held that results in a runoff election, each solicitation of runoff election contributions must expressly state that such contributions are being solicited only for a runoff election that may not occur.

(c) No single contribution may be divided between runoff and nonrunoff accounts, nor may a single contribution be accepted in an amount that exceeds the limit applicable for the primary and general election, or runoff election, under § 3-703(1)(f) or (h) of the Act.

(d) Contribution documentation

(i) Each contribution must be documented and disclosed to the Board as a runoff contribution.

(ii) For each contribution from an individual contributor that the candidate deposits into a runoff bank account, the candidate must maintain a contribution card, which shall be provided to the Board upon request.

(1) Runoff account contribution cards must include all information required pursuant to Section 4-01(b)(ii)(B)

(2) Runoff account contribution cards must state, above the line for the contributor's signature: "I understand that this entire contribution will be used only for a runoff election. I hereby affirm that I was not, nor, to my knowledge, was anyone else, reimbursed in any manner for this contribution; that this contribution is not being made as a loan; and that this contribution is being made from my personal funds or my personal account, which has no corporate affiliation."

(e) Public funds payments

(i) Flat grants for run-off elections. Pursuant to § 3-705(5)(a) of the Code, the Board shall pay to a candidate in a runoff primary election or runoff special election an amount equal to 25 cents for each one dollar of public funds paid to the candidate for the preceding election.

(ii) Timing

(A)) The Board shall make payments pursuant to this subdivision within four business days after the date of the preceding election, or as soon thereafter as practicable.

(B) No public funds shall be paid to candidates in a runoff election until the Board has determined

that such runoff election is reasonably anticipated to occur.

(f) Receipts accepted for a runoff primary or special election may not be commingled in any account with receipts accepted for another election, with receipts accepted for a TIE, or with personal or business funds, and may not be used for any other election held in the year that the runoff primary election is held or anticipated.

(g) Receipts accepted for an anticipated runoff election that is not held, or for a runoff election in which the candidate is not on the ballot, must be returned either to the contributors or to the Fund, after the earlier of (i) the first January 12 after the date of the election for which the runoff election was anticipated or (ii) the date upon which all the candidate's liabilities from the election for which the runoff election was anticipated have been extinguished. If the candidate received public funds for the runoff election, receipts must be returned to the Fund, up to the amount received in public funds, before any receipts are returned to contributors.

§ 16-06 Runoff bank accounts

(a) Runoff primary elections

(i) Candidates must use a runoff primary bank account. No disbursements, withdrawals, or transfers may be made from the runoff account prior to the day of the preceding primary election, except that contributions may be returned to contributors until the candidate first receives public funds for the runoff election.

(ii) A runoff bank account must be disclosed to the Board in an amendment to a Certification or Filer Registration.

(iii) Exchange of funds. Funds may be exchanged between an account maintained for a primary or the general election and an account maintained for a runoff primary for the following purposes only:

(A) exchanges from a primary or general election account to a runoff primary account made after the primary election so that the funds transferred may be spent in the runoff primary; and

(B) exchanges from a runoff primary account to a primary or general election account made after the runoff primary is held so that the funds exchanged may be spent in the general election.

(b) Runoff special elections

(i) A runoff special election may be held for the office of mayor, as provided by \$ 10(c)(10) of the Charter.

(ii) Candidates must use a runoff special bank account. No disbursements, withdrawals, or transfers may be made from the runoff account prior to the day of the preceding special election, except that contributions may be returned to contributors until the candidate first receives public funds for the runoff election.

(iii) A runoff bank account must be disclosed to the Board in an amendment to a Certification or Filer Registration.

(iv) Exchange of funds. Funds may be exchanged between an account maintained for a special election and an account maintained for a runoff special for the following purposes only:

> (A) exchanges from a special election account to a runoff special account made after the special election so that the funds transferred may be spent in the runoff special; and

(B) exchanges from a runoff special account to a primary or general election account made after the runoff special election is held so that the funds exchanged may be spent in the primary or general election.

§ 16-07 Independent expenditure disclosure

(a) The reporting period for the first disclosure statement in a runoff election begins on the day the Board issues a determination stating that such runoff election is reasonably anticipated.

(b) For a runoff primary election, a disclosure statement is due 10 days after the election.

(c) For a runoff special election, a disclosure statement is due 27 days after the election.

(d) During the 14 days before a runoff primary or special election, an independent spender must submit a disclosure statement to the Board within 24 hours of distributing any reportable communication, making any reportable expenditure, or receiving any reportable contribution.

Chapter 17: Voter Education and Engagement

Except as otherwise provided, the definitions set forth in Section 1-02 apply in this chapter. In addition, the following terms mean:

"Ballot proposal" means any proposition, referendum, or other question submitted to New York City voters pursuant to the Charter, the New York Municipal Home Rule Law, or any other law.

"Candidate print statement" means the document filed by a candidate containing biographical and other information requested by the Board, and a photograph of the candidate, for inclusion in the printed or online primary or general election Voter Guide.

"Candidate video statement" means a video-recorded statement by the candidate for inclusion in the video and online edition(s) of the primary or general election Voter Guide.

"Election" means any primary or general election for the office of mayor, public advocate, comptroller, borough president, or Council member, or a general election in which a ballot proposal is on the ballot, and does not include any special election held to fill a vacancy, runoff primary election, runoff special election, or election held pursuant to court order.

"Registered candidate" means an individual who has registered or filed a Certification with the Board pursuant to Section 2-01 or 2-02 and § 3-703 of the Code.

§ 17-02 Contents of the Voter Guide

(a) Generally. In addition to any information that the Board determines to be useful for promoting public awareness of the voting process, city government, and the candidates and ballot proposals in an election, the printed and online Voter Guides for an election shall contain: (1) the date of the election; (2) the hours during which the polls will be open; (3) an explanation of the voter registration process, including deadlines to register for both the primary and general elections; (4) an explanation of how to cast a vote, including write-in votes; (6) information about the boundaries of City Council districts to aid voters in determining their appropriate district; and (7) tables of contents, graphics, and other materials which the Board determines will make the Voter Guide easier to understand or more useful for the average voter.

(b) Candidate statements

(i) <u>Candidate print statements</u>

(A) Candidate print statements contain the following biographical information:

(1) the name of the candidate;

(2) the political party, if any, in which the candidate is enrolled, and for which party lines the candidate's name will appear on the ballot;

(3) the previous and current public offices held by the candidate;

(4) the current occupation and employer of the candidate;

(5) prior employment and positions held by the candidate;

(6) the experience the candidate has had in public service;

(7) the educational background of the candidate;

(8) a list of the candidate's major organizational affiliations;

(9) information about the candidate's principles, platform, or views, in a form prescribed by the Board; and

(10) such other information as may be determined by the Board and requested of the candidate.

(B) The candidate print statement must be submitted in English.

(C) The photograph of the candidate submitted as part of a candidate print statement must:

(1) be a recent photograph;

(2) have a plain background;

(3) show only the face or the head, neck, and shoulders of the candidate;

(4) not include the hands or anything held in the hands of the candidate;

(5) not show the candidate wearing any distinctive uniform, including but not limited to a judicial robe, or a military, police, or fraternal uniform; and (6) comply with the size and resolution requirements as determined by the Board.

(D) Candidate print statements may not:

(1) refer to any opposing candidate by name;

(2) contain profanity or statements that are

patently offensive, obscene, libelous, or defamatory; (3) assert facts that the candidate knows or should

know to be false; or (4) violate any city, state, or federal law, including

(4) violate any city, state, or federal law, including regulations of the New York State Public Service Commission.

(E) A candidate print statement that violates any of the requirements outlined in this chapter, as determined by the Board at its sole discretion, shall not be included in the Voter Guide.

(F) Timing of submission

(1) In the election year, all registered candidates considering filing designating petitions must submit their complete and final print statements in accordance with a deadline set by the Board.

(2) A candidate not named in a filed designating petition who anticipates filing an independent nominating petition for the general election must submit a candidate print statement on or before the "independent candidates" submission deadline set by the Board.

(3) Complete and final print statements for the general election Voter Guide must be submitted prior to the publication of the primary election Voter Guide.

(ii) Candidate video statements

(A) Candidate video statements must contain information regarding the candidate's platform and candidacy, and may contain such other information as the candidate may choose; provided, however, that the candidate may not:

(1) refer to any opposing candidate by name;

(2) use profanity, or statements, gestures, or materials that are patently offensive, obscene, or pornographic;

(3) make statements that are slanderous, or defamatory, or assert facts that the candidate knows or should know to be false;

(4) engage in any commercial programming or advertising;

(5) display any literature, graphs, or props; or

(6) violate any City, State, or Federal law, including regulations of the New York State Public Service <u>Commission and the Federal Communications</u> <u>Commission</u>.

(B) Candidates recording video statements may dress as they choose and are responsible for their own clothing, make-up and hairdressing; provided, however, that when recording a video statement, candidates may not:

(1) engage in full or partial nudity;

(2) wear any distinctive uniform, including but not limited to a judicial robe, or a military, police, or fraternal uniform; or

(3) violate any city, state or federal law, including regulations of the New York State Public Service Commission and the Federal Communications Commission.

(C) To ensure that candidate scripts meet the requirements of this section, candidate video statement scripts must be submitted for Board approval in advance of the candidate's scheduled recording session, and on or before the script submission deadline set by the Board. Candidates must follow their approved video statement script during the recording. Recorded statements shall not be edited by the Board or any entity participating in the production of the video edition of the Voter Guide, except that candidate identification and other election information may be displayed.

(D) Only the candidate may appear on camera, and only the candidate may record a candidate video statement.

(E) Candidates shall be allowed to sit or stand while

recording statements. Reasonable accommodations for candidates with special needs shall be made.

(F) Video statements shall be recorded in English. Candidates may record a portion of their video statements in a language other than English; provided, however, that the script submitted for Board approval contains both the English and non-English text, and an English translation of all non-English text. No additional time will be allotted for statements recorded in multiple languages.

(G) Candidate video statements that violate any of the requirements outlined in this chapter shall not be included in the Voter Guide.

(H) Timing of candidate video statement recordings. In the election year, the recording schedule for candidates' video statements shall be provided to registered candidates in advance. Appointments for candidate video statement recordings shall be made at a time within the prescribed production schedule. A candidate who fails to appear at the scheduled time shall be deemed to have waived participation in the video edition of the Voter Guide.

(iii) Inclusion of candidate statements in Voter Guide editions

(A) Primary election edition. Candidate print and video statements shall be included in primary election editions of the Voter Guide only for registered candidates who have met the requirements set forth in this chapter and are on the ballot in a contested primary. In the case of printed editions of the Voter Guide, print statements of candidates anticipated to appear on the ballot in a contested primary on the date that the primary election print edition goes to press shall be included, based on the Board's assessment of information available from the Board of Elections.

(B) General election edition. Candidate print and video statements shall be included in general election editions of the Voter Guide only for registered candidates who have met the requirements set forth in this chapter and are on the general election ballot. Candidates who are seeking nomination or election exclusively as write-in candidates shall not be included in the Voter Guide. In the case of printed editions of the Voter Guide, print statements of candidates anticipated to appear on the general election ballot on the date that the general election print edition goes to press shall be included, based on the Board's assessment of information available from the Board of Elections. Candidates running unopposed in the general election shall be included in general election being covered is uncontested, in which case the Board shall not produce or distribute print or video editions of the Voter Guide, but shall produce an online Voter Guide.

(C) If a candidate in the general election was included in the primary election Voter Guide, then that candidate's primary election Voter Guide statement shall be included in the general election Voter Guide. No modifications or additions to the original statement shall be accepted.

(D) Candidates' print statements shall be included in the primary and general election online editions in accordance with the requirements set forth in Paragraphs (i), (ii), and (iii).

(E) The Board shall not accept a candidate print or video statement unless it is submitted in a manner provided by the Board, includes any signatures or notarizations as may be required by the Board, and the candidate has verified that the contents of the submission are true to the best of such candidate's knowledge. The Board may, at its discretion, reject any candidate print or video statement, or portions thereof, it deems to contain matter that is obscene, libelous, slanderous, defamatory, or otherwise in violation of these rules.

(iv) Candidate statements must not exceed the length and space limitations provided by the Board. The Board may, at its discretion, require that candidate print statements follow a consistent format, and edit statements to achieve uniformity of presentation, conformance with length and space limitations, and consistency with existing law. Candidate video statements that exceed their allotted statement time, as determined by the Board, shall be cut off at the time limit.

(v) A candidate print statement or video script is a written instrument which, when filed, becomes part of the Board's records. A candidate may not include any false information in such candidate's print statement or video script. The candidate must verify that such candidate's print statement and video script are true, to the best of such candidate's knowledge. Knowingly filing a written instrument that contains a false statement or false information is a Class A misdemeanor under New York State Penal Law § 175.30.

(vi) With each candidate print statement, the Board shall publish one of the following notices:

(A) In the case of a participant: "Participant in the Campaign Finance Program" or language to like effect.

(B) In the case of a non-participant: "Not a participant in the Campaign Finance Program" or language to like effect.

(c) Ballot proposals

(i) The print and online editions of the Voter Guide for a general election in which a City ballot proposal is anticipated to appear on the ballot shall contain: (A) the form of each ballot proposal as it will appear on the ballot in the general election; (B) a plain-language abstract of each ballot proposal; and (C) to the extent feasible, the major arguments for and against the passage of each ballot proposal, clearly labeled as such. If feasible, the Board shall solicit and accept from the public statements for and against passage of each ballot proposal for possible inclusion in the Voter Guide for the general election.

(ii) A public statement shall not be accepted by the Board unless it: (A) is submitted in a form and manner provided by the Board and includes any signatures required by the Board; (B) conforms to the length and space limitations provided by the Board; (C) identifies the organization, if any, on whose behalf the statement is made; and (D) clearly argues for or against passage of the proposal. No person may submit more than one statement per ballot proposal pursuant to this paragraph.

(iii) Board determines whether to publish statements for and against ballot proposals. With respect to statements for or against passage of ballot proposals, the Board, at its discretion, may determine: (A) not to publish any such statements; (B) not to publish any statement submitted pursuant to Paragraph (i) of Subdivision (c); (C) to publish any portion of a statement submitted pursuant to Paragraph (i) of Subdivision (c); and (D) to compose and publish such statements as it deems appropriate.

(iv) State ballot proposals. The Board shall include information about state ballot proposals in Voter Guides for a covered office or a city ballot proposal. At its discretion, the Board may produce an online Voter Guide to provide information about state ballot proposals during an election for which no print Voter Guide is produced.

§ 17-03 Voter Guide publication and distribution

(a) The Board shall publish printed Voter Guides in English and Spanish, and in such other languages as may be required by law. The Voter Guide shall be distributed by mail to each City household in which there is at least one registered voter eligible to vote in the primary or general election, as applicable.

(b) The Board shall produce an online Voter Guide in English and make the translated versions of the printed editions available online.

(c) The Board shall make all reasonable efforts to produce a video edition of the Voter Guide for citywide elections, and shall seek partners for the production, marketing, and broadcasting of video editions of the Voter Guide. The Board shall post online the scripts provided pursuant to Section 17-02(b)(ii)(C), along with translations of those scripts into Spanish and such other languages as may be required by law.

(d) Any conflicts related to the submission or public release of candidate print or video statements shall be decided by the Board.

(e) All decisions made by the Board with respect to any edition of the Voter Guide, including resolution of conflicts, are final.

(f) The Board retains ownership of, and distribution rights to, all Voter Guide content, including candidate statements. Unedited candidate statements may be republished or broadcast with the Board's permission.

§ **17-04 Elections not held as scheduled.** Notwithstanding any other provision of this chapter, the Board shall take such actions as are practicable to prepare, publish, and distribute a Voter Guide in a timely manner for an election that is not held as initially scheduled.

Chapter 18: Public Access to Information

§ 18-01 Records available to the public. The New York State Freedom of Information Law (FOIL) (Public Officers Law, Article 6, § 84 et seq.) governs public access to the Board's records. The Board may deny access to records or portions of records that are exempted from disclosure by state or federal law.

§ 18-02 Records access officer. The Board's records access officer is designated by the Executive Director and is responsible for ensuring appropriate agency response to public requests for access to records.

§ 18-03 Requesting records

(a) A candidate may request access to records such candidate submitted to the Board by contacting the Candidate Guidance and Policy Unit, which may, at its discretion, provide access to such records without a FOIL request.

(b) To request access to Board records, a member of the public must:

(i) make a written FOIL request in person, by mail, or by email, addressed to the Board's records access officer;

(ii) reasonably describe the records sought;

(iii) provide the requestor's name and mailing or email address; and

(iv) specify preference for inspection of records or copies of records.

(c) Within five business days of receipt of a FOIL request made in accordance with subdivision (b) above, the Board will:

(i) grant or deny the request, in whole or in part, in writing; or (ii) provide:

(A) a written acknowledgment of the request and state the approximate date on which the request will be granted or denied; or

(B) where circumstances prevent granting or denying the request within 20 business days of the written acknowledgment,

> (1) a written statement of the reasons for the delay in making a determination; and

(2) a date, within a reasonable period depending on the circumstances, when the request will be granted or denied.

(d) Where the Board is unable to locate records responsive to the request, the Board, upon request, will certify that:

(i) the Board is not the custodian of such records; or

(ii) such records cannot be found after a diligent search.

(e) Where the request is granted, the Board will:

(i) make records available for inspection:

(A) between the hours of 10:00 A.M. and 4:00 p.m., on business days, Monday through Friday;

(B) at the offices of the Board or another location chosen by the Board;

(C) in quantities that may be limited to the amount available at the time; and

(D) contingent on the requester's promise that the records will not be removed, damaged, marked, or changed in any way during the inspection; or

(ii) make copies of records available in the medium requested, where practicable, upon payment of fees as described in this Chapter, and provide, on request, a certification that the copies are true copies;

(f) Where a request is denied, the Board will explain the reasons for the denial in writing and set forth the right to appeal.

§ 18-04 Appealing a denial of access to records

(a) To appeal a denial of access to records, the requester must, within 30 days of the denial, submit a written appeal to the Board's General Counsel including:

(i) a copy of the original request;

(ii) a reasonable description of the records to which access was denied; and

(iii) the name and address of the requester.

(b) Upon receipt of an appeal, the Board's General Counsel shall, within 10 business days:

(i) decide the appeal and send a copy of the written decision to the requester; and

(ii) send a copy of the appeal and a copy of the written decision to the Committee on Open Government of the Department of State of the State of New York.

§ **18-05 Fees.** The Board may require payment for copies of records, as follows:

(a) 25 cents per page for photocopies not exceeding 8-1/2 inches by 14 inches; or

(b) the actual cost of reproducing any other record, in accordance with § 87 of the New York Public Officers Law.

THE CITY RECORD

NEW YORK CITY LAW DEPARTMENT DIVISION OF LEGAL COUNSEL 100 CHURCH STREET NEW YORK, NY 10007 (212) 356-4028

CERTIFICATION PURSUANT TO CHARTER §1043(d)

RULE TITLE: Amendments to Campaign Finance Program Rules **REFERENCE NUMBER:** 2018 RG 102 **RULEMAKING AGENCY:** Campaign Finance Board

I certify that this office has reviewed the above-referenced proposed rule as required by Section 1043(d) of the New York City Charter, and that the above-referenced proposed rule:

- is drafted so as to accomplish the purpose of the authorizing provisions of law;
- (ii) is not in conflict with other applicable rules;
- (iii) to the extent practicable and appropriate, is narrowly drawn to achieve its stated purpose; and
- (iv) to the extent practicable and appropriate, contains a statement of basis and purpose that provides a clear explanation of the rule and the requirements imposed by the rule.

/s/ STEVEN GOULDEN Acting Corporation Counsel Date: March 5, 2019

NEW YORK CITY MAYOR'S OFFICE OF OPERATIONS 253 BROADWAY, 10th FLOOR NEW YORK, NY 10007 (212) 788-1400

CERTIFICATION/ANALYSIS PURSUANT TO CHARTER SECTION 1043(d)

RULE TITLE: Amendments to Campaign Finance Program Rules **REFERENCE NUMBER:** CFB- 9 **RULEMAKING AGENCY:** Campaign Finance Board

I certify that this office has analyzed the proposed rule referenced

above as required by Section 1043(d) of the New York City Charter, and that the proposed rule referenced above:

- (i) Included notification of and review by other affected agencies;
- (ii) Is understandable and written in plain language for the discrete regulated community or communities;
- (iii) Minimizes compliance costs for the discrete regulated community or communities consistent with achieving the stated purpose of the rule; and

The rules in general provide cure periods/mechanisms where appropriate, depending in many cases on temporal proximity to the election or other relevant dates; the nature of the violation or defect; and other pertinent factors.

<u>Bonda Lee-Cunningham</u> March 7, 2019 Mayor's Office of Operations Date

Accessibility questions: Chandler Hart-McGonigle (212) 409-1745, chart-mcgonigle@nyccfb.info, by: Monday, June 10, 2019, 5:00 P.M.

3i 69

🖝 a19

FINANCE

■ NOTICE

NOTICE OF RULE MAKING

Pursuant to the power vested in me as Commissioner of Finance by Sections 1043 and 1054 of the New York City Charter, I hereby promulgate the amendment to rules relating to residential real property held in the cooperative or condominium form of ownership. The rule was published in the proposed form on December 27, 2018. A hearing for public comment was held on January 29, 2019.

Jacques Jiha Commissioner of Finance

S/S

STATEMENT OF BASIS AND PURPOSE

The purpose of this rule amendment is to allow for a waiver of the current requirement that in order to be eligible for a partial tax abatement of real property taxes for qualifying property held in the cooperative or condominium form of ownership, the qualifying property

must be owned by an individual. This requirement will be waived if the apartment or its shares are owned by a LLC, or LP, one or more of the partners or members is a law enforcement officer and he or she demonstrates that there is an imminent and/or ongoing security threat that necessitates ownership by such an entity.

The enabling legislation for partial tax abatement for residential real property held in the cooperative or condominium form is set forth in Real Property Tax Law Section 467-a, which was originally enacted by Chapter 273 of the Laws of 1996 and as amended by Chapter 4 of the Laws of 2013.

Matter <u>underlined</u> is new. Matter in brackets [] is to be deleted.

"Shall" and "must" denote mandatory requirements and may be used interchangeably in the rules of this department unless otherwise specified or unless the context clearly indicates otherwise.

AMENDMENT TO RULES RELATING TO RESIDENTIAL REAL PROPERTY HELD IN THE COOPERATIVE OR CONDOMINIUM FORM OF OWNERSHIP

Section 1. Subdivision (j-1) of Section 50-02 of Chapter 50 of Title 19 of the Rules of the City of New York is added to read as follows:

"Law enforcement officer" means anyone who is, or was, employed as a Federal, State or local judge, prosecutor, State or Local police or peace officer or Federal law enforcement officer as defined by the United States Code.

Section 2. Subdivision (k) of Section 50-02 of Chapter 50 of Title 19 of the Rules of the City of New York is amended to read as follows:

(k) "Owner" means the owner, in whole or in part, of a dwelling unit in real property held in the condominium form of ownership, or a tenantstockholder of a cooperative apartment corporation who owns, in whole or in part, a dwelling unit, as represented by his or her shares of stock in such cooperative apartment corporation. For purposes of these rules, with respect to any dwelling unit, or the shares representing a dwelling unit, held in trust solely for the benefit of a person or persons who would otherwise be eligible for an abatement pursuant to these rules were such person or persons the owner or owners of such dwelling unit, such person or persons are each deemed to be an "owner" of the dwelling unit. With respect to any dwelling unit, or the shares representing a dwelling unit, held in trust, the trustee or trustees of the trust are each deemed to be an "owner" of the dwelling unit. The holder or holders of a life estate in a dwelling unit are deemed to be "owner(s)" of the dwelling unit. An "owner" can only be an individual and cannot be a corporation, limited liability company, partnership or other entity, unless a <u>waiver is granted</u> pursuant to Subdivision (d) of Section 50-05 for a limited liability company or limited partnership.

Section 3. Subdivision (d) of Section 50-05 of Chapter 50 of Title 19 of the Rules of the City of New York is re-lettered as subdivision (e), and a new subdivision (d) is added, to read as follows:

(d) Waiver of requirement that dwelling unit be owned by an individual. (1) A limited liability company or limited partnership may submit an application for a waiver of the requirement that an owner of a dwelling unit must be an individual to be eligible for the abatement. The application will be in a form and format and by a deadline determined by the Commissioner. An applicant will be eligible for a waiver if the application demonstrates that:

- A. One or more of the partners or members is a law enforcement officer and there is an imminent or ongoing security concerns threat which necessitates ownership by a limited liability company or a limited partnership because disclosure of an individual's residence could reasonably put the individual in danger; and
- B. the dwelling unit is not used for commercial purposes; and
 C. the dwelling unit serves as the primary residence of one or more of the partners or members; and
- D. the partners or members that reside in the dwelling unit personally pay all of the cooperative maintenance fees, property taxes and other costs associated with the property's ownership.

(2) An application must be filed on or before December 1 of the current tax year in order to be eligible for the abatement in the succeeding tax year. In order to be eligible to receive the abatement in a given tax year a waiver must be granted and in effect as of the taxable status date applicable to the tax year. The approved waiver must be submitted by the applicant to their condominium or cooperative board so that it can be submitted with the abatement application.

(3) An approved waiver is in effect for a term of one year. The applicant must submit an annual waiver renewal application to the Commissioner for approval. The applicant will be eligible for a renewal of the waiver if the applicant demonstrates either that the security threat that existed at the time of the initial application is still in existence or that there is a different security threat which necessitates ownership by a limited liability company or limited partnership.

(4) The Commissioner may request additional information if the Commissioner deems such information relevant to an application or renewal application. Such additional information will be provided within sixty days of the request. The Commissioner may deny an application for a waiver. The Commissioner will inform the applicant of the reasons for the denial in a written notice and advise the applicant that it has the right to appeal the denial. The appeal must be submitted to the Commissioner or his or her designee within 15 business days of the mailing of the notice.

Section 3. Subdivision (c) of Section 50-06 of Chapter 50 of Title 19 of the Rules of the City of New York is amended to read as follows:

(c) Ownership of dwelling unit by entity other than an individual. Notwithstanding any other provision of these rules, for purposes of this chapter and § 467-a of the Real Property Tax Law, a dwelling unit can be the primary residence only of individuals, and cannot be the primary residence of a corporation, limited liability company, partnership or any other entity <u>unless a waiver is granted pursuant to</u> <u>Subdivision (d) of Section 50-05.</u>

🕶 a19

SPECIAL MATERIALS

CITYWIDE ADMINISTRATIVE SERVICES

■ NOTICE

			OFFICI	AL FUEL PRICE (\$) SCHEI FUEL OIL AND KEROS			
CONTR. NO.	ITEM NO.	FUEL/OIL TYPE		DELIVERY	VENDOR	CHANGE (\$)	PRICE (\$) EFF. 4/15/2019
3687331	1.0	#2DULS		CITYWIDE BY TW	SPRAGUE	.0534 GAL.	2.2749 GAL.
3687331	2.0	#2DULS		PICK-UP	SPRAGUE	.0534 GAL.	2.1702 GAL.
3687331	3.0	#2DULS	WINTERIZED	CITYWIDE BY TW	SPRAGUE	.0534 GAL.	2.4732 GAL.
3687331	4.0	#2DULS	WINTERIZED	PICK-UP	SPRAGUE	.0534 GAL.	2.3684 GAL.
3687331	5.0	#1DULS		CITYWIDE BY TW	SPRAGUE	.0536 GAL.	2.5871 GAL.
3687331	6.0	#1DULS		PICK-UP	SPRAGUE	.0536 GAL.	2.4823 GAL.
3687331	7.0	#2DULS	>=80%	CITYWIDE BY TW	SPRAGUE	.0534 GAL.	2.3027 GAL.
3687331	8.0	#2DULS	WINTERIZED	CITYWIDE BY TW	SPRAGUE	.0534 GAL.	2.5937 GAL.
3687331	9.0	B100	B100<=20%	CITYWIDE BY TW	SPRAGUE	.0384 GAL.	2.6184 GAL.
3687331	10.0	#2DULS	>=80%	PICK-UP	SPRAGUE	.0534 GAL.	2.1979 GAL.
3687331	11.0	#2DULS	WINTERIZED	PICK-UP	SPRAGUE	.0534 GAL.	2.4889 GAL.
3687331	12.0	B100	B100 <=20%	PICK-UP	SPRAGUE	.0384 GAL.	2.5136 GAL.
3687331	13.0	#1DULS	>=80%	CITYWIDE BY TW	SPRAGUE	.0536 GAL.	2.5967 GAL.
3687331	14.0	B100	B100 <=20%	CITYWIDE BY TW	SPRAGUE	.0384 GAL.	2.6273 GAL.
3687331	15.0	#1DULS	>=80%	PICK-UP	SPRAGUE	.0536 GAL.	2.4919 GAL.
3687331	16.0	B100	B100 <=20%	PICK-UP	SPRAGUE	.0384 GAL.	2.5225 GAL.
3687331	17.0	#2DULS		BARGE MTF III & ST. WI	SPRAGUE	.0534 GAL.	2.2355 GAL.
3687331	17.1	#2DULS	WINTERIZED	BARGE MTF III & ST. WI	SPRAGUE	.0534 GAL.	2.5721 GAL.
3687192	1.0	JET		FLOYD BENNETT	SPRAGUE	.0433 GAL.	2.8595 GAL.
3587289	2.0	#4B5		MANHATTAN	UNITED METRO	.0470 GAL.	2.3033 GAL.
3587289	5.0	#4B5		BRONX	UNITED METRO	.0470 GAL.	2.3021 GAL.
3587289	8.0	#4B5		BROOKLYN	UNITED METRO	.0470 GAL.	2.2963 GAL.
3587289	11.0	#4B5		QUEENS	UNITED METRO	.0470 GAL.	2.3016 GAL.
3587289	14.0	#4B5		RICHMOND	UNITED METRO	.0470 GAL.	2.3870 GAL.
3687007	1.0	#2B5		MANHATTAN	SPRAGUE	.0526 GAL.	2.2078 GAL.
3687007	4.0	#2B5		BRONX	SPRAGUE	.0526 GAL.	2.1968 GAL.
3687007	7.0	#2B5		BROOKLYN	SPRAGUE	.0526 GAL.	2.2135 GAL.
3687007	10.0	#2B5		QUEENS	SPRAGUE	.0526 GAL.	2.2097 GAL.
3687007	13.0	#2B5		RICHMOND	SPRAGUE	.0526 GAL.	2.3741 GAL.
3687007		#2B5		RACK PICK-UP	SPRAGUE	.0526 GAL.	2.1356 GAL.
3687007	16.0	#2B10		CITYWIDE BY TW	SPRAGUE	.0519 GAL.	2.3678 GAL.

OFFICIAL FUEL PRICE (\$) SCHEDULE NO 8320

FRIDAY, APRIL, 19, 2019

THE CITY RECORD

- 1	a	5	7
	J	υ	

3687007	17.0	#2B20		CITYWIDE BY TW	SPRAGUE	.0504 GAL.	2.3921 GAL.
3787198	18.0	#2DULS		CITYWIDE BY TW	SPRAGUE	.0534 GAL.	2.4851 GAL.
3787198	19.0	B100		CITYWIDE BY TW	SPRAGUE	.0384 GAL.	3.0229 GAL.
3787198	20.0	#2DULS		PICK-UP	SPRAGUE	.0534 GAL.	2.3304 GAL.
3787198	21.0	B100		PICK-UP	SPRAGUE	.0384 GAL.	2.8682 GAL.
3687331	#2D	ULSB5	95% ITEM 7.0 & 5% ITEM 9.0	CITYWIDE BY TW	SPRAGUE	.0527 GAL.	2.3185 GAL.(A)
3687331	#2D	ULSB10	90% ITEM 7.0 & 10% ITEM 9.0	CITYWIDE BY TW	SPRAGUE	.0519 GAL.	2.3343 GAL.(B)
3687331	#2D1	ULSB20	80% ITEM 7.0 & 20% ITEM 9.0	CITYWIDE BY TW	SPRAGUE	.0504 GAL.	2.3658 GAL.(C)
3687331	#2D	ULSB5	95% ITEM 10.0 & 5% ITEM 12.0	P/U	SPRAGUE	.0527 GAL.	2.2137 GAL.(D)
3687331	#2D1	ULSB10	90% ITEM 10.0 & 10% ITEM 12.0	P/U	SPRAGUE	.0519 GAL.	2.2295 GAL.(E)
3687331	#2D1	ULSB20	80% ITEM 10.0 & 20% ITEM 12.0	P/U	SPRAGUE	.0504 GAL.	2.2610 GAL.(F)
3687331	#1D	ULSB20	80% ITEM 13.0 & 20% ITEM 14.0	CITYWIDE BY TW	SPRAGUE	.0506 GAL.	2.6028 GAL.
3687331	#1D	ULSB20	80% ITEM 15.0 & 20% ITEM 16.0	PICK-UP	SPRAGUE	.0506 GAL.	2.4980 GAL.
3787198	#2D1	ULSB50	50% ITEM 18.0 & 50% ITEM 19.0	CITYWIDE BY TW	SPRAGUE	.0459 GAL.	2.7540 GAL.
3787198	#2D1	ULSB50	50% ITEM 20.0 & 50% ITEM 21.0	PICK-UP	SPRAGUE	.0459 GAL.	2.5993 GAL.
			OFFIC	IAL FUEL PRICE (\$) SC FUEL OIL, PRIME AN			
CONTR. NO.	ITEM NO.	FUEL/OIL TYPE		DELIVERY	VENDOR	CHANGE (\$)	PRICE (\$) EFF. 4/15/2019
3787250	1.0	#2B5		ERP - CITYWIDE	PACIFIC ENERGY	.0526 GAL.	
			OFFIC	TAL FUEL PRICE (\$) SCI FUEL OIL AND RE			
CONTR. NO.		FUEL/OIL TYPE		DELIVERY	VENDOR	CHANGE (\$)	PRICE (\$) EFF. 4/15/2019
3787250	1.0	#2B5		CITYWIDE BY TW	PACIFIC ENERGY	.0526 GAL	2.2657 GAL.
3787250	2.0	#4B5		CITYWIDE BY TW	PACIFIC ENERGY	.0470 GAL	2.2200 GAL.
			OFFIC	CIAL FUEL PRICE (\$) SCI GASOLINE			
CONTR. NO.	ITEM NO.	FUEL/OIL TYPE		DELIVERY	VENDOR	CHANGE (\$)	PRICE (\$) EFF. 4/15/2019
3787120		REG UL		CITYWIDE BY TW	GLOBAL MONTELLO	.0864 GAL	2.0237 GAL.

3787120	1.0	REG UL	CITYWIDE BY TW	GLOBAL MONTELLO	.0864	GAL	2.0237 GAL.
3787120	2.0	PREM UL	CITYWIDE BY TW	GLOBAL MONTELLO	.0964	GAL	2.1480 GAL.
3787120	3.0	REG UL	PICK-UP	GLOBAL MONTELLO	.0864	GAL	1.9587 GAL.
3787120	4.0	PREM UL	PICK-UP	GLOBAL MONTELLO	.0964	GAL	2.0830 GAL.
3787121	5.0	E85 (SUMMER)	CITYWIDE BY DELIVERY	UNITED METRO	.0311	GAL	1.9455 GAL.

NOTE:

1. (A), (B) and (C) Contract 3687331, item 7.0 replaced item 8.0 (Winter Version) effective April 1, 2019

2. (D), (E) and (F) Contract 3687331, item 10.0 replaced item 11.0 (Winter Version) effective April 1, 2019

3. Contract 3787121, item 5.0 replaced item 6.0 (Winter Blend) effective April 1, 2019

4. As of February 9, 2018, the Bio-Diesel Blender Tax Credit was retroactively reinstated for calendar year 2017. Should the tax credit be further extended, contractors will resume deducting the tax credit as a separate line item on invoices.

5. Federal excise taxes are imposed on taxable fuels, (i.e., gasoline, kerosene, and diesel), when removed from a taxable fuel terminal. This fuel excise tax does not include Leaking Underground Storage Tank (LUST) tax. LUST tax applies to motor fuels for both diesel and gasoline invoices. Going forward, LUST Tax will appear as an additional fee at the rate of \$0.001 per gallon and will be shown as a separate line item on your invoice.

6. The National Oilheat Research Alliance (NORA) resumed operations in 2014. A related assessment of \$.002 per gallon has been added to the posted weekly fuel prices and will appear as a separate line item on invoices. This fee applies to heating oil only and since 2015 has included #4 heating oil. NORA has been authorized through February 2019. All other terms and conditions remain unchanged.

REMINDER FOR ALL AGENCIES:

All entities utilizing DCAS fuel contracts are reminded to pay their invoices on time to avoid interruption of service. <u>Please send inspection copy of receiving report for all gasoline (E70, UL & PREM) delivered by tank wagon to OCP/Bureau of Quality Assurance</u> (BQA), 1 Centre Street, 18th Floor, New York, NY 10007.

🕶 a17

HOUSING PRESERVATION AND DEVELOPMENT

■ NOTICE

REQUEST FOR COMMENT REGARDING AN APPLICATION FOR A CERTIFICATION OF NO HARASSMENT PILOT PROGRAM

Notice Date: April 12, 2019

To: Occupants, Former Occupants, and Other Interested Parties

Property: <u>Address</u>	Application #	Inquiry Period
15 Locust Street, Brooklyn	19/19	March 4, 2014 to Present
498 Putnam Avenue, Brooklyn	33/19	March 8, 2014 to Present
693 Flatbush Avenue, Brooklyn	39/19	March 14, 2014 to Present
505 McDonald Avenue, Brooklyn	35/19	March 18, 2014 to Present
Authonitry Dilot Duoguom	Administrativa	Codo 897 9009 1

Authority: Pilot Program Administrative Code §27-2093.1, §28-505.3

Before the Department of Buildings can issue a permit for the alteration or demolition of a multiple dwelling on the Certification of No Harassment Pilot Program building list, the owner must obtain a "Certification of No Harassment" from the Department of Housing Preservation and Development ("HPD") stating that there has not been harassment of the building's lawful occupants during a specified time period. Harassment is conduct by an owner that is intended to cause, or does cause, residents to leave or otherwise surrender any of their legal occupancy rights. It can include, but is not limited to, failure to provide essential services (such as heat, water, gas, or electricity), illegally locking out building residents, starting frivolous lawsuits, and using threats or physical force.

The owner of the building identified above has applied for a Certification of No Harassment. If you have any comments or evidence of harassment at this building, please notify HPD, at CONH Unit, 100 Gold Street, 6th Floor, New York, NY 10038, by letter postmarked not later than 45 days from the date of this notice, or by an in-person statement made within the same period. To schedule an appointment for an in-person statement, please call (212) 863-5277 or (212) 863-8211.

For the decision on the Certification of No Harassment Final Determination please visit our website at www.hpd.nyc.gov or call (212) 863-8266.

PETICIÓN DE COMENTARIO SOBRE UNA SOLICITUD PARA UN CERTIFICACIÓN DE NO ACOSO PROGRAMA PILOTO

Fecha de notificacion:

Para: Inquilinos, Inquilinos Anteriores, y Otras Personas Interesadas

Propiedad: Dire	ección:	Solicitud #:	Período de consulta:
15 Locust Stree Brooklyn	et,	19/19	March 4, 2014 to Present
498 Putnam Av Brooklyn	zenue,	33/19	March 8, 2014 to Present
693 Flatbush A Brooklyn	wenue,	39/19	March 14, 2014 to Present
505 McDonald Brooklyn	Avenue,	35/19	March 18, 2014 to Present

Autoridad: PILOT, Código Administrativo §27-2093.1, §28-505.3

Antes de que el Departamento de Edificios pueda conceder un permiso para la alteración o demolición de una vivienda múltiple de ocupación de cuartos individuales, el propietario debe obtener una "Certificación de No Acoso" del Departamento de Preservación y Desarrollo de la Vivienda ("HPD") que indique que tiene no haber sido hostigado a los ocupantes legales del edificio durante un período de tiempo especificado. El acoso es una conducta por parte de un dueño de edificio que pretende causar, o causa, que los residentes se vayan o renuncien a cualquiera de sus derechos legales de ocupación. Puede incluir, entre otros, no proporcionar servicios esenciales (como calefacción, agua, gas o electricidad), bloquear ilegalmente a los residentes del edificio, iniciar demandas frívolas y utilizar amenazas o fuerza física.

El dueño del edificio identificado anteriormente ha solicitado una Certificación de No Acoso. Si tiene algún comentario o evidencia de acoso en este edificio, notifique a HPD al CONH Unit, 100 Gold Street, 6th Floor, New York, NY 10038 por carta con matasellos no mas tarde que 45 días después de la fecha de este aviso o por una declaración en persona realizada dentro del mismo período. Para hacer una cita para una declaración en persona, llame al (212) 863-5277 o (212) 863-8211.

Para conocer la decisión final sobre la Certificación de No Acoso, visite nuestra pagina web en *www.hpd.nyc.gov* **o** *llame al* (212) 863-8266.

a12-22

REQUEST FOR COMMENT REGARDING AN APPLICATION FOR A CERTIFICATION OF NO HARASSMENT

Notice Date: April 12, 2019

To: Occupants, Former Occupants, and Other Interested Parties

Property: <u>Address</u>	Application #	Inquiry Period
116 Bedford Avenue, Brooklyn	32/19	October 4, 2004 to Present
157 Wythe Avenue, Brooklyn	34/19	October 4, 2004 to Present
159 Wythe Avenue, Brooklyn	40/19	October 4, 2004 to Present

Authority: Greenpoint-Williamsburg Anti-Harassment Area, Zoning Resolution §§23-013, 93-90

Before the Department of Buildings can issue a permit for the alteration or demolition of a multiple dwelling in certain areas designated in the Zoning Resolution, the owner must obtain a "Certification of No Harassment" from the Department of Housing Preservation and Development ("HPD") stating that there has not been harassment of the building's lawful occupants during a specified time period. Harassment is conduct by an owner that is intended to cause, or does cause, residents to leave or otherwise surrender any of their legal occupancy rights. It can include, but is not limited to, failure to provide essential services (such as heat, water, gas, or electricity), illegally locking out building residents, starting frivolous lawsuits, and using threats or physical force.

The owner of the building identified above has applied for a Certification of No Harassment. If you have any comments or evidence of harassment at this building, please notify HPD, at **CONH Unit**, **100 Gold Street**, **6th Floor, New York, NY 10038** by letter postmarked not later than 30 days from the date of this notice or by an in-person statement made within the same period. To schedule an appointment for an in-person statement, please call (**212**) **863-5277 or (212) 863-8211**.

For the decision on the Certification of No Harassment Final Determination, please visit our website, at www.hpd.nyc.gov, or call (212) 863-8266.

PETICIÓN DE COMENTARIO SOBRE UNA SOLICITUD PARA UN CERTIFICACIÓN DE NO ACOSO

Fecha de notificacion: April 12, 2019

Para: Inquilinos, Inquilinos Anteriores, y Otras Personas Interesadas

Propiedad:	Dirección:	<u>Solicitud #:</u>	Período de consulta:
116 Bedf	ford Avenue,	32/19	October 4, 2004
Brooklyn	1		to Present
157 Wytł	ne Avenue,	34/19	October 4, 2004
Brooklyn	1		to Present
159 Wytł	ne Avenue,	40/19	October 4, 2004
Brooklyn	1		to Present

Autoridad: Greenpoint-Williamsburg Anti-Harassment Area, Código Administrativo Zoning Resolution §§23-013, 93-90

Antes de que el Departamento de Edificios pueda conceder un permiso para la alteración o demolición de una vivienda múltiple de ocupación de cuartos individuales, el propietario debe obtener una "Certificación de No Acoso" del Departamento de Preservación y Desarrollo de la Vivienda ("HPD") que indique que tiene no haber sido hostigado a los ocupantes legales del edificio durante un período de tiempo especificado. El acoso es una conducta por parte de un dueño de edificio que pretende causar, o causa, que los residentes se vayan o renuncien a cualquiera de sus derechos legales de ocupación. Puede incluir, entre otros, no proporcionar servicios esenciales (como calefacción, agua, gas o electricidad), bloquear ilegalmente a los residentes del edificio, iniciar demandas frívolas y utilizar amenazas o fuerza física.

El dueño del edificio identificado anteriormente ha solicitado una Certificación de No Acoso. Si tiene algún comentario o evidencia de acoso en este edificio, notifique a HPD al CONH Unit, 100 Gold Street, 6th Floor, New York, NY 10038 por carta con matasellos no mas tarde que 30 días después de la fecha de este aviso o por una declaración en persona realizada dentro del mismo período. Para hacer una cita para una declaración en persona, llame al (212) 863-5277 o (212) 863-8211.

Para conocer la decisión final sobre la Certificación de No Acoso, visite nuestra pagina web en *www.hpd.nyc.gov o llame al (212) 863-8266.*

a12-22

REQUEST FOR COMMENT REGARDING AN APPLICATION FOR A CERTIFICATION OF NO HARASSMENT

Notice Date: April 12, 2019

To: Occupants, Former Occupants, and Other Interested Parties

Property:	Address	Application #	Inquiry Period
147 We	st 130 th Street,	36/19	March 7, 2016
Manha	ttan		to Present
914 St.	Nicholas Avenue,	37/19	March 8, 2016
Manha	ttan		to Present
216 We	st 23 rd Street,	38/19	March 11, 2016
Manha	ttan		to Present
358 We	st 123 rd Street,	41/19	March 27, 2016
Manha	ttan		to Present
132 6 th	Avenue,	21/19	March 26, 2016
Brookly	yn		to Present
376 Ste	erling Place,	22/19	March 28, 2016
Brookly	yn		to Present
54 2 nd F Brookly		23/19	March 29, 2016 to Present
2028 W Bronx	alton Avenue,	20/19	March 4, 2016 to Present

Authority: SRO, Administrative Code §27-2093

Before the Department of Buildings can issue a permit for the alteration or demolition of a single room occupancy multiple dwelling, the owner must obtain a "Certification of No Harassment" from the Department of Housing Preservation and Development ("HPD") stating that there has not been harassment of the building's lawful occupants during a specified time period. Harassment is conduct by an owner that is intended to cause, or does cause, residents to leave or otherwise surrender any of their legal occupancy rights. It can include, but is not limited to, failure to provide essential services (such as heat, water, gas, or electricity), illegally locking out building residents, starting frivolous lawsuits, and using threats or physical force. The owner of the building identified above has applied for a Certification of No Harassment. If you have any comments or evidence of harassment at this building, please notify HPD, at **CONH Unit**, **100 Gold Street**, **6**th **Floor**, **New York**, **NY 10038**, by letter postmarked not later than 30 days from the date of this notice or by an in-person statement made within the same period. To schedule an appointment for an in-person statement, please call **(212) 863-5277** or **(212) 863-8211**.

For the decision on the Certification of No Harassment Final Determination, please visit our website, at www.hpd.nyc.gov, or call (212) 863-8266.

PETICIÓN DE COMENTARIO SOBRE UNA SOLICITUD PARA UN CERTIFICACIÓN DE NO ACOSO PROGRAMA PILOTO

Fecha de notificacion:

Para: Inquilinos, Inquilinos Anteriores, y Otras Personas Interesadas

Propiedad:	Dirección:	Solicitud #:	Período de consulta:
147 West Manhatt	an 130 th Street,	36/19	March 7, 2016 to Present
914 St. N Manhatt	licholas Avenue, an	37/19	March 8, 2016 to Present
216 West Manhatt	an Street,	38/19	March 11, 2016 to Present
358 West Manhatt	t 123 rd Street, an	41/19	March 27, 2016 to Present
132 6 th A Brooklyr		21/19	March 26, 2016 to Present
376 Ster Brooklyr	ling Place, 1	22/19	March 28, 2016 to Present
54 2 nd Pla Brooklyr		23/19	March 29, 2016 to Present
2028 Wa Bronx	lton Avenue,	20/19	March 4, 2016 to Present

Autoridad: SRO, Código Administrativo §27-2093

Antes de que el Departamento de Edificios pueda conceder un permiso para la alteración o demolición de una vivienda múltiple de ocupación de cuartos individuales, el propietario debe obtener una "Certificación de No Acoso" del Departamento de Preservación y Desarrollo de la Vivienda ("HPD") que indique que tiene no haber sido hostigado a los ocupantes legales del edificio durante un período de tiempo especificado. El acoso es una conducta por parte de un dueño de edificio que pretende causar, o causa, que los residentes se vayan o renuncien a cualquiera de sus derechos legales de ocupación. Puede incluir, entre otros, no proporcionar servicios esenciales (como calefacción, agua, gas o electricidad), bloquear ilegalmente a los residentes del edificio, iniciar demandas frívolas y utilizar amenazas o fuerza física.

El dueño del edificio identificado anteriormente ha solicitado una Certificación de No Acoso. Si tiene algún comentario o evidencia de acoso en este edificio, notifique a HPD al CONH Unit, 100 Gold Street, 6th Floor, New York, NY 10038 por carta con matasellos no mas tarde que 30 días después de la fecha de este aviso o por una declaración en persona realizada dentro del mismo período. Para hacer una cita para una declaración en persona, llame al (212) 863-5277 o (212) 863-8211.

Para conocer la decisión final sobre la Certificación de No Acoso, visite nuestra pagina web en *www.hpd.nyc.gov o llame al (212) 863-8266*.

a12-22

CHANGES IN PERSONNEL

				POLICE DEPA	CE DEPARTMENT			
			F	OR PERIOD ENDIN	G 03/22/19			
			TITLE					
NAME			NUM	SALARY	ACTION	PROV	EFF DATE	AGENCY
BLAIZE	DAVID	Ν	70235	\$88945.0000	PROMOTED	NO	03/05/19	056
BOLLMANN	LAURA	А	70210	\$85292.0000	RETIRED	NO	10/01/18	056
BONANNO	MICHAEL	J	7021A	\$97324.0000	DISMISSED	NO	01/26/19	056
BOUHAFA	HALA	L	31175	\$53190.0000	RESIGNED	YES	03/09/19	056
BRAND	MIGUEL	А	70235	\$109360.0000	PROMOTED	NO	03/11/19	056
BREITFELLER	BRIAN	J	70235	\$88945.0000	PROMOTED	NO	03/11/19	056
BRIGHTON	SHAVONNA	D	71012	\$38183.0000	APPOINTED	NO	02/28/19	056

BROWN BROWNE	DONALD M HEATHER L	70210 71012	\$85292.0000 \$38183.0000	RETIRED APPOINTED	NO NO	10/01/18 02/28/19	056 056	GRANGE GUERRERO PONCE	COLETTE NAITZET I	10124 71012	\$52503.0000 \$38183.0000	PROMOTED APPOINTED	NO NO	12/21/18 02/28/19	056 056
BROWNE	NOEL	71651	\$38625.0000	DECEASED	NO	02/28/19	056	GUEVARA	CAROL	10124	\$54990.0000	PROMOTED	NO	12/21/18	056
BRUNO	ALEXIS	70235	\$88945.0000	PROMOTED	NO	03/05/19	056	GUTIERREZ	CESARIO	60817	\$46737.0000	RETIRED	NO	03/14/19	056
BRUZZESE	SALVATOR V		\$42500.0000	RESIGNED	NO	03/08/19	056	GUZMAN	ROBEY	70210	\$85292.0000	RETIRED	NO	03/09/19	056
BUAH	BENSON	71012	\$38183.0000	APPOINTED	NO	02/28/19	056	GUZZI	VINCENT F	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
BUTLER JR CABRERA	PATRICK B PHIVELIN D	10144 71012	\$35330.0000 \$38183.0000	RESIGNED APPOINTED	YES NO	03/05/19 02/28/19	056 056	HALL HAMMOND	CYNDA A ASHLEY M	71012 71012	\$38183.0000 \$38183.0000	APPOINTED APPOINTED	NO NO	02/28/19 02/28/19	056 056
CAESAR	KIRT B	60817	\$34570.0000	RESIGNED	NO	01/04/19	056	HENDRICKS	KRYSEDA	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
CAMELO	OMAR J	70235	\$88945.0000	PROMOTED	NO	03/11/19	056	HENRY	CHERISE P	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
CAMPBELL	YVETTE M		\$40920.0000	RETIRED	NO	03/13/19	056	HILEYGAR	JAMECA N	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
CANCELA	MARISOL	70235	\$88945.0000	PROMOTED	NO	03/11/19	056	HILL	SHANA N	71012	\$38183.0000	APPOINTED	NO	02/28/19	
CARMONA CARRIERI	JULIA MICHAEL J	70210 70235	\$42500.0000 \$88945.0000	RESIGNED PROMOTED	NO NO	03/06/19 03/11/19	056 056	HINES HINES JR.	BRIANA C ALONZO J	70205 71012	\$15.0000 \$39453.0000	RESIGNED RESIGNED	YES NO	01/12/19 03/05/19	056 056
CARUSO JR.	MICHAEL	70235	\$42500.0000	RESIGNED	NO	01/18/19	056	HINES OK. HOFFMAN	DANIEL J	70210	\$42500.0000	RESIGNED	NO	03/16/19	056
CASSEUS	SAMUEL	70210	\$42500.0000	RESIGNED	NO	01/18/19	056	HORNICK	JAMES C	70210	\$85292.0000	RETIRED	NO	10/19/18	056
CHASE	NICY L	70235	\$88945.0000	PROMOTED	NO	03/11/19	056	HOSAN	MD S	71651	\$38625.0000	RESIGNED	NO	03/02/19	056
CHIANFAGNA	MARK	70235	\$88945.0000	PROMOTED	NO	03/11/19	056	HOSEIN	KEVIN	71651	\$31320.0000	RESIGNED	NO	03/12/19	056
CHICO CHIU	KARRINA YUI K	71012 13632	\$38183.0000	APPOINTED INCREASE	NO NO	02/28/19	056 056	HOWARD	WILLIAM A YOUSSEF G	70210 71105	\$85292.0000	RETIRED APPOINTED	NO YES	10/01/18	056 056
CHOWDHURY	YUI K ALI E	70235	\$91499.0000 \$88945.0000	PROMOTED	NO	02/03/19 03/11/19	056	IBRAHIM INGUI	JOSEPH A	70210	\$29789.0000 \$54394.0000	RESIGNED	NO	03/03/19 03/01/19	056
CHOWDHURY	MUAZ A	71012	\$38183.0000	APPOINTED	NO	02/28/19	056	INNISS	TRACEY A	71012	\$48293.0000	DISMISSED	NO	02/22/19	056
CHOWDHURY	MUHAMMAD N	71651	\$38625.0000	RESIGNED	NO	10/01/18	056	ISLAM	FARZANA	71651	\$30706.0000	RESIGNED	NO	10/03/17	056
CHOWDHURY	ZUL	70235	\$88945.0000	PROMOTED	NO	03/11/19	056	JACK	MAYA	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
			POLICE DEPA	DUNENU				JAMES-REID	BRIANNA S	71012 60817	\$38183.0000	APPOINTED	NO NO	02/28/19	056
		,	FOR PERIOD ENDIN					JANACK JARRETT	DHANESHW MARION J	60817	\$34570.0000 \$32426.0000	RESIGNED INCREASE	NO	03/01/19 01/23/19	056 056
		TITLE	OK I DAIOD DADIA	0 03/22/19				JEAN	JEAN H	60817	\$46737.0000	DISMISSED	NO	02/08/19	056
NAME		NUM	SALARY	ACTION	PROV	EFF DATE	AGENCY	JIMENEZ	YESSENIA T	70210	\$54394.0000	DISMISSED	NO	03/07/19	056
CHU	HOWARD G	70235	\$88945.0000	PROMOTED	NO	03/11/19	056	JIMENEZ-ROMERO	ANGELA E	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
CHU CHAN	UT SEONG	90644	\$39424.0000	DECEASED	YES	02/05/19	056	JONES	CRYSTAL S	10124	\$59883.0000	PROMOTED	NO	12/21/18	056
CHUMAN CIPOLLA	JESSICA BRYAN S	70206 70235	\$16.3100 \$88945.0000	RESIGNED PROMOTED	YES NO	03/13/19 03/11/19	056 056	JONES JUDGE	RACHEL S PAULINE A	71012 71012	\$38183.0000 \$50562.0000	APPOINTED DECEASED	NO NO	02/28/19 01/09/18	056 056
COMPRES	EMMY 5	70235	\$15.0000	RESIGNED	YES	02/26/19	056	KABIR	HUMAYUN H	70235	\$88945.0000	PROMOTED	NO	01/03/18	056
CONCEPCION	FRED	60820	\$70419.0000	RETIRED	NO	03/06/19	056	KALIL	MATTHEW A	70235	\$109360.0000	PROMOTED	NO	03/11/19	056
COOK	WILLIAM F	70260	\$125531.0000	RETIRED	NO	10/01/18	056	KELLY	JOSHUA V	60817	\$33498.0000	RESIGNED	NO	02/26/19	056
COOPER	BRITTANY	71012	\$38183.0000	APPOINTED	NO	02/28/19	056	KHAZIN	VALENTIN	70235	\$109360.0000	RESIGNED	NO	03/16/19	056
CORWISE	DESTINY	60817	\$46737.0000	DISMISSED	NO	02/22/19	056								
CREMIN CRUZ	WILLIAM ENMANUEL	70210 70235	\$85292.0000 \$88945.0000	RETIRED PROMOTED	NO NO	10/01/18 03/11/19	056 056			FOI	POLICE DEPA PERIOD ENDIN				
CUMMINGS	RICHARD P		\$53443.0000	DISMISSED	NO	02/14/19	056			TITLE	FERIOD ENDIN	G 03/22/19			
CURRAN	KIMBERLY L	70210	\$85292.0000	RETIRED	NO	10/01/18	056	NAME		NUM	SALARY	ACTION	PROV	EFF DATE	AGENCY
CUSH	SHAKEISH A	71012	\$38183.0000	APPOINTED	NO	02/28/19	056	KIM	HWAN J	70265	\$128212.0000	PROMOTED	NO	03/05/19	056
DAVIS	TONYA N		\$51700.0000	DISMISSED	NO	02/22/19	056	KOROLEV	ALEXANDE	71651	\$30706.0000	RESIGNED	NO	09/30/17	056
DEAZA DEFREITAS	NUVIA E RICHARD	70205 71012	\$15.0000 \$38183.0000	RESIGNED APPOINTED	YES NO	02/05/19 02/28/19	056 056	KOSAK LAAP	RACHAEL JAAN J	70265 70210	\$128212.0000 \$85292.0000	PROMOTED RETIRED	NO NO	03/05/19 10/01/18	056 056
DELACRUZ	ELVIS J	70210	\$85292.0000	RETIRED	NO	10/01/18	056	LAMBERSON	ALEXANDR M	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
DEMARINO	NICOLE J	70235	\$88945.0000	PROMOTED	NO	03/11/19	056	LAWRENCE	ROXANNE	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
DESFORGES	KERVENS	71012	\$38183.0000	APPOINTED	NO	02/28/19	056	LAWRENCE JR.	RICHARD P	70235	\$109360.0000	RETIRED	NO	10/01/18	056
DIAZ		70235	\$88945.0000	PROMOTED	NO	03/11/19	056	LEE	BRYANT	60817	\$32426.0000	RESIGNED	NO	02/28/19	056
DIAZ DIAZ	SALVADOR TIMOTHY	70235 70210	\$88945.0000 \$85292.0000	PROMOTED RETIRED	NO NO	03/11/19 10/01/18	056 056	LEONE LEWANDOWSKI	EILEEN F THOMAS S	10124 70235	\$62794.0000 \$88945.0000	PROMOTED PROMOTED	NO NO	12/21/18 03/11/19	056 056
DIEUDONNE JR.	GEDEON	70210	\$88945.0000	PROMOTED	NO	03/11/19	056	LEWIS-COUTAIN	SANDIE N	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
DOWNER	CHEVELLE L		\$38183.0000	APPOINTED	NO	02/28/19	056	LOPEZ	EILEEN J	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
DOWNING	EILEEN T	70265	\$128212.0000	PROMOTED	NO	03/05/19	056	LOPEZ	JOANNE S	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
DOYLE	WILLIAM C	70260	\$125531.0000	RETIRED	NO	10/01/18	056	LOWE	BRITTANY A	70235	\$88945.0000	PROMOTED	NO	03/11/19	056
DRAKE	ERIKA L	70235	\$88945.0000	PROMOTED	NO	03/11/19	056	LUGO	TERESA S	71651	\$38625.0000	INCREASE	NO	06/15/18	056
DURANTE DYER	PATRICE N CHRISTIN	71012 71012	\$38183.0000 \$38183.0000	APPOINTED APPOINTED	NO NO	02/28/19 02/28/19	056 056	LYNCH MACRI	WILLIAM G DOMENICO	70235 90735	\$88945.0000 \$296.7300	PROMOTED RESIGNED	NO YES	03/11/19 12/09/18	056 056
ELIBOX		71012	\$38183.0000	RESIGNED	NO	02/24/19	056	MAHMOOD	ASGHAR	70235	\$88945.0000	PROMOTED	NO	03/11/19	
ELLIS	ASHLEY K		\$15.0000	RESIGNED	YES	02/12/19	056	MAHONES	KEVIN M	70235	\$88945.0000	PROMOTED	NO	03/11/19	
EPHRAIM		71012	420102 0000	APPOINTED	NO	02/28/19		MALAMED		70210	\$85292.0000			10/01/18	
ESPINAL	EDWIN		\$38183.0000									RETIRED	NO		
EVANS FALCON		70235	\$109360.0000	PROMOTED	NO	03/11/19	056	MALLON	BRENDAN M	70235	\$109360.0000	PROMOTED	NO	03/11/19	
	TYRONE	70205	\$109360.0000 \$15.0000	PROMOTED RESIGNED	NO YES	03/11/19 02/14/19	056	MALLON MANGD	BRENDAN M SHLOMO M	70235 70235	\$109360.0000 \$88945.0000	PROMOTED PROMOTED	NO NO	03/11/19	056
FIELD	ANGELIQU	70205 7021A	\$109360.0000 \$15.0000 \$94489.0000	PROMOTED RESIGNED RETIRED	NO YES NO	03/11/19 02/14/19 10/01/18	056 056	MALLON MANGD MARQUEZ	BRENDAN M SHLOMO M WILMER I	70235 70235 70235	\$109360.0000 \$88945.0000 \$88945.0000	PROMOTED PROMOTED PROMOTED	NO NO NO	03/11/19 03/11/19	056 056
FIELD FIGUEROA JR		70205 7021A 70235	\$109360.0000 \$15.0000	PROMOTED RESIGNED	NO YES	03/11/19 02/14/19	056 056 056	MALLON MANGD	BRENDAN M SHLOMO M	70235 70235	\$109360.0000 \$88945.0000	PROMOTED PROMOTED	NO NO	03/11/19	056 056 056
FIGUEROA JR FINNEY	ANGELIQU BRADLEY L ANDRE M SHAREE E	70205 7021A 70235 70210 71651	\$109360.0000 \$15.0000 \$94489.0000 \$88945.0000 \$42500.0000 \$38625.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED	NO YES NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 02/19/19	056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ	BRENDAN M SHLOMO M WILMER I JHONNY O ANTONIO J ELIJAH	70235 70235 70235 71651 70235 71012	\$109360.0000 \$88945.0000 \$88945.0000 \$30706.0000 \$88945.0000 \$38183.0000	PROMOTED PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED	NO NO NO NO NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19	056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P	70205 7021A 70235 70210 71651 12627	\$109360.0000 \$15.0000 \$94489.0000 \$88945.0000 \$42500.0000 \$38625.0000 \$95543.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RETIRED	NO YES NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 02/19/19 03/02/19	056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ	BRENDANMSHLOMOMWILMERIJHONNYOANTONIOJELIJAHFERNANDO	70235 70235 70235 71651 70235 71012 7021A	\$109360.0000 \$88945.0000 \$30706.0000 \$88945.0000 \$88945.0000 \$38183.0000 \$94489.0000	PROMOTED PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RETIRED	NO NO NO NO NO NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 10/01/18	056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA	70205 7021A 70235 70210 71651 12627 71012	\$109360.0000 \$15.0000 \$94489.0000 \$42500.0000 \$38625.0000 \$35543.0000 \$38183.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RETIRED APPOINTED	NO YES NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 02/19/19 03/02/19 02/28/19	056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ	BRENDANMSHLOMOMWILMERIJHONNYOANTONIOJELIJAHFERNANDORUBENA	70235 70235 70235 71651 70235 71012 7021A 7021A	\$109360.0000 \$88945.0000 \$30706.0000 \$88945.0000 \$88945.0000 \$38183.0000 \$94489.0000 \$15.0000	PROMOTED PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RETIRED RESIGNED	NO NO NO NO NO NO YES	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 10/01/18 02/26/19	056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P	70205 7021A 70235 70210 71651 12627 71012 70235	\$109360.0000 \$15.0000 \$94489.0000 \$88945.0000 \$42500.0000 \$38625.0000 \$95543.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RETIRED	NO YES NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 02/19/19 03/02/19	056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ	BRENDANMSHLOMOMWILMERIJHONNYOANTONIOJELIJAHFERNANDO	70235 70235 70235 71651 70235 71012 7021A	\$109360.0000 \$88945.0000 \$30706.0000 \$88945.0000 \$88945.0000 \$38183.0000 \$94489.0000	PROMOTED PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RETIRED	NO NO NO NO NO NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 10/01/18	056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W	70205 7021A 70235 70210 71651 12627 71012 70235 71012	\$109360.0000 \$15.0000 \$84489.0000 \$88945.0000 \$38625.0000 \$38625.0000 \$38183.0000 \$88945.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RETIRED APPOINTED PROMOTED	NO YES NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 02/19/19 03/02/19 02/28/19 03/11/19	056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTHUKUTTY	BRENDAN M SHLOMO M WILMER I JHONNY O ANTONIO J ELIJAH FERNANDO RUBEN A SHINTO	70235 70235 70235 71651 70235 71012 7021A 7021A 70205 92508	\$109360.0000 \$88945.0000 \$8945.0000 \$88945.0000 \$88945.0000 \$38183.0000 \$94489.0000 \$15.0000 \$33872.0000	PROMOTED PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RETIRED RESIGNED APPOINTED	NO NO NO NO NO YES NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19	056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCIS FRANCOIS	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71012 71012 71651	\$109360.0000 \$15.0000 \$94489.0000 \$42500.0000 \$36625.0000 \$35543.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RETIRED APPOINTED APPOINTED APPOINTED INCREASE	NO YES NO NO NO NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 02/19/19 03/02/19 03/02/19 02/28/19 03/11/19 02/28/19 02/28/19 02/28/19 06/15/18	056 056 056 056 056 056 056 056 056 056	MALLON MARQUEZ MARTE MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MATHUKUTTY MATIAS MAXIMOS MCCARTHY	BRENDAN M SHLOMO M VILURER I JHONNY O ANTONIO J ELIJAH FERNANDO RUBEN A ALFONSO A MENA M JOHN W	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 70235	\$109360.0000 \$88945.0000 \$870706.0000 \$88945.0000 \$387000 \$38183.0000 \$15.0000 \$33872.0000 \$338183.0000 \$109360.0000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RETIRED RESIGNED RESIGNED RESIGNED REMOTED RETIRED	NO NO NO NO NO YES NO NO NO NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19 01/15/19 03/11/19 10/01/18	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCIS FRANCOIS FRANGATOS	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SPIROS	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71012 71651 70235	\$109360.0000 \$15.0000 \$94489.0000 \$42500.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38625.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RETIRED APPOINTED APPOINTED APPOINTED INCREASE PROMOTED	NO YES NO NO NO NO NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 02/19/19 03/02/19 03/02/19 03/11/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MATINEZ MATIAS MAXIMOS MCCARTHY MCCORMACK	BRENDAN M SHLOMO M WILMER I JHONNY J LIJAH J FERJANDO RUBEN ALFONSO A MENOSO A JOHN M JOHN M ANDREW T	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 70235 70235	\$109360.0000 \$88945.0000 \$88945.0000 \$88945.0000 \$30706.0000 \$3183.0000 \$15.0000 \$3183.0000 \$109360.0000 \$109360.0000 \$109360.0000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED RESIGNED RETIRED	NO NO NO NO NO YES NO NO NO NO NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19 03/03/19 03/15/19 03/11/19 10/01/18 10/01/18	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORTUNE FRANCIS FRANCGIS FRANCATOS FRANZEL	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SPIROS STEVEN L	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71012 71651 70235 70235	\$109360.0000 \$15.0000 \$94489.0000 \$38695.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED INCREASE PROMOTED PROMOTED	NO YES NO NO NO NO NO NO NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 02/19/19 02/28/19 03/11/19 02/28/19 02/28/19 02/28/19 06/15/18 03/11/19 03/11/19	056 056 056 056 056 056 056 056 056 056	MALLON MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTAS MAXIMOS MCCARTHY MCCORMACK MCDERMOTT	BRENDAN M SHLOMO M VILLMER J JHONNY O ANTONIO J ELIJAH FERNAND RUBEN A ALFONSO A MENA M JOHRAW W ANDRERW R	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 70235 7021D 70235	\$109360.0000 \$88945.0000 \$88945.0000 \$38945.0000 \$38183.0000 \$15.0000 \$34883.0000 \$34883.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$48945.0000	PROMOTED PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RETIRED PROMOTED	NO NO NO NO NO YES NO NO NO NO NO NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19 01/15/19 03/03/11/19	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCIS FRANCOIS FRANCOIS FRANCATOS FRANZEL FRAY	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SPIROS STEVEN L JUDINE	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71012 71651 70235 70235 71012	\$109360.0000 \$15.0000 \$94489.0000 \$42500.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38625.0000 \$38625.0000 \$38625.0000 \$109360.0000 \$38183.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED APPOINTED APPOINTED INCREASE PROMOTED PROMOTED APPOINTED	NO YES NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 02/19/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 03/11/19 03/11/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINES MCCARTHY MCCORMACK MCDERMOTT MCDONALD	BRENDAN M SHLOMO M WILMER I JHONNY O ANTONIO J FERNANC R RUBEN A SHINON J ALFONSO A JOHN W JOHN W JOHN W JOHN W ANFONSO K FURMAS K SEAWN V	70235 70235 70235 71651 70215 71012 7021A 70205 92508 71012 70235 7021D 70235 7021D	\$109360.0000 \$88945.0000 \$88945.0000 \$30706.0000 \$88945.0000 \$94489.0000 \$15.0000 \$33872.0000 \$38183.0000 \$109360.0000 \$109360.0000 \$94489.0000 \$38183.0000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED RETIRED RETIRED RETIRED PROMOTED APPOINTED	NO NO NO NO NO YES NO NO NO NO NO NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19 03/03/19 03/11/19 10/01/18 10/01/18 10/01/18 03/11/19 02/28/19	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCIS FRANCIS FRANGATOS FRANGATOS FRANZEL FRAY GALLETTA	ANGELIQU RRADLEY L SHAREE E JANDRE M SHAREE E JJCHN W LYNDONNA B PEGGYANN K KERN SPIROS STEVEN L JUDINE MICHAEL A	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71012 71651 70235 70235 71012 70235 71012	\$109360.0000 \$15.0000 \$94489.0000 \$42500.0000 \$36625.0000 \$36625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38625.0000 \$38625.0000 \$38625.0000 \$19360.0000 \$128212.0000	PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RETIRED APPOINTED INCREASE PROMOTED PROMOTED PROMOTED PROMOTED	NO YES NO NO NO NO NO NO NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 03/02/19 03/02/19 03/02/19 03/11/19 02/28/19 02/28/19 06/15/18 03/11/19 03/11/19 03/28/19 03/05/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MATHUKUTTY MATIAS MAXIMOS MCCARTHY MCCORMACK MCDERMOTT MCDONALD MCGREEVY	BRENDAN M SHLOMO M WILMER I JHONNY O ANTONIO J ELIJAH FERNANDO KUBEN A SHINTO A SHINTO A JOHN M JOHN M ANDREW T THOMAS R SEAWN V WILLIAM J	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 70235 70235 70235 70235 70235 70235	\$109360.0000 \$89945.0000 \$89945.0000 \$88945.0000 \$30706.0000 \$3183.0000 \$15.0000 \$33872.0000 \$33872.0000 \$33872.0000 \$109360.0000 \$109360.0000 \$40945.0000 \$4489.0000 \$4489.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.0000 \$4193.00000 \$4193.00000 \$419	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RETIRED PROMOTED APPOINTED RETIRED PROMOTED RETIRED	NO NO NO NO NO YES NO NO NO NO NO NO NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19 01/15/19 03/11/19 10/01/18 10/01/18 03/11/19 02/28/19 10/01/18	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCIS FRANCOIS FRANCOIS FRANCATOS FRANZEL FRAY	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SSIROS SSIROS STEVEN L JUDINE MICHAEL A Z MARIA	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71012 71651 70235 70235 71012	\$109360.0000 \$15.0000 \$94489.0000 \$42500.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38625.0000 \$38625.0000 \$38625.0000 \$109360.0000 \$38183.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED APPOINTED APPOINTED INCREASE PROMOTED PROMOTED APPOINTED	NO YES NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 02/19/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 03/11/19 03/11/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINES MCCARTHY MCCORMACK MCDERMOTT MCDONALD	BRENDAN M SHLOMO M WILMER I JHONNY O ANTONIO J FERNANC R RUBEN A SHINON J ALFONSO A JOHN W JOHN W JOHN W JOHN W ANFONSO K FURMAS K SEAWN V	70235 70235 70235 71651 70215 71012 7021A 70205 92508 71012 70235 7021D 70235 7021D	\$109360.0000 \$88945.0000 \$88945.0000 \$30706.0000 \$88945.0000 \$94489.0000 \$15.0000 \$33872.0000 \$38183.0000 \$109360.0000 \$109360.0000 \$94489.0000 \$38183.0000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED RETIRED RETIRED RETIRED PROMOTED APPOINTED	NO NO NO NO NO YES NO NO NO NO NO NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19 03/03/19 03/11/19 10/01/18 10/01/18 10/01/18 03/11/19 02/28/19	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCOIS FRANGATOS FRANZEL FRAY GALLETTA GARCIA-GONZALE:	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SSIROS SSIROS STEVEN L JUDINE MICHAEL A Z MARIA	70205 7021A 70235 70210 71651 12627 71012 71012 71012 71651 70235 70235 71012 70265 70205	\$109360.0000 \$15.0000 \$94489.0000 \$42500.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38625.0000 \$109360.0000 \$128212.0000 \$15.0000 \$11.9700	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED APPOINTED APPOINTED INCREASE PROMOTED PROMOTED PROMOTED RESIGNED RESIGNED	NO YES NO NO NO NO NO NO NO NO NO YES	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 03/12/19 03/02/19 03/02/19 02/28/19 02/28/19 02/28/19 03/11/19 03/11/19 03/05/19 02/26/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MATHUKUTTY MATIAS MAXIMOS MCCARTHY MCCORMACK MCDERMOTT MCGORALD MCGREEVY MCGURAN MCKENZIE MCMANUS	BRENDAN M SHLOMO M WILMER I JHONNY O AMTONIO J FERNANC K RUBEN A SHIFONS A MENA M JOHNN V THOMAS R SEANN V WILLIAM J BRENAL J STATLAM J STATLAM J STATA S STEVEN V	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 70235 70235 70235 70235 70235 70235 70210 70218 70218 70218 70212 70210	\$109360.0000 \$89945.0000 \$89945.0000 \$88945.0000 \$30706.0000 \$94489.0000 \$15.0000 \$38183.0000 \$109360.0000 \$109360.0000 \$40945.0000 \$48945.0000 \$38183.0000 \$16175.0000 \$88945.0000 \$38183.0000 \$885292.0000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RETIRED PROMOTED APPOINTED RETIRED PROMOTED APPOINTED RETIRED	NO NO NO NO VES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 09/27/17 02/28/19 10/01/18 02/26/19 03/03/19 03/11/19 03/11/19 03/11/19 02/28/19 10/01/18 03/11/19 02/28/19 10/01/18	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCOIS FRANGATOS FRANZEL FRAY GALLETTA GARCIA-GONZALE:	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SSIROS SSIROS STEVEN L JUDINE MICHAEL A Z MARIA	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71651 70235 70235 70235 70205 70205	\$109360.0000 \$15.0000 \$94489.0000 \$38625.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$109360.0000 \$11.9700 POLICE DEPA	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RETIRED APPOINTED APPOINTED INCREASE PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED RESIGNED RESIGNED	NO YES NO NO NO NO NO NO NO NO NO YES	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 03/12/19 03/02/19 03/02/19 02/28/19 02/28/19 02/28/19 03/11/19 03/11/19 03/05/19 02/26/19	056 056 056 056 056 056 056 056 056 056	MALLON MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MAXIMOS MCCORMACK MCDERMOTT MCCORMACK MCDERMOTT MCGREVY MCGURRAN MCGREAV MCGURAN MCGURAN MCMANUS MENDOZA	BRENDAN M SHLOMO M VILLMER I JHONNY 0 ANTONIO J ELIJAH F FERNANDO A SHINTO A ALFONSO A MENA M JOHN W SHADREW T THOMAS R SEANN V WILLIAM J BRENDAN J STEVEN V NANCY G	70235 70235 70235 71651 70235 71012 70215 92508 71012 70235 70210 70235 70210 70218 70218 70218 70218 70210 70210 70210	\$109360.0000 \$88945.0000 \$88945.0000 \$88945.0000 \$330706.0000 \$33872.0000 \$3483.0000 \$15.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4489.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.0000 \$4894.	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RETIRED PROMOTED APPOINTED PROMOTED APPOINTED RETIRED PROMOTED APPOINTED	NO NO NO NO NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 03/03/19 01/15/19 03/03/19 03/11/19 10/01/18 03/11/19 02/28/19 00/11/18 03/11/19 02/28/19 02/28/19	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCOIS FRANGATOS FRANZEL FRAY GALLETTA GARCIA-GONZALE:	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SSIROS SSIROS STEVEN L JUDINE MICHAEL A Z MARIA	70205 7021A 70235 70210 71651 12627 71012 71012 71012 71012 71012 71012 71012 71055 70235 70235 70205 70205	\$109360.0000 \$15.0000 \$94489.0000 \$42500.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38625.0000 \$109360.0000 \$128212.0000 \$15.0000 \$11.9700	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RETIRED APPOINTED APPOINTED INCREASE PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED RESIGNED RESIGNED	NO YES NO NO NO NO NO NO NO NO NO YES	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 03/12/19 03/02/19 03/02/19 02/28/19 02/28/19 02/28/19 03/11/19 03/11/19 03/05/19 02/26/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MCORMOTT MCDONALD MCGREEVY MCGURRAN MCKENZIE MENDOZA MERA	BRENDAN M SHLOMO M VILLMER O JHONNY O ANTONIO J ELIJAH N KUBEN A ALFONSO A ALFONSO M ANDREW T FERNAND V BRENDAN J NIA S STEVEN V NIACY G ANDREW T MOLDAND N SERENDAN J NIA S STEVEN V NANCY G	70235 70235 71251 70235 71651 70212 70213 70205 92508 71012 70235 71012 70235 71012 70235 71012 70235 71012 70210 70210 70210	\$109360.0000 \$88945.0000 \$88945.0000 \$38183.0000 \$38183.0000 \$15.0000 \$15.0000 \$13183.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$108183.0000 \$38183.0000 \$88945.0000 \$38183.0000 \$38183.0000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RETIRED PROMOTED APPOINTED RETIRED APPOINTED RETIRED APPOINTED RETIRED PROMOTED APPOINTED RESIGNED	NO NO NO NO NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19 01/15/19 03/03/19 01/15/19 03/11/19 02/28/19 10/01/18 03/11/19 02/28/19 10/01/18	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCIS FRANCOIS FRANCOIS FRANCATOS FRANZEL FRAY GALLETTA GARCIA-GONZALE: GAROFALO	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SSIROS SSIROS STEVEN L JUDINE MICHAEL A Z MARIA	70205 7021A 70235 70210 71651 12627 71012 71012 71012 71012 71012 71012 71012 71012 70235 70205 70205 70205	\$109360.0000 \$15.0000 \$94489.0000 \$42500.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$128212.0000 \$128212.0000 \$11.9700 POLICE DEPA FOR PERIOD ENDIN	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED APPOINTED APPOINTED INCREASE PROMOTED APPOINTED PROMOTED PROMOTED RESIGNED RESIGNED RESIGNED	NO YES NO NO NO NO NO NO NO NO NO YES	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 02/19/19 03/02/19 03/02/19 02/28/19 03/02/19 02/28/19 03/02/19 06/15/18 03/11/19 02/28/19 03/15/19 02/28/19 03/15/19 02/16/19 09/24/16	056 056 056 056 056 056 056 056 056 056	MALLON MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MAXIMOS MCCORMACK MCDERMOTT MCCORMACK MCDERMOTT MCGREVY MCGURRAN MCGREAV MCGURAN MCGURAN MCMANUS MENDOZA	BRENDAN M SHLOMO M WILMER I JHONNY O ANTONIO J RUBEN A SHINTO V ALFONSO A JOHN M JOHN M JOHN M JOHN M JOHN M JOHN M SEAWN V WILLIAM J BRENDAM M NIA S STEVEN V NACTON G STEPHEN A	70235 70235 70235 71651 70235 71012 70215 92508 71012 70235 70210 70235 70210 70218 70218 70218 70218 70210 70210 70210	\$109360.0000 \$89345.0000 \$88945.0000 \$30706.0000 \$3183.0000 \$94489.0000 \$3183.0000 \$3183.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$48945.0000 \$88945.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.00000 \$38183.00000 \$38183.00000 \$38183.00000 \$38183.0000000 \$38183.00000000000 \$38183.0000000000000000000000000000000000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED RETIRED RETIRED PROMOTED APPOINTED RETIRED PROMOTED APPOINTED RETIRED PROMOTED APPOINTED RESIGNED PROMOTED	NO NO NO NO NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 09/27/17 02/28/19 10/01/18 02/26/19 03/13/19 10/01/18 10/01/18 03/11/19 10/01/18 03/11/19 10/01/18 03/11/19 10/01/18 03/11/19 02/28/19 10/01/18	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCOIS FRANGATOS FRANZEL FRAY GALLETTA GARCIA-GONZALE:	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SSIROS SSIROS STEVEN L JUDINE MICHAEL A Z MARIA	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71012 71051 70235 70235 70205 70205 70205	\$109360.0000 \$15.0000 \$94489.0000 \$38625.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$109360.0000 \$11.9700 POLICE DEPA	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RETIRED APPOINTED APPOINTED INCREASE PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED RESIGNED RESIGNED	NO YES NO NO NO NO NO NO NO NO NO YES YES	03/11/19 02/14/19 10/01/18 03/11/19 03/13/19 02/19/19 03/02/19 02/28/19 03/02/19 02/28/19 02/28/19 06/15/18 03/11/19 02/28/19 03/11/19 03/11/19 02/28/19 03/15/19 02/26/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MATHUKUTTY MATIAS MAXIMOS MCCARTHY MCCORMACK MCDERMOTT MCDONALD MCGREEVY MCGURRAN MCKENZIE MCMANUS MENDOZA MERA MESZAROS	BRENDAN M SHLOMO M WILMER I JHONNY O ANTONIO J RUBEN A SHINTO V ALFONSO A JOHN M JOHN M JOHN M JOHN M JOHN M JOHN M SEAWN V WILLIAM J BRENDAM M NIA S STEVEN V NACTON G STEPHEN A	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 70215 70235 71012 70235 71012 70218 70218 70210 71012 70210 71012 70213	\$109360.0000 \$88945.0000 \$88945.0000 \$38183.0000 \$38183.0000 \$15.0000 \$15.0000 \$13183.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$108183.0000 \$38183.0000 \$88945.0000 \$38183.0000 \$38183.0000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RETIRED PROMOTED APPOINTED RETIRED APPOINTED RETIRED APPOINTED RETIRED PROMOTED APPOINTED RESIGNED	NO NO NO NO NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 09/27/17 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19 01/15/19 03/03/19 01/15/19 03/11/19 02/28/19 10/01/18 03/11/19 02/28/19 10/01/18	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATICK FOLEY FORD FORTUNE FRANCIS FRANCIS FRANCIS FRANCATOS FRANZEL FRAY GALLETTA GARCIA-GONZALE: GAROFALO	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN STEVEN L JUDINE MICHAEL A MYRIAM P NYEEMA K NICKEISH O	70205 7021A 70235 70210 71651 12627 71012 71012 71651 70235 71012 70235 70235 70235 70235 70235 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 700000000	\$109360.0000 \$15.0000 \$94489.0000 \$42500.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$128212.0000 \$128212.0000 \$128212.0000 \$11.9700 POLICE DEPA FOR PERIOD ENDIN SALARY \$39453.0000 \$38183.0000	PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED INCREASE PROMOTED APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED	NO YES NO NO NO NO NO NO NO NO NO YES YES PROV NO	03/11/19 02/14/19 10/01/18 03/11/19 03/11/19 02/19/19 03/02/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 02/28/19 02/26/19 02/26/19 02/26/19 02/26/19 02/24/16	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MATINEZ MATINEZ MATINEZ MATIAS MAXIMOS MCCARTHY MCCORMACK MCDERMOTT MCDONALD MCGREVY MCGUREAN MCKENZIE MCMANUS MENDOZA MERA MESZAROS MIAH MILLER JR MINIER	BRENDAN M SHLOMO M VILLMER O JHONNY O ATONIO J FERNAND A JLIJAN V ALFONS A JOHN W JOHN W JOHN W JOHN W JOHN W SEAWN V WILLIAM J SRENDANI S STEVEN V NIA S JAZMIN G STEPHEN A IHANBIR A BEMUNDIR A LEDUNDR I ALBERKO F	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 71012 70235 71012 70235 71012 70210 71012 70210 71012 70235 71012	\$109360.0000 \$88945.0000 \$88945.0000 \$30706.0000 \$38183.0000 \$15.0000 \$15.0000 \$138183.0000 \$109360.0000 \$109360.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED RESIGNED PROMOTED APPOINTED RETIRED PROMOTED APPOINTED RETIRED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED	NG NG NG NG VG VES NG NG NG NG NG NG NG NG NG NG NG NG NG	03/11/19 03/11/19 03/21/17 03/21/17 02/28/19 10/01/18 02/26/19 03/03/19 10/01/18 10/01/18 03/11/19 02/28/19 03/11/19 02/28/19 02/28/19 03/11/19 02/28/19 03/03/11/19	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FIINBEY FITZPATRICK FOLEY FORD FORTUNE FRANCIS FRANCGIS FRANCATOS FRANCEL FRAY GALLETTA GARCIA-GONZALE: GAROFALO NAME GENERAL GITTENS GLOVER	ANGELIQU BRADLEY L ANDRE E SHAREE E JANDRE P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SPIROS STEVEN L MICHAEL A MYRIAM P NYEEMA K NICKEISH O	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71012 70235 70235 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205	\$109360.0000 \$15.0000 \$94489.0000 \$42500.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$128212.0000 \$128212.0000 \$128212.0000 \$11.9700 POLICE DEPA FOR PERIOD ENDIN SALARY \$3453.0000 \$38183.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RESIGNED APPOINTED PROMOTED APPOINTED INCREASE PROMOTED PROMOTED RESIGNED RESIGNED RESIGNED ACTION RESIGNED APPOINTED APPOINTED	NO YES NO NO NO NO NO NO NO NO NO YES YES PROV	03/11/19 02/14/19 10/01/18 03/11/19 03/11/19 02/28/19 02/28/19 03/02/19 02/28/19 02/28/19 02/28/19 03/11/19 02/28/19 03/11/19 02/28/19 02/26/19 02/28/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTOS MCCARTHY MCCORMACK MCDERMOTT MCCONALD MCGREEVY MCGURAN MCGREEVY MCGURAN MCKENZIE MCMANUS MENDOZA MERA MESZAROS MIAH MILLER JR MINIER MOLONEY	BRENDAN M SHLOMO M WILMER I JHONNY O AMTONIO J RUBEN A SHINTON V RUBEN M JALFONSO A MENA M JOHONNY M SHINTO V RUBAN M SOBAR J BRENA M VILLIAM J BRENA M NIA S STEPLEN A TAMDRIC G STEPHEN A ILMENZO M LUBERTO J LUBERTO J JOSEPH B	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 70235 70235 70235 71012 70218 70218 70210 71012 71012 71012 70210 71012 70235 71012	\$109360.0000 \$89345.0000 \$89345.0000 \$88945.0000 \$30706.0000 \$94489.0000 \$15.0000 \$3872.0000 \$3872.0000 \$3872.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$48945.0000 \$88945.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000000 \$48945.00000 \$48945.00000000000000000000000000000000000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED RESIGNED RETIRED RETIRED PROMOTED APPOINTED RETIRED RETIRED PROMOTED APPOINTED RESIGNED APPOINTED APPOINTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NO NO NO NO NO VES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 09/27/17 02/28/19 10/01/18 02/26/19 03/03/19 01/15/19 03/11/19 03/11/19 02/28/19 10/01/18 03/11/19 02/28/19 10/01/18 02/28/19 02/28/19 03/11/19 02/28/19 03/11/19	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FIINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCIS FRANCIS FRANCATOS FRANCATOS FRANZEL FRAY GALLETTA GARCIA-GONZALE: GAROFALO MAME GENERAL GITTENS GLOVER GODDARD	ANGELIQU BRADLEY L ANDRE E SIAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN STEVEN L JUDINE MICHAEL A MICHAEL A NYEEMA K NICKEISH O MECCA	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71012 70235 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205	\$109360.0000 \$15.0000 \$94489.0000 \$38625.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$109360.0000 \$11.9700 POLICE DEPA FOR PERIOD ENDIN SALARY \$39453.0000 \$38183.0000 \$38183.0000	PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED PROMOTED PROMOTED PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED	NO YES NO NO NO NO NO NO NO NO NO NO YES YES <u>PROV</u> NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/02/19 02/19/19 02/28/19 02/28/19 02/28/19 03/11/19 02/28/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/15/18 03/11/19 03/15/18 03/11/19 03/15/18 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/22/18/19 03/22/18/19 03/22/18/19 03/22/18/19 03/22/18/19 03/22/18/19 03/22/18/19 03/22/18/19 03/22/18/19 03/22/18/19 03/22/18/19 03/22/18/19 03/22/18/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTAS MAXIMOS MCCARHY MCCORMACK MCDERMOTT MCGORAN MCKENZIE MCGURAN MCKENZIE MCGURAN MCKENZIE MENAOZA MERA MESZAROS MIAH MILLER JR MINLER MOLONEY MORGAN	BRENDAN M SHLOMO M VILLMER I JHONNY O ANTONIO J ELIJAH F RUBARN M ALFONSO A ALFONSO A ALAFONSO M JOHN W JULAN S SEAMON T BRENDAN J NILLIAN S STEVEN V NANCY G JAZMIN G STEVEN N THANEIR A EQMUND D JLESENTO F JOSEPH A JOSEPN A	70235 70235 71255 71651 70215 71012 7021A 70205 92508 71012 70235 71012 70235 71012 70215 70210 70235 71012 70215 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012	\$109360.0000 \$8945.0000 \$89545.0000 \$30706.0000 \$38183.0000 \$15.0000 \$3183.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$40945.0000 \$88945.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.00000 \$38183.00000 \$38183.00000 \$38183.00000 \$38183.0000000 \$38183.00000 \$38183.00000000000000000000000000000	PROMOTED PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RETIRED RETIRED RETIRED RETIRED RETIRED RESIGNED PROMOTED RESIGNED PROMOTED APPOINTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED	NC NC NC NC NC NC NC NC NC NC NC NC NC N	03/11/19 03/11/19 09/27/17 02/28/19 10/01/18 02/26/19 03/03/19 01/15/19 03/11/19 10/01/18 03/11/19 10/01/18 03/11/19 02/28/19 10/01/18 02/28/19 10/01/18 02/28/19 02/28/19 02/27/19 03/11/19 03/11/19 03/11/19	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATICK FOLEY FORD FORTUNE FRANCIS FRANCIS FRANCIS FRANCATOS FRANZEL FRAY GALLETTA GARCIA-GONZALE: GAROFALO MAME CENERAL GITTENS GLOVER GODDARD GOLDEERG	ANGELIQU BRADLEY L ANDRE E SIAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SSIROS S SSIROS S SSIROS S STEVEN L JUDINE MICHAEL A MICHAEL A NYRIAM P	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71012 70235 71012 70235 70235 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205	\$109360.0000 \$15.0000 \$4489.0000 \$4489.0000 \$42500.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$11.9700 POLICE DEPA FOR PERIOD ENDIN SALARY \$39453.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000	PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED APPOINTED PROMOTED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED	NO YES NO NO NO NO NO NO NO NO YES YES PROV NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/11/19 02/19/19 03/02/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTAS MAXIMOS MCCARTHY MCCORMACK MCDERMOTT MCDONALD MCGREVY MCGURAN MCKENZIE MCMANUS MENDOZA MERA MESZAROS MIAH MILLER JR MINIER MONGGAN MURHEAD	BRENDAN M SHLOMO M WILLMER O JHONNY O LILJAH J RUBEN A KUBEN A JLFONSO A AMENA M JANDREW Y ANDREW Y SELIJAH W JANDREW Y BRENDAN Y NIA S STEPHEN J JAZMINE A THANAS Y STEPHEN J JAZMINE A EDMUND H JUBERTO J JALBERTO H JOSEFH B EARL T	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70210 71012 70235 71012 70235 71012 70235 71012	\$109360.0000 \$88945.0000 \$38945.0000 \$38183.0000 \$38183.0000 \$15.0000 \$15.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.00000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000000000000000000000000000000000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED APPOINTED PROMOTED APPOINTED RETIRED PROMOTED APPOINTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED	NC NC NC NC NC VES NC NC NC NC NC NC NC NC NC NC NC NC NC	03/11/19 03/11/19 03/21/17 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19 03/11/19 02/28/19 10/01/18 03/11/19 02/28/19 03/11/19 02/28/19 03/11/19 02/28/19 03/11/19 03/03/11/19 03/03/11/19	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCIS FRANCIS FRANCIS FRANCEL FRAY GALLETTA GARCIA-GONZALE: GAROFALO NAME GENERAL GITTENS GLOVER GODDARD GODDARD GODDERG GOMEZ	ANGELIQU RRADLEY L ANDRE E SHAREE E JANDRE P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SPIROS STEVEN L MICHAEL A MYRIAM P NYEEMA K NICKEISH O MECCA JASON A JENIFER	70205 7021A 70235 70210 71651 12627 71012 71012 71012 70235 70235 70235 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 7005 70	\$109360.0000 \$15.0000 \$94489.0000 \$4489.0000 \$42500.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$128212.0000 \$128212.0000 \$11.9700 POLICE DEPA FOR PERIOD ENDIN SALARY \$39453.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED INCREASE PROMOTED APPOINTED PROMOTED RESIGNED RESIGNED ACTION RESIGNED APPOINTED APPOINTED APPOINTED APPOINTED DECEASED	NO YES NO NO NO NO NO NO NO NO NO NO YES YES <u>PROV</u> NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/11/19 02/19/19 03/02/19 02/28/19 03/02/19 06/15/18 03/02/19 06/15/18 03/11/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MATINEZ MATINEZ MATINEZ MAXIMOS MCCARTHY MCCORMACK MCDERMOTT MCDONALD MCGREEVY MCGURRAN MCKENZIE MCMANUS MENDOZA MERA MESZAROS MIAH MILLER JR MINLER MOLONEY MOCGAN MURREAD MURREAD MURREAD MURREAD	BRENDAN M SHLOMO M WILMER I JHONNY J LELIJAH J RUBEN A SHINTO J RUBEN M JALFONSO A JUHONNY M JALFONSO A JOHN W JOHN W JOHN M JUANIN M JAZMIN A LBERTO F JOSEPH B EARL M FRANK M	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 70215 70235 71012 70235 71012 70218 70210 71012 70210 71012 70235 71012 70235 71012 70235 71012 70235 70210	\$109360.0000 \$89345.0000 \$88945.0000 \$30706.0000 \$38945.0000 \$31705.0000 \$348340.0000 \$3483.0000 \$34183.0000 \$409360.0000 \$409360.0000 \$409360.0000 \$4109360.0000 \$48945.0000 \$48945.0000 \$34183.0000 \$48292.0000 \$34183.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.00000 \$48945.00000 \$48945.00000 \$48945.00000 \$48945.000000000000000000000000000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED RETIRED RETIRED PROMOTED APPOINTED RETIRED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED APPOINTED PROMOTED RETIRED PROMOTED RETIRED PROMOTED RETIRED PROMOTED RETIRED	NC NC NC NC NC YES NC NC NC NC NC NC NC NC NC NC NC NC NC	03/11/19 03/11/19 09/27/17 02/28/19 10/01/18 02/26/19 03/13/19 10/01/18 10/01/18 03/11/19 02/28/19 10/01/18 03/11/19 02/28/19 10/01/18 02/28/19 10/01/18 02/28/19 02/21/19 03/11/19 02/28/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/10/19 03/10/19 03/11/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATICK FOLEY FORD FORTUNE FRANCIS FRANCIS FRANCIS FRANCATOS FRANZEL FRAY GALLETTA GARCIA-GONZALE: GAROFALO MAME CENERAL GITTENS GLOVER GODDARD GOLDEERG	ANGELIQU BRADLEY L ANDRE E SIAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SSIROS S SSIROS S SSIROS S STEVEN L JUDINE MICHAEL A MICHAEL A NYRIAM P	70205 7021A 70235 70210 71651 12627 71012 71012 71012 70235 70235 70235 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 7005 70	\$109360.0000 \$15.0000 \$4489.0000 \$4489.0000 \$42500.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$11.9700 POLICE DEPA FOR PERIOD ENDIN SALARY \$39453.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000	PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED APPOINTED PROMOTED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED	NO YES NO NO NO NO NO NO NO NO YES YES PROV NO NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/11/19 02/19/19 03/02/19 02/28/19 02/28/19 02/28/19 02/28/19 03/01/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTAS MAXIMOS MCCARTHY MCCORMACK MCDERMOTT MCDONALD MCGREVY MCGURAN MCKENZIE MCMANUS MENDOZA MERA MESZAROS MIAH MILLER JR MINIER MONGGAN MURHEAD	BRENDAN M SHLOMO M WILLMER O JHONNY O LILJAH J RUBEN A KUBEN A JLFONSO A AMENA M JANDREW Y ANDREW Y SELIJAH W JANDREW Y BRENDAN Y NIA S STEPHEN J JAZMINE A THANAS Y STEPHEN J JAZMINE A EDMUND H JUBERTO J JALBERTO H JOSEFH B EARL T	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70210 71012 70235 71012 70235 71012 70235 71012	\$109360.0000 \$88945.0000 \$38945.0000 \$38183.0000 \$38183.0000 \$15.0000 \$15.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.00000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000000000000000000000000000000000	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED APPOINTED PROMOTED APPOINTED RETIRED PROMOTED APPOINTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED	NC NC NC NC NC VES NC NC NC NC NC NC NC NC NC NC NC NC NC	03/11/19 03/11/19 03/21/17 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19 03/11/19 02/28/19 10/01/18 03/11/19 02/28/19 03/11/19 02/28/19 03/11/19 02/28/19 03/11/19 03/03/11/19 03/03/11/19	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FIINNEY FITZPATICK FOLEY FORD FORTUNE FRANCIS FRANCIS FRANCIS FRANCATOS FRANZEL FRAY GALLETTA GARCIA-GONZALE: GAROFALO NAME GENERAL GITTENS GLOVER GODDARD GODDERG GOMEZ GOMEZ GOMEZ GONCHAROV GONZALEZ	ANGELIQU BRADLEY L ANDRE E SHAREE E JANDRE P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SFIROS STEVEN L MICHAEL A NYRIAM P NYEEMA K NICKEISH O MECCA ROSHEL JASON A JENIFER YANELFI M MIKHAIL GLADYS I	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71012 70235 70235 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205	\$109360.0000 \$15.0000 \$94489.0000 \$42500.0000 \$386945.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$128212.0000 \$128212.0000 \$128212.0000 \$128453.0000 \$11.9700 POLICE DEPA FOR PERIOD ENDIN SALARY \$39453.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000	PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED	NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/11/19 02/19/19 03/02/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/26/19 02/26/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTAS MAXIMOS MCCORMACK MCDERMOTT MCCORMACK MCDERMOTT MCGOREVY MCGURAN MCGREVY MCGURAN MCGREVY MCGURAN MCGREVY MCGURAN MCGREVY MCGURAN MCGRAN MILLER JR MINLER MILLER JR MINLER MOLONEY MORGAN MURHEAD MURHEAD MURHEY NADIM	BRENDAN M SHLOMO M VILLMER J JHONNY O ANTONIO J ELIJAH K FERNAND M JOHNY A ALFONSO A MENA M JOHN W SHINTO J HENA M JOHRNW T BRENDAN J BRENDAN J STEVEN V STEVEN V STALMIN G STEVEN N HANDREW T THANBIR A EDMUND D JOSEPH J JOSEPH T FRANK T MU T	70235 70235 710235 71651 70213 70214 70210 70205 92506 71012 70235 71012 70235 71012 70215 70210 70210 70210 70215 71012 71012 71012 71012 71012 71012 71012 70235 71012 70235 71012 70215 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 70215 70215 70235 70215 70235 70215 70235 70215 70235 70215 70235 70215 70235 70215 70235 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215	\$109360.0000 \$89945.0000 \$88945.0000 \$30706.0000 \$38945.0000 \$31705.0000 \$15.0000 \$15.0000 \$194489.0000 \$109360.0000 \$109360.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.	PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED APPOINTED RESIGNED RESIGNED RESIGNED APPOINTED RETIRED PROMOTED APPOINTED RETIRED PROMOTED APPOINTED PROMOTED PROMOTED PROMOTED PROMOTED RETIRED PROMOTED RETIRED PROMOTED RETIRED INCREASE RESIGNED	NC NC NC NC NC NC NC NC NC NC NC NC NC N	03/11/19 03/11/19 09/27/17 02/28/19 10/01/18 02/26/19 03/03/19 01/15/19 03/11/19 10/01/18 10/01/18 10/01/18 03/11/19 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 02/28/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FIINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCIS FRANCGIS FRANGATOS FRANGATOS FRANGEL GALLETTA GARCIA-GONZALEY GAROFALO MAME GENERAL GITTENS GLOVER GODDARD GOLDBERG GOMEZ GOMEZ GONCHAROV GONZALEZ GONZALEZ	ANGELIQU BRADLEY L ANDRE M SHAREE E JANET P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN SPIROS STEVEN L JUDINE MICHAEL A MICHAEL A NICKEISH O NICKEISH O NICKEISH O NICKEISH O JASON A JSNIFR YANELFI M MIKHAIL GLADYS I JOSEPH F	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71012 70235 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205	\$109360.0000 \$15.0000 \$4489.0000 \$4489.0000 \$38625.0000 \$35625.0000 \$35625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$109360.0000 \$128212.0000 \$128212.0000 \$11.9700 POLICE DEPA FOR PERIOD ENDIN SALARY \$39453.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000	PROMOTED RESIGNED RETIRED PROMOTED RESIGNED RESIGNED RETIRED APPOINTED PROMOTED PROMOTED PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED PROMOTED PROMOTED PROMOTED	NO YES NO NO NO NO NO NO NO NO YES YES <u>PROV</u> NO NO NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 03/02/19 02/19/19 02/28/19 02/28/19 02/28/19 02/28/19 03/01/19 02/28/19 03/01/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTAS MAXIMOS MCCORMACK MCDERMOTT MCCORMACK MCDERMOTT MCGOREVY MCGURAN MCGREVY MCGURAN MCGREVY MCGURAN MCGREVY MCGURAN MCGREVY MCGURAN MCGRAN MILLER JR MINLER MILLER JR MINLER MOLONEY MORGAN MURHEAD MURHEAD MURHEY NADIM	BRENDAN M SHLOMO M VILLMER J JHONNY O ANTONIO J ELIJAH K FERNAND M JOHNY A ALFONSO A MENA M JOHN W SHINTO J HENA M JOHRNW T BRENDAN J BRENDAN J STEVEN V STEVEN V STALMIN G STEVEN N HANDREW T THANBIR A EDMUND D JOSEPH J JOSEPH T FRANK T MU T	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 71012 70235 71012 70235 71012 70215 71012 70210 71012 70210 71012 70210 71012 70210 71012 70210 71012 70210 71012 70210 71012	\$109360.0000 \$89345.0000 \$89345.0000 \$89345.0000 \$30706.0000 \$94489.0000 \$15.0000 \$348345.0000 \$34183.0000 \$109360.0000 \$109360.0000 \$48945.0000 \$48945.0000 \$38183.0000 \$48945.0000 \$38183.0000 \$48945.0000 \$38183.0000 \$48945.0000 \$38183.0000 \$48945.0000 \$38183.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$489	PROMOTED PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RETIRED PROMOTED APPOINTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PR	NC NC NC NC NC NC NC NC NC NC NC NC NC N	03/11/19 03/11/19 09/27/17 02/28/19 10/01/18 02/26/19 03/03/19 01/15/19 03/11/19 10/01/18 10/01/18 10/01/18 03/11/19 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 02/28/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCIS FRANCIS FRANCIS FRANZEL FRAY GARLETTA GARCIA-GONZALE: GAROFALO NAME GENERAL GITTENS GLOVER GODDARD GOLDBERG GOMEZ GONCLAEZ GONZALEZ GONZALEZ	ANGELIQU RRADLEY L ANDRE M SHAREE E JANTE P ESTELLA JOHN W LYNDONNA B PGGYANN K KEN STEVEN L JUDINE MICHAEL A MICHAEL A MICHAEL A NTYEEMA K NICKEISH O MECCA ROSHEL JANTER JANTER JANTER JANTER MIKHAIL GLADYS I JOSEPH F	70205 7021A 70235 70210 71651 12627 71012 70235 71012 71651 70235 71012 70255 70205 70205 70205 70205 70205 70205 70205 71012 71012 71012 71012 71012 71012 70210 71012 70210	\$109360.0000 \$15.0000 \$4489.0000 \$38695.0000 \$38625.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$109360.0000 \$11.9700 POLICE DEPA FOR PERIOD ENDIN SALARY \$39453.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.00000 \$38183.00000 \$38183.00000 \$38183.0000000 \$38183.00000 \$38183.0000000000000000000000000000000000	PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED INCREASE PROMOTED PROMOTED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED DECEASED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED	NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 02/19/19 03/02/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 03/11/19 03/11/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 02/28/19 03/11/19 02/28/19 03/11/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTAS MAXIMOS MCCORMACK MCDERMOTT MCCORMACK MCDERMOTT MCGOREVY MCGURAN MCGREVY MCGURAN MCGREVY MCGURAN MCGREVY MCGURAN MCGREVY MCGURAN MCGRAN MILLER JR MINLER MILLER JR MINLER MOLONEY MORGAN MURHEAD MURHEAD MURHEY NADIM	BRENDAN M SHLOMO M VILLMER J JHONNY O ANTONIO J ELIJAH K FERNAND M JOHNY A ALFONSO A MENA M JOHN W SHINTO J HENA M JOHRNW T BRENDAN J BRENDAN J STEVEN V STEVEN V STALMIN G STEVEN N HANDREW T THANBIR A EDMUND D JOSEPH J JOSEPH T FRANK T MU T	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 71012 71012 71012 71012 71012 71012 70235 70215 70215 71012 70235 70215 71012 70235 70215 71012 70235 70215 70215 71012 70235 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 7025 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 7	\$109360.0000 \$89945.0000 \$88945.0000 \$30706.0000 \$38945.0000 \$31705.0000 \$15.0000 \$15.0000 \$194489.0000 \$109360.0000 \$109360.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.	PROMOTED PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RETIRED PROMOTED APPOINTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PR	NC NC NC NC NC NC NC NC NC NC NC NC NC N	03/11/19 03/11/19 09/27/17 02/28/19 10/01/18 02/26/19 03/03/19 01/15/19 03/11/19 10/01/18 10/01/18 10/01/18 03/11/19 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 02/28/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FIINEY FITZPATICK FOLEY FORD FORTUNE FRANCIS FRANCIS FRANCIS FRANZEL FRAY GALLETTA GARCIA-GONZALE: GAROFALO MAME CENERAL GITTENS GLOVER GODDARD GOLDERG GOMEZ GOMEZ GONEAROV GONZALEZ GONZALEZ GONZALEZ	ANGELIQU BRADLEY L ANDRE E JANETE P ESTELLA JOHN W LYNDONNA B PEGGYANN K KERN S STEVEN L JUDINE MICHAEL A MICHAEL A NYEEMA K NICKEISH O NYEEMA K NICKEISH O JSNIFER JASON A JSNIFER JASON A JENIFER GLADYS I JOSEHF F KENNETH M RANDY A	70205 7021A 70235 70210 71651 12657 71012 70235 71012 71012 70235 70235 70205 70205 70205 70205 70205 70205 70205 70205 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205	\$109360.0000 \$15.0000 \$4489.0000 \$4489.0000 \$42500.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$109360.0000 \$11.9700 POLICE DEPA FOR PERIOD ENDIN SALARY \$39453.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000	PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED APPOINTED PROMOTED APPOINTED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED APPOINTED RESIGNED RESIGNED RESIGNED	NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 02/14/19 02/14/19 03/11/19 03/11/19 02/19/19 03/02/19 02/28/19 02/28/19 02/28/19 02/28/19 03/01/19 03/01/19 03/01/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 03/11/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTAS MAXIMOS MCCARTHY MCCORMACK MCDERMOTT MCDONALD MCGREVY MCGURAN MCKENZIE MCMANUS MENDOZA MERA MESZAROS MIAH MILLER JR MINIER MORGAN MURHEAD MURHEAD MURHEAD MURHEAD MURHEAD MURHEAD MURHEAD MURHEAD	BRENDAN M SHLOMO M VILLMER J JHONNY O ANTONIO J ELIJAH K FERNAND M JOHNY A ALFONSO A MENA M JOHN W SHINTO J HENA M JOHRNW T BRENDAN J BRENDAN J STEVEN V STEVEN V STALMIN G STEVEN N HANDREW T THANBIR A EDMUND D JOSEPH J JOSEPH T FRANK T MU T	70235 70235 71235 71651 70235 71012 7021A 70205 92508 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70210 71012 70235 71012 70235 71012 70235 71012 70235 70210 710235 70210 70235 70210 71012 70235 70215 70215 71012 70235 70215 70215 71012 70235 70215 70215 70215 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 71012 70235 71012 71012 70235 71012 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70210 70235 70210 70235 70210 70210 70235 70210 70235 70210 70235 70210 70215 70210 70235 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70215 70215 70210 70215 70215 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215	\$109360.0000 \$89345.0000 \$89345.0000 \$30706.0000 \$38945.0000 \$15.0000 \$15.0000 \$15.0000 \$13183.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$109360.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.	PROMOTED PROMOTED PROMOTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED APPOINTED RETIRED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RETIRED PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NC NC NC NC NC VES NC NC NC NC NC NC NC NC NC NC NC NC NC	03/11/19 03/11/19 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19 01/15/19 03/11/19 02/28/19 10/01/18 03/11/19 02/28/19 03/11/19 02/28/19 03/11/19 02/28/19 03/11/19 03/21/19 03/11/19 03/21/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/19 03/11/	056 056 056 056 056 056 056 056 056 056
FIGUEROA JR FINNEY FITZPATRICK FOLEY FORD FORTUNE FRANCIS FRANCIS FRANCIS FRANZEL FRAY GARLETTA GARCIA-GONZALE: GAROFALO NAME GENERAL GITTENS GLOVER GODDARD GOLDBERG GOMEZ GONCLAEZ GONZALEZ GONZALEZ	ANGELIQU RRADLEY L ANDRE M SHAREE E JANTE P ESTELLA JOHN W LYNDONNA B PGGYANN K KEN STEVEN L JUDINE MICHAEL A MICHAEL A MICHAEL A NTYEEMA K NICKEISH O MECCA ROSHEL JANTER JANTER JANTER JANTER MIKHAIL GLADYS I JOSEPH F	70205 7021A 70235 70210 71651 12627 71012 71012 71012 70235 70235 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70202 71012 71012 71012 71012 71012 71012 71012 71012 71012 71012 70216 70216 70216 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205	\$109360.0000 \$15.0000 \$4489.0000 \$38695.0000 \$38625.0000 \$38625.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$109360.0000 \$11.9700 POLICE DEPA FOR PERIOD ENDIN SALARY \$39453.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.00000 \$38183.00000 \$38183.00000 \$38183.0000000 \$38183.00000 \$38183.0000000000000000000000000000000000	PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED INCREASE PROMOTED PROMOTED RESIGNED RESIGNED APPOINTED APPOINTED APPOINTED DECEASED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED APPOINTED PROMOTED	NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 02/14/19 10/01/18 03/11/19 02/19/19 03/02/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 03/11/19 03/11/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 02/28/19 03/11/19 02/28/19 03/11/19	056 056 056 056 056 056 056 056 056 056	MALLON MANGD MARQUEZ MARTE MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTINEZ MARTAS MAXIMOS MCCORMACK MCDERMOTT MCCORMACK MCDERMOTT MCGOREVY MCGURAN MCGREVY MCGURAN MCGREVY MCGURAN MCGREVY MCGURAN MCGREVY MCGURAN MCGRAN MILLER JR MINLER MILLER JR MINLER MOLONEY MORGAN MURHEAD MURHEAD MURHEY NADIM	BRENDAN M SHLOMO M VILLMER J JHONNY O ANTONIO J ELIJAH K FERNAND M JOHNY A ALFONSO A MENA M JOHN W SHINTO J HENA M JOHRNW T BRENDAN J BRENDAN J STEVEN V STEVEN V STALMIN G STEVEN N HANDREW T THANBIR A EDMUND D JOSEPH J JOSEPH T FRANK T MU T	70235 70235 70235 71651 70235 71012 7021A 70205 92508 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 71012 71012 71012 71012 71012 71012 70235 70215 70215 71012 70235 70215 71012 70235 70215 71012 70235 70215 70215 71012 70235 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 7025 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 7	\$109360.0000 \$89345.0000 \$89345.0000 \$89345.0000 \$30706.0000 \$94489.0000 \$15.0000 \$348345.0000 \$34183.0000 \$109360.0000 \$109360.0000 \$48945.0000 \$48945.0000 \$38183.0000 \$48945.0000 \$38183.0000 \$48945.0000 \$38183.0000 \$48945.0000 \$38183.0000 \$48945.0000 \$38183.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$489	PROMOTED PROMOTED PROMOTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RETIRED PROMOTED APPOINTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PR	NC NC NC NC NC VES NC NC NC NC NC NC NC NC NC NC NC NC NC	03/11/19 03/11/19 03/11/19 02/28/19 10/01/18 02/26/19 03/03/19 01/15/19 03/11/19 02/28/19 10/01/18 03/11/19 02/28/19 03/11/19 02/28/19 03/11/19 02/28/19 03/11/19 03/28/19 03/11/19 03/28/19 03/11/19 03/11/19 03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19	056 056 055 055 056 056 056 056 056 056

THE CITY RECORD

,								
NIEDZWIECKI	ABRAHAM		70235	\$88945.0000	PROMOTED	NO	03/11/19	056
NOTARO	TARA	Е	70205	\$14.6400	RESIGNED	YES	12/18/18	056
NYLAND	PAUL	М	70210	\$85292.0000	RETIRED	NO	10/01/18	056
OECHSNER	LENHARD	J	7026B	\$138089.0000	RETIRED	NO	10/01/18	056
OGLE	THAWANN	Q	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
ONEIL	BRIGHID	т	70265	\$128212.0000	PROMOTED	NO	03/05/19	056
OSBOURNE	TRESHA	г	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
PARABDIN	RAMESH		70210	\$85292.0000	RETIRED	NO	10/01/18	056
PARKES	SEAN	P	70235	\$88945.0000	PROMOTED	NO	03/11/19	056
PATTERSON	BRIANNA	т	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
PAULSINGH	KEVIN		70235	\$88945.0000	PROMOTED	NO	03/11/19	056
PERSAUD	RAYMOND	V	70235	\$109360.0000	PROMOTED	NO	03/11/19	056
PETRASSI	JARED	В	70235	\$88945.0000	PROMOTED	NO	03/11/19	056
PHAIRE	ALTHEA	Е	71652	\$47238.0000	RETIRED	NO	03/08/19	056
PHYALL	CHARMAIN	D	10144	\$40709.0000	DECEASED	NO	10/12/18	056
PICKETT		S	71012 60817	\$38183.0000	APPOINTED	NO NO	02/28/19	056
PINDER PINEDA	DAHLIA JENNY	c	60817	\$42136.0000 \$32426.0000	RESIGNED INCREASE	NO	09/06/17 01/23/19	056 056
PODOLSKIY	MICHAEL	C	70235	\$88945.0000	PROMOTED	NO	01/23/19	056
POLANCO	MIGRITZA	п	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
PORTO	RICHARD	A	70265	\$128212.0000	PROMOTED	NO	03/05/19	056
QUITO	VICTORIA		71012	\$38183.0000	APPOINTED	NO	02/28/19	056
RAMOS	MILTON		71651	\$39449.0000	DECEASED	NO	03/07/19	056
REDMOND	FRANK	Ρ	70235	\$88945.0000	PROMOTED	NO	03/11/19	056
REEVES	MICHIAH	c	71012	\$51700.0000	DECEASED	NO	01/19/19	056
REGAN JR	THOMAS	J	20246	\$90000.0000	APPOINTED	YES	03/03/19	056
REYES	RAMON	-	70235	\$109360.0000	PROMOTED	NO	03/11/19	056
RICHARDS	DIANDRA	н	70205	\$15.0000	RESIGNED	YES	03/02/19	056
RIOS	MARLEEN	-	70205	\$15.0000	RESIGNED	YES	03/01/19	056
RIVERA	ADRIANO	A	70235	\$88945.0000	PROMOTED	NO	03/11/19	056
RIVERA	JOSE	L	70210	\$85292.0000	RETIRED	NO	10/01/18	056
RIVERA	KIARA	A	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
ROBERTS	AKIMBA	s	71651	\$39449.0000	RESIGNED	NO	03/12/19	056
ROBERTS-AJANWAC		С	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
RODRIGUEZ	ANTHONY	в	60817	\$33498.0000	RESIGNED	NO	02/22/19	056
RODRIGUEZ	ELVIS		70235	\$88945.0000	PROMOTED	NO	03/11/19	056
RODRIGUEZ	MELISSA		71012	\$38183.0000	APPOINTED	NO	02/28/19	056
RODRIGUEZ WHITE	RADZIVIL	I	71012	\$38183.0000	RESIGNED	NO	03/09/19	056
ROEHRIG	WILLIAM	ĸ	70235	\$88945.0000	PROMOTED	NO	03/11/19	056
ROMAN	SAMANTHA	Е	71651	\$30706.0000	RESIGNED	NO	09/27/17	056
ROMANO	VICTORIA	L	70206	\$16.3100	RESIGNED	YES	03/06/19	056
ROOPNARINE	REUEL	D	70210	\$54394.0000	RESIGNED	NO	03/07/19	056
	KURTIS	v	70265	\$128212.0000	PROMOTED	NO	03/05/19	056
	MERANI		70265	\$128212.0000	PROMOTED	NO	03/05/19	056
ROTTE					RESIGNED	YES	03/01/19	056
ROSE ROTTE RUBIN	PATRICIA	A	70205	\$15.0000				
ROTTE RUBIN RUGGIERO	PATRICIA DOMINIC		70235	\$88945.0000	PROMOTED	NO	03/11/19	056
ROTTE RUBIN RUGGIERO SAJDUK	PATRICIA DOMINIC MATEUSZ	A P	70235 70235	\$88945.0000 \$109360.0000	PROMOTED PROMOTED	NO	03/11/19	056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA	PATRICIA DOMINIC MATEUSZ JORGE	P	70235 70235 70235	\$88945.0000 \$109360.0000 \$88945.0000	PROMOTED PROMOTED PROMOTED	NO NO	03/11/19 03/11/19	056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANABRIA	PATRICIA DOMINIC MATEUSZ		70235 70235	\$88945.0000 \$109360.0000	PROMOTED PROMOTED	NO	03/11/19	056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANABRIA	PATRICIA DOMINIC MATEUSZ JORGE ERIC	P	70235 70235 70235 70235 70265	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000	PROMOTED PROMOTED PROMOTED PROMOTED	NO NO NO	03/11/19 03/11/19 03/05/19	056 056 056
ROTTE RUBIN RUGGIERO	PATRICIA DOMINIC MATEUSZ JORGE ERIC	P	70235 70235 70235 70265 60817	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RTMENT	NO NO NO	03/11/19 03/11/19 03/05/19	056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANABRIA	PATRICIA DOMINIC MATEUSZ JORGE ERIC	P	70235 70235 70235 70265 60817	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RTMENT	NO NO NO	03/11/19 03/11/19 03/05/19	056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANDSETH SANTANA NAME	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA	P S	70235 70235 70235 70265 60817 F TITLE NUM	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN SALARY	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RTMENT G 03/22/19 ACTION	NO NO NO PROV	03/11/19 03/11/19 03/05/19 03/12/19 EFF DATE	056 056 056 056 AGENCY
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANDSETH SANTANA NAME SANTIAGO	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY	P S L	70235 70235 70235 70265 60817 F TITLE NUM 60817	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA 'OR PERIOD ENDIN SALARY \$46737.0000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED RTMENT IG 03/22/19 ACTION DISMISSED	NO NO NO PROV	03/11/19 03/11/19 03/05/19 03/12/19 EFF DATE 02/13/19	056 056 056 056 <u>AGENCY</u> 056
ROTTE RUBIN RUGGIERO SAJDUK SANDREIA SANDSETH SANTANA NAME SANTIAGO SANTIAGO	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE	P S	70235 70235 70265 60817 F TITLE NUM 60817 71012	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED RETIRED ACTION DISMISSED RESIGNED	NO NO NO NO PROV NO NO	03/11/19 03/11/19 03/05/19 03/12/19 EFF DATE 02/13/19 03/14/19	056 056 056 056 <u>AGENCY</u> 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANDSETH SANTANA NAME SANTIAGO SANTIAGO SANTIAGO	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI	P S L M	70235 70235 70235 70265 60817 F TITLE NUM 60817 71012 70205	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$15.0000	PROMOTED PROMOTED PROMOTED RETIRED RTMENT G 03/22/19 <u>ACTION</u> DISMISSED RESIGNED	NO NO NO PROV NO YES	03/11/19 03/11/19 03/05/19 03/12/19 EFF DATE 02/13/19 03/14/19 03/08/19	056 056 056 056 <u>AGENCY</u> 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANDSETH SANTANA NAME SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTOS	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY	P S L M	70235 70235 70235 70265 60817 F TITLE NUM 60817 71012 70205 71012	\$88945.0000 \$109360.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$38183.0000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED RETIRED ACTION DISMISSED RESIGNED APPOINTED	NO NO NO PROV NO YES NO	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 02/13/19 03/14/19 03/08/19 02/28/19	056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANDRENA SANDSETH SANTANA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SEDA JR	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO	P S L M A	70235 70235 70235 70265 60817 F TITLE NUM 60817 71012 70205 71012 70210	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$15.0000 \$38183.0000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED RETINEN ACTION DISMISSED RESIGNED RESIGNED APPOINTED RETIRED	NO NO NO PROV NO YES NO NO	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 02/13/19 03/14/19 03/14/19 03/08/19 02/28/19 10/01/18	056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANDRIA SANDSETH SANTANA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SEDA JR SELBY	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHAREESE	P S L M A D	70235 70235 70235 70265 60817 F TITLE NUM 60817 71012 70205 71012 70210 71012	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA FOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$15.0000 \$38183.0000 \$38183.0000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED	NO NO NO PROV NO YES NO NO NO	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 03/14/19 03/14/19 03/08/19 02/28/19 00/01/18 02/28/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANDSETH SANTANA NAME SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SEDA JR SELBY SEWELL	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHARESSE ANDREA	P S L M A D M	70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 71012 71012 71012	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN <u>SALARY</u> \$46737.0000 \$39453.0000 \$38183.0000 \$38183.0000 \$33183.0000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED RETIRED DISMISSED RESIGNED RESIGNED APPOINTED RETIRED APPOINTED INCREASE	NO NO NO PROV NO NO YES NO NO NO NO	03/11/19 03/15/19 03/05/19 03/12/19 03/12/19 02/13/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 12/21/18	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANDSETH SANTDETH SANTANA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTOS SELBY SELBY SEWELL SHAHADAT	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHAREESE ANDREA MD	P S L M A D M A	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 70210 71012 71652 71651	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN <u>SALARY</u> \$46737.0000 \$39453.0000 \$15.0000 \$53443.0000 \$53443.0000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED RETIRED CISHISSED RESIGNED RESIGNED RETIRED APPOINTED INCREASE RESIGNED	NO NO NO PROV NO VES NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 12/21/18 11/07/17	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGIERO SAJDUK SANDRIA SANDSETH SANTANA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SEDA JR SELBY SELBY SEWELL SEWELL SHAHDAT SHAM	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA DAISY JASMINE NOEMI MILLENNY JULIO SHARESSE ANDREA MD HARVEY	P S M A D M A F	70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 70210 71012 70210 71012 71651 70235	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA FOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$39453.0000 \$38183.0000 \$35706.0000 \$109360.0000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED APPOINTED INCREASE RESIGNED PROMOTED	NO NO NO PROV NO YES NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 12/21/18 01/07/17	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANDSETH SANDSETH SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SEDA JR SELBY SEWELL SHAHADAT SIMPSON	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHARESE ANDREA MD HARVEY DANIEL	P S L M A D M A	70235 70235 70235 60817 70265 60817 71012 70205 71012 70210 71012 70210 71012 71652 71652 71655 71012	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN <u>SALARY</u> \$46737.0000 \$39453.0000 \$38183.0000 \$38183.0000 \$30706.0000 \$38183.0000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED RETIRED DISMISSED RESIGNED RESIGNED APPOINTED INCREASE RESIGNED APPOINTED INCREASE RESIGNED APPOINTED PROMOTED APPOINTED	NO NO NO PROV NO NO YES NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 02/13/19 03/14/19 03/08/19 02/28/19 10/01/18 11/07/17 03/11/19 02/28/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANDSETH SANTER SANTER SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTOS SEDA JR SELBY SEWELL SHAHASTMENS SHAPSON SIMPSON SIMPSON	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA DAISY JASMIE NOEMI MIALENNY JULIO SHAREESE ANDREA MD HARVEY DANIEL DOODNAUT	P S L M A D M A F J	70235 70235 70235 70265 60817 70265 60817 71012 70205 71012 71012 71020 71012 71651 70215 71012 71651 70235 71012 70210	\$88945.0000 \$109360.0000 \$18945.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$15.0000 \$38183.0000 \$38183.0000 \$30706.0000 \$19360.0000 \$38183.0000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED RETIRED APPOINTED INCREASE RESIGNED PROMOTED RESIGNED RESIGNED	NO NO NO PROV NO YES NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/14/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 12/21/18 11/07/17 03/11/19 02/28/19 10/27/18	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGIERO SAJDUK SANABRIA SANDSETH SANTSAN SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SENDA JR SELBY SELBY SEWELL SENDA JR SELBY SEWELL SIMCH SINGH SINGLETON	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA DAISY JASMINE NOEMI MILLENNY JULIO SHARESE ANDREA MD HARVEY DANIEL DOODNAUT DANEISHA	P S L M A D M A F J	70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 71652 71651 70235 71651 70235 71012 70210 71012	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$15.0000 \$38183.0000 \$109360.0000 \$42500.0000 \$38183.0000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED INCREASE RESIGNED PROMOTED APPOINTED APPOINTED APPOINTED APPOINTED	NO NO NO PROV NO YES NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 12/21/18 11/07/17 03/11/19 02/28/19 02/28/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANDRIA SANDSETH SANTANA NAME SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SINTIAGO SINTAGEN SELBY SEWELL SHAHADAT SINGLETON SINGLETON SMALL	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHAREESE ANDREA MD HARVEY DANIEL DOODNAUT DANEISHA KEMIE	P S L M A D M A F J D	70235 70235 70225 60817 TITLE NUM 60817 71012 70205 71012 70205 71012 70210 71012 70210 71052 71651 70235 71012 70235 71012 70235	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA FOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$39453.0000 \$38183.0000 \$30706.0000 \$109360.0000 \$38183.0000 \$38183.0000 \$34470.0000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED INCREASE RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED RESIGNED RESIGNED	NO NO NO PROV NO NO YES NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 12/21/18 02/28/19 12/21/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANDSETH SANTANA NAME SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SELBY SEWELL SELBY SEWELL SHAHADAT SHAM SINGETON SINGETON SINGE	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHARESSE ANDREA HARVEY DANIEL DONDAUT DANIELSHA KEMIE ONEIL	P S L M A D M A F J O	70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 71652 71651 70235 71651 70235 71012 70210 71012	\$88945.0000 \$109360.0000 \$199360.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN <u>SALARY</u> \$46737.0000 \$3453.0000 \$38183.0000 \$53443.0000 \$30706.0000 \$109360.0000 \$38183.0000 \$42500.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.0000 \$34132.00000 \$34132.00000 \$34132.00000 \$34132.000000000000000000000000000000000000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED PROMOTED	NO NO NO NO YES NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 02/13/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 12/21/18 11/07/17 03/11/19 02/28/19 02/28/19 02/28/19 02/24/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGIERO SAJDUK SANDSETH SANDSETH SANDSETH SANTANA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SEDA JR SELBY SEWELL SIMHADAT SIMPSON SINGH SINGLETON SMALL SINGLETON SMITHWICK SOTO SOSA	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHAREESE ANDREA MD HARVEY DANIEL DOOINAUT DANIEISHA KEMIE SHERYL	P S L M A D M A F J D	70235 70235 70225 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 71651 70235 71651 70235 71651 70235 71012 70210 71012 70210 71012 70210 71012	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN <u>SALARY</u> \$46737.0000 \$39453.0000 \$19453.0000 \$53443.0000 \$53443.0000 \$19360.0000 \$42500.0000 \$42500.0000 \$42500.0000 \$42570.0000 \$18183.0000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED CONTENT PROMOTED RETIRED DISMISSED RESIGNED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED	NO NO NO NO NO NO NO NO NO NO NO NO NO N	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/14/19 03/08/19 03/08/19 02/28/19 10/01/18 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUBGIERO SAJDUK SANDARIA SANDSETH SANTANA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SINGLES SELBY SEWELL SHAHADAT SHAHADAT SINGLETON SINGLETON SMALL SNIGLETON SMALL SMITHWICK SOTO SOSA SPARKMAN	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMING NOEMI MILLENNY JULIO SHAREESE ANDREA HARVEY DANIEL DOODNAUT DANEISEA KEMIE ONEIL SHERYL ROGER	P S L M A D M A F J O	70235 70235 70255 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 71651 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 70235 71012 71654	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$3183.0000 \$3183.0000 \$31813.0000 \$42500.0000 \$31813.0000 \$44570.0000 \$3183.0000 \$44570.0000 \$3183.0000 \$44570.0000 \$3183.0000 \$34570.0000 \$3183.0000 \$34570.0000 \$3183.0000 \$34570.0000 \$3183.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.00000 \$34570.00000 \$34570.00000 \$34570.00000000000000000000000000000000000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED PROMOTED APPOINTED INCREASE RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED	NO NO NO NO NO NO NO NO NO NO NO NO NO N	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 02/13/19 03/14/19 03/08/19 02/28/19 12/21/18 02/28/19 10/01/18 02/28/19 10/27/18 02/28/19 02/24/19 02/24/19 02/24/19 02/24/19 02/24/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANDSETH SANTSAN SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SAN	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHAREESE ANDREA MD HARVEY DANIEL DOOINAUT DANIEISHA KEMIE SHERYL	P S L M A D M A F J O	70235 70235 70225 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 71651 70235 71651 70235 71651 70235 71012 70210 71012 70210 71012 70210 71012	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN <u>SALARY</u> \$46737.0000 \$39453.0000 \$19453.0000 \$53443.0000 \$53443.0000 \$19360.0000 \$42500.0000 \$42500.0000 \$42500.0000 \$42570.0000 \$18183.0000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED CONTENT PROMOTED RETIRED DISMISSED RESIGNED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED	NO NO NO NO NO NO NO NO NO NO NO NO NO N	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/14/19 03/08/19 03/08/19 02/28/19 10/01/18 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGIERO SAJDUK SANDRIA SANDSETH SANTSAN SANDSETH SANTANA SANTGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SELBY SEWELL SELL SELL SELL SIMPSON SINGH SINGLETON SMALL SINGLETON SMITHWICK SOTO SOSA SPARKMAN SPERRAZZA STEIN	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHARESE ANDREA MD HARVEY DANIEL DODNAUT DANEILS SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL	P S L M A D M A F J O	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 71052 71651 70215 71651 70210 71652 71651 70210 71012 70210 71012 70210 71012 70210 71012 70210 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71020 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 71000 710000 7100000 7100000000	\$88945.0000 \$109360.0000 \$199360.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$38183.0000 \$38183.0000 \$30706.0000 \$30706.0000 \$34183.0000 \$42500.0000 \$38183.0000 \$42500.0000 \$38183.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.0000 \$3455.00000 \$3455.0000 \$3455.0000 \$3455.0000 \$3	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED RETIRED DISMISSED RESIGNED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED	NO NO NO NO NO NO NO NO NO NO NO NO NO N	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 02/13/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 10/21/19 02/28/19 02/28/19 02/28/19 03/05/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGIERO SAJDUK SANDSETH SANDSETH SANDSETH SANDSETH SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SIN	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MILLENNY JULIO SHAREESE ANDREA MD HARVEY DANIEL DOODNAUT DANEISHA KEMIE ONEIL SHERYL ROGER STEPHEN HELEN	P S L M A D M A F J D S	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 71651 70235 71012 70210 71012 70255 71012 70215 71012 70255 71012	\$88945.0000 \$109360.0000 \$18935.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$3143.0000 \$35143.0000 \$38183.0000 \$38183.0000 \$42500.0000 \$38183.0000 \$42500.0000 \$38183.0000 \$4570.0000 \$38183.0000 \$46385.0000 \$128212.0000 \$38183.0000 \$46385.0000 \$128212.0000 \$128122.0000 \$128122.0000 \$12813.0000 \$128122.0000 \$12813.0000 \$128122.0000 \$12813.0000 \$128122.0000 \$12813.0000 \$128122.0000 \$128122.0000 \$12813.0000 \$128122.0000 \$12813.0000 \$128122.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.0000 \$12813.00000 \$12813.00000 \$12813.00000 \$12	PROMOTED PROMOTED PROMOTED RETIRED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NO NO NO PROV NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANDARIA SANDSETH SANTANA NAME SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAGO SINTIAG	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MILLENNY JULIO SHAREESE ANDREA HARVEY DANIEL DOODNAUT DANEISHA KEMIE ONEIL SHERYL ROGER STEPHEN HELEN AKIL	P S L M A D M A F J D S A	70235 70235 70255 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 71651 70235 71012 71652 71012 71652 71012 71654 70215 71012 71658 71012 71658	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$19453.0000 \$38183.0000 \$3076.0000 \$109360.0000 \$42500.0000 \$42500.0000 \$4570.0000 \$38183.0000 \$45385.0000 \$128212.0000 \$38183.0000 \$45385.0000	PROMOTED PROMOTED PROMOTED RETIRED RETIRED RETIRED DISMISSED RESIGNED RESIGNED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RETIRED RETIRED RESIGNED	NO NO NO NO NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 12/21/18 01/07/18 02/28/19 10/27/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANDSETH SANTANA NAME SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTAGO SANTAGO STEIN STEVART SYENTORAITIS SYED	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHARESSE ANDREA MD HARVEY DANIEL DOODNJUT DANEISHA KEMIE ONEIL SHERYL ROGER STEPHEN HELEN AKIL STEPHEN	P S L M A D M A F J D S A	70235 70235 70235 70265 60817 FTITLE 60817 71012 70205 71012 71022 71652 71012 71651 71022 71651 71022 71651 70210 71012 60817 70265 71012 70205 71020 70205 70205 70205 70205 70205	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$38183.0000 \$38183.0000 \$3443.0000 \$3443.0000 \$3443.0000 \$3443.0000 \$3453.0000 \$3453.0000 \$3453.0000 \$3453.0000 \$3453.0000 \$3453.0000 \$3453.0000 \$3453.0000 \$3453.0000 \$3453.0000 \$3453.0000 \$4535.0000 \$55344.0000 \$55344.0000 \$55344.0000 \$5534.0000 \$5534.0000 \$5534.0000 \$5534.0000 \$5534.0000 \$5534.0000 \$5534.0000 \$5534.0000 \$5534.0000 \$5534.0000 \$5534.0000 \$5534.0000 \$5534.0000 \$5534.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5535.0000 \$5555.00000 \$5555.00000 \$5555.000000000000000000000000000000000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED RETIRED DISMISSED RESIGNED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED	NO NO NO PROV NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/12/19 02/13/19 03/14/19 03/08/19 02/28/19 12/21/18 02/28/19 12/21/18 11/07/17 03/11/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/15/19 03/15/19 03/15/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGIERO SAJDUK SANDERIA SANDERIA SANDERH SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SEDA JR SELBY SEWELL SEWALL SHAHADAT SHAHADAT SINGLETON SMAHLU SINGLETON SMAHLU SINGLETON SMAHLU SINGLETON SMAHLU SINGLETON SMAHLU SINGLETON SMAHLU SINGLETON SMAHLU STREYART STENN STEWART SVENDCRAITIS SVED	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHARESE ANDREA ANDREA ANDREA ANDREA ENTE DOMIEL DANIEL DANIEL DANIEL SHERYL ROGER STEPHEN HELEN AKIL STEPHEN NADIA	P S L M A D M A F J D S A	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 71012 71012 71012 71020 71012 71651 70210 71012 71651 70210 71012 70210 71012 71655 70210 70205 70210	\$88945.0000 \$109360.0000 \$18945.0000 \$46737.0000 POLICE DEPA 'OR PERIOD ENDIN <u>SALARY</u> \$46737.0000 \$39453.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$138183.0000 \$42500.0000 \$38183.0000 \$3457.0000 \$3457.0000 \$3457.0000 \$3457.0000 \$3457.0000 \$3457.0000 \$3457.0000 \$46385.0000 \$128212.0000 \$128121.0000 \$46385.0000 \$46805.0000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED RESIGNED RETIRED PROMOTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RETIRED RETIRED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NO NO NO NO NO NO NO NO NO NO NO NO NO N	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANJBRIA SANJBRIA SANJSETH SANTANA	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA DAISY JASMINE NOEMI NOEMI MIALENNY JULIO SHAREESE ANDREA MD HARVEY DANIEL DAOIRAU DAOIRAU SHERYL ROGER STEPHEN NADIA SURFRAZ	P S LM MADMAFJ D OS AP	70235 70235 70255 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 71652 71651 70235 71012 71655 71012 71655 71012 71655 71012 70205 71012 70210 71025 71012 70205 71012 70205 71012 70205 70205 70200 70205 70200 70205 70200 70205 70200 70205 70200 70205 70200 70205 70200 70205 70200 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70200 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70005 70005 70005 70005 7005 7	\$88945.0000 \$109360.0000 \$88945.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$19453.0000 \$53443.0000 \$53443.0000 \$19360.0000 \$42500.0000 \$42500.0000 \$48183.0000 \$44570.0000 \$38183.0000 \$46385.0000 \$54394.0000 \$54394.0000 \$54394.0000 \$8945.0000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED	NO NO NO NO NO NO NO NO NO NO NO NO NO N	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/15/19 03/05/19 03/05/19 03/05/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANABRIA SANDSETH SANTANA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SEDA JR SELBY SEWELL SHAHADAT SINGLETON SMALL SMITHWICK SOTO SOSA STRUE SINGLETON SMALL SMITHWICK SOTO SOSA STEIN STEVART STEIN STEVART SVENTORAITIS SVED TAVERAS TERRAZAS	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO NOEMI MIALENNY JULIO SHARESSE ANDREA SHARESSE ANDREA SHARESSE ANDREA HARVEY DANIELS DOOINJUT DANEISHA KEMIE ONELL SHERYL ROOGR STEPHEN HELEN AKIL STEPHEN NDIA SURFRAZ JOSE	P S L M A D M A F J D O S A P J	70235 70225 70225 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 71651 70215 71012 71652 71012 71652 71012 71654 70205 71012 71654 70205 71012 71654 70205 70210 70235 70210	\$88945.0000 \$109360.0000 \$199360.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN SALARY \$46737.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$3443.0000 \$38183.0000 \$38183.0000 \$4535.0000 \$4635.0000 \$128212.0000 \$128183.0000 \$4635.0000 \$4685.0000 \$128212.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.00000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.0000 \$4695.00000 \$4695.00000 \$4695.00000 \$469	PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RETIRED RETIRED RETIRED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED	NO NO NO NO NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 02/13/19 03/05/19 02/28/19 02/28/19 10/01/18 02/28/19 10/27/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUBIN RUGGIERO SAJDUK SANDSETH SANDSETH SANDSETH SANDSETH SANDSETH SANDSETH SANDS SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHARESZ ANDREA MD DANIEL DODNAUT DANIEL DODNAUT DANEIL SHERYL ROGER SHEPHEN HELEN AKIL STEPHEN NADIA SUEFRAZ JOSE ERAYAN	P S L M M A D M A F J D O S A P J A	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70205 71012 71651 70235 71651 70210 71012 70210 71012 70210 71012 70210 71012 70210 71012 7025 71012 7025 70210 7025 70210 70235 70210 70235 70210	\$88945.0000 \$109360.0000 \$189365.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$139360.0000 \$38183.0000 \$42500.0000 \$38183.0000 \$42500.0000 \$38183.0000 \$42500.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$34570.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.0000 \$35222.00000 \$35222.00000 \$35222.00000 \$35222.00000 \$35222.000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED RESIGNED PROMOTED RETIRED PROMOTED RETIRED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED	NO NO NO PROV NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/11/19 03/05/19 03/05/19 03/05/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGIERO SAJDUK SANDARIA SANDSETH SANTSAN SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SINGLES SELBY SELBY SELBY SELBY SELBY SELBY SELBY SELBY SELBY SELBY SELBY SELBY SELBY SELBY SELBY SINGLETON SMALL SINGLETON SMALL SINGLETON SMALL SINGLETON SMALL SINGLETON SMALL SURTORA SPERNZZA STELN STEWART SVED TAVERAS TERRAZAS THIBOU THOMAS	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY MOEMI MIALENNY JULIO SHARESE ANDREA MD HARVEY DANIEL DOONAUT DANIEL DOONAUT DANEILS SHARESE ANDREA MD HARVEY DANIEL SHARESE STEPHEN HELEN AKIL STEPHEN NADLA SURFRAZ JOSE BRAYAN TEMESHA ADRIAN	P S LM MADMAFJ D OS AP JAJN	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70205 71012 71651 70235 71012 70210 71652 71655 71012 70265 71012 70265 71012 70265 70210 70235 70210 70235 70210 70235 70210 70235 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70000000000	\$88945.0000 \$109360.0000 \$199360.0000 \$18945.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$3453.0000 \$3453.0000 \$3453.0000 \$3483.0000 \$34343.0000 \$3483.0000 \$1343.0000 \$3483.0000 \$42500.0000 \$3483.0000 \$3483.0000 \$3483.0000 \$3483.0000 \$3483.0000 \$4250.0000 \$4635.0000 \$4635.0000 \$4895.0000 \$4895.0000 \$4895.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$4895.0000 \$4813.0000 \$4537.0000 \$4537.0000 \$4577.00000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.00	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NG NG NG PROV NG YES NG NG NG NG NG NG NG NG NG NG NG NG NG	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 02/13/19 03/14/19 03/08/19 02/28/19 12/21/18 02/28/19 12/21/18 11/07/17 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUBIN RUGGIERO SAJDUK SANDSETH SANDSETH SANDSETH SANDSETH SANDSETH SANDSETH SANDSETH SANDSETH SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SEDA JR SELBY SEWELL SIMGLETON SMALL SIMPSON SINGH SINGLETON SMALL SINGLETON SMALL SINGLETON SMITHWICK SOTO SOSA SPARTMAN SPERRAZA STEIN STEWATORAITIS SYED TAVERAS THERAZAS THIBOU THOMAS THOMAS	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHARESE ANDREA ANDREA ANDREA ANDREA DANIEL DOMIEL DANIEL DANIEL DANIEL DANIEL SHERYL ROGER STEPHEN HELEN AKIL SHERYL ROGER STEPHEN HELEN AKIL STEPHEN HELEN AKIL SHERYL ROSE STEPHEN HELEN AKIL SHERYL ROSE STEPHEN AKIL SHERYL ROSE STEPHEN AKIL SHERYL ROSE STEPHEN AKIL SHERYL ROSE STEPHEN AKIL SHERYL SHERYL ROSE STEPHEN AKIL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERYL SHERY	P S L M A D M A F J D O S A P JAJN D	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70210 71012 70210 71012 71651 70215 71012 71651 70210 71012 70210 71012 70210 71012 70205 70205 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70210 70215 70210 70210 70215 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210	\$88945.0000 \$109360.0000 \$18935.0000 \$128212.0000 \$46737.0000 POLICE DEPA 'OR PERIOD ENDIN <u>SALARY</u> \$46737.0000 \$39453.0000 \$3183.0000 \$35183.0000 \$35183.0000 \$31813.0000 \$42500.0000 \$38183.0000 \$42500.0000 \$38183.0000 \$42500.0000 \$38183.0000 \$4250.0000 \$128212.0000 \$128212.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$4695.0000 \$46805.0000 \$4695.00000 \$4695.00000 \$4695.00000 \$4695.00000 \$4695.00000 \$4695.000000 \$4695.00000 \$4695.000000000000000000000000000000000000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED DISMISSED	NO NO NO NO NO NO NO NO NO NO NO NO NO N	03/11/19 03/11/19 03/05/19 03/05/19 03/05/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/08/19 03/08/19 03/08/19 03/01/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUBIN RUGGIERO SAJDUK SANDARIA SANDERIA SANDERIA SANDERIA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO STALA STALA STALA STALA STALA STALA STENARI STEVARI STEVARI STEVARI STEVARI STEVARI STEVARI STEVARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STENARI STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STALA STA	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MILLENNY JULIO SHAREESE ANDREA HARVEY DANTEL DOODNAUT DANTISLA KEMIE ONEIL SHERYL ROGER STEPHEN HELEN STEPHEN HELEN STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN ALIS SURFRAZ JOSE ERAYAN TENESHA	P S LM MADMAFJ D OS AP JAJN DT	70235 70225 70225 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 71652 71651 70235 71012 71652 71651 70235 71012 71654 70210 71012 71658 70210 70210 70225 70210 71012 71658 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70210 70225 70210 70225 70210 70225 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70225 70210 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70220 70225 70225 70220 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225	\$88945.0000 \$109360.0000 \$199360.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$39453.0000 \$38183.0000 \$53443.0000 \$53443.0000 \$30706.0000 \$19360.0000 \$42500.0000 \$42500.0000 \$48183.0000 \$46385.0000 \$128212.0000 \$128212.0000 \$54394.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46395.0000 \$46395.0000 \$468945.0000 \$48945.0000 \$46737.0000 \$128222.0000 \$3183.0000 \$46737.0000 \$120210.0000	PROMOTED PROMOTED PROMOTED REMOTED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NO NO NO NO NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 00/05/19 00/05/19 00/05/19 00/05/19 00/05/19 00/05/19 00/05/19 000	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANDERIA SANDERIA SANDERIA SANDERIA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SAN	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIC SHARESZ ANDRA MD HARVEY DANIEL DOODNJUT DANEIL SHARESZ KEMIE ONEIL SHARESZ STEPHEN HELEN AKIL STEPHEN NADIA SURFRAZ JOSE ERAYAN TENESHA ALYSSA FUENAN PULIKEZZ PULIKEZZ	P S LM MADMAFJ D OS AP JAJN DTT	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70205 71012 71651 70235 71012 70210 71651 70210 71652 71651 70210 71012 7025 71012 7025 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70205 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70225 70210 70210 70225 70210 70225 70210 70210 70225 70210 70225 70210 70210 70225 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 70200 700000000	\$88945.0000 \$109360.0000 \$109360.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$3183.0000 \$3183.0000 \$3243.0000 \$3443.0000 \$3443.0000 \$3443.0000 \$3443.0000 \$34570.0000 \$42500.0000 \$43535.0000 \$46805.0000 \$5344.0000 \$45394.0000 \$45394.0000 \$45395.0000 \$46805.0000 \$46873.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.0000 \$4673.00000 \$4673.0000 \$4673.0000 \$4673.00000 \$4673.000000 \$4	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NG NG NG NG NG NG NG NG NG NG NG NG NG N	03/11/19 03/11/19 03/05/19 03/12/19 03/12/19 02/13/19 03/08/19 02/28/19 12/21/18 02/28/19 12/21/18 02/28/19 12/21/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/11/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/03/14/19 03/14/19 03/14/19 03/14/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUBIN RUGGIERO SAJJUK SANTGETO SANTGETO SANTSA SANDSETH SANTSA SANTSA SANTOS SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SELBY SEWELL SELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELL SEWELLS SEWELL SEWELLS SEWELL SEWELLS SEWELLS SEWELLS	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY MOEMI MIALENNY JULIC SHERKJE ANDREA MD DANIEL DONDAUT DANIEL DONDAUT DANTEL SHERKJE SHERKJE SHERKJE AKIL STEPHEN HELEN AKIL STEPHEN NADIA STEPHEN NADIA STEPHEN NADIA STEPHEN NADIA STEPHEN NADIA STEPHEN NADIA STEPHEN NADIA STEPHEN NADIA STEPHEN NADIA STEPHEN NADIA STEPHEN NADIA STEPHEN NADIA SUFRZZ JOSE ERAYAN TENESHA ALYSSA	P S L M M A D M A F J D O S A P J A J N D T T T	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70205 71012 71651 70205 71651 70212 71651 70212 70210 71012 70210 71012 70210 71012 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70102 70102 70102 70102 70102 70102 70102 70102 70102 70102	\$88945.0000 \$109360.0000 \$199360.0000 \$199360.0000 \$128212.0000 \$46737.0000 POLICE DEPA 'OR PERIOD ENDIN <u>SALARY</u> \$46737.0000 \$38183.0000 \$38183.0000 \$138183.0000 \$109360.0000 \$138183.0000 \$42500.0000 \$38183.0000 \$42500.0000 \$318183.0000 \$46385.0000 \$128212.0000 \$128212.0000 \$128121.0000 \$46805.0000 \$128212.0000 \$128212.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$468945.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED APPOINTED RESIGNED DISMISSED RESIGNED APPOINTED RESIGNED DISMISSED RETIRED APPOINTED RESIGNED DISMISSED RETIRED APPOINTED	NO NO NO NO NO NO NO NO NO NO NO NO NO N	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/05/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/03/14/19 03/14/19 03/14/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUBIN RUGGIERO SAJDUK SANDARIA SANDERIA SANDERIA SANDERIA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SEDA JR SELEY SEWELL SEWALL SHAHADAT SHAHADAT SHAHADAT SINGLETON SMALL SINGH SINGLETON SMALL SINGH SINGLETON SMALL SMITHWICK SOTO SOSA SPARMAN SPERAZZA STEIN STEWART STEWART STEWART STEWART STEWART STEN STEWART STEN STEWART STEN STEWART STEN STEN STEN STEN STEN STEN STEN STE	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MILLENNY JULIO SHARESSE ANDRA MD HARVEY DANLEL DOODNAUT DANEISHA KEMIE OODILU SHERYL ROGER STEPHEN HARVEY DANEISH SHERYL ROGER STEPHEN HARVEY DANEISH SHERYL SUBFRAZ JOSE BRAYAN TEMESKA ADIAN ADIAS SUBFRAZ JOSE BRAYAN TEMESKA ADIAS SUBFRAZ JOSE BRAYAN TEMESKA ADIAS SUFRAZ JOSE BRAYAN	P S L M M A D M A F J D O S A P J A J N D T T T T M	70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 71651 70235 71012 71651 70235 71012 71655 71012 71654 70210 71012 70205 70210 70215 70210 70225 70210 70235 70210 70235 70210 70235	\$88945.0000 \$109360.0000 \$19360.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$15.0000 \$38183.0000 \$38183.0000 \$42500.0000 \$128212.0000 \$128212.0000 \$128212.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.00000 \$46737.00000000000000000000000000000000000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED DROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RES	NO NO NO NO NO NO YES NO NO NO NO NO NO NO NO NO NO NO NO NO	03/11/19 03/11/19 03/05/19 03/05/19 03/05/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 12/21/18 11/07/17 03/11/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/14/19 03/14/19 03/14/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANDERIA SANDSETH SANDSETH SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO STALL STHEMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE SHARESSE ANDRA SHARESSE ANDRA SHARESSE ANDRA MD HARVEY DANIEL DOONAUT DANEISHA KUL SHERYL ROGER STEPHEN HELEN SHERYL STEPHEN HELEN STEPHEN HELEN SURFRAZ SURFRAZ SURFRAZ JOSE ERAYAN TENESHA ALRIAN SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ SURFRAZ	P S L M M A D M A F J D O S A P J A J N D T T T T M S	70235 70235 70235 70265 60817 TITLE 60817 71012 70205 71012 71022 71652 71012 71651 70210 71012 71655 71012 71655 71012 70210 71012 70205 71012 70205 70210 70205 70205 70205 70205 70205 70205 70205 70205 70205 70205 70210 70235 70205 70210 70235 70210 70235 70210 70215 70210 70215 70210 70215 70215 70215 70210 70215 70215 70215 70215 70210 70215 70215 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70000000000	\$88945.0000 \$109360.0000 \$199360.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN SALARY \$46737.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$42500.0000 \$438183.0000 \$42500.0000 \$438183.0000 \$4470.0000 \$4635.0000 \$128212.0000 \$128212.0000 \$45394.0000 \$45394.0000 \$45394.0000 \$45394.0000 \$46737.0000 \$46737.0000 \$1270.0000 \$46737.0000 \$45394.0000 \$46737.0000 \$45394.0000 \$46737.0000 \$45394.0000 \$46737.0000 \$46737.0000 \$45394.0000 \$46737.0000 \$45394.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.00000 \$4675.00000 \$4675.00000000 \$4675.00000 \$4	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED RESIGNED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED DISMISSED RESIGNED DISMISSED RETIRED RETIRED RETIRED RETIRED RETIRED RETIRED RETIRED RETIRED RETIRED APPOINTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED DISMISSED RETIRED RETIRED RETIRED RETIRED RETIRED APPOINTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED DISMISSED RETIRED RETIRED RETIRED RETIRED RETIRED RETIRED RETIRED RETIRED RETIRED	NG NG NG NG NG NG NG NG NG NG NG NG NG N	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 12/21/18 02/28/19 12/21/18 02/28/19 02/28/19 02/28/19 03/11/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/14/19 03/14/19 03/14/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANDARIA SANDERIA SANDERIA SANDERIA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA JASMINE NOEMI MIALENNY JULIO SHARESZ ANDRA MD HARVEY DONIEL SHARESZ ANDRA MD HARVEY DONIEL SHARESZ SHARESZ ANDRA MD HARVEY DONIEL SHERYL ROGER STEPHEN HELEN AKIL STEPHEN NADIA SURFRZ JOSE BRAYAN TENESIA ADRIAN SURFRZ JOSE BRAYAN FULKEZ FULMAN PULIKEZ TRAVIS SEAN PULIKEZ TATYANA BENJAMIN	P S L M M A D M A F J D O S A P J A J N D T T T M S J	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70205 71012 71651 70235 71651 70210 71012 70210 71012 70210 71012 70210 71012 70210 71012 7025 70210 70215 70210 7025 70210 7025 70210 7025 70210 7025 70210 7025 70210 7025 70210 7025 70210 7025 70210 7025 70210 7025 70210 7025 70210 7025 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70215 70210 70210 70215 70210 70215 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70102 70210 70102 70210 70102 70210 70102 7025 70210 70102 7025 70210 70102 7025 70210 70102 7025 70210 70102 7025 7025 70210 7025 7025 70210 7025 70210 70025 70210 7025 70210 70025 70210 7025 7025 70210 7025 7025 7025 7025 7025 7025 70210 7025 7025 7025 7025 7025 7025 7025 702	\$88945.0000 \$109360.0000 \$199360.0000 \$199360.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$38183.0000 \$38183.0000 \$331706.0000 \$138183.0000 \$42500.0000 \$34183.0000 \$42500.0000 \$34183.0000 \$42500.0000 \$34183.0000 \$4685.0000 \$128212.0000 \$128212.0000 \$4685.0000 \$46895.0000 \$46895.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$466737.0000 \$466737.0000 \$466737.0000 \$466737.0000 \$466737.0000 \$466737.0000 \$466737.0000 \$466737.0000 \$466737.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.0000 \$46605.00000 \$4	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED RESIGNED PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED PROMOTED RESIGNED RESIGNED PROMOTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NG NG NG NG NG NG NG NG NG NG NG NG NG N	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/03/11/19 02/28/19 03/14/19 03/14/19 03/14/19 03/14/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUBIN RUGGIERO SAJDUK SANDSEH SANTANA NAME SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SEDA JR SELEY SEWELL SENAHADAT SHAM SIMPSON SINGH SINGLETON SMALL SINGLETON SMITHWICK SOTO SOSA SPARTMAN SPERAZZA STEIN STEWART STEWART STEWART STEWART STEWART STENA STENA STENA STEIN STEWART STENA STENA STENA STEIN STEWART STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STENA STE	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MILLENNY JULIO SHARESSE ANDRA MD HARVEY DANIEL DOODNAUT DANEISHA KEMIE OODIL SHERYL NOGER STEPHEN HARVEY DANEISH SHERYL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STEPHEN AKIL STE	P S L M M A D M A F J D O S A P J A J N D T T T T M S J P	70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 71651 70235 71012 71651 70235 71012 71655 70210 71012 70210 70210 70210 70210 70225 70210 70225 70210 70225 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70235	\$88945.0000 \$109360.0000 \$19360.0000 \$128212.0000 \$46737.0000 POLICE DEPA 'OR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$15.0000 \$38183.0000 \$354343.0000 \$38183.0000 \$42500.0000 \$38183.0000 \$42500.0000 \$38183.0000 \$42500.0000 \$128212.0000 \$128212.0000 \$128212.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$463737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$48945.0000 \$40789.0000 \$40789.0000 \$40789.0000 \$40789.0000 \$40789.0000 \$40789.0000 \$40789.0000 \$40789.0000 \$40789.0000 \$40789.0000 \$48945.0000 \$88945.0000 \$88945.0000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED APPOINTED RESIGNED DISMISSED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NG NG NG NG NG NG NG NG NG NG NG NG NG N	03/11/19 03/11/19 03/05/19 03/05/19 03/05/19 03/12/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/11/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGIERO SAJDUK SANDARIA SANDSETH SANTANA NAME SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SEDA JR SELBY SEULJ SINGLETON SMALL SMITHWICK SOTO SOSA SPARKMAN SPERNAZIA STEVART SVENTORAITIS SYED TAVERAS TERMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MIALENNY JULIO SHARESSE ANDRA SHARESSE ANDRA SHARESSE ANDRA HARVEY DANIEL DOOINAUT DANRISKA KEMIE ONELL SHERYL SHERYL SHERYL SHERYL STEPHEN HELEN SURFRAZ JOSE ERAYAN TENESHA ALVISA FURMAN FURMAN PULIKEZ PULIKEZ PULIKEZ PULIKEZ PULIKEZ PULIKEZ PULIKEZ SEAN TATYANA BENJAMIN DANEL	P S L M M A D M A F J D O S A P J A J N D T T T T M S J P	70235 70235 70235 70265 60817 FTITLE 60817 71012 70205 71012 71022 71652 71012 71651 70210 71012 70205 71012 70210 71012 70205 71012 70205 70210 70205 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215 70215	\$88945.0000 \$109360.0000 \$109360.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN SALARY \$46737.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$3443.0000 \$3443.0000 \$3443.0000 \$3443.0000 \$109360.0000 \$34570.0000 \$4635.0000 \$128212.0000 \$128183.0000 \$4635.0000 \$128212.0000 \$4635.0000 \$128212.0000 \$4635.0000 \$128212.0000 \$4635.0000 \$128212.0000 \$45394.0000 \$4635.0000 \$48945.0000 \$46737.0000 \$46737.0000 \$46737.0000 \$4673.0000 \$4673.0000 \$4675.0000 \$48945.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.00000 \$4675.00000000 \$4675.00000 \$	PROMOTED PROMOTED PROMOTED REMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED DISMISSED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED DISMISSED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED RESIGNED PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NG NG NG NG NG NG NG NG NG NG NG NG NG N	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 12/21/18 02/28/19 12/21/18 02/28/19 02/28/19 03/05/19 03/16/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/10/19 02/28/19 03/14/19 03/14/19 11/07/18 03/05/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANDARIA SANDERIA SANDERIA SANDERIA SANDERIA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SELBY SEWALL SHAHADAT SINGLETON SMALL SINGLETON SMALL SINGLETON SMALL SINGLETON SMALL SINGLETON SMALL SUMITWICK SOTO SOSA SPERRAZAS STEIN STEUN STENAS STEIN STEUN STEVART SVED TAVERAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY MOEMI MIALENNY JULIO SHARESZ ANDREA MD HARVEY DANIEL DODNAUT DANIEL COGER STEPHEN HELEN AKIL SHERYL ROGER STEPHEN HELEN AKIL SHERYL ROGER STEPHEN HELEN AKIL SURFRAZ JOSE BRAYAN TENESIA ADRIAN SURFRAZ JOSE BRAYAN TENESIA ADRIAN SURFRAZ SEAN PULIKEZ TRAVIS SEAN PULIKEZ TATYANA BENJAMIN DANIEL TENESIA	P S L M MAADMAAFJ D OS AP JAJN DTTTTMSJPD	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70205 71012 70205 71012 7025 71052 71651 70210 71052 71651 70215 70210 70205 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70257 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70257 70210 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70257 70577 70577 70577777 7057777777777	\$88945.0000 \$109360.0000 \$199360.0000 \$199360.0000 \$128212.0000 \$46737.0000 POLICE DEPA YOR PERIOD ENDIN SALARY \$46737.0000 \$39453.0000 \$38183.0000 \$39453.0000 \$39453.0000 \$139360.0000 \$30706.0000 \$138183.0000 \$42500.0000 \$34183.0000 \$453430.0000 \$453434.0000 \$4635.0000 \$4635.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$46895.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4677.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.0000 \$4577.00000 \$4577.00000 \$4577.00000 \$4577.00000000 \$4	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED RESIGNED RESIGNED PROMOTED RESIGNED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED DISMISSED RETIRED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED DISMISSED RETIRED RESIGNED APPOINTED RETIRED RESIGNED APPOINTED RETIRED RETIRED RETIRED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED	NG NG NG PROV NG NG YES NG	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/11/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/05/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/12/19 03/	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUBIN RUGGIERO SAJJUK SANDSETH SANTANA NAME SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SINGH SINGLEU SIMPSON SINGH SINGLETON SMALL SIMPSON SINGLETON SMALL SINGH SINGLETON SMALL SUPSON SINGLETON SMALL SUPSON SINGH SINGLETON SMALL SUPSON SINGLETON SMALL SUPSON SINGLETON SMALL SUPSON SINGH SINGLETON SMALL SUPSON STEIN STEWART SVENTORAITIS SVED SYED SYED SYED TAVERAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS	PATRICIA DOMINIC MATEUSZ JORCE ERIC DOMINGA JASMINE NOEMI MIALENNY JULIO SHARESE ANDREA MD DANIEL DONIEL DANIEL DANIEL DANIEL COER STEPHEN HELEN SHERYL ROGER STEPHEN HELEN AKIL SHERYL SHERYL SHERYL SHERYL SHERYL JOSE STEPHEN HELEN AKIL JOSE STEPHEN NADIA SUEFRZ JOSE STEPHEN NADIA SUEFRZ JOSE STEPHEN NADIA SUEFRZ JOSE STEPHEN NADIA SUEFRZ JOSE STEPHEN NADIA SUEFRZ JOSE STEPHEN NADIA SUEFRZ JOSE STEPHEN NADIA SUEFRZ JOSE STEPHEN NADIA SUEFRZ JOSE SERYAN TENESHA ADRIA SEAN TATYAN ENLIKEZ SEAN TATYAN	P S LM MAADMAAFJ D OS AP JAJN DTTTTMSJPD M	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70210 71012 71012 71022 71651 70212 71651 70212 71651 70212 70210 71012 70210 71012 70210 71012 70210 71012 7025 70205 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70000000000	\$88945.0000 \$109360.0000 \$19360.0000 \$128212.0000 \$46737.0000 POLICE DEPA 'OR PERIOD ENDIN <u>SALARY</u> \$46737.0000 \$39453.0000 \$3183.0000 \$38183.0000 \$318183.0000 \$34570.0000 \$318183.0000 \$42500.0000 \$34183.0000 \$42500.0000 \$34183.0000 \$46805.0000 \$128212.0000 \$128212.0000 \$128212.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$46895.0000 \$48945.0000 \$48945.0000 \$48183.0000 \$48945.0000 \$48183.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$38183.0000 \$8945.0000 \$38183.0000 \$8945.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.00000 \$38183.00000 \$38183.00000 \$38183.0000000000000000000000000000000000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED R R R R R R R R R R R R R R R R R R R	NG NG NG NG NG NG NG NG NG NG NG NG NG N	03/11/19 03/11/19 03/05/19 03/05/19 03/05/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 10/01/18 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/08/19 03/08/19 03/10/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/11/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/14/19 03/	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUBIN RUGGIERO SAJDUK SANDARIA SANDERIA SANDERIA SANDERIA SANDERIA SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO STALAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO STALAGO STALAGO STALAGO STALAGO STALAGO STEIN STEWART SYED STEUN STEWART SYED STENAS TERRAZAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JAGNIDE SHARESS ANDRA SHARESSE ANDRA HARVEY DAILE SHARESSE ANDRA HARVEY DANIEL SHERYL ONBILL SHERYL SHERYL SHERYL SHERYL SHERYL STEPHEN HELEN SURPRAZ JOSE BRAYAN TENESHA AKIL STEPHEN NADIA SURPRAZ JOSE BRAYAN TENESHA ALNISA SEAN TATANA BENJANIN DANIEL SEAN	P S L M M A D M A F J D O S A P J A J N D T T T T M S J P D M F	70235 70225 70265 60817 TITLE NUM 60817 71012 70205 71012 70210 71012 70210 71012 7025 71012 71651 70235 71012 71651 70210 71012 70255 71012 71055 70210 70210 70205 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70225 70210 70235 70225 70210 70235 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 70225 7025 70	\$88945.0000 \$109360.0000 \$19360.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN \$3453.0000 \$15.0000 \$38183.0000 \$38183.0000 \$34570.0000 \$42500.0000 \$128212.0000 \$128212.0000 \$128212.0000 \$12813.0000 \$46385.0000 \$128212.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$46385.0000 \$128212.0000 \$128212.0000 \$12813.0000 \$46385.0000 \$46385.0000 \$128212.0000 \$128212.0000 \$128212.0000 \$12813.0000 \$46385.0000 \$46737.0000 \$38183.0000 \$46737.0000 \$38183.0000 \$46737.0000 \$38183.0000 \$46737.0000 \$38183.0000 \$46737.0000 \$12000 \$120000 \$120000 \$120000 \$38183.0000 \$46737.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$40786.0000 \$8945.0000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RES	NG NG PROV NG PROV NG YES NG	03/11/19 03/11/19 03/05/19 03/05/19 03/05/19 03/12/19 03/12/19 03/14/19 03/08/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/08/19 03/05/19 03/08/19 03/08/19 03/08/19 03/08/19 03/08/19 03/08/19 03/08/19 03/08/19 03/08/19 03/08/19 03/12/19 03/11/19 03/03/12 19 03/03/12 19 03/03/12 19 03/03/19 03/03/12 19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANDERIA SANDERIA SANDSETH SANTANA SANDSETH SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO STALL STHOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMA	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MIJAENNY JULIC SHARESSE ANDRA SHARESSE ANDRA MD HARVEY DANIEL SHARESSE ANDRA HARVEY DANIEL SHARESSE SHARESSE NORA SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE PULIKEEZ TRAVIS SEANAN PULIKEEZ FURMAN PULIKEZ FURMAN PULIKEZ FURMAN SEANAN ALYSSA FURMAN ALYSSA FURMAN SEANAN ALYSSA FURMAN SEANAN TENESIA SEANAN ALYSSA FURMAN SEANAN SEANAN ALYSSA FURMAN SEANAN SEANAN SEANAN ALYSSA FURMAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEAN	P S LM MADMAFJ D OS AP JAJN DTTTTMSJPD MFC	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70205 71012 71651 70215 71012 70210 71651 70215 71012 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70257 70257 70257 70257 70257 70257 70257 70257 70210 70257 70257 70257 70210 70257 70210 70257 70257 70210 70257 70210 70257 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70210 70257 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70257 70012 70257 70012 70025 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 700257 700257 700257 700257 700257 700257 700257 7000	\$88945.0000 \$109360.0000 \$109360.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN SALARY \$46737.0000 \$15.0000 \$38183.0000 \$38183.0000 \$3443.0000 \$3443.0000 \$3443.0000 \$128212.0000 \$128212.0000 \$128212.0000 \$42550.0000 \$44805.0000 \$43843.0000 \$42500.0000 \$43843.0000 \$42500.0000 \$4635.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$48945.0000 \$46805.0000 \$48945.0000 \$46737.0000 \$46737.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46730.0000 \$46730.0000 \$46730.0000 \$46730.0000 \$46730.0000 \$46730.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.000000 \$475.0000 \$475.00000 \$475.000000000000000000000000	PROMOTED PROMOTED PROMOTED PROMOTED RESTIRED ACTION DISMISSED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED	NG NG NG PROV NG NG YES NG	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 02/13/19 03/08/19 02/28/19 12/21/18 02/28/19 12/21/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/11/19 03/14/19 03/14/19 03/14/19 03/11/19 03/03/119 03/03/119 03/03/119 03/03/119 03/03/119 03/03/119 03/03/119	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUBIN RUGGIERO SAJDUK SANDSETH SANTANA SANDSETH SANTANA SANDSETH SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SINGH SINGLEON SINGH SINGLEON SINGH SINGLETON SMALL SIMPSON SINGH SINGLETON SMALL SINGH SINGLETON SMALL SUPSON STRUCK SOTO SOSA SPARMAN SPERAZZA STEIN STEWART SVENTORAITIS SVED TAVERAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA JASMINE NOEMI MIALENNY JULIO SHARESZ ANDREA MD DANIEL DODNIJU DANIEL DODNIJU DANIEL SHARESZ ANDREA MD DANIEL SHARESZ STEPHEN HELEN AKIL STEPHEN HELEN AKIL STEPHEN NADIA SUEFRZ JOSE BRAYAN TENESIA ADRIA SUEFRZ JOSE BRAYAN TENESIA ADRIA SUEFRZ JOSE BRAYAN TENESIA ADRIA SUEFRZ JOSE BRAYAN TENESIA ADRIA SUEFRZ JOSE BRAYAN TENESIA ADRIA SUEFRZ JOSE BRAYAN TENESIA ADRIA SENJANIN DANIEL TENECIA LIZA	P S L M M A D M A F J D O S A P J A J N D T T T T M S J P D M F	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 71025 71012 71651 70205 71012 71651 70205 71012 70210 71012 70210 71012 70210 71012 70255 70210 70205 70210 70235 70210 70235 70210 70235 70210 71012 60817 70215 70210 70235 70210 70235 70210 71012 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70235 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70225 70102 70025 70012 70025 70012 70025 70012 70025 70012 70025 70012 70025 70012 70025 70012 70025 70012 70025 70012 70025 70012 70025 70012 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025 70025	\$88945.0000 \$109360.0000 \$19360.0000 \$128212.0000 \$46737.0000 POLICE DEPA 'OR PERIOD ENDIN <u>SALARY</u> \$46737.0000 \$39453.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$109360.0000 \$128212.0000 \$128122.0000 \$128122.0000 \$128122.0000 \$128122.0000 \$128122.0000 \$128122.0000 \$128122.0000 \$128120000 \$128122.0000 \$128120000 \$12812.0000 \$46805.0000 \$46805.0000 \$46895.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$48945.0000 \$138183.0000 \$40786.0000 \$38183.0000 \$40786.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000 \$38183.0000	PROMOTED PROMOTED PROMOTED PROMOTED RETIRED RETIRED ACTION DISMISSED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED APPOINTED RESIGNED DISMISSED RETIRED PROMOTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED DISMISSED RETIRED PROMOTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED RESIGNED APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESI	NG NG NG NG NG NG NG NG NG NG NG NG NG N	03/11/19 03/11/19 03/05/19 03/05/19 03/05/19 03/12/19 03/12/19 03/08/19 03/08/19 02/28/19 10/01/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/08/19 03/08/19 03/08/19 03/08/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/11/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/11/19 03/03/12/19 03/03/12/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/11/19 03/03/11/19 03/03/12/19 03/03/11/19 03/03/11/19 03/03/12/19 03/03/12/19 03/03/11/19 03/03/11/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/12/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/	056 056 056 056 056 056 056 056 056 056
ROTTE RUBIN RUGGIERO SAJDUK SANDERIA SANDERIA SANDSETH SANTANA SANDSETH SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO SANTIAGO STALL STHOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMAS THOMA	PATRICIA DOMINIC MATEUSZ JORGE ERIC DOMINGA DAISY JASMINE NOEMI MIJAENNY JULIC SHARESSE ANDRA SHARESSE ANDRA MD HARVEY DANIEL SHARESSE ANDRA HARVEY DANIEL SHARESSE SHARESSE NORA SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE SHARESSE PULIKEEZ TRAVIS SEANAN PULIKEEZ FURMAN PULIKEZ FURMAN PULIKEZ FURMAN SEANAN ALYSSA FURMAN ALYSSA FURMAN SEANAN ALYSSA FURMAN SEANAN TENESIA SEANAN ALYSSA FURMAN SEANAN TENESIA SEANAN ALYSSA FURMAN SEANAN SEANAN ALYSSA FURMAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEANAN SEAN	P S LM MADMAFJ D OS AP JAJN DTTTTMSJPD MFC	70235 70235 70235 70265 60817 TITLE NUM 60817 71012 70205 71012 70205 71012 71651 70215 71012 70210 71651 70215 71012 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70255 70210 70257 70257 70257 70257 70257 70257 70257 70257 70210 70257 70257 70257 70210 70257 70210 70257 70257 70210 70257 70210 70257 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70257 70210 70210 70257 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70210 70257 70012 70257 70012 70025 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 70012 700257 700257 700257 700257 700257 700257 700257 700257 7000	\$88945.0000 \$109360.0000 \$109360.0000 \$128212.0000 \$46737.0000 POLICE DEPA OR PERIOD ENDIN SALARY \$46737.0000 \$15.0000 \$38183.0000 \$38183.0000 \$3443.0000 \$3443.0000 \$3443.0000 \$128212.0000 \$128212.0000 \$128212.0000 \$42550.0000 \$44805.0000 \$43843.0000 \$42500.0000 \$43843.0000 \$42500.0000 \$4635.0000 \$46805.0000 \$46805.0000 \$46805.0000 \$48945.0000 \$46805.0000 \$48945.0000 \$46737.0000 \$46737.0000 \$46629.0000 \$46629.0000 \$46629.0000 \$46730.0000 \$46730.0000 \$46730.0000 \$46730.0000 \$46730.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$4675.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.0000 \$475.00000 \$475.0000 \$475.000000000000000000000000000000000000	PROMOTED PROMOTED PROMOTED PROMOTED RESTIRED ACTION DISMISSED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED PROMOTED	NG NG NG PROV NG NG YES NG	03/11/19 03/11/19 03/05/19 03/05/19 03/12/19 02/13/19 03/08/19 02/28/19 12/21/18 02/28/19 12/21/18 02/28/19 02/28/19 02/28/19 02/28/19 02/28/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/05/19 03/11/19 03/14/19 03/14/19 03/14/19 03/11/19 03/03/119 03/03/119 03/03/119 03/03/119 03/03/119 03/03/119 03/03/119	056 056 056 056 056 056 056 056 056 056

ALTERS	CAMILLE	A	71012	\$38183.0000	APPOINTED	NO	02/28/19	056
ASHINGTON	MARTIS		70205	\$15.0000	RESIGNED	YES	02/27/19	056
ATSON	BREANNA	L	71012	\$38183.0000	APPOINTED	NO	03/14/19	056
HEATLEY	KISHA		71012	\$38183.0000	APPOINTED	NO	02/28/19	056
ILLIAMS	MARCY FA		71012	\$39453.0000	RESIGNED	NO	03/02/19	056
VILLIAMS-JOHNSO			71012	\$38183.0000	APPOINTED	NO	02/28/19	056
ITHAM	MICHAEL	в	82800	\$154500.0000	APPOINTED	YES	03/03/19	056
				POLICE DEPA	RTMENT			
			FC	R PERIOD ENDIN				
			TITLE					
IAME			NUM	SALARY	ACTION		EFF DATE	AGENCY
IONG IREN	VICKILYN	М	70205 60817	\$15.0000	RESIGNED	YES	02/12/19 02/08/19	056
RIGHT	LETESHA WINSTON	J	70235	\$46737.0000 \$88945.0000	DISMISSED PROMOTED	NO NO	02/08/19	056 056
π.	GUI YAN	č	10144	\$35330.0000	DECEASED	NO	12/16/18	056
ASIN	AROOSA	A	70235	\$88945.0000	PROMOTED	NO	03/11/19	056
EH	JACK	S	71651	\$38625.0000	INCREASE	NO	06/15/18	056
IIMMERMAN UNIGA	CHRISTOP APRIL	С	7026B 70235	\$138089.0000 \$88945.0000	RETIRED PROMOTED	NO NO	10/01/18 03/11/19	056 056
IUNIGA	AFRID		/0255	\$00545.0000	FROMOTED	NO	05/11/15	050
				FIRE DEPAR	TMENT			
				R PERIOD ENDIN	G 03/22/19			
			TITLE	CALADY	ACTION	DROW		AGENOV
IAME AGOSTA	MICHAEL	J	NUM 70360	\$90457.0000	ACTION PROMOTED	NO	EFF DATE 03/02/19	AGENCY 057
KAN	KENNETH	L	70392	\$95527.0000	PROMOTED	NO	03/02/19	057
LTMAN	SCOTT	R	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
MATO	SHAWNA	N	53053	\$50604.0000	RESIGNED	NO	03/01/19	057
NDOH	ISAAC	J	31662	\$59872.0000	INCREASE	YES	02/17/19	057
TWARU	BALKRISH		10124	\$50763.0000	APPOINTED	YES	11/20/18	057
BAKRADZE BALCOMBE	SOFIA EVAN	в	10124 70365	\$75000.0000 \$110395.0000	INCREASE PROMOTED	NO NO	03/10/19 03/02/19	057 057
BOWLES	DAVID	R	31105	\$52008.0000	TRANSFER	NO	01/06/19	057
BUSH JR	WILLIAM	М	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
CAPOBIANCO	RICHARD		70360	\$90457.0000	PROMOTED	NO	03/02/19	057
CASSIDY	MATTHEW	D	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
CAVARRETTA CLOUGH	JOSEPH PETER	R	70365 53053	\$110395.0000 \$37914.0000	PROMOTED	NO NO	03/02/19 03/13/19	057 057
COLLIER	BRIAN	J	70365	\$110395.0000	DECREASE PROMOTED	NO	03/02/19	057
CONNOLLY	JOSEPH	s	70392	\$95527.0000	PROMOTED	NO	03/02/19	057
COOK	MICHAEL	J	70365	\$110395.0000	PROMOTED	NO	03/02/19	057
CORRIGAN	JAMES	М	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
RESPO	LUIS	E	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
CRICHLOW	CRAIG CHRISTOP	K D	70365 70392	\$110395.0000 \$95527.0000	PROMOTED PROMOTED	NO NO	03/02/19 03/02/19	057 057
CUCCIA	MARK	D	70392	\$95527.0000	PROMOTED	NO	03/02/19	057
CURCIO	KATHRYN	М	10251	\$50021.0000	INCREASE	NO	02/24/19	057
DARETANY	JOSEPH	Ρ	70392	\$95527.0000	PROMOTED	NO	03/02/19	057
DAVID	MICHAEL	R	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
DEANGELIS DELARABA JR	MICHAEL GARY	D	70365 70365	\$110395.0000 \$110395.0000	PROMOTED PROMOTED	NO NO	03/02/19 03/02/19	057 057
ELLETT	JOSHUA	c	70392	\$95527.0000	PROMOTED	NO	03/02/19	057
IGIULIO	CHRISTOP	J	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
DRAGUC	ADRIAN	С	70392	\$95527.0000	PROMOTED	NO	03/02/19	057
YCHKOWSKI	JARED	M	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
FERGUSON FERNANDEZ	RACHEL RICK	S	71010 70392	\$38403.0000 \$95527.0000	RESIGNED PROMOTED	YES NO	03/09/19 03/02/19	057 057
				FIRE DEPAR			00,02,25	
				R PERIOD ENDIN	G 03/22/19			
			TITLE		1 (111)	DDO11		1 011101
NAME FOGELSTROM JR.	DAVID	Ρ	NUM 70365	SALARY \$110395.0000	ACTION PROMOTED	NO	EFF DATE 03/02/19	AGENCY 057
RANK	SHAUN	M	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
AGLIANO	MICHAEL	F	92587	\$65320.0000	APPOINTED	YES	03/03/19	057
AMBINO	STEFANO		70360	\$90457.0000	PROMOTED	NO	03/02/19	057
EORGE	LARRY	_	70392	\$95527.0000	PROMOTED	NO	03/02/19	057
GODSEY GONZALEZ	CHRISTOP MARK	R D	53053	\$53163.0000 \$90457.0000	RESIGNED	NO NO	03/08/19	057 057
GONZALEZ	GARRETT	F	70360 70392	\$95527.0000	PROMOTED PROMOTED	NO	03/02/19 03/02/19	057
IAAS	MICHAEL	A	70392	\$95527.0000	PROMOTED	NO	03/02/19	057
IANSEN	GREGORY	М	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
IANSEN	LEIF	C	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
IORNBROOK	STEVEN	C M	53054	\$65226.0000	RESIGNED	NO	02/28/19	057
IOY IUSSEIN	WAYNE REHAM	М	70365 12749	\$110395.0000 \$45123.0000	PROMOTED APPOINTED	NO YES	03/02/19 01/10/19	057 057
VERRY	MICKEY		91232	\$56305.0000	RETIRED	NO	09/28/18	057
IOST III	CARL		70365	\$110395.0000	PROMOTED	NO	03/02/19	057
ELLY	ROBERT	J	70392	\$95527.0000	PROMOTED	NO	03/02/19	057
CENNEDY	CHRISTOP	,	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
KESSLER KHODAI	SAMUEL ALEX	A P	70365 70392	\$110395.0000 \$95527.0000	PROMOTED PROMOTED	NO NO	03/02/19 03/02/19	057 057
LITT	SEAN	F	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
OSTER	KEVIN		70360	\$90457.0000	PROMOTED	NO	03/02/19	057
ABARBERA	TRACEY	E	53054	\$65866.0000	RESIGNED	NO	03/12/19	057
ARSEN	KEITH	D	70360	\$90457.0000	PROMOTED	NO	03/02/19	057
JEVY	JARRETT NANCY	J B	70392 53055	\$95527.0000 \$71202.0000	PROMOTED RETIRED	NO NO	03/02/19 02/13/19	057 057
JOWE	TAMEKA	ь Y	21744	\$84301.0000	APPOINTED	YES	02/13/19 03/03/19	057
JOWRY	BRENDAN	D	70392	\$95527.0000	PROMOTED	NO	03/02/19	057
UCARELLI	CHRISTOP		70392	\$95527.0000	PROMOTED	NO	03/02/19	057
IAHONEY	THOMAS	P	70365	\$110395.0000	PROMOTED	NO	03/02/19	057
IANNING IARCANO JR	JOHN JESUS	E	70392 70360	\$95527.0000 \$90457.0000	PROMOTED PROMOTED	NO NO	03/02/19 03/02/19	057 057
IARIN	KEVIN		70380	\$38403.0000	RESIGNED	YES	03/02/19	057
ARTIN	KENNETH	G	70365	\$110395.0000	PROMOTED	NO	03/02/19	057
ASCIA	JOSEPH	М	70365	\$110395.0000	PROMOTED	NO	03/02/19	057
ATTHEWS	JOSHUA	т	90733	\$391.6000	APPOINTED	YES	03/10/19	057
MAZARAKIS JR.	GEORGE		70392	\$95527.0000	PROMOTED	NO	03/02/19	057

THE CITY RECORD

FRIDAY, APRIL, 19, 2019

MCCARTHY	BRENDAN I	70392	\$95527.0000	PROMOTED	NO	03/02/19	057	GILBERT	BRANDY	708	\$46737.0000	RESIGNED	NO	02/26/19	067
MCGOVERN	JOSEPH P	70365	\$110395.0000	PROMOTED	NO	03/02/19	057	GILBERT	TODD I	102	\$40629.0000	APPOINTED	NO	02/10/19	067
MCGROARTY JR	CHARLES D	90723	\$236.8000	APPOINTED	YES	03/10/19	057	GLASGOW	RHONDA N	523	\$47250.000	APPOINTED	NO	08/05/18	067
MEANEY	MARTIN P	70365	\$110395.0000	PROMOTED	NO	03/02/19	057	GRANT	MICHAEL A	522	\$44426.000	RESIGNED	YES	03/12/19	067
MICHAEL	KATERINA	53053	\$35254.0000	APPOINTED	NO	02/19/19	057	GREEN	ROBIN V	7 560	\$36327.0000	RETIRED	YES	03/03/19	067
MOLA	ROBERT	31121	\$67138.0000	INCREASE	NO	03/03/19	057	HANLEY	NYITA A	100	C \$85000.0000	INCREASE	NO	03/10/19	067
MOSHER	STEPHEN M	91972	\$387.0300	PROMOTED	NO	03/03/19	057	HOWELL	CARESSE E	5230	\$53519.0000	TERMINATED	NO	02/10/19	067
MULLIN	CLETUS S	70392	\$95527.0000	PROMOTED	NO	03/02/19	057	IRWIN	MARTIN A	5304	\$170.0000	INCREASE	YES	03/03/19	067
MURPHY	CHRISTOP S	70365	\$110395.0000	PROMOTED	NO	03/02/19	057	JOHNSON	COLIN N	1 300	\$67523.0000	INCREASE	YES	02/10/19	067
NAPOLITANO	THOMAS P	70365	\$110395.0000	PROMOTED	NO	03/02/19	057	JOHNSON	PHYLLIS F	\$ 523	\$83700.0000	RETIRED	NO	03/03/19	067
PANEBIANCO	DAVID	70365	\$110395.0000	PROMOTED	NO	03/02/19	057	KAYHAREE	NERVANA F	102	\$45168.0000	INCREASE	NO	03/10/19	067
PATRONI	MICHAEL S		\$90457.0000	PROMOTED	NO	03/02/19	057	LEE	LEUNG S				NO	12/20/17	067
PEREZ	KELVIN	70360	\$90457.0000	PROMOTED	NO	03/02/19	057	LEWIS	ASHLEY	560			YES	03/10/19	067
PERGAMO	MICHAEL P	70360	\$90457.0000	PROMOTED	NO	03/02/19	057	LUCIEN		\$ 5230	-		NO	03/03/19	067
								MARTINEZ	JULYSSA A		-		YES	03/10/19	067
			FIRE DEPAR					MASI	ROZALYN N		•	INCREASE	YES	03/03/19	067
			FOR PERIOD ENDIN	IG 03/22/19				MAY	MARGARET S				YES	02/24/19	067
		TITLE						MCKELLAR JR	DAVID I		-	TERMINATED	YES	03/06/19	067
NAME		NUM	SALARY	ACTION		EFF DATE	AGENCY	MCLEAN	BRITTANY N		-	TERMINATED	NO	03/01/19	067
PERSAUD	MICHAEL G		\$59872.0000	INCREASE	YES	02/17/19	057	MENENDEZ	DANIELLE N			RESIGNED	YES	03/01/19	067
PLANT	JOHN T		\$90457.0000	PROMOTED	NO	03/02/19	057	MORAN MORRIS	ROSEMARY KENDRA A	300	-	RETIRED	YES NO	02/22/19	067 067
RICH		53055	\$71202.0000	RETIRED	NO	03/07/19	057				-			02/10/19	
RICHARDSON	THOMAS J		\$235139.0000	INCREASE	NO	02/27/19	057	MORRIS		7 5228 7 3009	-		YES	03/03/19	067 067
ROFRANO	THOMAS J		\$37914.0000	DECREASE	NO	03/13/19	057	NATELSON	RACHEL I		-		NO	03/10/19	
SANDOVAL UBES	FATIMA P		\$30924.0000	TRANSFER	NO	07/09/17	057	PALMORE	VALERIE DATRICIA I	100 523		INCREASE	NO NO	02/24/19	067 067
SANTINI	GREGORY P		\$95527.0000	PROMOTED	NO	03/02/19	057		PATRICIA I			TERMINATED RESIGNED		03/07/19	067
SEVER	RIZA N		\$43901.0000	RESIGNED	NO	03/01/19	057	PERDOCH QUINTERO	JOHN F WILLIAM C		-		YES	03/12/19 03/10/19	
SEWELL	KEVIN N	53053	\$37914.0000	DECREASE	NO	03/13/19	057	RAMOS RIVERA	WILLIAM (ENGEL A	•			NO YES	03/10/19	067 067
SHEA SMITH	THOMAS J JEFFREY P		\$95527.0000 \$95527.0000	PROMOTED PROMOTED	NO NO	03/02/19	057 057	ARIOD KIVERA	H LIGHT	. 010	.5 .57054.0000	HLLOINIED	100	33/10/19	007
SODANO	SALVATOR L		\$90457.0000	PROMOTED	NO	03/02/19 03/02/19	057				ADMIN FOR CHIL	DREN'S SVCS			
SUDNIK	JOHN	70380	\$235526.0000	INCREASE	YES		057				FOR PERIOD ENDI				
TOBIASSEN	THOMAS C		\$90457.0000	PROMOTED	NO	02/27/19 03/02/19	057			TIT		NG 03/22/23			
TYSON				RETIRED			057	NAME		NUI		ACTION	PROV	EFF DATE	AGE
	HERMAN J JOHN M		\$85292.0000 \$95527.0000	PROMOTED	NO NO	06/16/18		RAO	SUCHET S				YES	03/03/19	067
URRICO	JOHN M	10392	\$95527.0000												
113 00 3	MA TM	70265	611030E 0000			03/02/19	057				•				067
	NAIM	70365	\$110395.0000	PROMOTED	NO	03/02/19	057	ROSENMAN	ROGER I	300	\$85029.0000	RESIGNED	NO	03/03/19	067
VERNON	RICHARD G	70360	\$90457.0000	PROMOTED PROMOTED	NO NO	03/02/19 03/02/19	057 057	ROSENMAN SCOTT	ROGER I CHIFFON F	2 300 5 560	\$85029.0000 \$8 \$60403.0000	RESIGNED INCREASE	NO YES	03/03/19 03/03/19	067
VERNON WAYNE	RICHARD G JUSTIN E	70360 70360	\$90457.0000 \$90457.0000	PROMOTED PROMOTED PROMOTED	NO NO NO	03/02/19 03/02/19 03/02/19	057 057 057	ROSENMAN SCOTT SIMONS	ROGER F CHIFFON F RONALD C	9 3008 5609 7 1029	37 \$85029.0000 58 \$60403.0000 51 \$40629.0000	RESIGNED INCREASE APPOINTED	NO YES NO	03/03/19 03/03/19 02/10/19	067 067
VERNON WAYNE WEIHE JR.	RICHARD G JUSTIN E GERARD A	70360 70360 70392	\$90457.0000 \$90457.0000 \$95527.0000	PROMOTED PROMOTED PROMOTED PROMOTED	NO NO NO NO	03/02/19 03/02/19 03/02/19 03/02/19	057 057 057 057	ROSENMAN SCOTT SIMONS SMALL	ROGER F CHIFFON F RONALD C KEMIE	2 300 560 102 522	\$85029.0000 \$85029.0000 \$60403.0000 \$1 \$40629.0000 \$7 \$44426.0000	RESIGNED INCREASE APPOINTED APPOINTED	NO YES NO YES	03/03/19 03/03/19 02/10/19 02/24/19	067 067 067
VERNON WAYNE WEIHE JR. WILSON	RICHARD G JUSTIN E GERARD A MICHAEL R	70360 70360 70392 70360	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED	NO NO NO NO	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19	057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH	ROGER E CHIFFON F RONALD C KEMIE ANNMARIE	2 3000 5609 1029 5220 0672	37 \$85029.0000 58 \$60403.0000 51 \$40629.0000 37 \$44426.0000 21 \$64863.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT	70360 70360 70392 70360 70360	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$90457.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED	NO NO NO NO NO	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 03/02/19	057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL	ROGER F CHIFFON F RONALD C KEMIE	2 3000 5609 51029 5220 0672	37 \$85029.0000 58 \$60403.0000 51 \$40629.0000 37 \$44426.0000 21 \$64863.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES	03/03/19 03/03/19 02/10/19 02/24/19	067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT	RICHARD G JUSTIN E GERARD A MICHAEL R	70360 70360 70392 70360	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED	NO NO NO NO	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19	057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH	ROGER E CHIFFON F RONALD C KEMIE ANNMARIE	2 3000 5609 1029 5220 0672	37 \$85029.0000 58 \$60403.0000 51 \$40629.0000 37 \$44426.0000 21 \$64863.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT SAMUEL	70360 70360 70392 70360 70360 53053	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$90457.0000 \$35254.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED	NO NO NO NO NO	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19	057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH	ROGER E CHIFFON F RONALD C KEMIE ANNMARIE	2 3000 5609 1029 5220 0672	37 \$85029.0000 58 \$60403.0000 51 \$40629.0000 37 \$44426.0000 21 \$64863.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT SAMUEL	70360 70392 70360 70360 53053 53053	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$90457.0000 \$35254.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED	NO NO NO NO NO NO	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19	057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER E CHIFFON E RONALD C KEMIE ANNMARIE SANDY E	2 3008 5609 7 1029 5228 067 1 5238	\$85029.0000 \$8 \$60403.0000 \$1 \$40629.0000 \$7 \$44426.0000 \$1 \$64863.0000 \$6 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT SAMUEL	70360 70360 70392 70360 70360 53053 53053	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED	NO NO NO NO NO NO	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19	057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH	ROGER E CHIFFON E RONALD C KEMIE ANNMARIE SANDY E	2 3008 5609 7 1029 5228 067 1 5238	\$85029.0000 \$8 \$60403.0000 \$1 \$40629.0000 \$7 \$44426.0000 \$1 \$64863.0000 \$6 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT SAMUEL	70360 70360 70392 70360 70360 53053 53053	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$30457.0000 \$35254.0000 \$35254.0000 \$35254.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED	NO NO NO NO NO NO	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19	057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER E CHIFFON E RONALD C KEMIE ANNMARIE SANDY E	2 3008 5609 7 1029 5228 067 1 5238	\$85029.0000 \$8 \$60403.0000 \$1 \$40629.0000 \$7 \$44426.0000 \$1 \$64863.0000 \$6 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT SAMUEL	70360 70392 70360 70360 53053 53053 NY	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$30457.0000 \$35254.0000 \$35254.0000 \$35254.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED	NO NO NO NO NO	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19	057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER E CHIFFON E RONALD C KEMIE ANNMARIE SANDY E	2 3008 5609 7 1029 5228 067 1 5238	\$85029.0000 \$8 \$60403.0000 \$1 \$40629.0000 \$7 \$44426.0000 \$1 \$64863.0000 \$6 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT SAMUEL RONALD S	70360 70360 70392 70360 70360 53053 53053 53053 NY TITLE NUM 95615	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 C DEPT OF VETERA FOR PERIOD ENDIN SALARY \$120000.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED ACTION RESIGNED	NO NO NO NO NO PROV	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 EFF DATE 04/15/18	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER E CHIFFON E RONALD C KEMIE ANNMARIE SANDY E	2 3008 5609 7 1029 5228 067 1 5238	\$85029.0000 \$8 \$60403.0000 \$1 \$40629.0000 \$7 \$44426.0000 \$1 \$64863.0000 \$6 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT SAMUEL RONALD	70360 70360 70392 70360 53053 53053 53053 NY TITLE NUM	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 C DEPT OF VETERA FOR PERIOD ENDIN SALARY \$120000.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED NS' SERVICES NG 03/22/19 ACTION	NO NO NO NO NO PROV	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 EFF DATE	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER E CHIFFON E RONALD C KEMIE ANNMARIE SANDY E	2 3008 5609 7 1029 5228 067 1 5238	\$85029.0000 \$8 \$60403.0000 \$1 \$40629.0000 \$7 \$44426.0000 \$1 \$64863.0000 \$6 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT S SAMUEL RONALD S VENKAT	70360 70360 70392 70360 70360 53053 53053 53053 NY TITLE NUM 95615	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 C DEPT OF VETERA FOR PERIOD ENDIN SALARY \$120000.0000 \$139571.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED ACTION RESIGNED RESIGNED	NO NO NO NO NO PROV	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 EFF DATE 04/15/18	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER E CHIFFON E RONALD C KEMIE ANNMARIE SANDY E	2 3008 5609 7 1029 5228 067 1 5238	\$85029.0000 \$8 \$60403.0000 \$1 \$40629.0000 \$7 \$44426.0000 \$1 \$64863.0000 \$6 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT S SAMUEL RONALD S VENKAT	70360 70360 70392 70360 53053 53053 53053 NY TITLE <u>NUM</u> 95615 0668A	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 C DEPT OF VETERA FOR PERIOD ENDIN <u>SALARY</u> \$120000.0000 \$139571.0000 ADMIN FOR CHILD	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED NS' SERVICES IG 03/22/19 ACTION RESIGNED RESIGNED RESIGNED	NO NO NO NO NO PROV	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 EFF DATE 04/15/18	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER E CHIFFON E RONALD C KEMIE ANNMARIE SANDY E	2 3008 5609 7 1029 5228 067 1 5238	\$85029.0000 \$8 \$60403.0000 \$1 \$40629.0000 \$7 \$44426.0000 \$1 \$64863.0000 \$6 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT S SAMUEL RONALD S VENKAT	70360 70360 70392 70360 53053 53053 53053 NY TITLE <u>NUM</u> 95615 0668A	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 C DEPT OF VETERA FOR PERIOD ENDIN SALARY \$120000.0000 \$139571.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED NS' SERVICES IG 03/22/19 ACTION RESIGNED RESIGNED RESIGNED	NO NO NO NO NO PROV	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 EFF DATE 04/15/18	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER H CHIFFON H RONALD C KEMIE ANNMARIE SANDY N	9 3001 5603 5224 0677 1 5234	77 \$85029.0000 58 \$60403.0000 51 \$40629.0000 57 \$44426.0000 56 \$48195.0000 D	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV <u>NAME</u> MOTUPALLI OTHMAN	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT S SAMUEL RONALD S VENKAT	70360 70360 70392 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 C DEPT OF VETERA FOR PERIOD ENDIN \$12000.0000 \$139571.0000 ADMIN FOR CHILD FOR PERIOD ENDIN	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED NS' SERVICES IG 03/22/19 ACTION RESIGNED RESIGNED DREN'S SVCS IG 03/22/19	NO NO NO NO NO NO PROV YES YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 EFF DATE 04/15/18 03/10/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER H CHIFFON H RONALD C KEMIE ANNMARIE SANDY N	9 3001 5603 5224 0677 1 5234	37 \$85029.0000 58 \$60403.0000 51 \$40629.0000 57 \$44426.0000 57 \$64863.0000 56 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT SAMUEL RONALD F VENKAT JAMAL	70360 70360 70392 70360 70360 53053 53053 NY TITLE NUM TITLE NUM	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 C DEPT OF VETERA FOR PERIOD ENDIN \$120000.0000 \$139571.0000 ADMIN FOR CHILL FOR PERIOD ENDIN SALARY	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED NS' SERVICES IG 03/22/19 ACTION RESIGNED DREN'S SVCS IG 03/22/19 ACTION	NO NO NO NO NO NO PROV YES YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 02/26/19 02/26/19 04/15/18 03/10/19 EFF DATE	057 057 057 057 057 057 057 057 057 063 063	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER H CHIFFON H RONALD C KEMIE ANNMARIE SANDY N	9 3001 5603 5224 0677 1 5234	37 \$85029.0000 58 \$60403.0000 51 \$40629.0000 57 \$44426.0000 57 \$64863.0000 56 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN	RICHARD G JUSTIN E GERARD A MICHAEL R SCOTT 5 AMUEL 7 RONALD 7 VENKAT 1 JAMAL 7 SHAHIDA	70360 70360 70392 70360 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 C DEPT OF VETERA FOR PERIOD ENDIN \$120000.0000 \$139571.0000 ADMIN FOR CHILL FOR PERIOD ENDIN SALARY \$29.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED NS' SERVICES NG 03/22/19 ACTION RESIGNED PREN'S SVCS NG 03/22/19 ACTION INCREASE	NO NO NO NO NO NO YES YES PROV YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 02/26/19 02/26/19 02/26/19 02/15/18 03/10/19 EFF DATE 03/03/19	057 057 057 057 057 057 057 057 057 053 063 063	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER H CHIFFON H RONALD C KEMIE ANNMARIE SANDY N	9 3001 5603 5224 0677 1 5234	37 \$85029.0000 58 \$60403.0000 51 \$40629.0000 57 \$44426.0000 57 \$64863.0000 56 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED	NO YES NO YES YES	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN NAME ABDUL-SALAM ALSTON-MOORE	RICHARD G JUSTIN E GERARD A NICHAEL R SCOTT 5 AMUEL 7 NONALD 7 VENKAT 1 JAMAL 7 SHAHIDA 8	70360 70360 70392 70360 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 C DEPT OF VETERA FOR PERIOD ENDIN \$120000.0000 \$139571.0000 \$139571.0000 \$DMIN FOR CHILD FOR PERIOD ENDIN SALARY \$29.0000 \$67523.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED APPOINTED RESIGNED ACTION RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NO NO NO NO NO NO YES YES PROV YES YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 <u>EFF DATE</u> 04/15/18 03/10/19 <u>EFF DATE</u> 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER H CHIFFON H RONALD C KEMTE ANNMARTE SANDY D	2 3001 5 560 7 102 5221 067 6 5230	37 \$85029.0000 58 \$60403.0000 51 \$40629.0000 57 \$44426.0000 56 \$48195.0000 56 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED	NO YES NO YES YES NO	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18	067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN NAME ABDUL-SALAM ALSTON-MOORE ANDERSON	RICHARD G JUSTIN E GERARD A NICHAEL R SCOTT S AMUEL C SAMUEL C SAMUEL C SAMUEL C SAMUEL C SAMUEL C SHAHIDA C AYO R	70360 70360 70392 70360 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 C DEPT OF VETERA FOR PERIOD ENDIN SALARY \$120000.0000 \$139571.0000 ADMIN FOR CHILD FOR PERIOD ENDIN SALARY \$29.0000 \$67523.0000 \$70900.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES NG 03/22/19 ACTION RESIGNED RESIGNED RESIGNED RESIGNED DREN'S SVCS NG 03/22/19 ACTION INCREASE INCREASE APPOINTED	NO NO NO NO NO NO YES YES YES YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 <u>EFF DATE</u> 04/15/18 03/10/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO	ROGER H CHIFFON H RONALD C KEMIE ANNMARIE SANDY H NOT	2 3001 5601 7 1021 5224 0677 4 5234	37 \$85029.0000 58 \$60403.0000 51 \$40629.0000 57 \$44426.0000 56 \$48195.0000 56 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED resigned	NO YES NO YES YES NO	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18	067 067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN NAME ABDUL-SALAM ALSTON-MOORE ANDERSON ASIS	RICHARD G JUSTIN E GERARD A SCOTT S SAMUEL R NONALD H VENKAT J JAMAL S SHAHIDA R AYO R ARIEL B	70360 70360 70392 70360 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$0000 \$0000 \$67523.0000 \$70900.0000 \$276.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES NG 03/22/19 ACTION RESIGNED RESIGNED RESIGNED NCREASE INCREASE APPOINTED RETIRED	NO NO NO NO NO NO YES YES YES YES YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 EFF DATE 03/10/19 03/03/19 03/03/19 03/03/19 02/28/19	057 057 057 057 057 057 057 057 057 063 063 063 063	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO ILATIC BOARD BOARD D NOTICE The New Yor April 23, 201	ROGER H CHIFFON H RONALD C KEMIE ANNMARIE SANDY M NOT OF COF	2 3001 5603 7 1023 5224 0677 4 5234	37 \$85029.0000 58 \$60403.0000 51 \$40629.0000 57 \$44426.0000 56 \$48195.0000 56 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C	NO YES NO YES YES NO Speci	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18	067 067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN NAME ABDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK	RICHARD 6 JUSTIN 6 GERARD 4 SCOTT 5 SAMUEL 7 NONALD 7 VENKAT 7 JAMAL 7 SHAHIDA 8 AYO 8 ANTHONY 7	70360 70360 70392 70360 70360 53053 53053 53053 NY TITLE NUM 56058 0668A TITLE NUM 56058 30087 52408 90702 52366	\$90457.0000 \$90457.0000 \$95527.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$0000 \$57523.0000 \$70900.0000 \$276.0000 \$53519.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES G 03/22/19 ACTION RESIGNED RESIGNED NCREASE INCREASE APPOINTED RESIGNED	NO NO NO NO NO NO YES YES YES YES YES YES YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 EFF DATE 04/15/18 03/10/19 03/03/19 03/03/19 03/03/19 03/03/19 02/28/19 03/10/19	057 057 057 057 057 057 057 057 057 063 063 063 063	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO ILATIE BOARD BOARD D NOTICE The New You April 23, 201 Avenue (1st	ROGER I CHIFFON P RONALD C KEMTE ANNMARIE SANDY D NOT OF COF	2 3001 3 5603 7 1023 5224 067' 1 5234 1 67' 1 7' 1 7	37 \$85029.0000 38 \$60403.0000 31 \$40629.0000 37 \$44426.0000 37 \$44426.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 37 \$64863.0000 36 \$48195.0000 36 \$48195.0000 37 \$60000 38 \$60000 39 \$60000 30 \$60000 30 \$60000 31 \$60000 32 \$60000 33 \$60000 34 \$60000 34 \$60000 34 \$60000 35 \$60000 36 \$74000 37 \$74000 38 \$76000 39 \$76000 39 \$76000 39 \$76000 39 \$76000 39 \$76000 39 \$76000 39 \$76000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C	NO YES NO YES YES NO Speci Cente	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18	067 067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN NAME ABDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK BASS	RICHARD 6 JUSTIN 6 GERARD 4 SCOTT 5 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUE 7 SHAHIDA 7 AXO 8 ANTHONY 7 SARAH 8	70360 70360 70392 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$0000 \$67523.0000 \$70900.0000 \$70900.0000 \$53519.0000 \$109122.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES G 03/22/19 ACTION RESIGNED RESIGNED INCREASE INCREASE APPOINTED RESIGNED RESIGNED RESIGNED	NO NO NO NO NO NO YES YES YES YES YES YES YES YES NO	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 EFF DATE 04/15/18 03/10/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/02/28/19 03/10/19	057 057 057 057 057 057 057 057 057 063 063 063 063	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO ILATIE BOARD BOARD D NOTICE The New You April 23, 201 Avenue (1st	ROGER I CHIFFON P RONALD C KEMTE ANNMARIE SANDY D NOT OF COF	2 3001 3 5603 7 1023 5224 067' 1 5234 1 67' 1 7' 1 7	37 \$85029.0000 58 \$60403.0000 51 \$40629.0000 57 \$44426.0000 56 \$48195.0000 56 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C	NO YES NO YES YES NO Speci Cente	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18	063 065 065 065 065 065
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN ABDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN	RICHARD 6 JUSTIN 6 GERARD 4 SCOTT 5 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUE 7 SAMUE 7 SAMIDA 8 SARIEL 8 ANTHONY 7 SARAH 8	70360 70360 70392 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 C DEPT OF VETERA FOR PERIOD ENDIN \$120000.0000 \$139571.0000 ADMIN FOR CHILD FOR PERIOD ENDIN \$ALARY \$29.0000 \$67523.0000 \$70900.0000 \$70900.0000 \$53519.0000 \$109122.0000 \$61938.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES G 03/22/19 ACTION RESIGNED RESIGNED NCREASE INCREASE INCREASE APPOINTED RESIGNED RESIGNED PROMOTED	NO NO NO NO NO NO NO YES YES YES YES YES YES YES NO NO	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 02/26/19 02/26/19 02/26/19 03/10/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/01/19 03/01/19 03/01/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO ILAIN D BOARD BOARD MOTICE The New Yor April 23, 201 Avenue (1st the Special I	ROGER I CHIFFON F RONALD C KEMTE ANNMARIE SANDY F NOT OF COF	9 3001 5601 5224 0677 1029 5224 0677 1 5234	37 \$85029.0000 38 \$60403.0000 31 \$40629.0000 37 \$44426.0000 37 \$44426.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 37 \$64863.0000 36 \$48195.0000 36 \$48195.0000 37 \$60000 38 \$60000 39 \$60000 30 \$60000 30 \$60000 31 \$60000 32 \$60000 33 \$60000 34 \$60000 34 \$60000 34 \$60000 35 \$60000 36 \$74000 37 \$74000 38 \$76000 39 \$76000 39 \$76000 39 \$76000 39 \$76000 39 \$76000 39 \$76000 39 \$76000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED resigned rill hold, a S Children's C C, NY 10016 oard's webs	NO YES NO YES NO Speci Cente 3. Th.	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18	067 067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN ABDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI	RICHARD (JUSTIN) GERARD A SCOTT (SAMUEL) NONALD (NONALD) NUENKAT (JAMAL) AYO R ARIEL 0 ANTHONY 1 NICTORIA J SARAH 1 MARINA M	70360 70360 70392 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$07900.0000 \$70900.0000 \$70900.0000 \$7523.0000 \$109122.0000 \$61938.0000 \$82924.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES IG 03/22/19 ACTION RESIGNED RESIGNED NCREASE APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED PROMOTED RETIRED	NO NO NO NO NO NO NO YES YES YES YES YES YES YES YES NO NO NO	03/02/19 03/02/19 03/02/19 03/02/19 02/19/09 02/19/19 02/26/19 02/26/19 02/26/19 03/10/19 03/10/19 03/03/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19 03/10/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO ILATIE BOARD BOARD BOARD The New Yor April 23, 201 Avenue (1st the Special I The Board w	ROGER H CHIFFON H RONALD C KEMTE ANNMARTE SANDY H NOOT OF COH rk City Bo 19, at 9:00 Floor Auc Hearing is vill be disc	9 3000 5560 5224 0677 1029 5224 0677 1 5234 1067 1 5234 1067 1 5234 1067 1 5234 1 667 1 5234 1 667 1 5234 1 667 1 5234 1 667 1 5234 1 667 1 5234 1 5244 1 52	37 \$85029.0000 38 \$60403.0000 31 \$40629.0000 37 \$44426.0000 36 \$48195.0000 36 \$48195.0000 37 \$64863.0000 36 \$48195.0000 36 \$48195.0000 37 \$68863.0000 36 \$48195.0000 37 \$68863.0000 36 \$48195.0000 37 \$68863.0000 36 \$48195.0000 37 \$68863.0000 36 \$48195.0000 37 \$68863.0000 38 \$60000 39 \$60000 39 \$60000 39 \$60000 39 \$60000 39 \$60000 300000 \$60000 3000000 \$60000 3000000 \$60000 30000000 \$60000 30000000 \$60000 30000000 \$60000 300000000 \$600000 300000000000 \$600000 <	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a & Children's C C, NY 10016 oard's webs	NO YES NO YES NO Specia Cente 5. Th site.	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18 ial Heari r, 492 Fi e agenda on's	063 065 065 065 065 065
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN ABDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI BRADLEY	RICHARD 6 JUSTIN 6 GERARD 4 SCOTT 6 SAMUEL 7 SAMUEL 7 SAM	70360 70360 70392 70360 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631 52366	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$67523.0000 \$70900.0000 \$7523.0000 \$70900.0000 \$109122.0000 \$61938.0000 \$82924.0000 \$61184.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES IG 03/22/19 ACTION RESIGNED RESIGNED NCREASE INCREASE INCREASE APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NO NO NO NO NO NO NO YES YES YES YES YES YES NO NO NO NO	03/02/19 03/02/19 03/02/19 03/02/19 02/19/09 02/19/19 02/26/19 02/26/19 02/26/19 03/10/19 03/03/19 03/03/19 03/03/19 03/10/19 11/08/18 02/19/19 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO ILATIC BOARD BOARD BOARD The New You April 23, 201 Avenue (1st the Special I The Board w implementar	ROGER H CHIFFON H RONALD C KEMIE ANNMARIE SANDY M NOT OF COF OF COF TR City Bo 19, at 9:00 Floor Aud Hearing is vill be diso	2 3001 3 5603 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 1029 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 5247 52	37 \$85029.0000 38 \$60403.0000 31 \$40629.0000 37 \$44426.0000 36 \$44195.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 36 \$48195.0000 37 \$48195.0000 36 \$48195.0000 37 \$48195.0000 38 \$60000000 39 \$6000000000 39 \$60000000000 44100000000000000000000000000000000000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a \$ Children's C c, NY 10016 oard's webs ment of Cor	NO YES NO YES YES NO Specie Cente S. Th site. rrecti ards	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18 ial Heari er, 492 Fi e agenda on's on the	ng,
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN ABDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI BRADLEY BRUNSON	RICHARD (JUSTIN) (GERARD) (SCOTT) (SAMUEL) (NONALD) (NONALD) (NONALD) (NONAL) (NONAL	70360 70360 70392 70360 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 8297A 13631 52366 52287	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$139571.0000 \$67523.0000 \$67523.0000 \$67523.0000 \$53519.0000 \$53519.0000 \$109122.0000 \$61938.0000 \$82924.0000 \$61184.0000 \$44426.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES G 03/22/19 ACTION RESIGNED RESIGNED NCREASE INCREASE INCREASE APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RETIRED RETIRED RETIRED RESIGNED	NO NO NO NO NO NO NO YES YES YES YES YES YES NO NO NO NO NO	03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 02/26/19 02/26/19 02/26/19 03/10/19 03/10/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO ILATIC BOARD BOARD D NOTICE The New You April 23, 201 Avenue (1st the Special I The Board w implementa: Elimination	ROGER I CHIFFON P RONALD C KEMIE ANNMARIE SANDY D NOT OF COF OF COF TR City Bo 19, at 9:00 Floor Aud Hearing is vill be disc tion and c of Sexual	2 3000 3 5603 5224 067' 1 223 5224 067' 1 223 1 224 067' 1 223 067' 1 224 067' 1 224 0 0 0 0 0 0 0 0 0 0 0 0 0	37 \$85029.0000 38 \$60403.0000 31 \$40629.0000 37 \$44426.0000 37 \$44426.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 36 \$48195.0000 37 \$48195.0000 38 \$60000000 39 \$48195.0000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 31 \$48195.0000 32 \$48195.0000 33 \$48195.0000 34 \$48195.0000 34 \$48195.0000 35 \$48195.0000 36 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C c, NY 10016 oard's webs ment of Cor num Stand.	NO YES NO YES YES NO Speci Cente 3. Th site. rrecti ards nt in	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18 ial Heari er, 492 Fi e agenda on's on the Correctio	067 067 067 067 067
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN ABDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI BRADLEY BRUNSON CALIXTE	RICHARD 6 JUSTIN 6 GERARD 4 SCOTT 6 SAMUEL 7 SAMUEL 7 SAMUEL 7 SHAHIDA 7 ARIEL 8 SHAHIDA 8 ANTIEL 9 SARAH 10 SARAH 10 S SARAH 10 S S S SARAH 10 S SARAH 10 S SARAH 10 S SARAH 10 S SARAH 10 S SARAH 10 S SARAH 10 S SARAH	70360 70360 70392 70360 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631 52366	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$139571.0000 \$67523.0000 \$67523.0000 \$67523.0000 \$53519.0000 \$53519.0000 \$53519.0000 \$61938.0000 \$61938.0000 \$61184.0000 \$61184.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES IG 03/22/19 ACTION RESIGNED RESIGNED INCREASE APPOINTED RESIGNED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED APPOINTED	NO NO NO NO NO NO YES YES YES YES NO NO NO NO NO NO YES	03/02/19 03/02/19 03/02/19 03/02/19 02/02/19 02/19/19 02/26/19 02/26/19 02/26/19 02/26/19 02/26/19 02/26/19 02/26/19 03/10/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO ILATIE BOARD BOARD INOTICE The New You April 23, 201 Avenue (1st the Special I The Board w implementat Elimination Facilities. Th	ROGER I CHIFFON R RONALD C KEMTE ANNMARIE SANDY D NOT OF COF OF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF	2 3000 3 5603 7 1023 5224 067' 1 5234 1 67' 1 67'	37 \$85029.0000 38 \$60403.0000 31 \$40629.0000 37 \$44426.0000 36 \$44426.0000 36 \$44195.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 36 \$48195.0000 37 \$48195.0000 36 \$48195.0000 37 \$48195.0000 38 \$48195.0000 39 \$48195.0000 39 \$48195.0000 30 \$48195.0000 30 \$48195.0000 31 \$48195.0000 32 \$48195.0000 33 \$48195.0000 34 \$48195.0000 35 \$48195.0000 36 \$48195.0000 37 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C c, NY 10016 oard's webs ment of Cor num Stand. Harassmer went into e	NO YES NO YES YES NO Speci- Cente 3. This ite. Trecti ards at in ffect	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 ial Heari er, 492 Fi e agenda on's on the Correctii in Janua	ng, for
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN ADDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI BRADLEY BRUNSON CALIXTE CAMPBELL	RICHARD (JUSTIN) (GERARD) (SCOTT) (SAMUEL) (NONALD) (NONALD) (NONALD) (NONALD) (NONAL) (NONA	70360 70360 70392 70360 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631 52366 52287	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$139571.0000 \$67523.0000 \$67523.0000 \$57523.0000 \$53519.0000 \$53519.0000 \$53519.0000 \$61938.0000 \$61938.0000 \$61184.0000 \$61184.0000 \$55000.0000 \$55000.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RETIRED RESIGNED RETIRED RESIGNED RESIGNED PROMOTED RETIRED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NO NO NO NO NO NO YES YES YES YES YES YES YES YES YES YES	03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 02/26/19 02/26/19 02/26/19 02/26/19 02/26/19 02/26/19 03/03/10/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO ILATIC BOARD NOTICE The New You April 23, 200 Avenue (1st the Special I The Board w implementat Elimination Facilities. TH 2017, are de	ROGER I CHIFFON P RONALD C KEMTE ANNMARIE SANDY P NOOT OF COF	a 3000 5600 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000	37 \$85029.0000 38 \$60403.0000 31 \$40629.0000 37 \$44426.0000 37 \$44426.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 36 \$48195.0000 37 \$44100.000 36 \$48195.0000 37 \$44126.0000 36 \$48195.0000 37 \$44126.0000 38 \$60401.000 39 \$60000.000 39 \$60000.000 30 \$48195.0000 30 \$60000.000 30 \$48195.0000 30 \$60000.000 30 \$60000.000 30 \$60000.000 30 \$60000.000 30 \$60000.000 30 \$60000.000 30 \$60000.000 30 \$60000.000 30 \$60000.000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C c, NY 10016 oard's webs nent of Cor num Stand: Harassmer went into e l respond to	NO YES NO YES YES NO Speci Cente 3. Th site. Trecti ards at in ffect o sex	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18 ial Heari r, 492 Fi e agenda on's on the Correctii in Janua ual abus	ng, ona ona ary e a
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN NAME ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI BRADLEY BRUNSON CALIXTE CAMPBELL CANTARERO	RICHARD (JUSTIN) GERARD (SCOTT) SAMUEL) CONALD) CON	70360 70360 70392 70360 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631 52366 52287 56058	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$67523.0000 \$67523.0000 \$67523.0000 \$67523.0000 \$6153519.0000 \$109122.0000 \$61938.0000 \$61938.0000 \$61938.0000 \$61184.0000 \$61184.0000 \$55000.0000 \$55000.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES NG 03/22/19 ACTION RESIGNED RESIGNED NCREASE APPOINTED RESIGNED RESIGNED RESIGNED PROMOTED RESIGNED PROMOTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED	NO NO NO NO NO NO NO YES YES YES YES YES YES YES YES YES YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 02/26/19 02/26/19 02/26/19 02/26/19 03/03/10 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO ILATIC BOARD NOTICE The New You April 23, 200 Avenue (1st the Special I The Board w implementat Elimination Facilities. TH 2017, are de	ROGER I CHIFFON P RONALD C KEMTE ANNMARIE SANDY D NOOT OF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF	a 3000 5600 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5220 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5230 067 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5300 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000	37 \$85029.0000 38 \$60403.0000 31 \$40629.0000 37 \$44426.0000 37 \$44426.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 38 \$60403.0000 39 \$60403.0000 39 \$60403.0000 30 \$44195.0000 30 \$44195.0000 30 \$44195.0000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 31 \$48195.0000 32 \$48195.0000 32 \$48195.0000 33 \$48195.0000 34 \$48195.0000 34 \$48195.0000 35 \$48195.0000 36 \$48195.0000 37 \$48195.0000 <td>RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C c, NY 10016 oard's webs nent of Cor num Stand: Harassmer went into e l respond to</td> <td>NO YES NO YES YES NO Speci Cente 3. Th site. Trecti ards at in ffect o sex</td> <td>03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18 ial Heari r, 492 Fi e agenda on's on the Correctii in Janua ual abus</td> <td>ng, rst</td>	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C c, NY 10016 oard's webs nent of Cor num Stand: Harassmer went into e l respond to	NO YES NO YES YES NO Speci Cente 3. Th site. Trecti ards at in ffect o sex	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18 ial Heari r, 492 Fi e agenda on's on the Correctii in Janua ual abus	ng, rst
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN NAME ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI BRADLEY BRADLEY BRADLEY BRADLEY CAMPBELL CANTARERO CUMMINGS	RICHARD (JUSTIN) GERARD (SCOTT) SAMUEL (SAMUEL) (NUENKAT) JAMAL (JAMAL) SAMAL (MARINA) SARAH 0 MARINA) MARINA) MARINA 1 SARAH 0 MARINA 1 SARAH 1 SARAH 1 SARAH 1 SARAH 1 SARAH 1 SARAH 2 SARAH 1 SARAH 1 SARAH 1 SARAH 1 SARAH 2 SARAH 1 SARAH 1	70360 70360 70392 70360 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631 52366 52287 56058 52366 56058	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$617523.0000 \$67523.0000 \$67523.0000 \$67523.0000 \$109122.0000 \$61938.0000 \$61938.0000 \$61938.0000 \$61938.0000 \$55000.0000 \$55707.0000 \$55707.0000 \$60000.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES IG 03/22/19 ACTION RESIGNED RESIGNED RESIGNED RETIRED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED	NO NO NO NO NO NO YES YES YES YES YES YES YES YES YES YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 02/26/19 02/26/19 02/26/19 02/26/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO ILATIC BOARD BOARD D NOTICE The New You April 23, 201 Avenue (1st the Special I The Board w implementat Elimination Facilities. Th 2017, are de sexual haras operated by	ROGER I CHIFFON F RONALD C KEMIE ANNMARIE SANDY M NOT OF COF OF COF OF COF I, at 9:00 Floor Aud Hearing is vill be disc tion and c of Sexual ne Board's signed to ssment of DOC.	2 3000 3 5603 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5224 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 0677 1029 5234 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 10	37 \$85029.0000 38 \$60403.0000 31 \$40629.0000 37 \$44426.0000 36 \$44426.0000 36 \$44195.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 37 \$44126.0000 36 \$48195.0000 37 \$48195.0000 38 \$64863.0000 39 \$64863.0000 39 \$64863.0000 39 \$64863.0000 39 \$64863.0000 39 \$64863.0000 39 \$64863.0000 39 \$64863.0000 39 \$64863.0000 39 \$64863.0000 39 \$64863.0000 39 \$64863.0000 39 \$64863.0000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C c, NY 10016 oard's webs nent of Cor num Stand. Harassmer went into e l respond to ed in jails a	NO YES NO YES YES NO Specie Cente S. Th site. rrecti ards at in ffect o sex and o	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18 ial Heari er, 492 Fi e agenda on's on the Correcti in Janua ual abus ther facil	ng, for
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN NAME ABDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI ERADLEY BRUNSON CLALIXTE CANTARERO CUMMINGS DANIEL	RICHARD (JUSTIN) GERARD (SCOTT) SAMUEL) NONALD (NONALD) NUENKAT (JAMAL) NUENKAT (JAMAL) NUENKAT (NUENKAT) NUENKAT) NUENKAT (NUENKAT) NUENKAT (NUENKAT) NUENKAT) NUENKAT (NUENKAT) NUENKAT	70360 70360 70360 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631 52287 56058 52366 52287 56058	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$139571.0000 \$617523.0000 \$7523.0000 \$7523.0000 \$7523.0000 \$109122.0000 \$61938.0000 \$61938.0000 \$61184.0000 \$4426.0000 \$5500.0000 \$55707.0000 \$55707.0000 \$5000.0000 \$55000.0000 \$55000.0000 \$55000.0000 \$55000.0000 \$55000.0000 \$55000.0000 \$5105121.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED APPOINTED RESIGNED RESIGNED RESIGNED RETIRED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED	NG NG NG NG NG NG YES YES YES YES YES YES YES YES YES NG NG NG NG NG YES YES NG YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 02/26/19 02/26/19 02/26/19 02/26/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 063 063 063 063 063 067 067 067 067 067 067 067 067 067 067	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO LATIE BOARD NOTICE The New You April 23, 20: Avenue (1st the Special I The Board w implementa: Elimination Facilities. Th 2017, are de sexual haras operated by The public w	ROGER I CHIFFON R RONALD C KEMTE ANNMARIE SANDY D NOT OF COF OF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF	2 3000 3 5603 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 5224 067' 1023 1023 1023 1023 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 1024 10	37 \$85029.0000 38 \$60403.0000 31 \$40629.0000 37 \$44426.0000 37 \$44426.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 36 \$48195.0000 36 \$48195.0000 37 \$44195.0000 38 \$64863.0000 39 \$48195.0000 30 \$48195.0000 30 \$48195.0000 31 \$48195.0000 32 \$48195.0000 32 \$48195.0000 33 \$48195.0000 34 \$48195.0000 34 \$48195.0000 35 \$48195.0000 36 \$48195.0000 37 \$48195.0000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C c, NY 10016 oard's webs nent of Cor num Stand. Harassmer went into e I respond to ed in jails a mment on	NO YES NO YES YES NO Speci Cente 6. Thisite. Trecti ards int in fffect o sex and o these	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 ial Hearri er, 492 Fi e agenda on's on the Correctii in Janua ual abus ther facil e issues.	ng, rst ona litic Pul
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHWAN NAME ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI BRADLEY BRUNSON CALIXTE CANTARERO CUMMINGS DANIEL DESOUZA	RICHARD (JUSTIN) (GERARD) (SOUT) (SAMUEL) (SAMUEL) (SAMUEL) (SAMUEL) (SAMUEL) (SAMUEL) (SAMAL) (SAMAL) (SAMAL) (SARAH (S	70360 70360 70360 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631 52366 52287 56058 52366 52287 56058 52366 52287 56058	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$419571.0000 \$617523.0000 \$7523.0000 \$7523.0000 \$7523.0000 \$7523.0000 \$109122.0000 \$6138.0000 \$6138.0000 \$61184.0000 \$55000.0000 \$557070.0000 \$55000.0000 \$55000.0000 \$54093.0000 \$105121.0000 \$41849.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES G 03/22/19 ACTION RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED	NO NO NO NO NO NO YES YES YES YES YES YES YES YES YES NO NO NO NO NO NO NO NO NO NO YES YES NO YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 02/26/19 02/26/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 063 063 063 063 063 067 067 067 067 067 067 067 067 067 067	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO LATIE BOARD NOTICE The New You April 23, 201 Avenue (1st the Special I The Board w implementat Elimination Facilities. Th 2017, are de sexual haras operated by The public w comment on	ROGER I CHIFFON R RONALD C KEMTE ANNMARIE SANDY D NOT OF COF OF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF	 3000 \$5603 \$5224 \$6677 \$677 \$677	37 \$85029.0000 38 \$60403.0000 31 \$40629.0000 37 \$44426.0000 37 \$44426.0000 36 \$44195.0000 36 \$44195.0000 36 \$48195.0000 36 \$48195.0000 37 \$44426.0000 36 \$48195.0000 36 \$48195.0000 37 \$44195.0000 36 \$48195.0000 37 \$44126.0000 36 \$48195.0000 36 \$48195.0000 37 \$44126.0000 38 \$60401.000 39 \$6001.000 39 \$6001.000 39 \$6001.000 39 \$6001.000 39 \$6001.000 39 \$6001.000 30 \$6001.000 30 \$6001.000 30 \$6001.000 30 \$6001.000 30 \$6001.000 30 \$6001.000 30 \$6001.000	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C c, NY 10016 oard's webs ment of Cor num Stand. Harassmer went into e I respond to ed in jails a mment on assment w	NO YES NO YES YES NO Speci Cente 3. This iste. Trecti ards at in ffect o sex and o these ill be	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 ial Heari er, 492 Fi e agenda on's on the Correctii in Janua ual abus ther facil e issues.	ng, rst litic Pul
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN NAME ABDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI BRADLEY BRUNSON CALIXTE CAMPBELL CAMTRERO CUMMINGS DANIEL DESOUZA DICKSON-WILLIAM	RICHARD G JUSTIN 6 GERARD 4 SCOTT 5 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SARAH 7	70360 70360 70360 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631 52366 52287 5638 52366 56058 52369 30087 90235 52366	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$67523.0000 \$67523.0000 \$67523.0000 \$61938.0000 \$61938.0000 \$61184.0000 \$55519.0000 \$55000.0000 \$55000.0000 \$55000.0000 \$55000.0000 \$54093.0000 \$105121.0000 \$41849.0000 \$53519.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED APPOINTED RESIGNED RESIGNED RESIGNED ACTION INCREASE INCREASE INCREASE APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED	NO NO NO NO NO NO YES YES YES YES YES YES YES YES NO NO NO NO NO NO NO YES YES NO YES NO YES NO	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 02/26/19 02/26/19 03/03/10 03/10/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO ILATIC BOARD NOTICE The New You April 23, 201 Avenue (1st the Special I The Board w implementa: Elimination Facilities. Th 2017, are de sexual haras operated by The public w comment on minutes per	ROGER I CHIFFON R RONALD C KEMTE ANNMARTE SANDY D CNOT OF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF	 3000 5601 5224 667 5224 667 5224 667 5224 667 5224 667 677 5224 667 677 5224 667 677 6	37 \$85029.0000 38 \$60403.0000 31 \$40629.0000 31 \$44426.0000 32 \$44426.0000 33 \$44426.0000 34 \$44426.0000 35 \$44426.0000 36 \$44195.0000 36 \$48195.0000 36 \$48195.0000 37 \$4426.0000 38 \$60401.000 39 \$60401.000 39 \$60401.000 30 \$44195.0000 30 \$6001.000 30 \$44195.0000 30 \$6001.000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 30 \$48195.0000 30 \$6001.000 30 \$6001.000 30 \$6001.000 30 \$6001.000 30 \$6001.000 30 \$6001.000 <t< td=""><td>RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C C, NY 10016 oard's webs nent of Cor Harassmer went into e l respond to ed in jails a mment on ' assment w</td><td>NO YES NO YES YES NO Special Cented S. The site. Trectil ards thin fifect o sex und o these ill be</td><td>03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18 ial Heari or, 492 Fi e agenda on's on the Correction in Janua ual abus ther facil e issues. b limited by callin</td><td>ng, rst litic Pul</td></t<>	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C C, NY 10016 oard's webs nent of Cor Harassmer went into e l respond to ed in jails a mment on ' assment w	NO YES NO YES YES NO Special Cented S. The site. Trectil ards thin fifect o sex und o these ill be	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18 ial Heari or, 492 Fi e agenda on's on the Correction in Janua ual abus ther facil e issues. b limited by callin	ng, rst litic Pul
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN ADDUL-SALAM ALSTON-MOORE ALSTON-MOORE ALSTON-MOORE ANDERSON ASIS BAIK BASS BAIK BASS BERNAN BOKHARI BRADLEY BRUNSON CALIXTE CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBELL CAMPBE	RICHARD G JUSTIN 6 GERARD 4 SCOTT 5 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUEL 7 SAMUE 7 SAMUE 7 SARAH 7 S	70360 70360 70392 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631 52366 52287 56058 52366 56058 52366 56058 52366 56058	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$617523.0000 \$67523.0000 \$70900.0000 \$70900.0000 \$109122.0000 \$6138.0000 \$6138.0000 \$61184.0000 \$55000.0000 \$55000.0000 \$55003.0000 \$5000.0000 \$544426.0000 \$55003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000 \$5003.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES G 03/22/19 ACTION RESIGNED RESIGNED NCREASE APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED APPOINTED RESIGNED TERMINATED INCREASE APPOINTED	NO NO NO NO NO NO NO YES YES YES YES YES YES YES NO NO NO NO NO NO YES YES NO YES YES NO YES YES	03/02/19 03/02/19 03/02/19 03/02/19 02/19/09 02/19/19 02/26/19 02/26/19 02/26/19 02/26/19 03/01/19 03/03/10/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO LATTE BOARD NOTICE The New Yor April 23, 201 Avenue (1st the Special I The Board w implementat Elimination Facilities. Th 2017, are de sexual haras operated by The public w comment on minutes per (212) 669-79	ROGER I CHIFFON R RONALD C KEMTE ANNMARTE SANDY D CNOT OF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF	a 3000 5600 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5220 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 667 5230 677 520 677 520 677 520 677 520 677 520 677 520 677 520 677 520 677 520 677 520 677 507 507 507 507 507 507 507 5	T \$5029.0000 \$50403.0000 \$51 \$40629.0000 \$75 \$44426.0000 \$75 \$44426.0000 \$75 \$44426.0000 \$75 \$44125.0000 \$75 \$44195.0000 \$75 \$48195.0000 \$75 \$48195.00000 \$75 \$48195.00000 \$75 \$48195.00000 \$75 \$48195.00000 \$75 \$4819	RESIGNED INCREASE APPOINTED APPOINTED RESIGNED rill hold, a S Children's C C, NY 10016 oard's webs ment of Corr tum Standa Harassmer went into e l respond to ed in jails a mment on ' assment w ore the hea gov. You can	NO YES NO YES YES NO Special Scente 3. This ite. Trecti and so these ill be ring n also	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 ial Heari er, 492 Fi e agenda on's on the Correction in Janua ual abus ther facil e issues. b limited by callin o sign up	ng, rst litie Puk to s g in
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN ADDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI BRADLEY BRUNSON CALIXTE CAMPBELL CANTARERO CUMMINGS DANIEL DESOUZA DINIEL DESOUZA DINIEL DESOUZA	RICHARD G JUSTIN 6 GERARD 4 SCOTT 5 SAMUEL 7 SAMUEL 7 SAM	70360 70360 70360 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631 52366 52287 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 56058 52366 52370	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$607523.0000 \$70900.0000 \$70900.0000 \$109122.0000 \$61938.0000 \$61938.0000 \$61938.0000 \$61938.0000 \$61938.0000 \$55500.0000 \$61938.0000 \$61938.0000 \$61938.0000 \$5500.0000 \$61938.0000 \$54093.0000 \$54093.0000 \$105121.0000 \$41849.0000 \$32918.0000 \$32918.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES G 03/22/19 ACTION RESIGNED RESIGNED NCREASE APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED TREMINATED RESIGNED TERMINATED INCREASE APPOINTED RESIGNED TERMINATED INCREASE APPOINTED	NO NO NO NO NO NO NO YES YES YES YES YES YES YES YES NO NO NO YES NO YES NO YES NO YES NO YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 02/26/19 02/26/19 02/26/19 03/10/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO BOARD BOARD NOTICE The New Yor April 23, 201 Avenue (1st the Special I The Board w implementai Elimination Facilities. Th 2017, are de sexual harass operated by The public w comment on minutes per (212) 669-79 the hearing	ROGER I CHIFFON R RONALD C KEMTE ANNMARTE SANDY D CNOT OF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF	2 3000 3 5601 5220 667 5220 667 67 5220 67 67 67 67 67 67 67 67 67 67	SCTION f Correction w , at the ACS C um), New York lable on the B ng the Departri- tiance of Minim se and Sexual dards, which ' t, prevent, and mus incarcerate portunity to co und sexual har an sign up bef boc@boc.nyc., ay of the heari	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED RESIGNED rill hold, a S Children's C C, NY 10016 oard's webs ment of Corr hum Standa Harassmer Went into e l respond tr ed in jails a mment on assment w ore the hea gov. You car ng. Additio	NO YES NO YES YES NO Specia Centa Streeti ards nt in ffect o sex und o these ill be rring n alsy nally	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18 ial Heari or, 492 Fi e agenda con's on the Correctii in Janua ual abus ther facil e issues. E limited by callin p sign up y, the Boa	ng, rst ona litic ona ury e a litic ona
VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN ABDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI BRADLEY BRINSON CALIXTE CAMPBELL CANTARERO CUMMINGS DANIEL DISOUZA DICKSON-WILLIAM DUGAN EDWARDS FOX	RICHARD (JUSTIN) (GERARD) (SCOTT) (SAMUEL) (SAMUEL) (SAMUEL) (SAMUEL) (SAMUEL) (SAMUEL) (SAMUEL) (SAMUEL) (SAMATION) (SARAH)	70360 70360 70360 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631 52366 52287 56058 52366 52287 56058 52366 52287 56058 52366 52287 52366 52366 52287 52366	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$6000.0000 \$70900.0000 \$70900.0000 \$70900.0000 \$109122.0000 \$61938.0000 \$61938.0000 \$61938.0000 \$5500.0000 \$5500.0000 \$5500.0000 \$5500.0000 \$5500.0000 \$5500.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.0000 \$54093.00000 \$54093.00000000000000000000000000000000000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES G 03/22/19 ACTION RESIGNED RESIGNED NCREASE APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED TERMINATED RESIGNED TERMINATED RESIGNED TERMINATED RESIGNED TERMINATED NCREASE APPOINTED RESIGNED TERMINATED INCREASE APPOINTED RESIGNED	NO NO NO NO NO NO NO YES YES YES YES YES NO NO YES NO YES NO YES NO YES NO YES NO YES NO YES NO	03/02/19 03/02/19 03/02/19 03/02/19 02/02/19 02/19/19 02/26/19 02/26/19 02/26/19 02/26/19 02/26/19 02/26/19 02/26/19 03/10/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO BOARD C NOTICE The New Yor April 23, 201 Avenue (1st the Special I The Board w implementar Elimination Facilities. Th 2017, are de sexual haras operated by The public w comment on minutes per (212) 669-79 the hearing welcomes with	ROGER I CHIFFON F RONALD C KEMTE ANNMARTE SANDY F NOOT OF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF	a 3000 5600 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 1029 5220 1029 5220 1029 5220 1029 5200 1029 5200 1029 5200 1029 5200 1029 5200 1029 5200 1029 5200 1029 5200 1029 5200 1029 5200 1029 5200 1029 5200 1029 5200 1029 5200 1029 5200 1029 5000 1029 5000 1029 5000 1029 5000 1029 5000 1029 5000 1029 5000 1029 5000 1029 5000 1029 5000 1029 5000 1029 5000 1029 5000 1029 5000 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1020	Section w and Sexual and Sexual bar the ACS (and Sexual bar the content of the form and sexual contunity to co bar the heari cortunity to co and sexual han bar the heari and sexual han bar the heari and sexual han bar the heari and sexual han bar the heari bar the heari an sign up bef bac boc.nyc. ay of the heari s prior to or a	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED RESIGNED rill hold, a S Children's C Children's C C Children's C C Children's C C Children's C C C C C C C C C C C C C C C C C C C	NO YES NO YES NO Special Scale Special Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Sca	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18 ial Heari r, 492 Fi e agenda on's on the Correcti in Janua ual abus ther facil e issues. • e limited by callin e o sign up r, the Boa c. Writter	ng, rst litie ona ury e a litie g in
VATA VERNON WAYNE WEIHE JR. WILSON WORONTZOFF WRIGHT ZAVALUNOV NAME MOTUPALLI OTHMAN NAME ABDUL-SALAM ALSTON-MOORE ANDERSON ASIS BAIK BASS BERNAN BOKHARI BRADLEY BRUNSON CULLIYTE CAMPBELL CANTARERO CUMMINGS DANIEL DESOUZA DICKSON-WILLIAM DUGAN EDWARDS FOX FUNG GEDEON	RICHARD G JUSTIN 6 GERARD 4 SCOTT 5 SAMUEL 7 SAMUEL 7 SAM	70360 70360 70360 70360 53053 53053 53053 NY TITLE NUM 95615 0668A TITLE NUM 56058 30087 52408 90702 52366 10056 8297A 13631 52366 52287 56058 52366 52287 56058 52366 52287 56058 52366 52287 52366 52366 52287 52366	\$90457.0000 \$90457.0000 \$90457.0000 \$90457.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$35254.0000 \$139571.0000 \$139571.0000 \$607523.0000 \$70900.0000 \$70900.0000 \$109122.0000 \$61938.0000 \$61938.0000 \$61938.0000 \$61938.0000 \$61938.0000 \$55500.0000 \$61938.0000 \$61938.0000 \$61938.0000 \$5500.0000 \$61938.0000 \$54093.0000 \$54093.0000 \$105121.0000 \$41849.0000 \$32918.0000 \$32918.0000	PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED PROMOTED RESIGNED NS' SERVICES G 03/22/19 ACTION RESIGNED RESIGNED NCREASE APPOINTED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED RESIGNED TREMINATED RESIGNED TERMINATED INCREASE APPOINTED RESIGNED TERMINATED INCREASE APPOINTED	NO NO NO NO NO NO NO YES YES YES YES YES YES YES YES NO NO NO YES NO YES NO YES NO YES NO YES	03/02/19 03/02/19 03/02/19 03/02/19 03/02/19 02/19/19 02/26/19 02/26/19 02/26/19 02/26/19 03/10/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19 03/03/19	057 057 057 057 057 057 057 057 057 057	ROSENMAN SCOTT SIMONS SMALL SMITH SOTO BOARD C NOTICE The New Yor April 23, 201 Avenue (1st the Special I The Board w implementar Elimination Facilities. Th 2017, are de sexual haras operated by The public w comment on minutes per (212) 669-79 the hearing welcomes with	ROGER I CHIFFON F RONALD C KEMTE ANNMARTE SANDY F NOT OF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF COF	a 3000 5600 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5220 0677 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5230 1029 5200 1029 5200 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1029 1020	T \$5029.0000 \$50403.0000 \$50403.0000 \$504029.0000 \$70544426.0000 \$5055000 \$5055000 \$50550000 \$50550000 \$50550000 \$50550000 \$505500000 \$505500000 \$5055000000 \$5055000000 \$50550000000000	RESIGNED INCREASE APPOINTED APPOINTED APPOINTED RESIGNED rill hold, a S Children's C Children's C C Children's C C Children's C C Children's C C C C C C C C C C C C C C C C C C C	NO YES NO YES NO Special Scale Special Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Scale Sca	03/03/19 03/03/19 02/10/19 02/24/19 03/03/19 07/08/18 07/08/18 ial Heari r, 492 Fi e agenda on's on the Correcti in Janua ual abus ther facil e issues. • e limited by callin e o sign up r, the Boa c. Writter	ng, rst for ona ary e an litie g in urd

FRIDAY, APRIL, 19, 2019

THE CITY RECORD

Please inform the Board if you need a reasonable accommodation of a disability at the Special Hearing. Please also inform us if you need a language interpreter. You can inform us by telephone at (212) 669-7900, or by email at boc@boc.nyc.gov. Please inform us by the close of business on April 16, 2019, so that we have sufficient time to arrange the accommodation.

🖝 a19

DISTRICT ATTORNEY - NEW YORK COUNTY

■ SOLICITATION

Construction / Construction Services

PLUMBING- 3RD FL AND BASEMENT INTERIOR

RENOVATION - Competitive Sealed Bids - PIN# 20190900015 - Due 5-17-19 at 3:00 P.M.

Renovation of the 3rd floor and basement, at 80 Centre Street. DANY is requesting the services of a licensed plumbing contractor, to provide plumbing work for the interior renovation of space.

Please note that this is one contract for two renovation projects, which may happen on different schedules. The contractor shall review two separate bid packages and fill in two bid pages.

• FIRE SPRINKLER- 80 CENTRE BASEMENT INTERIOR RENOVATION - Competitive Sealed Bids - PIN# 20190900012 -Due 5-17-19 at 3:00 P.M.

Licensed Fire Sprinkler Contractors, to provide and install sprinkler systems.

• MECHANICAL (HVAC)- 80 CENTRE BASEMENT INTERIOR RENOVATION - Competitive Sealed Bids - PIN# 20190900011 -Due 5-17-19 at 3:00 P.M.

Licensed Mechanical (HVAC) Contractor, to provide and install all mechanical (HVAC) systems.

Please download and save all documents from the FTP account. Note, the FTP account is temporary. Please save prior to 5/24/19.

FTP Account http://download.manhattanda.org Login: 50000664/PW: 50000664

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor prequalification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above.

District Attorney - New York County, 1 Hogan Place, New York, NY 10013. Louise Pettiford (212) 335-3459; pettifordl@dany.nyc.gov

🖝 a19

HUMAN RESOURCES ADMINISTRATION

OFFICE OF CONTRACTS

AWARD

Services (other than human services)

HASA TRAINING ACADEMY VII, CITYWIDE - Competitive Sealed Bids - PIN# 17BPEHA02301 - AMT: \$918,013.00 - TO: Fordham University, 441 East Fordham Road, Bronx, NY 10458.

🖝 a19

LAW DEPARTMENT

■ INTENT TO AWARD

Services (other than human services)

NOTICE OF INTENT TO ENTER INTO SOLE SOURCE NEGOTIATIONS WITH MITRATECH HOLDINGS, INC. - Sole Source - Available only from a single source - PIN# 02519X003795 -Due 5-8-19 at 5:00 P.M.

Notice of Intent to enter into Sole Source ("Sole Source"), Negotiations with Mitratech Holdings, Inc., d/b/a Law Manager Inc., for Provision of Software Maintenance Services, for Case Management System PIN 02519X003795; E-PIN 02519S0003).

IT IS THE INTENT of the New York City Law Department ("Department"), to enter into negotiations for a one-year contract with Mitratech Holdings, Inc., ("Mitratech"), d/b/a Law Manager Inc., ("Law Manager"), pursuant to PPB Rules Section 3-05(a) for the provision of software maintenance services for the Law Manager case management system which is proprietary to Mitratech.

Based upon information obtained from Law Manager, the Department's Agency Chief Contracting Officer ("ACCO"), has determined that there is only one source for the required service.

Firms that believe they are qualified to provide these services and wish to be considered for future procurements for the same or similar services, should send an expression of interest to the Department at the following address: Esther S. Tak, Senior Counsel, New York City Law Department, 100 Church Street, Room 5-208, New York, NY 10007; Phone (212) 356-1122; Fax (212) 356-1148; Email etak@law.nyc.gov.

Use the following address unless otherwise specified in notice, to secure, examine or submit bid/proposal documents, vendor prequalification and other forms; specifications/blueprints; other information; and for opening and reading of bids at date and time specified above.

Law Department, 100 Church Street, Room 5-208, New York, NY 10007. Esther Tak (212) 356-1122; Fax: (212) 356-1148; etak@law.nyc.gov

• a19-25

OFFICE OF THE MAYOR

■ NOTICE

OFFICE OF THE MAYOR NOTICE OF A PUBLIC HEARING ON PROPOSED LOCAL LAW

PURSUANT TO STATUTORY REQUIREMENT, NOTICE IS HEREBY GIVEN that proposed Local Law numbered and titled hereinafter, has been passed by the Council, and that a public hearing on such proposed Local Law, will be held in the Blue Room, at City Hall, Borough of Manhattan, New York City, on April 24, 2019 at 3:00 P.M.:

Int. 276-A - A Local Law to amend the New York City building code, in relation to requiring that the roofs of certain buildings be partially covered in green roof or solar photovoltaic electricity generating systems.

Int. 1031-A - A Local Law to amend the administrative code of the City of New York, in relation to posting information regarding green roofs on the website of the department of buildings.

Int. 1032-A - A Local Law to amend the administrative code of the City of New York and the New York City building code, in relation to requiring that the roofs of certain buildings be covered in green roofs or solar photovoltaic electricity generating systems.

Int. 1251-A - A Local Law to amend the administrative code of the City of New York, in relation to a building energy efficiency grade.

Int. 1252-A - A Local Law to amend the administrative code of the City of New York, in relation to establishing a sustainable energy loan program.

Int. 1253-C - A Local Law to amend the New York City charter and the administrative code of the City of New York, in relation to the commitment to achieve certain reductions in greenhouse gas emissions by 2050.

Int. 1317-A - A Local Law to amend the administrative code of the City of New York, and the New York City building code, in relation to large wind turbines.

Int. 1318-A - A Local Law to amend the administrative code of the City of New York, in relation to an assessment of the replacement of gasfired power plants and to amend Local Law number 248 for the year 2017, in relation to the completion date of the long-term energy plan.

Preconsidered Int.____- A Local Law to amend the administrative code of the City of New York, in relation to a paper carryout bag reduction fee, and to repeal Chapter 4-F of Title 16 of such code, relating to carryout bag reduction.

Bill de Blasio Mayor

NOTE: Individuals requesting Sign Language Interpreters should contact the Mayor's Office of City Legislative Affairs, 253 Broadway, 4th Floor, New York, NY 10007, (212) 788-3678, no later than <u>five days</u> prior to the public hearing.

COURT NOTICE MAP FOR 142ND STREET AT ITS INTERSECTION WITH THE SOUTHWEST CORNER OF 135TH AVENUE

